

WT/DS400/R WT/DS401/R

25 November 2013

Page: 1/186

Original: English

EUROPEAN COMMUNITIES – MEASURES PROHIBITING THE IMPORTATION AND MARKETING OF SEAL PRODUCTS

REPORTS OF THE PANEL

Note by the Secretariat:

(13-6374)

These Panel Reports are in the form of a single document constituting two separate Panel Reports: WT/DS400/R and WT/DS401/R. The cover page, preliminary pages, sections 1 through 7 are common to both Reports. The page header throughout the document bears the two document symbols WT/DS400/R and WT/DS401/R, with the following exceptions: section 8 on pages CAN-183 and CAN-184, which bears the document symbol for and contains the Panel's conclusions and recommendations in the Panel Report WT/DS400/R; and section 8 on pages NOR-185 and NOR-186, which bears the document symbol for and contains the Panel's conclusions and recommendations in the Panel Report WT/DS401/R. The annexes, which are a part of the Panel Reports, are circulated in a separate document (WT/DS400/R/Add.1 and WT/DS401/R/Add.1).

- 2 -

TABLE OF CONTENTS

1	INTRODUCTION1	2	
1.1	Complaints by Canada and Norway	12	
1.2	Panel establishment and composition	12	
1.3	Panel proceedings	13	
1.3	1 General	13	
1.3	2 Procedures for open hearings	13	
1.3	3 Requests for enhanced third-party rights	13	
1.3	4 Amicus curiae submissions	14	
1.3	5 Preliminary ruling	14	
1.3	6 Request under Article 13 of the DSU	15	
2	FACTUAL ASPECTS 1	5	
2.1	Measures at issue	15	
2.2	Products at issue	16	
3	PARTIES' REQUESTS FOR FINDINGS AND RECOMMENDATIONS 1	6	
3.1	Canada	16	
3.2	Norway	17	
3.3	European Union	17	
4	ARGUMENTS OF THE PARTIES 1	8	
5	ARGUMENTS OF THE THIRD PARTIES 1	8	
6	INTERIM REVIEW 1	8	
6.1	General comments	18	
6.2	Preliminary question on commercial seal hunts (Section 7.3.2.3.2)	18	
6.3	Specific comments on other parts of the Reports	22	
6.4 Inte	Additional documents submitted by the European Union with its comments on the prime Reports	24	
7	FINDINGS	25	
7.1	Overview of the dispute	25	
7.2	Preliminary matters	26	
7.2	1 Description of the measures at issue	26	
7.2	1.1 The Basic Regulation	27	
7.2	1.2 The Implementing Regulation	28	
7.2	2 Consideration of the measures at issue	29	
7.2	2.1 Single or multiple measure(s)	29	
7.2	2.2 Characterization of the measure at issue	30	
7.2	7.2.2.2.1 Main arguments of the parties		
7.2.2.2.2 Analysis by the Panel			
7.2.3 Order of analysis			
	3 Order of analysis	35	
7.2			

7.2.4 Norway's request under Article 13 of the DSU
7.3 Claims under the TBT Agreement40
7.3.1 Whether the EU Seal Regime is a technical regulation within the meaning of the TBT Agreement40
7.3.1.1 Whether the EU Seal Regime lays down one or more characteristics of the products or their related PPMs, including applicable administrative provisions40
7.3.1.1.1 Main arguments of the parties40
7.3.1.1.2 Analysis by the Panel
7.3.1.2 Whether the EU Seal Regime applies to an identifiable product or group of products
7.3.1.3 Whether compliance with the EU Seal Regime is mandatory
7.3.2 Canada's claim under Article 2.147
7.3.2.1 Like products
7.3.2.2 Whether the EU Seal Regime causes a detrimental impact on imported products49
7.3.2.2.1 Main argument of the parties
7.3.2.2.2 Analysis by the Panel
7.3.2.3 Whether the detrimental impact caused by the EU Seal Regime "stems exclusively from legitimate regulatory distinctions"
7.3.2.3.1 "Regulatory distinctions" drawn under the EU Seal Regime
7.3.2.3.2 Preliminary question on commercial seal hunts
7.3.2.3.3 Whether the distinction between commercial hunts and IC hunts is legitimate83
7.3.2.3.4 Whether the distinction between commercial hunts and MRM hunts is legitimate
7.3.3 Article 2.2
7.3.3.1 Identification of the objective(s) pursued through the EU Seal Regime
7.3.3.1.1 Main arguments of the parties110
7.3.3.1.2 Analysis by the Panel
7.3.3.2 Legitimacy of the identified objective
7.3.3.2.1 Main arguments of the parties124
7.3.3.2.2 Analysis by the Panel
7.3.3.3 Whether the EU Seal Regime is more trade restrictive than necessary to fulfil a legitimate objective, taking account of the risks non-fulfilment would create
7.3.3.3.1 Whether the EU Seal Regime is trade restrictive
7.3.3.3.2 Degree of the EU Seal Regime's fulfilment of the identified objective
7.3.3.3.3 Risks non-fulfilment of the objective would create
7.3.3.3.4 Whether a less trade-restrictive alternative measure is reasonably available, taking account of the risks non-fulfilment would create
7.3.3.3.5 Conclusion
7.3.4 Article 5
7.3.4.1 Conformity assessment procedure(s) (CAP): whether the EU Seal Regime is a CAP within the meaning of Annex 1
7.3.4.2 Article 5.1.2: whether the CAP creates an unnecessary obstacle to international trade

- 3 -

7.3.4.2.1 Whether the CAP creates an unnecessary obstacle to trade by failing to ensure the existence of a body to perform the CAP
7.3.4.2.2 Whether the CAP creates an unnecessary obstacle to trade because it is "more strict or applied more strictly than is necessary to give the importing Member adequate confidence" of conformity
7.3.4.3 Article 5.2.1: whether the EU Seal Regime is "undertaken and completed as expeditiously as possible"
7.3.4.3.1 Main argument of the parties154
7.3.4.3.2 Analysis by the Panel
7.4 Non-discrimination claims under the GATT 1994159
7.4.1 Relationship between the GATT 1994 and the TBT Agreement
7.4.2 Article I:1
7.4.2.1 Main arguments of the parties161
7.4.2.2 Analysis by the Panel
7.4.3 Article III:4
7.4.3.1 Main arguments of the parties
7.4.3.2 Analysis by the Panel
7.4.4 The European Union's justification of the EU Seal Regime under Article XX
7.4.4.1 Introduction
7.4.4.2 Aspects of the EU Seal Regime to be considered
7.4.4.2.1 Main arguments of the parties166
7.4.4.2.2 Analysis by the Panel
7.4.4.3 Paragraph (a) of Article XX169
7.4.4.3.1 Main arguments of the parties169
7.4.4.3.2 Analysis by the Panel
7.4.4.4 Paragraph (b) of Article XX172
7.4.4.5 Chapeau of Article XX
7.4.4.5.1 Main arguments of the parties172
7.4.4.5.2 Analysis by the Panel
7.5 Claims under Article XI:1 of the GATT 1994 and Article 4.2 of the Agreement on Agriculture
7.5.1 Article XI:1 of the GATT 1994
7.5.1.1 Summary of the parties' arguments
7.5.1.2 Analysis by the Panel
7.5.2 Article 4:2 of the Agreement on Agriculture (Norway)
7.6 Non-violation claim under Article XXIII:1(b) of the GATT 1994
7.6.1 Main arguments of the parties
7.6.2 Analysis by the Panel
8 CONCLUSIONS AND RECOMMENDATIONS
8.1 Complaint by Canada (DS400): Conclusions and recommendationsCAN-183
8.1 Complaint by Norway (DS401): Conclusions and recommendationsNOR-185

- 5 -

LIST OF ANNEXES

ANNEX A

WORKING PROCEDURES OF THE PANEL

Contents		
Annex A-1	Working Procedures of the Panel	A-2

ANNEX B

ARGUMENTS OF THE PARTIES

	Contents	Page
Annex B-1	Integrated executive summary of the arguments of Canada	B-2
Annex B-2	Integrated executive summary of the arguments of Norway	B-24
Annex B-3	Integrated executive summary of the arguments of the European Union	B-40

ANNEX C

ARGUMENTS OF THE THIRD PARTIES

	Contents	Page
Annex C-1	Integrated executive summary of the arguments of Colombia	C-2
Annex C-2	Integrated executive summary of the arguments of Iceland	C-5
Annex C-3	Integrated executive summary of the arguments of Japan	C-6
Annex C-4	Integrated executive summary of the arguments of Mexico	C-12
Annex C-5	Integrated executive summary of the arguments of Namibia	C-17
Annex C-6	Integrated executive summary of the arguments of the United States	C-22

CASES CITED IN THESE REPORTS

Short Title	Full Case Title and Citation
Argentina – Textiles and Apparel	Appellate Body Report, <i>Argentina – Measures Affecting Imports of Footwear, Textiles, Apparel and Other Items</i> , WT/DS56/AB/R and Corr.1, adopted 22 April 1998, DSR 1998:III, p. 1003
Argentina – Textiles and Apparel	Panel Report, Argentina – Measures Affecting Imports of Footwear, Textiles, Apparel and Other Items, WT/DS56/R, adopted 22 April 1998, as modified by Appellate Body Report WT/DS56/AB/R, DSR 1998:III, p. 1033
Brazil – Retreaded Tyres	Appellate Body Report, <i>Brazil – Measures Affecting Imports of Retreaded Tyres</i> , WT/DS332/AB/R, adopted 17 December 2007, DSR 2007:IV, p. 1527
Brazil – Retreaded Tyres	Panel Report, <i>Brazil – Measures Affecting Imports of Retreaded Tyres</i> , WT/DS332/R, adopted 17 December 2007, as modified by Appellate Body Report WT/DS332/AB/R, DSR 2007:V, p. 1649
Canada – Aircraft	Appellate Body Report, <i>Canada – Measures Affecting the Export of Civilian Aircraft</i> , WT/DS70/AB/R, adopted 20 August 1999, DSR 1999:III, p. 1377
Canada – Aircraft	Panel Report, <i>Canada – Measures Affecting the Export of Civilian Aircraft</i> , WT/DS70/R, adopted 20 August 1999, upheld by Appellate Body Report WT/DS70/AB/R, DSR 1999:IV, p. 1443
Canada – Autos	Appellate Body Report, <i>Canada – Certain Measures Affecting the Automotive Industry</i> , WT/DS139/AB/R, WT/DS142/AB/R, adopted 19 June 2000, DSR 2000:VI, p. 2985
Canada – Autos	Panel Report, <i>Canada – Certain Measures Affecting the Automotive Industry</i> , WT/DS139/R, WT/DS142/R, adopted 19 June 2000, as modified by Appellate Body Report WT/DS139/AB/R, WT/DS142/AB/R, DSR 2000:VII, p. 3043
Canada – Wheat Exports and Grain Imports	Appellate Body Report, <i>Canada – Measures Relating to Exports of Wheat and Treatment of Imported Grain</i> , WT/DS276/AB/R, adopted 27 September 2004, DSR 2004:VI, p. 2739
Canada – Wheat Exports and Grain Imports	Panel Report, <i>Canada – Measures Relating to Exports of Wheat and Treatment of Imported Grain</i> , WT/DS276/R, adopted 27 September 2004, upheld by Appellate Body Report WT/DS276/AB/R, DSR 2004:VI, p. 2817
China – Publications and Audiovisual Products	Appellate Body Report, <i>China – Measures Affecting Trading Rights and Distribution Services for Certain Publications and Audiovisual Entertainment Products</i> , WT/DS363/AB/R, adopted 19 January 2010, DSR 2010:I, p. 3
China – Publications and Audiovisual Products	Panel Report, <i>China – Measures Affecting Trading Rights and Distribution</i> <i>Services for Certain Publications and Audiovisual Entertainment Products</i> , WT/DS363/R and Corr.1, adopted 19 January 2010, as modified by Appellate Body Report WT/DS363/AB/R, DSR 2010:II, p. 261
China – Raw Materials	Appellate Body Reports, <i>China – Measures Related to the Exportation of Various Raw Materials</i> , WT/DS394/AB/R / WT/DS395/AB/R / WT/DS398/AB/R, adopted 22 February 2012
China – Raw Materials	Panel Reports, <i>China – Measures Related to the Exportation of Various Raw</i> <i>Materials</i> , WT/DS394/R / WT/DS395/R / WT/DS398/R / and Corr.1, adopted 22 February 2012, as modified by Appellate Body Reports WT/DS394/AB/R / WT/DS395/AB/R / WT/DS398/AB/R
Colombia – Ports of Entry	Panel Report, <i>Colombia – Indicative Prices and Restrictions on Ports of Entry</i> , WT/DS366/R and Corr.1, adopted 20 May 2009, DSR 2009:VI, p. 2535
EC – Approval and Marketing of Biotech Products	Panel Reports, European Communities – Measures Affecting the Approval and Marketing of Biotech Products, WT/DS291/R / WT/DS292/R / WT/DS293/R, Add.1 to Add.9, and Corr.1, adopted 21 November 2006, DSR 2006:III, p. 847
EC – Asbestos	Appellate Body Report, <i>European Communities – Measures Affecting</i> <i>Asbestos and Asbestos-Containing Products</i> , WT/DS135/AB/R, adopted 5 April 2001, DSR 2001:VII, p. 3243

WT/DS400/R • WT/DS401/R

- 7 -

Short Title	Full Case Title and Citation
EC – Asbestos	Panel Report, European Communities – Measures Affecting Asbestos and Asbestos-Containing Products, WT/DS135/R and Add.1, adopted 5 April 2001, as modified by Appellate Body Report WT/DS135/AB/R, DSR 2001:VIII, p. 3305
EC – Bananas III	Appellate Body Report, <i>European Communities – Regime for the Importation, Sale and Distribution of Bananas</i> , WT/DS27/AB/R, adopted 25 September 1997, DSR 1997:II, p. 591
EC – Bananas III (Ecuador)	Panel Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas, Complaint by Ecuador, WT/DS27/R/ECU, adopted 25 September 1997, as modified by Appellate Body Report WT/DS27/AB/R, DSR 1997:III, p. 1085
EC – Bananas III (Guatemala and Honduras)	Panel Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas, Complaint by Guatemala and Honduras, WT/DS27/R/GTM, WT/DS27/R/HND, adopted 25 September 1997, as modified by Appellate Body Report WT/DS27/AB/R, DSR 1997:II, p. 695
EC – Bananas III (Mexico)	Panel Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas, Complaint by Mexico, WT/DS27/R/MEX, adopted 25 September 1997, as modified by Appellate Body Report WT/DS27/AB/R, DSR 1997:II, p. 803
EC – Bananas III (US)	Panel Report, European Communities – Regime for the Importation, Sale and Distribution of Bananas, Complaint by the United States, WT/DS27/R/USA, adopted 25 September 1997, as modified by Appellate Body Report WT/DS27/AB/R, DSR 1997:II, p. 943
EC – Sardines	Appellate Body Report, <i>European Communities – Trade Description of Sardines</i> , WT/DS231/AB/R, adopted 23 October 2002, DSR 2002:VIII, p. 3359
EC – Sardines	Panel Report, <i>European Communities – Trade Description of Sardines</i> , WT/DS231/R and Corr.1, adopted 23 October 2002, as modified by Appellate Body Report WT/DS231/AB/R, DSR 2002:VIII, p. 3451
EC – Tariff Preferences	Panel Report, <i>European Communities – Conditions for the Granting of Tariff</i> <i>Preferences to Developing Countries</i> , WT/DS246/R, adopted 20 April 2004, as modified by Appellate Body Report WT/DS246/AB/R, DSR 2004:III, p. 1009
EC – Trademarks and Geographical Indications (Australia)	Panel Report, European Communities – Protection of Trademarks and Geographical Indications for Agricultural Products and Foodstuffs, Complaint by Australia, WT/DS290/R, adopted 20 April 2005, DSR 2005:X, p. 4603
India – Autos	Panel Report, India – Measures Affecting the Automotive Sector, WT/DS146/R, WT/DS175/R and Corr.1, adopted 5 April 2002, DSR 2002:V, p. 1827
Japan – Agricultural Products II	Appellate Body Report, <i>Japan – Measures Affecting Agricultural Products</i> , WT/DS76/AB/R, adopted 19 March 1999, DSR 1999:I, p. 277
Japan – Agricultural Products II	Panel Report, <i>Japan – Measures Affecting Agricultural Products</i> , WT/DS76/R, adopted 19 March 1999, as modified by Appellate Body Report WT/DS76/AB/R, DSR 1999:I, p. 315
Japan – Alcoholic Beverages II	Appellate Body Report, <i>Japan – Taxes on Alcoholic Beverages</i> , WT/DS8/AB/R, WT/DS10/AB/R, WT/DS11/AB/R, adopted 1 November 1996, DSR 1996:I, p. 97
Japan – Alcoholic Beverages II	Panel Report, Japan – Taxes on Alcoholic Beverages, WT/DS8/R, WT/DS10/R, WT/DS11/R, adopted 1 November 1996, as modified by Appellate Body Report WT/DS8/AB/R, WT/DS10/AB/R, WT/DS11/AB/R, DSR 1996:I, p. 125
Japan – Film	Panel Report, Japan – Measures Affecting Consumer Photographic Film and Paper, WT/DS44/R, adopted 22 April 1998, DSR 1998:IV, p. 1179
Korea – Various Measures on Beef	Appellate Body Report, <i>Korea – Measures Affecting Imports of Fresh, Chilled and Frozen Beef</i> , WT/DS161/AB/R, WT/DS169/AB/R, adopted 10 January 2001, DSR 2001:I, p. 5

WT/DS400/R • WT/DS401/R

- 8 -

Short Title	Full Case Title and Citation
Korea – Various Measures on Beef	Panel Report, <i>Korea – Measures Affecting Imports of Fresh, Chilled and Frozen Beef</i> , WT/DS161/R, WT/DS169/R, adopted 10 January 2001, as modified by Appellate Body Report WT/DS161/AB/R, WT/DS169/AB/R, DSR 2001:I, p. 59
Thailand – Cigarettes (Philippines)	Appellate Body Report, <i>Thailand – Customs and Fiscal Measures on Cigarettes from the Philippines</i> , WT/DS371/AB/R, adopted 15 July 2011, DSR 2011:IV, p. 2203
Thailand – Cigarettes (Philippines)	Panel Report, <i>Thailand – Customs and Fiscal Measures on Cigarettes from the Philippines</i> , WT/DS371/R, adopted 15 July 2011, as modified by Appellate Body Report WT/DS371/AB/R, DSR 2011:IV, p. 2299
US – 1916 Act (EC)	Panel Report, United States – Anti-Dumping Act of 1916, Complaint by the European Communities, WT/DS136/R and Corr.1, adopted 26 September 2000, upheld by Appellate Body Report WT/DS136/AB/R, WT/DS162/AB/R, DSR 2000:X, p. 4593
US – Clove Cigarettes	Appellate Body Report, United States – Measures Affecting the Production and Sale of Clove Cigarettes, WT/DS406/AB/R, adopted 24 April 2012
US – Clove Cigarettes	Panel Report, <i>United States – Measures Affecting the Production and Sale of Clove Cigarettes</i> , WT/DS406/R, adopted 24 April 2012, as modified by Appellate Body Report WT/DS406/AB/R
US – Continued Zeroing	Appellate Body Report, <i>United States – Continued Existence and Application of Zeroing Methodology</i> , WT/DS350/AB/R, adopted 19 February 2009, DSR 2009:III, p. 1291
US – Continued Zeroing	Panel Report, United States – Continued Existence and Application of Zeroing Methodology, WT/DS350/R, adopted 19 February 2009, as modified as Appellate Body Report WT/DS350/AB/R, DSR 2009:III, p. 1481
US – COOL	Appellate Body Reports, <i>United States – Certain Country of Origin Labelling</i> (<i>COOL</i>) Requirements, WT/DS384/AB/R / WT/DS386/AB/R, adopted 23 July 2012
US – COOL	Panel Reports, United States – Certain Country of Origin Labelling (COOL) Requirements, WT/DS384/R / WT/DS386/R, adopted 23 July 2012, as modified by Appellate Body Reports WT/DS384/AB/R / WT/DS386/AB/R
US – Gambling	Appellate Body Report, <i>United States – Measures Affecting the Cross-Border Supply of Gambling and Betting Services</i> , WT/DS285/AB/R, adopted 20 April 2005, DSR 2005:XII, p. 5663 (Corr.1, DSR 2006:XII, p. 5475)
US – Gambling	Panel Report, United States – Measures Affecting the Cross-Border Supply of Gambling and Betting Services, WT/DS285/R, adopted 20 April 2005, as modified by Appellate Body Report WT/DS285/AB/R, DSR 2005:XII, p. 5797
US – Gasoline	Appellate Body Report, <i>United States – Standards for Reformulated and Conventional Gasoline</i> , WT/DS2/AB/R, adopted 20 May 1996, DSR 1996:I, p. 3
US – Gasoline	Panel Report, <i>United States – Standards for Reformulated and Conventional Gasoline</i> , WT/DS2/R, adopted 20 May 1996, as modified by Appellate Body Report WT/DS2/AB/R, DSR 1996:I, p. 29
US – Large Civil Aircraft (2 nd complaint)	Appellate Body Report, United States – Measures Affecting Trade in Large Civil Aircraft (Second Complaint), WT/DS353/AB/R, adopted 23 March 2012
US – Large Civil Aircraft (2 nd complaint)	Panel Report, United States – Measures Affecting Trade in Large Civil Aircraft (Second Complaint), WT/DS353/R, adopted 23 March 2012, as modified by Appellate Body Report WT/DS353/AB/R
US – Section 337 Tariff Act	GATT Panel Report, United States Section 337 of the Tariff Act of 1930, L/6439, adopted 7 November 1989, BISD 36S/345
US – Shrimp	Appellate Body Report, <i>United States – Import Prohibition of Certain Shrimp and Shrimp Products</i> , WT/DS58/AB/R, adopted 6 November 1998, DSR 1998:VII, p. 2755
US – Shrimp	Panel Report, United States – Import Prohibition of Certain Shrimp and Shrimp Products, WT/DS58/R and Corr.1, adopted 6 November 1998, as modified by Appellate Body Report WT/DS58/AB/R, DSR 1998:VII, p. 2821

WT/DS400/R • WT/DS401/R

- 9 -

Short Title	Full Case Title and Citation
US – Shrimp (Article 21.5 – Malaysia)	Appellate Body Report, <i>United States – Import Prohibition of Certain Shrimp</i> <i>and Shrimp Products – Recourse to Article 21.5 of the DSU by Malaysia</i> , WT/DS58/AB/RW, adopted 21 November 2001, DSR 2001:XIII, p. 6481
US – Shrimp (Article 21.5 – Malaysia)	Panel Report, United States – Import Prohibition of Certain Shrimp and Shrimp Products – Recourse to Article 21.5 of the DSU by Malaysia, WT/DS58/RW, adopted 21 November 2001, upheld by Appellate Body Report WT/DS58/AB/RW, DSR 2001:XIII, p. 6529
US – Tuna II (Mexico)	Appellate Body Report, United States – Measures Concerning the Importation, Marketing and Sale of Tuna and Tuna Products, WT/DS381/AB/R, adopted 13 June 2012
US – Tuna II (Mexico)	Panel Report, United States – Measures Concerning the Importation, Marketing and Sale of Tuna and Tuna Products, WT/DS381/R, adopted 13 June 2012, as modified by Appellate Body Report WT/DS381/AB/R
US – Upland Cotton	Appellate Body Report, United States – Subsidies on Upland Cotton, WT/DS267/AB/R, adopted 21 March 2005, DSR 2005:I, p. 3
US – Upland Cotton	Panel Report, United States – Subsidies on Upland Cotton, WT/DS267/R, Corr.1, and Add.1 to Add.3, adopted 21 March 2005, as modified by Appellate Body Report WT/DS267/AB/R, DSR 2005:II, p. 299
US – Zeroing (EC) (Article 21.5 – EC)	Appellate Body Report, United States – Laws, Regulations and Methodology for Calculating Dumping Margins ("Zeroing") – Recourse to Article 21.5 of the DSU by the European Communities, WT/DS294/AB/RW and Corr.1, adopted 11 June 2009, DSR 2009:VII, p. 2911
US – Zeroing (EC) (Article 21.5 – EC)	Panel Report, United States – Laws, Regulations and Methodology for Calculating Dumping Margins ("Zeroing") – Recourse to Article 21.5 of the DSU by the European Communities, WT/DS294/RW, adopted 11 June 2009, as modified by Appellate Body Report WT/DS294/AB/RW, DSR 2009:VII, p. 3117

- 10 -

LIST OF FREQUENTLY CITED EXHIBITS

Exhibit	Short Title	Full Title
JE-1	Basic Regulation	Regulation (EC) No. 1007/2009 of the European Parliament and of the Council on trade in seal products, adopted on 16 September 2009
JE-2	Implementing Regulation	Commission Regulation (EU) No. 737/2010 laying down detailed rules for the implementation of the Basic Regulation, adopted on 10 August 2010
JE-20	COWI 2008 Report	COWI, Assessment of the Potential Impact of a Ban of Products Derived from Seal Species (April 2008)
JE-21	COWI 2010 Report	COWI, Study on Implementing Measures for Trade in Seal Products, Final Report (January 2010)
JE-22	EFSA Scientific Opinion	European Food Safety Authority ("EFSA"), Panel on Animal Health and Welfare, <i>Scientific Opinion on Animal Welfare Aspects of the</i> <i>Killing and Skinning of Seals</i> , The EFSA Journal (2007) 610, pp. 1-122 (6 December 2007)
JE-24	NAMMCO Report (2009)	NAMMCO Expert Group, Report on the Meeting on Best Practices in the Hunting and Killing of Seals (February 2009)
JE-31	VKM Scientific Opinion	Norwegian Scientific Committee for Food Safety ("VKM"), Panel on Animal Health and Welfare, <i>Scientific Opinion on Animal Welfare</i> <i>Aspects of the Killing and Skinning in the Norwegian Seal Hunt</i> (8 October 2007)
CDA-33	IVWG Report (2005)	Smith, B. et al., "Improving Humane Practice in the Canadian Harp Seal Hunt: A Report of the Independent Veterinarians' Working Group on the Canadian Harp Seal Hunt" (August 2005)
CDA-34	Daoust (2012)	Daoust, P-Y., and Caraguel, C., "The Canadian harp seal hunt: observations on the effectiveness of procedures to avoid poor animal welfare outcomes", <i>Animal Welfare</i> , vol. 21, pp. 445-455 (2012)
EU-31	Burdon (2001)	Burdon, R.L. et al., "Veterinary report: Canadian commercial seal hunt, Prince Edward Island" (March 2001)
EU-32	Daoust (2002)	Daoust, PY., Crook, A., Bollinger, T.K., Campbell, K.G. and Wong G., "Animal Welfare and the harp seal hunt in Atlantic Canada", <i>Canadian Veterinary Journal</i> , vol. 43, pp. 687-694 (2002)
EU-34	Butterworth (2007)	Butterworth, A. et al., "Welfare aspects of the Canadian seal hunt" (31 August 2007)
EU-36	Richardson (2007)	Richardson, M., Inherently Inhumane (August 2007)
EU-37	Butterworth (2012)	Butterworth A., Richardson M., "A Review of animal welfare implications of the commercial Canadian seal hunt", <i>Marine Policy</i> (2012)
EU-43	NOAH Report (2012)	Martinsen, S., <i>Sealing in Norway – Welfare Aspects, report for NOAH</i> , (6 December 2012)

- 11 -

ABBREVIATIONS USED IN THESE REPORTS

Abbreviation	Description
ЗРСАР	Third party conformity assessment
САР	Conformity assessment procedure(s)
CN	Combined Nomenclature
DSB	Dispute Settlement Body
DSU	Understanding on Rules and Procedures Governing the Settlement of Disputes
EFSA	European Food Safety Authority
EU Seal Regime	The Basic Regulation and the Implementing Regulation combined together
GATT 1994	General Agreement on Tariffs and Trade 1994
IC	Inuit or other indigenous communities
NGOs	Non-governmental organizations
MFN	Most-favoured nation
MRM	Marine resource management
OIE	Office International des Epizooties
PPMs	Processes and production methods
SPS Agreement	Agreement on the Application of Sanitary and Phytosanitary Measures
TBT Agreement	Agreement on Technical Barriers to Trade
TAC	Total allowable catch
Vienna Convention	Vienna Convention on the Law of Treaties, Done at Vienna, 23 May 1969, 1155 UNTS 331; 8 International Legal Materials 679
WTO	World Trade Organization

1 INTRODUCTION

1.1 Complaints by Canada and Norway

1.1. On 2 November 2009, Canada requested consultations with the European Union¹ pursuant to Article 4 of the Understanding on Rules and Procedures Governing the Settlement of Disputes (DSU), Article XXII of the General Agreement on Tariffs and Trade (GATT 1994), and Article 14.1 of the Agreement on Technical Barriers to Trade (TBT Agreement), with respect to the measures and claims set out below.² On 18 October 2010, Canada requested supplementary consultations with the European Union.³

1.2. On 5 November 2009, Norway requested consultations with the European Union pursuant to Article 4 of the DSU, Article XXII of the GATT 1994, Article 14 of the TBT Agreement, and Article 19 of the Agreement on Agriculture, with respect to the measures and claims set out below.⁴ On 20 November 2009, Canada requested, pursuant to Article 4.11 of the DSU, to join in the consultations requested by Norway on 5 November 2009.⁵ On 19 October 2010, Norway requested supplementary consultations with the European Union.⁶

1.3. On 28 and 29 October 2010, respectively, Canada and Norway requested to join each other's supplementary consultations.⁷

1.4. Consultations were held on 15 December 2009, and supplementary consultations were held on 1 December 2010. None of these consultations led to a mutually satisfactory resolution.⁸

1.2 Panel establishment and composition

1.5. On 11 February and 14 March 2011, respectively, Canada and Norway requested the establishment of a panel pursuant to Article 6 of the DSU with standard terms of reference.⁹

1.6. At its meeting on 25 March 2011, the Dispute Settlement Body (DSB) established a panel pursuant to the request of Canada in document WT/DS400/4, in accordance with Article 6 of the DSU.¹⁰ At its meeting on 21 April 2011, the DSB established a panel pursuant to the request of Norway in document WT/DS401/5, in accordance with Article 6 of the DSU, and agreed, as provided for in Article 9 of the DSU in respect of multiple complainants, that the panel established to examine the complaint by Canada would also examine the complaint by Norway.¹¹

1.7. The Panel's terms of reference are the following:

To examine, in the light of the relevant provisions of the covered agreements cited by the parties to the dispute, the matter referred to the DSB by Canada in document WT/DS400/4, and by Norway in document WT/DS401/5, and to make such findings as will assist the DSB in making the recommendations or in giving the rulings provided for in those agreements.¹²

1.8. On 24 September 2012, Canada and Norway requested the Director-General to determine the composition of the panel, pursuant to Article 8.7 of the DSU.

¹ For consistency and ease of reference, these Reports will refer to "the European Union" or "EU" for all events regardless of their date of occurrence.

² WT/DS400/1, Canada's request for consultations.

³ WT/DS400/1/Add. 1, Canada's request for consultations. Canada made its supplemental request pursuant to the same provisions as its original request for consultations with the addition of Article 1 of the DSU, and Article 14.1 of the TBT Agreement instead of Article 14 of the TBT Agreement.

⁴ WT/DS401/1, Norway's request for consultations.

⁵ WT/DS401/3.

⁶ WT/DS401/1/Add. 1, Norway's request for consultations.

⁷ WT/DS400/3 and WT/DS401/4.

⁸ WT/DS400/4, Canada's request for the establishment of a panel; WT/DS401/5, Norway's request for the establishment of a panel.

⁹ Canada's request for the establishment of a panel; Norway's request for the establishment of a panel. ¹⁰ See WT/DSB/M/294, para. 73.

¹¹ See WT/DSB/M/295, para. 73, WT/DS400/5 and WT/DS401/6.

¹² WT/DS400/5 and WT/DS401/6.

1.9. On 4 October 2012, the Director-General accordingly composed the Panel as follows:

Chairperson:	Mr Luzius Wasescha
Members:	Ms Elizabeth Chelliah Ms Patricia Holmes

1.10. Argentina, Canada (for WT/DS401), China, Colombia, Ecuador, Iceland, Japan, Mexico, Namibia (for WT/DS401), Norway (for WT/DS400), the Russian Federation¹³, and the United States notified their interest in participating in the Panel proceedings as third parties.

1.3 Panel proceedings

1.3.1 General

1.11. After consultation with the parties, the Panel adopted its Working Procedures¹⁴ and timetable on 23 October 2012. Upon request of the parties, the Panel modified the timetable on 4 March 2013 and 8 May 2013.

1.12. The Panel held a first substantive meeting with the parties on 18-20 February 2013. A session with the third parties took place on 19 February 2013. The Panel held a second substantive meeting with the parties on 29-30 April 2013.

1.13. On 19 June 2013, the Panel issued the descriptive part of its Reports to the parties. The Panel issued its Interim Reports to the parties on 3 September 2013. The Panel issued its Final Reports to the parties on 8 October 2013.

1.3.2 Procedures for open hearings

1.14. At the organizational meeting held on 15 October 2012, the parties requested and the Panel agreed that the substantive meetings with the Panel would be open to public viewing subject to additional procedures to ensure the security and orderly conduct of the proceedings. On 4 December 2012, the Panel submitted proposed additional working procedures to the parties for comment. After it had received comments from the parties, the Panel adopted on 20 December 2012 additional Working Procedures for its open hearings at the first and second substantive meetings of the Panel, providing for public viewing by means of simultaneous closed-circuit television broadcasting of the proceedings to a separate room.

1.3.3 Requests for enhanced third-party rights

1.15. At the organizational meeting held on 15 October 2012, Canada made a request for enhanced third-party rights to allow third-party access to both substantive meetings and all written submissions. The European Union objected to Canada's request on the grounds that no third party had submitted such a request. After considering Canada's request and the views of the other parties, the Panel informed the parties on 23 October 2012 that it had decided to decline Canada's request. In reaching its decision, the Panel took particular note of the fact that Canada's request was made by a party to the dispute, and that no third party had made a request for enhanced rights. Furthermore, because the substantive meetings were to be open to public viewing and thus would serve to provide third-party access to the Panel's substantive meetings, the Panel did not consider it necessary to grant the enhanced third-party rights requested by Canada.

1.16. Following the first substantive meeting with the parties on 18-20 February 2013, the Panel received on 6 March 2013 a request from Namibia "to participate in the second substantive meeting" in order to rebut comments made by the European Union at the first substantive meeting regarding the Namibian seal hunt. After consulting the parties on Namibia's request, the Panel

¹³ On 18 October 2012, the Russian Federation notified its interest to participate as a third party in the dispute. After receiving the parties' views on this notification, the Panel indicated on 5 November 2012 that the Russian Federation would be added to the list of third parties. See also WT/DS400/5/Rev.1 and WT/DS401/6/Rev.1, para. 5.

¹⁴ See the Panel's Working Procedures in Annex A-1.

informed Namibia that it had considered Namibia's request, taking into account Namibia's status as a developing country and the material on the record relating to Namibia, which included Namibia's third-party written submission, oral statement at the first substantive meeting and its written responses to the Panel's questions. On the basis of its review and in light of the parties' comments, the Panel decided that there was no need to provide Namibia with an opportunity for further rebuttal and therefore declined Namibia's request to participate in the second substantive meeting.

1.3.4 *Amicus curiae* submissions

1.17. On 25 January 2013, the Panel received an unsolicited *amicus curiae* submission from a group of non-governmental organizations (NGOs).¹⁵ On 29 January 2013, the Panel notified the parties of the unsolicited *amicus curiae* submission and advised the parties that any *amicus curiae* submission it received would be immediately forwarded to the parties. The parties would be invited to provide their views on the admissibility and relevance of any *amicus curiae* submission either at the first or second substantive meeting. The Panel further informed the parties that any *amicus curiae* brief submitted to the Panel after the second substantive meeting would be automatically rejected, as the Panel was of the view that the consideration of any new information at that stage of the proceedings would risk causing undue delays.

1.18. Subsequently, the Panel received four additional unsolicited *amicus curiae* submissions prior to the second substantive meeting with the parties.¹⁶

1.19. During the first substantive meeting with the parties, the European Union indicated that it had incorporated the *amicus curiae* submission provided by the group of NGOs on 25 January 2013 as an integral part of its written submissions to the Panel.¹⁷

1.3.5 Preliminary ruling

1.20. On 19 December 2012, the European Union filed a request for a preliminary ruling to remove two exhibits from the record. On 16 January 2013, both Canada and Norway responded to the European Union's preliminary ruling request. In addition, the United States provided comments on the European Union's request in its third-party written submission pursuant to the Panel's invitation to all third parties to do so.

1.21. The Panel issued its preliminary ruling to the parties, with a copy to third parties, on 29 January 2013, granting the European Union's request to remove the exhibits from the record and inviting the complainants to submit replacement exhibits. After consulting with the parties, the Panel requested the Chairperson of the DSB to circulate its preliminary ruling to all WTO Members. The Panel further decided that the circulated ruling would be incorporated as an integral part of the Panel's findings in its Reports.¹⁸ The Panel's preliminary ruling was circulated on 5 February 2013 in documents WT/DS400/6 and WT/DS401/7.

¹⁵ This submission was made jointly by a group of the following organizations: Anima, Animal Rights Action Network (ARAN), Animalia, Bont Voor Dieren (BVD), Change for Animals Foundation (CFAF), Compassion in World Farming (CIWF), Djurens Rätt (Animal Rights Sweden), Eurogroup for Animals, Fondation Brigitte Bardot (FFB), Fondation Franz Weber (FFW), Four Paws, Global Action in the Interest of Animals (GAIA), Humane Society of the United States/Humane Society International (HSUS/HSI), International Fund for Animal Welfare (IFAW), Lega Anti Vivisezione (LAV), Prijatelji životinja (Animal Friends Croatia), Respect for Animals, Royal Society for the Prevention of Cruelty to Animals (RSPCA), Svoboda zvířat and World Society for the Protection of Animals (WSPA).

¹⁶ The Panel received submissions from Robert Howse, Joanna Langille, and Katie Sykes, dated 11 February 2013; Pamela Anderson on behalf of People for the Ethical Treatment of Animals (PETA), dated 12 February 2013; the International Fur Trade Federation, dated 28 March 2013; and Jude Law, received 20 April 2013.

¹⁷ This submission is included on the record as Exhibit EU-81.

¹⁸ In its preliminary ruling, the Panel reserved the right to modify its ruling and observed that the ruling would be incorporated as an integral part of the Panel's findings in its modified form if any modifications were made.

1.3.6 Request under Article 13 of the DSU

1.22. On 16 January 2013, the same day that the complainants provided their comments on the European Union's request for preliminary ruling, Norway requested the Panel to exercise its power under Article 13 of the DSU to seek copies of the two documents that were the object of the European Union's request for removal from the record. Further to the Panel's invitation, the Panel received on 8 February 2013 comments from the European Union and Canada on Norway's request, as well as third-party comments from the United States. On 8 April 2013, the Panel informed the parties that the Panel did not consider it necessary to seek the information requested by Norway. Consequently, the Panel denied Norway's request for the Panel to exercise its authority under Article 13 of the DSU and indicated that the Panel would provide the reasons for its decision in its Reports.¹⁹

2 FACTUAL ASPECTS

2.1 Measures at issue²⁰

2.1. The claims brought by Canada and Norway concern the European Union's measures relating to seal products.

- 2.2. Canada submits that the measures at issue are the following²¹:
 - a. Regulation (EC) No. 1007/2009 of the European Parliament and of the Council, of 16 September 2009 on trade in seal products;
 - Regulation (EU) No 737/2010 laying down detailed rules for the implementation of Regulation (EC) No. 1007/2009 of the European Parliament and of the Council on trade in seal products; and
 - c. For each of the measures referred to above, any amendments, replacements, extensions, implementing measures or other related measures, administrative orders, directives, or customs guidelines including those issued by individual European Union member States.

2.3. Canada refers to the Basic Regulation and the Implementing Regulation together as the "EU Seal Regime".

- 2.4. Norway submits that the measures at issue are the following²²:
 - a. Regulation (EC) No. 1007/2009 of the European Parliament and of the Council on trade in seal products, adopted on 16 September 2009 (the "Basic Regulation");
 - Commission Regulation (EU) No. 737/2010, laying down detailed rules for the implementation of the Basic Regulation, adopted on 10 August 2010 (the "Implementing Regulation");
 - c. Omissions to adopt adequate procedures for establishing that seal products conforming to the relevant conditions, set forth in exceptions in the EU seal regime, may be placed on the EU market; and
 - d. Any other related measures adopted by the EU or its member States that provide guidance on, amend, supplement, replace, and/or implement the rules set forth in the Basic Regulation and Implementing Regulation, whether adopted pursuant to these regulations or otherwise.

¹⁹ The reasoning for the Panel's decision is provided in section 7 of these Reports.

²⁰ The Panel's use of the term "measures" in this Section does not prejudge any disputed factual or legal issues relating to that term.

²¹ Canada's request for the establishment of a panel.

²² Norway's request for the establishment of a panel.

- 16 -

2.5. Norway refers to the Basic Regulation and the Implementing Regulation together as the "EU Seal Regime".

2.2 Products at issue

2.6. This dispute concerns products, either processed or unprocessed, deriving or obtained from seals, including meat, oil, blubber, organs, raw fur skins and tanned fur skins, as well as articles (such as clothing and accessories, and omega-3 capsules) made from fur skins and oil.²³

2.7. In accordance with Article 3(3) of Regulation (EC) No. 1007/2009 of the European Parliament and of the Council, the EU Commission issued a "Technical Guidance Note Setting Out an Indicative List of the Codes of the Combined Nomenclature that May Cover Prohibited Seal Products" (Technical Guidance Note).²⁴ The Technical Guidance Note includes products listed according to their Combined Nomenclature (CN) codes. In its foreword, the Note explains that only "those CN codes with the greatest likelihood of covering products subject to prohibition" are included therein, and those listed are indicative.²⁵ For example, it includes the following sections of the CN codes:

- "live animals, animal products" (Section I);
- "animal or vegetable fats and oils and their cleavage products ..." (Section III);
- "prepared foodstuffs, beverages, spirits and vinegar ..." (Section IV);
- "products of the chemical or allied industries" (Section VI);
- "rawhides and skins, leather, fur skins and articles thereof; ... handbags and similar containers" (Section VIII);
- "textiles and textile articles" (Section XI);
- "footwear, headgear ..." (Section XII);
- "... precious metals, metals clad with precious metal, and articles thereof; imitation jewellery ..." (Section XIV);
- "optical, photographic, ..., medical or surgical instruments and apparatus; clocks and watches; musical instruments; parts and accessories thereof" (Section XVIII);
- "miscellaneous manufactured articles" (Section XX); and
- "works of art, collectors' pieces and antiques" (Section XXI).

3 PARTIES' REQUESTS FOR FINDINGS AND RECOMMENDATIONS

3.1 Canada

- 3.1. Canada requests that the Panel find that the "EU Seal Regime":
 - a. is a technical regulation in the sense of Annex 1.1 of the TBT Agreement;
 - b. is inconsistent with the European Union's obligations under the TBT Agreement, in particular Articles 2.1, 2.2, 5.1.2, and 5.2.1;

²³ See Article 2(2) of the Regulation (EC) No. 1007/2009 of the European Parliament and of the Council; Canada's first written submission, paras. 61-70; Norway's first written submission, paras. 86-102.

²⁴ Technical guidance note setting out an indicative list of the codes of the combined nomenclature that may cover prohibited seal products, Official Journal of the European Union, C Series, No. 356

⁽²⁹ December 2010), (Exhibit JE-3).

²⁵ Ibid. p. 44.

- c. is inconsistent with the European Union's obligations under the GATT 1994, in particular Articles I:1, III:4, and XI:1; and
- d. is not justified by Article XX(a) or XX(b) of the GATT 1994.²⁶

3.2. Canada requests, pursuant to Article 19.1 of the DSU, that the Panel recommend to the DSB that it request the European Union to bring its measures into conformity with its obligations under the TBT Agreement and the GATT 1994.²⁷

3.3. In the event that the European Union's measures are not found to violate the European Union's obligations under the TBT Agreement or the GATT 1994, Canada requests that the Panel find that the EU Seal Regime has nullified and impaired benefits accruing to Canada in the sense of Article XXIII:1(b) of the GATT 1994, and that the Panel recommend to the DSB that it request the European Union to make a mutually satisfactory adjustment as required by Article 26.1 of the DSU.²⁸

3.2 Norway

3.4. Norway requests that the Panel find that the "EU Seal Regime":

- a. violates Articles I:1, III:4 and XI:1 of the GATT 1994;
- b. is not justified by Article XX(a) or (b) of the GATT 1994;
- c. violates Article 4.2 of the Agreement on Agriculture;
- d. is a technical regulation in the sense of Annex 1.1 of the TBT Agreement;
- e. violates Articles 2.2, 5.1.2 and 5.2.1 of the TBT Agreement; and
- f. nullifies or impairs benefits accruing to Norway in the sense of Article XXIII:1(b) of the GATT 1994, whether or not it conflicts with relevant provisions.²⁹

3.5. Norway therefore requests the Panel, pursuant to Article 19.1 of the DSU, to recommend that the DSB request that the European Union bring the EU Seal Regime into conformity with the European Union's obligations under the GATT 1994, the TBT Agreement and the Agreement on Agriculture.³⁰

3.6. If, and to the extent, that the Panel finds that the EU Seal Regime does not conflict with relevant WTO provisions, but nonetheless finds that the measures nullify or impair benefits accruing to Norway in the sense of Article XXIII:1(b) of the GATT 1994, Norway requests the Panel to recommend that the DSB request the European Union to make a mutually satisfactory adjustment as required by Article 26.1 of the DSU.³¹

3.3 European Union

3.7. The European Union requests the Panel to reject all the claims submitted by Canada and Norway against the "EU Seal Regime". 32

²⁹ Norway's first written submission, para. 1039; Norway's second written submission, para. 439.

²⁶ Canada's first written submission, para. 752; Canada's second written submission, para. 359.

²⁷ Canada's first written submission, para. 753; Canada's second written submission, para. 360.

²⁸ Canada's first written submission, para. 754; Canada's second written submission, para. 361.

³⁰ Norway's first written submission, para. 1040; Norway's second written submission, para. 440.

³¹ Norway's first written submission, para. 1041; Norway's second written submission, para. 441.

³² European Union's first written submission, para. 628; European Union's second written submission, para. 387. The European Union refers to its regulations at issue in this dispute as the "EU Seal Regime". (See European Union's first written submission, para. 1).

- 18 -

4 ARGUMENTS OF THE PARTIES

4.1. The arguments of the parties are reflected in their executive summaries, provided to the Panel in accordance with paragraph 19 of the Working Procedures adopted by the Panel (see Annexes B-1, B-2 and B-3).

5 ARGUMENTS OF THE THIRD PARTIES

5.1. The arguments of Colombia, Iceland, Japan, Mexico, Namibia and the United States are reflected in their executive summaries, provided in accordance with paragraph 20 of the Working Procedures adopted by the Panel (see Annexes C-1, C-2, C-3, C-4, C-5, and C-6). Argentina, China, Ecuador and the Russian Federation did not submit written or oral arguments to the Panel.

6 INTERIM REVIEW

6.1. On 3 September 2013, the Panel submitted its Interim Panel Reports to the parties. On 17 September 2013, Canada, Norway, and the European Union each submitted written requests for the review of precise aspects of the Interim Reports. On 24 September 2013, Canada, Norway, and the European Union submitted comments on a number of requests for review presented by the other parties. None of the parties requested an interim review meeting.

6.2. In accordance with Article 15.3 of the DSU, this section of the Panel Reports sets out the Panel's response to the arguments made at the interim review stage, providing explanations where necessary. The Panel has modified aspects of its reports in light of the parties' comments where it considered it appropriate to do so, as explained below. The Panel has also made certain technical and editorial corrections and revisions to the Interim Panel Reports for the purposes of clarity and accuracy. References to sections, paragraph numbers, and footnotes in this section relate to the Interim Panel Reports, except as otherwise noted.

6.1 General comments

Reference to "Greenland"

6.3. The complainants observed that the Interim Reports variously refer to "Greenland", "Greenland (Denmark)", and "Denmark (Greenland)" and requested that the Panel follow a consistent approach. Specifically, Canada proposed that the Panel uniformly refer to "Greenland (Denmark)" in the Reports. For its part, the European Union noted that the term "Denmark (Greenland)" does not accurately describe the constitutional relationship between Denmark and Greenland and requested that "Greenland" be used instead.

6.4. The Panel has used the term "Greenland" consistently throughout the Reports.

6.2 Preliminary question on commercial seal hunts (Section 7.3.2.3.2)

6.5. Norway made both general and specific comments regarding the emphasis given to certain aspects of its hunting practices and regulations as well as practices in other seal hunts. Norway expressed concern with the Panel's portrayal and characterization of the risks of inhumane killing in seal hunts. In particular, Norway stated that evidence from the Norwegian hunt of "compliance with humane seal hunting ... is omitted or downplayed" and that, conversely, aspects of other hunts "that demonstrate a failure to mitigate risks of inhumane killing ... are underrepresented".³³

6.6. The Panel addresses below the parties' comments on section 7.3.2.3.2 of the Panel Reports, additionally addressing comments made with respect to similar subjects in other parts of the Reports where relevant to the Panel's review.

Use of the term "commercial hunts"

6.7. Norway requested the Panel to review the nomenclature that it adopts to distinguish seal hunts conforming to the IC and MRM exceptions and those that do not. Norway argued that the

³³ Norway's comments on the Interim Panel Reports, para. 40.

- 19 -

use of the terms "commercial hunts" and "IC and MRM hunts" could be taken "to reflect a moral judgment with respect to the different hunts".³⁴ Norway further argued that "the Panel's own findings demonstrate the falseness of the distinction created by the Panel between 'commercial' and the other types of hunts".³⁵ Norway suggested that the Panel adopt neutral language to reflect the basis for its distinction, such as "non-conforming hunts", or the "Canadian East Coast hunt" and "Norwegian West Ice hunt". Canada suggested that the phrase "non-conforming hunts" be used instead of the term "commercial hunts".

6.8. The European Union disagreed with this request and submitted that, based on the Panel's usage of terms in different parts of its analysis, the use of the term "commercial hunts" is not confusing.

6.9. The Panel determined that "commercial hunts" are those having commercial profit (rather than direct use or consumption of seal products) as the sole or primary objective, along with various other factual characteristics described in section 7.3.2.3.2.2 of the Reports. The Panel further took note of evidence that such hunts are distinctly designated by a variety of sources as "commercial", including within Canada and Norway. Where appropriate and as relevant to its analysis, the Panel noted the existence of a commercial element in IC and MRM hunts. This does not negate other relevant factual characteristics of hunts that conform to the IC and MRM exceptions (such as the identity of the hunter or scale of the hunt). We therefore decline to revise the terminology used in these Reports to distinguish different seal hunts.

Paragraph 7.184

6.10. Norway disputed the assertion that "relatively little information is provided regarding the actual seal hunting conducted in ... sealing countries" other than Norway and Canada. Norway contended that substantial evidence is available dealing with seal hunting in Greenland, and requested that this evidence be noted and fully reflected by the Panel.

6.11. The European Union considered that the referenced passage deals with scientific information concerning the animal welfare aspects of seal hunting. According to the European Union, the documents cited by Norway provide some limited information on certain aspects of seal hunting in Greenland, but the European Union maintained that there is "hardly any scientific evidence concerning the animal welfare outcomes of the IC hunts, including those conducted in Greenland".³⁶

6.12. The Panel recalls EFSA's observations that "[v]ery little robust information is available ... on the efficacy of" different killing methods employed in seal hunts around the world³⁷ and that "the vast majority of available data is from commercial hunts".³⁸ The Panel observes that it was provided with a proportionally greater amount of information in the form of scientific and empirical studies on the Canadian and Norwegian hunts.³⁹ Further, the Panel referred to available material on seal hunting in Greenland *inter alia* in the context of its assessment of the characteristics of IC hunts and the occurrence of struck and lost seals. In light of the parties' comments, the Panel made slight amendments to this paragraph for clarity.

Paragraph 7.188

6.13. Canada requested that the Panel clarify if its conclusion in paragraph 7.188 is that the physical conditions of seal hunting are not only distinct from other wildlife hunts and the commercial slaughter of farmed animals, but also pose challenges that are not present in these other types of hunts. In particular, Canada inquired whether seal hunting is more challenging than other types of wildlife hunts because of the physical conditions in which it is carried out.

Opinion, (Exhibit JE-31); Burdon (2001), (Exhibit EU-31); Daoust (2002), (Exhibit EU-32); and NOAH Report (2012), (Exhibit EU-43).

³⁴ Norway's comments on the Interim Panel Reports, para. 37.

³⁵ Norway's comments on the Interim Panel Reports, para. 38.

³⁶ European Union's comments on the Interim Panel Reports, para. 15.

³⁷ EFSA Scientific Opinion, p. 24.

³⁸ EFSA Scientific Opinion, p. 13.

³⁹ See, e.g. IVWG Report (2005), (Exhibit CDA-33); Daoust (2012), (Exhibit CDA-34); VKM Scientific

6.14. The Panel addressed characteristics of the physical environment of seal hunts in connection with the risks of poor animal welfare arising in that specific context. The Panel also explained the limitations on comparing the risks of poor animal welfare in seal hunts, which have been examined in detail, to those in other wildlife hunts and commercial abattoirs. Therefore, the Panel considers further comparison to the killing of other animals to be unnecessary.

Paragraph 7.191

6.15. Norway commented that other evidence indicating reasons for targeting the seal's head may be more fully reflected in this paragraph, including for purposes of effective stunning and preservation of pelt value. Canada took issue with the citation to the length of the limbs as a reason for targeting the seal's head.

6.16. The reasons mentioned in paragraph 7.191 for targeting the seal's head are not exhaustive, and correspond to the distinct characteristics of seals. The targeting of the head for purposes of effective stunning and preservation of the value of skins is addressed elsewhere in the Reports where appropriate. The Panel slightly amended this paragraph to reflect the referenced material in light of the parties' comments.

Use of the term "clubbing instrument" (Paragraphs 7.193, footnote 262, 7.200, and 7.204)

6.17. Norway requested that the Panel not use the term "clubbing instrument" to describe a wide category of tools including both simple clubs and the hakapik prescribed by Norwegian regulations for the effective stunning of seals. Norway considered that this term does not convey the different features, animal welfare implications, and regulatory treatment of distinct instruments to which it is applied. Accordingly, Norway requested the Panel to replace the term "clubbing instrument" with a more specific reference to the type of instrument to which the Panel refers.

6.18. The Panel employed the term "clubbing" to refer to the physical act of striking, notwithstanding different dimensions and specific features of the tools used, in the same fashion that "shooting" is a general term for the action of employing a firearm, notwithstanding the use of rifles and ammunition of different power. The Panel therefore used the general designation "clubbing instrument" to address those tools with which the action of "clubbing" is carried out, without prejudice to the Panel's observation of the different features of hakapiks and clubs. Relevant sections (e.g. regarding the application of stunning methods) address factors and risks that pertain to the action of clubbing, including the use of hakapiks and slagkroks. The Panel made revisions to clarify that clubs are not permitted as a stunning instrument in Norway, and considers that further changes are not necessary.

Paragraphs 7.197-7.198 and 7.222

6.19. Norway proposed noting that not all seal hunting occurs under regulatory conditions that require application of a humane killing method, specifically referring to the practice of trapping or netting in Greenland, and requested reference to its regulatory efforts to minimize poor animal welfare in seal hunts.

6.20. The European Union argued that it has shown that neither Canada's nor Norway's regulations prescribe genuinely humane killing methods. Therefore, the European Union contended that the Panel should reject Norway's request.

6.21. The Panel notes that, as indicated in the heading preceding paragraphs 7.197-7.198, this sub-section concerns the application of humane killing methods in seal hunts. Consistent with this focus, the Panel considered evidence relating to sealing regulations insofar as it provides insight into the conduct of seal hunts, especially regarding the challenges of applying humane killing methods and the risks of poor animal welfare. The Panel additionally points out that it addressed the history and framework of Norway's seal hunting regulations under the organization and control of commercial seal hunts, which includes reference to Norway's mandatory training of hunters and inspectors. The practice of trapping seals underwater in certain hunts as well as the implications of such practice for animal welfare have been noted in other parts of the Reports as appropriate. Therefore, the Panel does not consider additional references to this evidence to be necessary.

- 21 -

Paragraphs 7.216, 7.236, 7.268, 7.273, and 7.333

6.22. Norway made various comments concerning hunting regulations and practices within Norway and elsewhere. Norway requested that the Panel include reference in paragraph 7.216 to the prohibition in its regulations of shooting seals in water, and further requested the Panel to clarify the link between shooting seals in open water and the struck and lost rates in the Greenlandic hunt. Norway also requested clarification of its regulatory scheme in paragraph 7.236, particularly the prohibitions against the use of clubs, nets, as well as the shooting of seals in water. Norway requested explicit reference in paragraph 7.268 to the animal welfare problems related to open water hunting and trapping and netting. Norway requested revision of paragraph 7.273, which it considered to imply that "the IC hunts described by the Panel are 'no different' than other seal hunts such as the Norwegian and Canadian commercial hunts are not "similar", specifically citing its prohibition against the use of nets and shooting seals in water. Finally, Norway requested modification of paragraph 7.333 to reflect that clubs are prohibited in Norway.

6.23. The European Union commented that the fact that Norway's regulations prohibit shooting seals in water does not imply that struck and lost is not a problem in the Norwegian hunt. The European Union added that shooting seals near water is not prohibited, and that the Panel should specify that shooting seals in water is allowed in Canada's commercial hunt. With respect to paragraph 7.273, the European Union considered that the difference highlighted by Norway is already mentioned and therefore no amendment is needed.

6.24. The Panel explained its reference to and assessment of seal hunting regulations in connection with Norway's comments on paragraphs 7.197-7.198 and 7.222. In light of the parties' comments, the Panel added reference to the Norwegian prohibition on shooting seals in water in footnote 324 of the Panel Reports as well as a cross-reference in footnote 329 of the Panel Reports to the Panel's discussion of hunting methods in Greenland. The Panel has also modified paragraphs 7.236 and 7.333 to reflect the use of different instruments in different countries. Finally, the Panel does not consider revision to paragraph 7.273 to be necessary.

Footnote 308 to paragraph 7.216 (as well as footnote 259 to paragraph 7.196; footnote 304 to paragraph 7.214; and footnote 324 to paragraph 7.221)

6.25. Norway requested that these references to the inspector's report in Appendix K of the NOAH Report (2012) be supplemented to mention that the referenced voyage "involved exceptional conduct" that resulted in criminal prosecutions.⁴¹ The European Union submitted that it rebutted Norway's assertion that the inspection report in question "involved exceptional conduct" and therefore called on the Panel to reject Norway's request.

6.26. The Panel notes that Norway's comments concern footnote references to an inspection report from a Norwegian hunting expedition of the Kvitungen vessel. The principal statements to which each footnoted reference corresponds explain the nature of the reference being made. Thus, paragraph 7.216 states that "there are varying indications from sealing inspectors of the extent to which struck and lost is a problem in the Norwegian hunt", citing the inspection report in question along with the report of another sealing inspector. In the footnote to paragraph 7.196, the Kvitungen report is cited amongst several others after the statement that "[i]ndications from both participants in the hunts and veterinary experts recognize the heavy demands and difficult conditions of seal hunts". Similarly, in paragraphs 7.214 and 7.221, reference to the Kvitungen report corresponds to statements for which the report provides support and is made in conjunction with multiple other cited sources of evidence. Therefore, the Panel does not consider it necessary to supplement these references.

Footnote 317 to paragraph 7.218

6.27. With respect to hooking/gaffing seals aboard vessels, Norway requested that the Panel refer to the specific conditions under which this practice may occur according to Norwegian regulations.

⁴⁰ Norway's comments on the Interim Panel Reports, para. 63.

⁴¹ Norway's comments on the Interim Panel Reports, para. 53.

Norway also requested inclusion of its explanation of the conclusion of the Ministry of Fisheries and Coastal Affairs referenced in this footnote.

6.28. The European Union submitted that it had rebutted Norway's alleged motivation for not amending the provisions of its hunting regulations on the practices of hooking/gaffing seals. The European Union requested that, should the Panel accede to Norway's request, it also reflect the European Union's submissions in this regard.

6.29. The Panel referred to the relevant conditions under which seals may be hooked aboard vessels prior to exsanguination under Norwegian regulations and made a minor amendment on the basis of Norway's comments. Further, the Panel referred to the conclusion of the Norwegian Ministry of Fisheries and Coastal Affairs to indicate the ultimate disposition of this matter. The Panel therefore does not find it necessary to revise this footnote.

Paragraphs 7.219-7.221

6.30. Norway requested revision of these paragraphs to reflect "the distinct approach taken to monitoring under Norway's sealing regulations".⁴² Norway also requested reference to evidence that greater oversight leads to a lower likelihood of animal welfare problems. Finally, Norway requested replacement of the term "government inspector" in paragraph 7.220 with the term "independent veterinary inspector".

6.31. The European Union argued that the inspectors on board Norwegian vessels are government employees who represent the Norwegian government and take direct orders from the Fisheries Directorate. According to the European Union, therefore, such inspectors cannot be considered "independent".

6.32. The Panel added reference in these paragraphs regarding the animal welfare benefit of monitoring and enforcement as well as monitoring in Greenland. The Panel notes that these paragraphs primarily concern the feasibility and/or difficulty of monitoring and enforcement of the application of humane killing methods, and specifically draw upon evidence pertaining to the Norwegian hunt with added reference to the comments of Mr Danielsson. The Panel therefore does not consider further revision of these paragraphs to be necessary.

6.33. Additionally, as a factual matter, inspectors are "government-mandated" and report to the Norwegian Directorate of Fisheries.⁴³ Apart from its factual accuracy, the current wording conveys the authority of the inspector (as distinct from, for instance, an independent observer⁴⁴). Therefore, no change was made in this respect.

6.3 Specific comments on other parts of the Reports

Paragraphs 7.154, 7.597, and 7.608

6.34. The complainants requested the Panel to review paragraph 7.154 of the Interim Reports to note explicitly that the groups of products to be compared in Table 1 are those contained in cells C+H (*all* Canadian seal products), cells A+F (*all* domestic seal products) and cells D+I (*all* seal products from Greenland). Norway requested that appropriate references also be made to the cells in Table 1 in the context of the Panel's analysis of its claims under Articles I:1 and III:4 of the GATT 1994.

6.35. Further to the complainants' comments, the Panel made modifications in paragraphs 7.154, 7.597, and 7.608 of its Reports.

Paragraphs 7.159, 7.161, and footnotes 195, 891

6.36. The complainants noted that different figures were used in the Panel Reports to describe the proportion of seal products originating in Canada that could qualify under the IC exception.

⁴² Norway's comments on the Interim Panel Reports, para. 56.

⁴³ See Norway's second written submission, para. 297.

⁴⁴ See Norway's response to Panel question No. 61.

In particular, Canada requested the Panel to refer to its submissions and evidence referencing Canada's official statistical data. The European Union expressed reservations regarding the revision of the figures in the Interim Reports but did not object to the Panel adding a reference to Canada's submission to provide a more complete description of Canada's arguments.

6.37. As the figures at issue relate to the proportion of IC hunts in Canada, the Panel modified the above-mentioned paragraphs and footnotes of its Reports to make specific reference to the data provided by Canada.

Paragraph 7.164

6.38. Norway requested the Panel to complete its findings in paragraph 7.164 with additional evidence, in particular regarding how the IC requirements apply specifically to Greenland. The European Union requested the Panel to reject Norway's request because the conditions of the IC exception are origin-neutral and do not apply specifically to Greenland; it is therefore unnecessary for the Panel to make more factual findings on an issue that is not disputed by the European Union.

6.39. The Panel notes that evidence on Greenland is referenced in a number of paragraphs of the Reports. Moreover, given our finding that all, or virtually all, seal products from Greenland may be eligible under the IC exception, we decline Norway's request to add references to evidence on Greenland in paragraph 7.164 of the Interim Reports.

Paragraph 7.275

6.40. Norway suggested replacement of the word "can" with the word "do" in the third line of this paragraph. The European Union considered that using the verb "do" would convey the impression that all IC hunts, by definition, cause the pain and suffering referenced in paragraph 7.275. The European Union argued that the fact that some hunting methods used by Inuit communities are not consistent with humane killing methods does not mean that all IC hunts in every single case result in poor animal welfare.

6.41. The Panel agrees with the European Union that Norway's suggested revision would alter the meaning of the sentence as currently phrased. The Panel therefore declines to make the requested change.

Paragraphs 7.358, 7.363, 7.366, 7.376, 7.421, and 7.629

6.42. Norway made various comments relating to the characterization and summarization of its arguments, in particular with respect to the objective pursued by the EU Seal Regime.

6.43. The Panel made modifications to paragraphs 7.363, 7.366, 7.421, and accompanying footnotes to reflect Norway's comments.

Paragraph 7.386

6.44. Norway drew attention to recital (21) of the preamble of the Basic Regulation and requested reflection of the text of this recital in the Panel's analysis. The European Union submitted that the passage of recital (21) cited by Norway has a limited purpose of setting out the justification of the Basic Regulation in light of the principle of subsidiarity under EU law. According to the European Union, the recital does not purport to explain why the EU legislators chose the harmonizing measures provided in the Basic Regulation, rather than other possible harmonizing measures.

6.45. The Panel notes that the objective of internal market harmonization is addressed in paragraph 7.371, as amended by the Panel in the course of the interim review. Therefore, the Panel does not consider further reference to this objective to be necessary.

- 24 -

Section 7.3.3.1.2 (paragraphs 7.372-7.411)

6.46. Norway sought inclusion of a reference to the European General Court decision in Inuit Tapiriit Kanatami and others v. Commission with respect to the objective of the EU Seal Regime. The European Union considered this judgment of the European General Court to be of very limited relevance for this dispute and, in particular, with regard to the issue raised by Norway.

6.47. In Inuit Tapiriit Kanatami and others v. Commission (Case T-526/10), the European General Court addressed claims as to the alleged "illegality" of the Basic Regulation under EU law. More particularly, the Court examined the sufficiency of the Basic Regulation's legal basis under a specific provision of EU law and whether the objective of the Basic Regulation was such that it could legitimately be adopted on the basis of that provision.⁴⁵ Because the European General Court examined the objective of the Basic Regulation in the context of a substantively distinct inquiry with different claims and legal provisions at issue than those in the present case, the Panel declines to accept Norway's request.

Paragraphs 7.458 and 7.478

6.48. Canada noted in its comments that the last sentences of paragraphs 7.458 and 7.478 were in its view contradictory. Canada requested the Panel to harmonize the two paragraphs and indicated its preference for the formulation in paragraph 7.458. The European Union argued that there is no contradiction between the sentence in 7.458 and the findings reported in paragraphs 7.459, 7.460, and 7.478.

6.49. The Panel does not consider the statements in these paragraphs to be contradictory; the Panel made minor revisions to paragraph 7.458 for clarity.

Paragraphs 7.588-7.609 (Sections 7.4.2 and 7.4.3)

6.50. Norway argued that the Panel's decision to address Canada's claim under Article 2.1 of the TBT Agreement before addressing the complainants' claims under Articles I:1 and III:4 of the GATT 1994 resulted in Norway's position being overlooked in both parts of the Reports. Specifically, Norway expressed concern regarding the Panel's treatment and disposition of its discrimination claims under Articles I:1 and III:4 of the GATT 1994 in view of the cross-referencing made in paragraphs 7.594 and 7.597 to the Panel's analysis of Canada's claim under Article 2.1 of the TBT Agreement.

6.51. The Panel added references to the arguments and evidence submitted by Norway in support of its claims under Articles I:1 and III:4 of the GATT 1994 in paragraphs 7.594 and 7.597 of the Reports, and deleted some of the cross-references to Section 7.3.2 of the Reports.

Other paragraphs and footnotes

6.52. Finally, the Panel made a number of additional modifications to its Reports further to the comments by Canada⁴⁶ and Norway⁴⁷, in cases where the Panel considered that the proposed changes improved the clarity of the parties' arguments or the overall accuracy of the Reports.

6.4 Additional documents submitted by the European Union with its comments on the **Interim Reports**

6.53. The European Union submitted with its comments on the Interim Reports additional documents consisting of letters between officials from the European Commission on the one hand,

⁴⁵ See, e.g. paras. 22 and 64 of the General Court Judgment.

⁴⁶ Changes were made in paragraphs 1.3, 1.15, 1.16, 7.2, 7.4, 7.11, 7.30, 7.31, 7.51, 7.90, 7.91, 7.130, 7.144, 7.154, 7.168, 7.191, 7.200, 7.206, 7.216, 7.220, 7.229, 7.248, 7.266, 7.272, 7.275, 7.285, 7.294, 7.311, 7.331, 7.335, 7.336, 7.342, 7.381, 7.400, 7.408, 7.423, 7.425, 7.463, 7.518, 7.522, 7.587,

^{7.588, 7.594, 7.597, 7.608, 7.612, 7.616, 7.622, 7.650, 7.658,} and 7.668; and footnotes 44, 58, 70, 75, 76, 99, 109, 111, 113, 157, 158, 163, 201, 222, 240, 254, 264, 276, 321, 350, 356, 415, 420, 462, 517, 535,

^{536, 547, 624, 655, 711, 765, 808, 918, 919,} and 964. ⁴⁷ Changes were made in paragraphs 7.4, 7.40, 7.166, 7.192, 7.194, 7.331, 7.363, 7.366, 7.371, 7.421, 7.470, and 7.617; and footnotes 53, 202, 598, 604, and 960.

and the Canadian Government as well as the authorities of Nunavut on the other hand. The European Union noted that these documents reflected recent developments of relevance to the state of the requests made by entities to become recognized bodies entitled to deliver attesting documents for placing seal products on the EU market. The European Union requested the Panel to make reference to these new documents in its reports.⁴⁸

6.54. The complainants objected to the European Union's request on the grounds that the information provided by the European Union constituted new evidence that should have been introduced earlier in the proceedings in accordance with the Panel's working procedures. The complainants further argued that the submission of evidence at this late stage of the proceedings was inconsistent with the requirements of due process, and that the European Union's failure to abide by such requirements could not be remedied at the interim review stage.

6.55. The Panel considers that the documents attached to the European Union's comments on the Interim Reports constitute new evidence. The letters submitted by the European Union are dated May 2013 and, as such, they could have been introduced at an earlier time in the proceedings in accordance with the Panel's working procedures.⁴⁹ Further, the Panel recalls that in *EC - Sardines*, the Appellate Body stated that "[t]he interim review stage is not an appropriate time to introduce new evidence".⁵⁰ For these reasons, the Panel declines to amend relevant parts of its Reports to reflect the aforementioned information submitted by the European Union.

7 FINDINGS

7.1 Overview of the dispute

7.1. This dispute concerns a 2009 European Union measure relating to the sale of seal products (EU Seal Regime). Under the measure, the placing of seal products on the market is prohibited in the European Union unless they satisfy certain conditions. One such condition applies to seal products obtained from seals hunted by Inuit or indigenous communities (IC condition). The other applies to seal products obtained from seals hunted for marine resource management (MRM condition). Travellers may also be able to bring seal products into the European Union in limited circumstances (Travellers condition). The Regime lays down specific requirements for all three conditions.

7.2. Canada and Norway claim that the EU Seal Regime violates the European Union's various obligations under the GATT 1994 and the TBT Agreement. First, the complainants allege that the IC and MRM conditions of the EU Seal Regime violate the non-discrimination obligations under Articles I:1 and III:4 of the GATT 1994. Canada also presented a claim under Article 2.1 of the TBT Agreement with respect to the IC and MRM conditions. The complainants argue that the IC and MRM conditions accord seal products from Canada and Norway (imported products) treatment less favourable than that accorded to like seal products of domestic origin, mainly from Sweden and Finland (domestic products) as well as those of other foreign origin, particularly from Greenland (other foreign products). Second, the complainants argue that the EU Seal Regime creates an unnecessary obstacle to trade that is inconsistent with Article 2.2 of the TBT Agreement because it is more trade restrictive than necessary to fulfil a legitimate objective. Third, the complainants argue that certain procedural requirements under the EU Seal Regime violate the requirements for conformity assessment under Article 5 of the TBT Agreement. Fourth, the complainants claim that each of the IC, MRM, and Travellers conditions of the EU Seal Regime impose quantitative restrictions on trade inconsistently with Article XI:1 of the GATT 1994. Finally, the complainants submit that the application of the EU Seal Regime nullifies or impairs benefits accruing to them under the covered agreements within the meaning of Article XXIII:1(b) of the GATT 1994.

⁴⁸ The European Union proposed that a reference to the new documents be added for instance in footnote 198 to paragraph 7.162 of the Interim Reports.

⁴⁹ According to Rule 7 of the Panel's working procedures, "[e]ach party shall submit all factual evidence to the Panel no later than during the first substantive meeting, except with respect to evidence necessary for purposes of rebuttal, answers to questions or comments on answers provided by the other party(ies). Exceptions to this procedure shall be granted upon a showing of good cause."

⁵⁰ Appellate Body Report, *EC* – *Sardines*, para. 301.

7.3. The European Union asserts that the measure is fully consistent with its WTO obligations. The European Union claims that the EU Seal Regime is aimed at addressing public moral concerns on the welfare of seals. The EU Seal Regime is thus not based on conservation concerns. The complainants contest the objective of the measure as put forward by the European Union. According to the complainants, the measure pursues a multiplicity of objectives such as the protection of seal welfare; the protection of the social and economic interests of Inuit or indigenous communities; and the promotion of sustainable marine resource management. Based on its identified objective, the European Union argues that any inconsistencies of the measure under the GATT 1994 should be justified under the general exceptions provisions of the GATT 1994, namely Articles XX(a) and XX(b), because the measure is necessary to protect public morals (regarding the welfare of seals) and to protect seals' health, respectively. Further, the European Union argues that any distinction made under the EU Seal Regime, for instance a distinction based on the type and purpose of the hunt, is legitimate within the meaning of Article 2.1 of the TBT Agreement. The European Union also contends that no other measure can protect its public moral concerns on seals at the same level as does the current Regime.

7.4. Factually, the parties debated extensively whether humane killing methods can be applied, monitored, and enforced in seal hunts. The European Union's justification of its measure is based on the premise that the application and enforcement of humane killing methods in seal hunting are not always feasible because of *inter alia* the unique environmental conditions in which the hunting takes place. The European Union asserts that due to the "inherent" inhumane nature of the hunts, particularly hunts conducted for commercial purposes, the European public is ethically and morally repelled by the presence on the EU market of seal products. Hence, a general ban as designed under the current measure is the only effective way to protect the public moral concerns. The complainants argue that humane killing methods can be properly enforced in seal hunts. Further, they underline that as the current measure does not condition market access on the humaneness with which seals are killed, seal products derived from seals killed inhumanely may be allowed on the EU market. This, in their view, proves that the current measure is not capable of protecting the welfare of seals. Both sides have submitted a voluminous amount of evidence, mostly based on scientific studies and expert statements, pertaining to whether the application and monitoring of humane killing methods can be enforced in seal hunting practices.

7.5. Legally, the Panel is presented with the task of *inter alia* examining the obligations under, as well as the relationship between, the GATT 1994 and the TBT Agreement. In the current dispute, the two complainants brought claims under both agreements, namely Articles I:1, III:4, and XI:1 of the GATT 1994 as well as Articles 2.2, 5.1.2 and 5.2.1 of the TBT Agreement. Canada also brought a claim under Article 2.1 the TBT Agreement.

7.6. These Reports are structured in the following order: (i) preliminary matters; (ii) the measure's qualification as a technical regulation; (iii) claims under the TBT Agreement; (iv) claims under the GATT 1994 and the Agreement on Agriculture; (v) non-violation claim under Article XXIII:1(b) of the GATT 1994; and (vi) our conclusions and recommendations.

7.2 Preliminary matters

7.2.1 Description of the measures at issue

7.7. As described in the Factual Aspects section above, Canada and Norway are challenging the following two EU legal instruments in this dispute:

- Regulation (EC) No. 1007/2009 of the European Parliament and of the Council of 16 September 2009 on trade in seal products, Official Journal of the European Union, L Series, No. 286 (31 October 2009)⁵¹; and
- b. Commission Regulation (EU) No. 737/2010 of 10 August 2010 laying down detailed rules for the implementation of Regulation (EC) No. 1007/2009 of the European Parliament

⁵¹ Exhibit JE-1.

and of the Council on trade in seal products, Official Journal of the European Union, L Series, No. 216 (17 August 2010). 52

7.8. For the purpose of this dispute, the Panel will use the following terms: "the Basic Regulation" for Regulation (EC) No. 1007/2009, "the Implementing Regulation" for Commission Regulation (EU) No. 737/2010, and "the EU Seal Regime" for these two legal instruments combined together.⁵³

7.9. We briefly describe the contents of the two Regulations below.

7.2.1.1 The Basic Regulation

7.10. The Basic Regulation consists of a preamble (21 recitals) and eight provisions ((1) "Subject matter"; (2) "Definitions"; (3) "Conditions for placing on the market"; (4) "Free movement"; (5) "Committee procedure"; (6) "Penalties and enforcement"; (7) "Reporting"; and (8) "Entry into force and application").

7.11. The preamble of the Basic Regulation refers to *inter alia* concerns and observations on seal hunting as well as seal products resulting from such hunts. The Panel will examine specific parts of the preamble in the context of its examination of the parties' claims and arguments.

7.12. Article 3 of the Basic Regulation lays down the rules regarding "conditions for placing on the market" of seal products:

Article 3

Conditions for placing on the market

1. The placing on the market of seal products shall be allowed only where the seal products result from hunts traditionally conducted by Inuit and other indigenous communities and contribute to their subsistence. These conditions shall apply at the time or point of import for imported products.

2. By way of derogation from paragraph 1:

(a) the import of seal products shall also be allowed where it is of an occasional nature and consists exclusively of goods for the personal use of travellers or their families. The nature and quantity of such goods shall not be such as to indicate that they are being imported for commercial reasons;

(b) the placing on the market of seal products shall also be allowed where the seal products result from by-products of hunting that is regulated by national law and conducted for the sole purpose of the sustainable management of marine resources. Such placing on the market shall be allowed only on a non-profit basis. The nature and quantity of the seal products shall not be such as to indicate that they are being placed on the market for commercial reasons.

The application of this paragraph shall not undermine the achievement of the objective of this Regulation.

...

7.13. Article 3(1) of the Basic Regulation prescribes that the *placing on the market* of seal products is allowed "*only* where the seal products result from hunts traditionally conducted by Inuit and other indigenous communities and contribute to their subsistence" (IC hunts). The provision also states that for imported products, these conditions are applied at the time or point of import.

⁵² Exhibit JE-2.

 $^{^{\}rm 53}$ See section 3 above regarding the parties' usage of "EU Seal Regime" to refer to the measures at issue.

7.14. Article 3(2) describes two situations where the condition set out in paragraph 1 does not apply: first, the *import* of seal products is allowed where it is of an occasional nature and consists exclusively of goods for the personal use of travellers or their families (Travellers imports); second, the *placing on the market* of seal products is allowed where the seal products result from by-products of hunting that is regulated by national law and conducted for the sole purpose of the sustainable management of marine resources (MRM hunts).

7.15. Specific requirements for each of the three conditions for importing and/or placing seal products on the market are elaborated in the Implementing Regulation.

7.2.1.2 The Implementing Regulation

7.16. The Implementing Regulation comprises a preamble (13 recitals) and twelve provisions.

7.17. The preamble refers to the need to specify detailed requirements for the import and placing on the market of certain seal products and the principles to be applied in setting out procedures for adequate verification of compliance with such requirements, as well as for the control of attesting documents.

7.18. Unlike the Basic Regulation, a specific title is not assigned to each of the provisions in the Implementing Regulation. The purpose of the Implementing Regulation is set forth in Article 1: to "lay [] down detailed rules for the placing on the market of seal products pursuant to Article 3" of the Basic Regulation.

7.19. Articles 3, 4, and 5 of the Implementing Regulation address the specific requirements for each of the three conditions mentioned in Articles 3(1) and 3(2) of the Basic Regulation.

7.20. Specifically, Article 3 sets out that, to fall under the *IC hunts* category, seal products must originate from seal hunts that satisfy the following three conditions:

- a. seal hunts conducted by Inuit⁵⁴ or other indigenous communities⁵⁵ which have a tradition of seal hunting in the community and in the geographical region;
- b. seal hunts the products of which are at least partly used, consumed or processed within the communities according to their traditions; and
- c. seal hunts which contribute to the subsistence of the community.

7.21. Article 4 sets out that, to fall under the *Travellers imports* category, one of the following three requirements must be fulfilled:

- a. the seal products are either worn by the travellers, or carried or contained in their personal luggage;
- b. the seal products are contained in the personal property of a natural person transferring his normal place of residence from a third country to the Union; or
- c. the seal products are acquired on site in a third country by travellers and imported by those travellers at a later date, provided that ... those travellers present to the customs authorities ... the following documents:
 - i. a written notification of import; and

⁵⁴ "Inuit" is defined as "indigenous members of the Inuit homeland, namely those arctic and subarctic areas where, presently or traditionally, Inuit have aboriginal rights and interests, recognised by Inuit as being members of their people and includes Inupiat, Yupik (Alaska), Inuit, Inuvialuit (Canada), Kalaallit (Greenland) and Yupik (Russia)". (Basic Regulation, Article 2(4)).

⁵⁵ "Other indigenous communities" is defined as "communities in independent countries who are regarded as indigenous on account of their descent from the populations which inhabited the country, or a geographical region to which the country belongs, at the time of conquest or colonisation or the establishment of present State boundaries and who, irrespective of their legal status, retain some or all of their own social, economic, cultural and political institutions". (Implementing Regulation, Article 2(1)).

ii. a document giving evidence that the products were acquired in the third country concerned.

7.22. Article 5 provides that, to fall under the *MRM hunts* category, seal products must originate from seal hunts that satisfy the following three conditions:

- seal hunts conducted under a national or regional natural resources management plan which uses scientific population models of marine resources and applies the ecosystembased approach;
- b. seal hunts which does [*sic*] not exceed the total allowable catch quota established in accordance with the plan referred to in point (a); and
- c. seal hunts the by-products of which are placed on the market in a non-systematic way on a non-profit basis.⁵⁶

7.23. Articles 3(2) and 5(2) require that, at the time of placing on the market pursuant to Articles 3(1) and 3(2)(b) of the Basic Regulation and Articles 3 and 5 of the Implementing Regulation (i.e. products resulting from IC and MRM hunts), the seal product be accompanied by the attesting document referred to in Article 7(1) of the Implementing Regulation.

7.24. Articles 6 through 10 prescribe the procedural requirements that must be met to place seal products on the market. For a seal product to be placed on the market, it must be accompanied by an attesting document (Article 7) issued by a recognized body (Article 6). A reference to the attesting document number must be included in any further invoice (Article 7(4)). A model attesting document is attached as an annex to the Implementing Regulation.

7.2.2 Consideration of the measures at issue

7.2.2.1 Single or multiple measure(s)

7.25. As noted above, the EU Seal Regime consists of the Basic Regulation and the Implementing Regulation. The Basic Regulation was adopted by the European Parliament and Council of the European Union on 16 September 2009; it sets forth the "conditions for placing on the market" of seal products. Pursuant to Article 3(4) of the Basic Regulation, the Implementing Regulation was subsequently adopted by the European Commission on 10 August 2010. The Implementing Regulation lays down the specific requirements necessary for implementing the rules in the Basic Regulation.

7.26. Therefore, the Basic Regulation and the Implementing Regulation operate in conjunction with each other in governing the importation and the placing of seal products on the EU market. In particular, the Implementing Regulation does not operate on its own as it is a regulation adopted to "implement" the rules in the Basic Regulation. The parties agree that the EU Seal Regime should be treated as a single measure.⁵⁷ Accordingly, we consider that these two legal instruments must be examined as an integrated whole.

7.27. Treating both the Basic and Implementing Regulations as a single measure does not mean that different aspects of the EU Seal Regime cannot be challenged under different provisions of the WTO covered agreements. In fact, we note that in presenting their claims that the EU Seal Regime is inconsistent with several provisions of the GATT 1994 and the TBT Agreement, the complainants focus at times on specific aspects of the EU Seal Regime and at other times on the EU Seal Regime as a whole.⁵⁸

⁵⁶ "Placing on the market on a non-profit basis" is defined as "placing on the market for a price less than or equal to the recovery of the costs borne by the hunter reduced by the amount of any subsidies received in relation to the hunt". (Implementing Regulation, Article 2(2)).

The Regulations do not provide a definition for "placing on the market in a non-systematic way".

⁵⁷ Parties' responses to Panel question No. 2.

⁵⁸ For example, the complainants focus on the IC hunts category under the measure with respect to their claim under Article I:1 of the GATT 1994 and on the MRM hunts category with respect to their claim under

7.2.2.2 Characterization of the measure at issue

7.28. Despite their common understanding that the EU Seal Regime should be treated as a single measure, the parties disagree on how the EU Seal Regime should be characterized for the purpose of this dispute. Briefly stated, the complainants argue that the EU Seal Regime provides for three sets of specific requirements concerning the importation and/or the placing on the market of seal products. The respondent submits that the EU Seal Regime consists of a general ban on seal products with certain exceptions.

7.29. As it is important for the Panel to start its analysis with the proper understanding of the measure at issue, we now turn to the question of how the EU Seal Regime must be characterized.

7.2.2.2.1 Main arguments of the parties

7.2.2.2.1.1 Complainants

7.30. Canada argues that the EU Seal Regime should be viewed as setting out requirements concerning the importation of seal products.⁵⁹ Article 3 of the Basic Regulation and Articles 3, 4, and 5 of the Implementing Regulation establish a comprehensive regime of conditions that is simultaneously restrictive and permissive and determines when seal products may be imported and placed on the market in the European Union, or are prevented from accessing it. According to Canada, these provisions, when read together based on their design, structure, and expected operation, lead to the conclusion that the EU Seal Regime is accurately described as "requirements concerning the importation of seal products".⁶⁰ The texts of the Regulations do not support the characterization of the measure as a general ban with certain exceptions as the term "prohibition" or "ban" is not used.61

7.31. Canada submits that the conditions under the EU Seal Regime are divided into three categories or sets of requirements: seal products derived from IC hunts; seal products derived from MRM hunts; and seal products imported as Travellers imports.⁶² Seal products that satisfy the conditions pertaining to the Travellers imports are eligible to be imported; in the case of IC hunts and MRM hunts, seal products are eligible to be imported and placed on the market. Seal products that do not fall within any of these categories are not eligible to be imported or placed on the market, although neither the Basic Regulation nor the Implementing Regulation states expressly that such products are prohibited from importation or from being placed on the market.

7.32. Norway submits that the EU Seal Regime does not comprise a ban (on the sale or import of seal products) with separate exceptions.⁶³ Rather, the Regulations combine permissive and prohibitive elements both formally and in substance, laying down three sets of market access conditions that constitute restrictive gateways for the sale and importation of seal products. Particularly, neither the Basic Regulation nor the Implementing Regulation contains the term "General Ban" as distinct from "exceptions".

7.33. Norway submits that, in assessing whether a measure imposes a trade "restriction", neither the generality of a rule nor its association with an exception is important.⁶⁴ Rather, what matters is whether a measure imposes conditions that, by nature or effect, place limits on trade.⁶⁵ Hence, in characterizing a measure for purposes of WTO obligations addressing trade restrictions, a panel must ascertain whether a measure imposes a "limiting condition" and, if so, assess whether that condition is WTO-consistent. Each of the three requirements in the EU Seal Regime includes a

⁶³ Norway's first written submission, paras. 158-170; response to Panel question No. 1; opening statement at the first meeting of the Panel, paras. 7-16. ⁶⁴ Norway's opening statement at the first meeting of the Panel, paras. 14-15.

Article III:4 of the GATT 1994. As regards their claim under Article 2.2 of the TBT Agreement, the complainants focus on the EU Seal Regime as a whole.

⁵⁹ Canada's response to Panel question No. 1.

⁶⁰ Canada's response to Panel question No. 1, para. 24

⁶¹ Canada's response to Panel question No. 1.

⁶² The parties use different terms for these three conditions contained in the Regulations. For the

purpose of these Reports, the Panel will use the terms set out in paras. 7.13-7.14 and 7.56 for these conditions.

⁶⁵ Norway's opening statement at the first meeting of the Panel, para. 14 (referring to Appellate Body Reports, China – Raw Materials, para. 319; US – Tuna II (Mexico), para. 319).

series of specific requirements that place limits on EU market access with respect to both the placing on the market and the import of products. According to Norway, the subject matter of this dispute is these specific limiting conditions, which are the legal source of the market access restrictions on Norwegian seal products.

7.34. In response to a question from the Panel on their understanding of the measure as described in their respective panel requests, the complainants submit that their position on the characterization of the measure does not substantively differ from that contained in the panel requests.66

7.2.2.2.1.2 Respondent

7.35. The European Union argues that the EU Seal Regime should be considered as a "General Ban coupled with three exceptions".⁶⁷ Article 3(1) of the Basic Regulation sets forth the "General Ban" on the placing on the market of seal products, applied at the point of importation in the case of imported products, together with the main exception (the IC exception) and two additional exceptions (the MRM exception and the Travellers' exception).

7.36. The European Union argues that the complainants' proposition (i.e. three trade-restrictive requirements) is overly formalistic. From a logical point of view, the Basic Regulation's conditional authorization ("the placing on the market of seal products shall be allowed only where") has the same meaning as other propositions such as "the placing on the market of seal products shall not be allowed unless" or "the placing on the market of seal products shall be prohibited except where". The European Union also refers to the legislative history and the recitals of the Basic Regulation, which in its view confirm that the EU legislators sought to enact a general ban subject to certain exceptions.68

7.37. The European Union asserts that the complainants' position is based on their belief that it allows them to claim that the EU Seal Regime makes no contribution to the overarching policy objective pursued by the European Union (addressing public moral concerns related to animal welfare). For the European Union, the complainants' argument is a "manifest non sequitur"⁶⁹ in that it suggests that the EU Seal Regime as a whole makes no contribution to its stated objective because none of the three requirements (in reality, exceptions) contributes to the objective. The European Union maintains that there can be no question that the EU Seal Regime has the effect of prohibiting the placing on the market of seal products where none of the three so-called "requirements" is met.

7.38. According to the European Union, the complainants' position is also at odds with the characterization of the measure included in the panel requests. If the EU Seal Regime did not provide for a ban subject to exceptions, but rather for three self-standing "requirements", both Norway's and Canada's panel requests were manifestly incorrect and deficient in that they failed to "present the problem clearly". Thus, were the Panel to decide that the EU Seal Regime cannot be characterized as a General Ban subject to three exceptions, the European Union requests the Panel to find that the complainants' panel requests do not meet the requirements of Article 6.2 of the DSU and to reject all the claims submitted by them.⁷

7.2.2.2.2 Analysis by the Panel

7.39. We begin our examination of the measure with a consideration of the text of the Basic Regulation and the Implementing Regulation.

7.40. Article 1 of the Basic Regulation, entitled "Subject matter", states that the Regulation "establishes harmonised rules concerning the placing on the market of seal products". Article 3, entitled "Conditions for placing on the market", starts with a paragraph prescribing that the placing

⁶⁶ Complainants' responses to Panel question No. 102.

⁶⁷ European Union's first written submission, paras. 1, 11, 35, and 357-358; response to Panel question

Nos.1, 2, and 100. ⁶⁸ For example, the European Union refers to recital (10) of the Basic Regulation where the phrase "the placing on the market should ... not be allowed" is used. ⁶⁹ European Union's response to Panel question No. 1.

 $^{^{\}rm 70}$ European Union's response to Panel question No. 1, para. 11.

on the market of seal products shall be allowed *only* where the seal products result from IC hunts. It also explains that the conditions in the first paragraph shall apply at the time or point of *import* for imported products.⁷¹ The wording of the first paragraph of Article 3 thus indicates that this is the only situation where seal products can be placed on the EU market.⁷²

7.41. The second paragraph of Article 3 begins with the phrase "by way of derogation from paragraph 1" and provides for two situations where derogation from paragraph 1 is allowed. First, the *placing on the market* of seal products on a non-profit basis is allowed where a seal product is derived from MRM hunts and is *not* being placed on the market for "commercial reasons" (Article 3(2)(b)).⁷³ Second, the *import* by travellers of a seal product is allowed (Article 3(2)(a)) to the extent that it is not "for commercial reasons".⁷⁴ The Regulations do not define the term "commercial reasons".⁷⁵

7.42. Based on the text of Article 3 of the Basic Regulation, therefore, we understand that the measure operates as follows:

- seal products derived from IC hunts may be imported and/or placed on the EU market;
- seal products derived from MRM hunts may be placed on the EU market when it is on a non-profit basis and is not for commercial reasons; the text does not indicate whether the conditions also apply to imported products; and
- seal products for personal use of travellers or their families may be imported for non-commercial reasons; however, the placing on the market of such products is prohibited.

7.43. Therefore, although seal products of both Travellers imports and IC hunts, along with seal products allowed for transit and other commercial activities under the measure⁷⁶, may be *imported* into the European Union (i.e. enter "the customs territory of the Community")⁷⁷, only the products of IC hunts may also be placed on the EU market.

7.44. Further, despite the absence of any reference to imported products, the complainants consider that, like in the case of seal products obtained from IC hunts, the conditions governing the placing on the market of seal products of MRM hunts also apply at the point or time of

⁷¹ The Basic Regulation defines "import" as "any entry of goods into the customs territory of the Community". (Basic Regulation, Article 2(5).

The European Union explained further the scope and meaning of "import" as defined by Article 2(5): "The Basic Regulation is being interpreted by the customs authorities of the Member States as applying to goods which have been released for free circulation in the customs territory of the European Union after payment of the duties to which they are liable. The EU Commission has not taken any official position on this issue. Nor has this matter been resolved by the EU Court of Justice." (European Union's response to Panel question No. 79, para. 231). The European General Court had occasion to address the meaning of the term "import" under the Basic Regulation in its decision of 25 April 2013 in *Inuit Tapiriit Kanatami and others v. Commission*, observing that the Basic Regulation "does not prevent the entry, warehousing, processing or manufacture of seal products in the Union, if they are intended for export and are never released for free circulation in the Union". (European General Court, Case T-526/10, Inuit Tapiriit Kanatami and others v. Commission (2013), para. 70).

⁷² The Basic Regulation defines "placing on the market" as "introducing on the Community market, thereby making available to third parties, in exchange for payment". (Basic Regulation, Article 2(3)).

⁷³ Article 3(2)(b) of the Basic Regulation states at the end, "The nature and quantity of the seal products shall not be such as to indicate that they are being *placed on the market* for commercial reasons."

⁷⁴ More specifically, the import of seal products is allowed "where it is of an occasional nature and consists exclusively of goods for the personal use of travellers or their families". (Basic Regulation, Article 3(2)(a)). It further adds that the nature and quantity of such goods shall not be such as to indicate that they are being *imported* for commercial reasons.

⁷⁵ In response to a question from the Panel regarding the MRM requirements, the European Union explained that "[t]he systematic and repeated way in placing those products on the EU's market would indicate that the *purpose of the hunt in question was commercial*. Indeed, if seal products are repeatedly and systematically placed on the EU's market, e.g. in certain periods of the year or through the same channels of commerce, this would indicate that they are being *sold for a commercial purpose*." (European Union's response to Panel question No. 123, para. 84) (emphasis added)

⁷⁶ See para. 7.53.

⁷⁷ Article 2(5) of the Basic Regulation.

importation for seal products of foreign origin.⁷⁸ The European Union has also confirmed that this is a correct understanding.⁷⁹

7.45. Overall, the practical implication of Article 3 is that seal products derived from hunts other than IC or MRM hunts cannot be imported and/or placed on the EU market. Canada and Norway claim, and the European Union does not dispute, that most of the seal products from Canada and Norway are derived from hunts that are *not* IC or MRM hunts as defined by the measure and are consequently prevented from accessing the EU market.⁸⁰

7.46. We note that the operative part of the Basic Regulation does not use words such as "prohibit" or "ban". Rather, as described in the preceding paragraphs, it prescribes the specific conditions under which the import or placing on the EU market of seal products is *allowed*. Nevertheless, the use of the word "only" in the first paragraph of Article 3, combined with the phrase "by way of derogation" in the second paragraph, signifies that the import and placing on the EU market of seal products are *not allowed* other than in the situation specified in the first paragraph, plus two further circumstances set out in the second paragraph of Article 3. In other words, having regard to the design and structure of the Basic Regulation, and in the light of the text of that Regulation, the measure effectively operates as a prohibition on seal products that do not meet the conditions under the measure.

7.47. The preamble of the Basic Regulation, however, refers explicitly to the notion of a ban, particularly an import ban, on seal products in several places. For example, recital (1) mentions the European Parliament's 2006 declaration on banning seal products in the European Union, which led to the current EU Seal Regime, and the Parliament's request to the Commission to draft a regulation to ban the import, export and sale of all harp and hooded seal products. Recital (5) references several EU member States' adoption or intention to adopt legislation regulating trade in seal products by prohibiting the import and production of such products, and recital (21) refers to "harmonising national bans concerning trade in seal products". Recital (10) makes clear that "as a general rule", seal products would "not be allowed":

In order to counter barriers to the free movement of products concerned in an effective and proportionate fashion, *the placing on the market of seal products should, as a general rule, not be allowed* in order to restore consumer confidence while, at the same time, ensuring that animal welfare concerns are fully met. ... In order to ensure effective enforcement, the harmonised rules should be enforced at the time or point of import for imported products. (emphasis added)

7.48. Several recitals in the preamble also address the economic and social interests of Inuit communities engaged in seal hunting and the desire to empower the Commission to define the three conditions as currently reflected in the Basic Regulation for the placing on the market or import of seal products. The preamble thus describes *inter alia* the circumstances leading to the adoption of the Basic Regulation, the general rule under the measure not to allow seal products, and the need to define the three conditions under which seal products are allowed.

7.49. Turning to the Implementing Regulation, we note that it lays down detailed rules, including procedural requirements, for the placing on the market of seal products of both domestic and foreign origin under the three conditions specified in the Basic Regulation. Apart from that, the Implementing Regulation does not provide any further indication to assist us in understanding the character of the EU Seal Regime as a whole.

⁷⁸ Complainants' responses to Panel question No. 1.

⁷⁹ European Union's response to Panel question No. 77.

⁸⁰ In response to a question from the Panel, the European Union confirms that commercial imports are prohibited unless they qualify for the IC or MRM exceptions. (European Union's response to Panel question No. 78). We note that the term "commercial" in this context was used without any specific definition. Nevertheless, regarding the scope and meaning of the term "commercial hunting of seals" as referenced in recital (10) of the Basic Regulation, the European Union states that it "should be understood as referring to hunts which are conducted exclusively or primarily for the purpose of obtaining products, such as skins or oil, which are subsequently marketed for profit". (European Union's response to Panel question No. 76). Taken together, the term "commercial imports" appears to be referring to seal products obtained from "commercial hunting of seals" as defined by the European Union in its responses to questions from the Panel.

7.50. Having examined the texts of the Basic Regulation and the Implementing Regulation, we also observe that the respective panel requests of the complainants describe the EU Seal Regime as the "trade ban"⁸¹ or "general prohibition"⁸² on the importation and sale of seal products, with certain exceptions. This suggests to us that the complainants understood the Regime to function as a ban with exceptions. For the complainants, regardless of the form of the Regime, it operates as a ban with respect to their seal products, while it does not operate as such with respect to other seal products, in particular those from the European Union and Greenland.⁸³

7.51. In their written submissions, both complainants emphasize that the EU Seal Regime combines permissive and prohibitive elements, laying down three sets of market access conditions that determine when seal products may or may not be imported and/or placed on the EU market.⁸⁴ In other words, for the complainants, the EU Seal Regime essentially allows certain seal products and prohibits all other seal products. Canada asserts that although the EU Seal Regime is on its face framed as a measure governing the placement of seal products on the EU market, in practical terms it does little more than impede imports of seal products from Canada, Norway, and other WTO Members, while continuing to allow seal products from favoured exporting countries such as Greenland and domestic seal products.⁸⁵ Similarly, Norway submits that it characterizes the measure in the same way the European Union did when notifying the Implementing Regulation, namely, the measure establishes three sets of requirements that specify conditions that must be fulfilled for seal products to be placed on the EU market. Norway notes that these "three sets of requirements simultaneously combine, both in form and in substance, the prohibitive and permissive elements of the measure".⁸⁶

7.52. Like the complainants, we also consider that it is the three conditions set out in Article 3 of the Basic Regulation that, taken together, both allow and prohibit the placing on the market of seal products. By allowing seal products only under a defined condition complemented by two derogations, the measure effectively prohibits all seal products that do not fit into the specifications of those three requirements. For imported products that do not meet the conditions for one of the three requirements, therefore, the measure as a whole effectively works as an import ban. The fact that the measure is phrased in a positive form does not change the substantive character of the measure as both prohibiting seal products and allowing them upon meeting certain specific conditions.

7.53. Additionally, we observe that under the EU Seal Regime, seal products may also enter the territory of the Community in the following circumstances: (a) seal products may transit across the European Union; (b) seal products may be processed in the European Union for export under an inward processing scheme⁸⁷, using seal inputs regardless of their source; and (c) seal products may be sold for export at EU auction houses. Therefore, in addition to the *explicit exceptions* enshrined in the Regulations (i.e. IC, MRM, and Travellers exceptions), the EU Seal Regime also creates *implicit exceptions* for seal products for transit, inward processing, and importation for auction and re-export.⁸⁸

7.54. To the extent that the complainants' contention about the nature of the measure is related to their view that the measure cannot be described as a "general" ban as described by the European Union because certain seal products are allowed under the measure, we recall the

⁸¹ Canada's request for the establishment of a panel, p. 2.

⁸² Norway's request for the establishment of a panel, p. 2.

⁸³ See, below sections 7.3.2; 7.4.2; and 7.4.3.

⁸⁴ For example, Canada submits that seal products that do not fall within any of these categories are, in effect, not eligible to be imported or placed on the market, although neither the Basic Regulation nor the Implementing Regulation states expressly that such products are prohibited.

⁸⁵ Canada also argues that the measure operates *de facto* as a border measure imposing a discriminatory restriction on the importation of seal products, in violation of Articles I:1 and XI:1 of the GATT 1994, rather than as an internal measure. (Canada's response to Panel question No. 1; see also Canada's response to Panel question No. 102).

⁸⁶ Norway's response to Panel question No. 102.

⁸⁷ The European Union notes that the EU Seal Regime does not allow the "processing" of seal products in general but only "inward processing", which it defines as "the processing under customs control of imported inputs into products intended for export". (European Union's response to Panel question No. 131; see also European Union's response to Panel question Nos. 75, 79, and 101; complainants' comments on the European Union's response to Panel question Nos. 100 and 101).

⁸⁸ See European Union's response to Panel question Nos. 75, 101, 131, and 177; and Canada's and Norway's comments on European Union's response to Panel question Nos. 75, 101, 131 and 177.

Appellate Body's finding in *EC-Asbestos* that the measure in that dispute was not a *total* prohibition on asbestos fibres because it also included provisions that permitted the use of asbestos in certain situations.⁸⁹ The Appellate Body stated further that to characterize the measure simply as a general prohibition, and to examine it as such, would overlook the complexities of the measure, which included both prohibitive and permissive elements. Similarly, given the exceptions under the EU Seal Regime, we do not consider the EU Seal Regime to constitute a "total" or "general" ban on seal products; rather, the Regime consists of both prohibitive and permissive components and should be examined as such.

7.55. The measure would have been clearer in expressing its intended purpose and function as a ban on seal products if it had explicitly prohibited the import and placing on the EU market of seal products.⁹⁰ However, insofar as we can discern the true character of the measure from its design, structure, and expected operation, we need not second-guess the precise reason why the measure was formulated in the present manner.

7.56. The considerations above, taken together, demonstrate that the EU Seal Regime in its entirety operates as a ban on seal products, combined with an exception and two derogations, forming three conditions prescribed in Article 3 of the Basic Regulation (i.e. seal products obtained from IC hunts, MRM hunts, and those imported under the Travellers imports category). In this connection, for ease of reference, these Reports will refer to these three conditions using the following terms: the "IC hunts/category/exception/requirements" (for the condition in Article 3(1) of the Basic Regulation and Article 3 of the Implementing Regulation); the "MRM hunts/category/exception/requirements" (for the condition in Article 3(2)(b) of the Basic Regulation and Article 5 of the Implementing Regulation); and the "Travellers imports/category/exception/requirements" (for the condition in Article 3(2)(a) of the Basic Regulation and Article 4 of the Implementing Regulation).⁹¹

7.2.3 Order of analysis

7.2.3.1 Main arguments of the parties

7.57. While acknowledging that panels should normally first examine the measure in relation to the agreement that deals specifically, and in detail, with the subject matter addressed by the measure at issue, the complainants suggest, for the reasons set out below, that it is open to the Panel to follow the sequence of claims and arguments set out in the complainants' first written submissions in this dispute.⁹²

7.58. Both complainants have presented their claims and arguments under the GATT 1994 first, followed by their claims and arguments under the TBT Agreement. Nothing in these claims and arguments would require the Panel to examine the complainants' claims and arguments under the TBT Agreement before it examines their claims and arguments under the GATT 1994. Thus, a panel may begin with the claims that are common to both parties, therefore examining the GATT 1994 first in this dispute.

7.59. Moreover, Canada invites the Panel to examine the claims under the GATT 1994 first and, should it find that the EU Seal Regime violates Articles I:1 and III:4 of the GATT 1994 and that

⁸⁹ Appellate Body Report, *EC – Asbestos*, para. 64.

⁹⁰ For example, and in contrast to the EU Seal Regime, other EU measures in similar contexts include an explicit ban and exceptions in the text of the measure. (See, e.g. Council Directive 83/129/EEC of 28 March 1983 concerning the importation into Member States of skins of certain seal pups and products derived therefrom, Official Journal of the European Communities, L Series, No. 91 (9 April 1983), (Exhibit CDA-12), Articles 1 and 3; Regulation (EC) No. 1523/2007 of the European Parliament and of the Council of 11 December 2007 banning the placing on the market and the import to, or export from, the Community of cat and dog fur, and products containing such fur, Official Journal of the European Union, L Series, No. 343 (27 December 2007), (Exhibit EU-6), Articles 3 and 4).

We observe that the European Commission Services advised that the Regulation be framed in positive terms so as to make it an internal measure rather than a border measure as this would make a potential challenge at the WTO more difficult. (Non-paper of the Commission Services on the Proposed legislation on trade in seals – WTO Issues (17 April 2009), (Exhibit NOR-28)).

⁹¹ Where appropriate, the Panel will also use these shorthand terms in the sections addressing the main arguments of the parties.

⁹² Complainants' responses to Panel question No. 4; Norway's response to Panel question Nos. 5 and 6.

those violations cannot be justified under Article XX thereof, it may be possible for the Panel to exercise judicial economy with respect to Canada's claims under Article 2.1 of the TBT Agreement.93

7.60. The European Union suggests that the Panel start its analysis with the claims under the TBT Agreement followed by those under the GATT 1994, leaving the analysis under the Agreement on Agriculture for last.⁹⁴ The European Union does not consider that the Panel should take into account the difference in the scope of claims between the complainants in deciding the order of its analysis.95

7.2.3.2 Analysis by the Panel

7.61. The complainants in this dispute raised claims under both the GATT 1994 and the TBT Agreement. Specifically, both complainants brought claims under Articles I:1, III:4, and XI:1 of the GATT 1994 and Articles 2.2, 5.1.2, and 5.2.1 of the TBT Agreement. Additionally, Canada presented a claim under Article 2.1 of the TBT Agreement, and Norway a claim under Article 4.2 of the Agreement on Agriculture.

7.62. The complainants consider that, given the discrepancy in the scope of claims between the two complainants (i.e. Article 2.1 of the TBT Agreement invoked only by Canada), the Panel may wish to start with the GATT 1994 as this is the agreement under which most of the complainants' common claims are presented. Further, in their view, it could give the Panel the possibility to exercise judicial economy with respect to Canada's claim under Article 2.1 of the TBT Agreement.

7.63. The Appellate Body has stated that, as a general rule, panels are free to structure the order of their analysis as they see fit.⁹⁶ Based on this general approach, it is the "structure and logic" of the provisions at issue in each dispute that decide the proper sequence of steps in the panel's analysis, whether the panel's examination involves one provision, or more than one provision or WTO agreement.⁹⁷ In other words, unless there exists a mandatory sequence of analysis which, if not followed, would amount to an error of law and/or affect the substance of the analysis itself, panels have the discretion to structure the order of their analysis.

7.64. Here, the Panel is presented with no such mandatory sequence of analysis. We thus need to determine the order of our analysis by focusing on the "structure and logic" of the provisions at issue in this dispute. We are also mindful that it may be useful for panels to take account of the manner in which a claim is presented to them by a complainant Member.⁹⁸ However, as the Appellate Body has clarified, a panel may also depart from the sequential order suggested by a complaining party.99

7.65. First, turning to our analysis of the "structure and logic" of the provisions at issue, we recall the Appellate Body's statement in *EC-Asbestos*:

We observe that, although the TBT Agreement is intended to "further the objectives of GATT 1994", it does so through a specialized legal regime that applies solely to a limited class of measures. For these measures, the TBT Agreement imposes obligations on Members that seem to be *different* from, and *additional* to, the obligations imposed on Members under the GATT 1994.¹⁰⁰ (emphasis original)

⁹³ Canada's response to Panel question Nos. 4 and 6; Norway's response to Panel question No. 6.

⁹⁴ European Union's response to Panel question No. 4. The European Union refers to the Appellate Body's statement in EC-Bananas III that the provision from the agreement that "deals specifically, and in detail" with the measures at issue should "normally" be analysed first.

Canada also refers to this Appellate Body Report but adds that this discretion did not amount to an inflexible rule that must be followed without regard to the specific circumstances of a given dispute. (See Canada's response to Panel question No. 4, para. 36).

European Union's response to Panel question No. 6.

⁹⁶ Appellate Body Report, *Canada – Wheat Exports and Grain Imports*, paras. 126-129.

⁹⁷ Appellate Body Reports, Canada – Autos, para. 151; and Canada – Wheat Exports and Grain Imports, para. 109.

 ⁹⁸ Appellate Body Report, *Canada – Wheat Exports and Grain Imports*, paras. 126-129.
 ⁹⁹ Appellate Body Report, *US – Zeroing (EC) (Article 21.5 – EC)*, paras. 277-279.

¹⁰⁰ Appellate Body Report, *EC – Asbestos*, para. 80.

- 37 -

7.66. We thus consider that if a measure at issue is found to fall within the scope of the TBT Agreement, it is reasonable for such measure to be examined first under the obligations set out in that agreement. In previous disputes where claims were made under two WTO agreements, panels usually addressed first the claim under the more specific and detailed agreement, in accordance with the guidance from the Appellate Body in EC – Bananas III.¹⁰¹ Following this guidance, in EC - Sardines where claims were made under both the GATT 1994 and the TBT Agreement, the panel considered that "if the [measure at issue] is a technical regulation, then the analysis under the TBT Agreement would precede any examination under the GATT 1994."102 The same approach was followed in all three recent TBT disputes: all three panels addressed nondiscrimination claims under the TBT Agreement first, exercising judicial economy on the complainants' non-discrimination claims under the GATT 1994 where considered appropriate.¹⁰³

7.67. Second, the complainants' challenges against the measure at issue under the TBT Agreement extend to Articles 2.2, 5.1.2, and 5.2.1 and thus are as broad in scope as their common claims under the GATT 1994, which are Articles I:1, III:4, and XI:1. As such, it would not necessarily prove more efficient to proceed with one agreement rather than another because the claims under both agreements are extensive.

7.68. Finally, we are not persuaded of the possibility of exercising judicial economy with respect to Canada's claim under Article 2.1 of the TBT Agreement in this case. This is particularly so when there are several claims under each agreement and given that, as mentioned by the Appellate Body in US - Tuna II (Mexico), the obligations under the non-discrimination provisions of the TBT Agreement and the GATT 1994 cannot be assumed to be the same.

7.69. In light of the above, we do not consider that starting with the complainants' claims under the GATT 1994 would be the most logical or economical order of analysis under the circumstances of this dispute. We therefore consider it appropriate to start our analysis with the complainants' claims under the TBT Agreement, followed by those under the GATT 1994.

7.70. Before turning to our examination of the claims made with respect to the EU Seal Regime, we address the preliminary matter of Norway's request to the Panel under Article 13 of the DSU.

7.2.4 Norway's request under Article 13 of the DSU

7.71. As described in Section 1, on 16 January 2013, Norway submitted a request for the Panel to exercise its authority under Article 13 of the DSU to seek copies of two legal opinions of the Legal Service of the Council of the European Union (the Opinions). These were also the documents at issue in the European Union's request for a preliminary ruling.¹⁰⁴ In its letter of 8 April 2013, the Panel informed the parties of its decision to deny Norway's request. As indicated in the letter, we are providing the reasons for our decision in these Reports.¹⁰⁵

7.72. In its request, Norway contends that the legal opinions would "help complete the record"¹⁰⁶ in that publicly available material on the record refers to the Opinions and the original documents would allow the Panel to confirm attributed statements and provide proper context. Norway also argues that it is "fair"¹⁰⁷ to request the Opinions from the European Union because it is not seeking *public* disclosure but rather disclosure within *confidential* WTO proceedings.¹⁰⁸ In addition, Norway

¹⁰¹ Appellate Body Report, *EC – Bananas III*, para. 204: "Although Article X:3(a) of the GATT 1994 and Article 1.3 of the Licensing Agreement both apply, the Panel, in our view, should have applied the Licensing Agreement first, since this agreement deals specifically, and in detail, with the administration of import licensing procedures. If the Panel had done so, then there would have been no need for it to address the alleged inconsistency with Article X:3(a) of the GATT 1994." See also Panel Reports, EC – Sardines, paras. 7.15-7.16; and *EC – Asbestos*, paras. 8.16-8.17. ¹⁰² Panel Reports, *EC – Sardines*, para. 7.15-7.16; and *EC – Asbestos*, paras. 8.16-8.17.

¹⁰³ Panel Reports, US – Clove Cigarettes; US – Tuna II (Mexico); and US – COOL.

¹⁰⁴ See, above Section 1, paras. 1.20-1.21. In the preliminary ruling, the Panel granted the

European Union's request to remove the documents from the record as exhibits and directed the parties as to how to proceed in light of the rulings. (WT/DS400/6 and WT/DS401/7).

¹⁰⁵ See, above Section 1, para. 1.22.

¹⁰⁶ Norway's Request under Article 13 of the DSU, section III.A, paras. 10-11.

¹⁰⁷ Ibid. section III.B, paras. 12-33.

¹⁰⁸ In connection with this argument, Norway disputes the confidential nature of the Opinions under EU law and stresses the public circulation of the Opinions evidenced in other public documents.

asserts that the Opinions are in the possession of the European Union and that "the most reasonable means available to the Panel"¹⁰⁹ to access these documents would be to request them from the European Union. Finally, Norway contends that disclosure of the Opinions is necessary to ensure due process and proper adjudication of Norway's claims. Norway primarily claims that the Opinions are "evidence of certain facts relevant to Norway's claims"¹¹⁰ and therefore are not being relied upon for their legal conclusions.

7.73. The European Union responds that the Council of the European Union had not authorized disclosure of the Opinions, and therefore persons or entities in possession of the Opinions had obtained them unlawfully.¹¹¹ Further, the European Union argues that the Opinions are not necessary for the proper adjudication of the dispute because the Council Legal Service, which prepared the Opinions, lacks both the authority and capacity to make factual findings. Although Norway portrays the Opinions as constituting factual evidence, the European Union counters that all factual material is derivative of information supplied to the Council Legal Service and already on the record before the Panel. Beyond this, the issues for which Norway cites factual relevance are in reality issues of legal characterization and conclusion rather than factual matters. The European Union also contends that it would be "unfair" to request the Opinions as it "could be required to take position against the legal advice received in confidence from one of its legal services"¹¹², which could risk undermining its right to a fair hearing.

7.74. Canada considers that Norway's request for information is well-founded and that "the designation or classification of that information as confidential by the party in possession of it is not a barrier to the Panel requesting it".113

7.75. Further to the Panel's invitation for comments from third parties, the United States emphasizes that a panel should not use its authority under Article 13 of the DSU to make a party's prima facie case and considers that Norway's statements regarding the Opinions' factual relevance to its claims would raise questions in this regard.¹¹⁴ The United States also notes the very sensitive issues implicated by Norway's request, such as the relevance of domestic law and evidentiary status, and that under the present circumstances believes it neither necessary nor appropriate to address arguments or make findings on such issues.¹¹⁵

7.76. Article 13 of the DSU provides in relevant part:

Article 13: Right to Seek Information

Each panel shall have the right to seek information and technical advice from 1. any individual or body which it deems appropriate. However, before a panel seeks such information or advice from any individual or body within the jurisdiction of a Member it shall inform the authorities of that Member. A Member should respond promptly and fully to any request by a panel for such information as the panel considers necessary and appropriate. Confidential information which is provided shall not be revealed without formal authorization from the individual, body, or authorities of the Member providing the information.

2. Panels may seek information from any relevant source ...

7.77. The Appellate Body has interpreted Article 13 of the DSU as "a grant of discretionary authority" for panels that enables them to seek information from any source, including from a Member who is party to a dispute.¹¹⁶ Moreover, "a panel is vested with ample and extensive

¹¹⁴ United States' comments on Norway's Request, dated 8 February 2013, para. 3.

¹¹⁵ Ibid. para. 5.

¹⁰⁹ Norway's Request under Article 13 of the DSU, para. 37.

¹¹⁰ Ibid. para. 39.

¹¹¹ European Union's comments on Norway's Request, dated 8 February 2013, paras. 3-10. The European Union also described the status of the documents in relation to the unauthorized disclosure and challenged various contentions by Norway with respect to EU law. ¹¹² Ibid. para. 28.

¹¹³ Canada's comments on Norway's Request, dated 8 February 2013 (these comments contain no paragraph numbering).

¹¹⁶ Appellate Body Report, Canada – Aircraft, paras. 184-185 (citing Appellate Body Report, Argentina – Textiles and Apparel, para. 84).

discretionary authority to determine *when* it needs information and *what* information it needs".¹¹⁷ Thus, the exercise of authority under Article 13 of the DSU is to be made with regard to the particular facts and circumstances of each case, including "what information is needed to complete the record, whose possession it lies within, what other reasonable means might be used to procure it, why it has not been produced, whether it is fair to request the party in possession of the information to submit it, and whether the information or evidence in question is likely to be necessary to ensure due process and a proper adjudication of the relevant claim(s)".¹¹⁸

7.78. In assessing Norway's request, therefore, the Panel considered Norway's request in the light of the Appellate Body's guidance, finding particularly relevant in this case the considerations of the need for the requested information for the Panel's assessment of the matter before it and the consistency with due process for all parties.

7.79. Regarding the need for the information requested by Norway, we recall that Norway submitted replacement exhibits for the Opinions following the Panel's preliminary ruling expunging them from the record. Indeed, Norway noted prior to the Panel issuing its preliminary ruling that "the facts revealed by the relevant exhibits can also be demonstrated, albeit in a less direct and immediate manner, by other evidence".¹¹⁹ Having reviewed the replacement exhibits filed by Norway in light of the claims before us, we did not consider that the requested information was needed to complete the record.¹²⁰ We also noted Norway's additional arguments that a proper adjudication of the relevant claims requires disclosure of these documents. While Norway highlighted various points concerning the factual relevance of the views of the Council Legal Service to Norway's claims, it did so by reference to publicly available information (including that found in its replacement exhibits) and specific arguments as to what it considered such public information to reveal. To the extent, if at all, that the views of the Council Legal Service may be factually relevant to Norway's claims, we did not consider that requesting the Opinions from the European Union was necessary for the Panel to make an objective assessment of the matter before it.

7.80. As to due process, we observed in our preliminary ruling that the complainants had agreed to withdraw the exhibits at issue and that their due process rights would not be affected by the removal of the exhibits from the record. We further determined that the complainants would have an opportunity to provide replacement evidence.¹²¹ Norway specifically referred to its due process concerns when it requested that the Panel allow replacement of the evidence it had agreed to withdraw and "an opportunity to address the resulting incomplete sentences and paragraphs in its submission".¹²² In our preliminary ruling, therefore, we invited the complainants to provide "a brief explanation of [the replacement exhibits'] relevance to the complainants' arguments, referring to the relevant paragraphs in their respective first written submissions".¹²³ Pursuant to this invitation, Norway filed its replacement exhibits and explanation of their relevance with cross-references to its first written submission. We therefore considered that Norway had been "permitted to make its case before the Panel"¹²⁴ and that the requirements of due process had been fully satisfied.

7.81. In light of the above, we did not consider the circumstances of Norway's request to warrant the exercise of authority under Article 13 of the DSU and thus decided to deny Norway's request.

¹¹⁷ Appellate Body Report, *Canada – Aircraft*, para. 192. (emphasis original)

The Appellate Body has also underscored the "comprehensive nature" of this authority and consistently affirmed that the grant of discretionary authority under Article 13 of the DSU is "indispensably necessary" to enable a panel to discharge the duty imposed by Article 11 of the DSU to "make an objective assessment of the matter before it". See Appellate Body Reports, *US* – *Shrimp*, paras. 104, 106; *Japan – Agricultural Products II*, para. 127; and *US – Continued Zeroing*, para. 345.

II, para. 127; and US – Continued Zeroing, para. 345.
 ¹¹⁸ Appellate Body Report, US – Large Civil Aircraft (Second Complaint), para. 1140, cited in Norway's Request under Article 13 of the DSU, para. 6.

¹¹⁹ Norway's letter of 16 January 2013 to the Panel.

¹²⁰ With specific regard to Norway's argument that disclosure would provide proper context and confirm attributed statements in public materials, we note that Norway has already explained the nature and relevance of its replacement exhibits and the contents thereof have not been called into question by any other party.

¹²¹ WT/DS400/6 and WT/DS401/7, Preliminary Ruling of the Panel, para. 3.3.

¹²² Norway's letter of 16 January 2013 to the Panel.

¹²³ WT/DS400/6 and WT/DS401/7, Preliminary Ruling of the Panel, para. 3.7.

 $^{^{124}}$ Norway's Request under Article 13 of the DSU, para. 38.

7.3 Claims under the TBT Agreement

7.82. In this section, the Panel examines the complainants' claims under the TBT Agreement. Both Canada and Norway presented claims under Articles 2.2, 5.1.2, and 5.2.1. Additionally, Canada presented a claim under Article 2.1.

7.83. Before considering the complainants' claims under the TBT Agreement , the Panel must first determine whether the EU Seal Regime constitutes a "technical regulation" within the meaning of Annex 1:1 of the TBT Agreement and thus falls within the scope of the Agreement.

7.3.1 Whether the EU Seal Regime is a technical regulation within the meaning of the TBT Agreement

7.84. The term "technical regulation" is defined in Annex 1.1 of the TBT Agreement as follows:

Document which lays down product characteristics or their related processes and production methods, including the applicable administrative provisions, with which compliance is mandatory. It may also include or deal exclusively with terminology, symbols, packaging, marking or labelling requirements as they apply to a product, process or production method. (explanatory note omitted)

7.85. Based on this definition, the Appellate Body has developed a three-tier test to establish whether a document qualifies as a technical regulation:

[There are] *three criteria* that a document must meet to fall within the definition of 'technical regulation' in the *TBT Agreement. First*, the document must apply to an identifiable product or group of products. The *identifiable* product or group of products need not, however, be expressly *identified* in the document. *Second*, the document must lay down one or more characteristics of the product. These product characteristics may be intrinsic, or they may be related to the product. They may be prescribed or imposed in either a positive or a negative form. *Third*, compliance with the product characteristics must be mandatory. [T]hese three criteria are derived from the wording of the definition in Annex 1.1.¹²⁵ (emphasis original)

7.86. The parties do not contest that the EU Seal Regime meets the first and third criteria of the definition, i.e. that it applies to an identifiable group of products and that compliance with the measure is mandatory. The parties disagree, however, on the second criterion of the definition, namely whether the EU Seal Regime "lays down product characteristics or their related processes and production methods [(PPMs)], including applicable administrative provisions".

7.87. The criteria of the definition are cumulative, and there is no particular order of analysis that we need to follow in assessing whether the EU Seal Regime is a technical regulation.¹²⁶ Accordingly, we start our analysis with the second criterion of the definition, which is the main issue of contention as to whether the measure at issue qualifies as a technical regulation. We then address the two other elements of the three-tier test that are not disputed by the parties.

7.3.1.1 Whether the EU Seal Regime lays down one or more characteristics of the products or their related PPMs, including applicable administrative provisions

7.3.1.1.1 Main arguments of the parties

7.3.1.1.1.1 Complainants

7.88. The complainants argue that the EU Seal Regime lays down product characteristics in both positive and negative form. If a product meets the requirements of the IC, MRM, or Travellers categories (i.e. the exceptions), it may possess the characteristic of containing seal. Conversely, if

¹²⁵ Appellate Body Report, *EC* – *Sardines*, para. 176 (summarizing the Appellate Body's interpretation of the definition of "technical regulation" in *EC* – *Asbestos*, paras. 66-70). The same test was applied by the Appellate Body in *US* – *Clove Cigarettes*, *US* – *Tuna II (Mexico)*, and *US* – *COOL*.

¹²⁶ Panel Report, US – COOL, para. 7.149.

the product does not meet such requirements, then it may not contain seal.¹²⁷ In determining whether the EU Seal Regime lays down product characteristics, it is not necessary for the exceptions themselves to prescribe product characteristics. In the complainants' view, the issue is whether the exceptions, combined with other elements of the measure, lay down product characteristics.¹²⁸

7.89. The complainants further argue that because products falling within one of the three categories must satisfy certain administrative requirements set out in the Implementing Regulation in order to be placed on the European Union market, the EU Seal Regime also sets forth "applicable administrative provisions" within the meaning of Annex 1.1 of the TBT Agreement.¹²⁹

7.90. As an alternative to the argument that the measure lays down product characteristics, Norway argues that the EU Seal Regime prescribes related PPMs within the meaning of Annex 1.1. Based on the ordinary meaning of the terms of the definition¹³⁰, Norway asserts that a PPM is laid down through the IC and MRM exceptions.¹³¹ In particular, with respect to the IC category, Norway argues that the IC requirements prescribe a "process" involving a particular course of action (a traditional hunt by specified persons) with a defined end (the production of seal products for community subsistence). Regarding the MRM category, Norway argues that the measure imposes a particular course of action relating to the purpose of the hunt (sustainable marine resource management); the way in which the hunt is conducted (regulated at national level pursuant to a resource management plan); and the way in which the seal products are marketed (not-for-profit, non-commercial nature and quantity). Furthermore, the action also has a defined end (the sale of MRM by-products).¹³²

7.91. Canada argues for its part that the identity of the producers of a product could be a relevant factor in the identification of a PPM. In particular, Canada notes that "certain elements of the Inuit Communities category could be characterized as processes or production methods".¹³³

7.3.1.1.2 Respondent

7.92. The European Union contests that the EU Seal Regime lays down product characteristics pursuant to the definition set out in Annex 1.1.¹³⁴ The European Union first argues that the EU Seal Regime prohibits the placing on the market of products which consist exclusively of seal, such as "pure" seal meat, oil, blubber, organs and fur skins, whether processed or not.¹³⁵ The European Union asserts that this prohibition under the EU Seal Regime is similar to the prohibition of asbestos fibres "as such" in the measure at issue in *EC – Asbestos*, which the Appellate Body found did not constitute a technical regulation.

7.93. As regards products containing seal and other ingredients ("mixed" products), the European Union argues that it would be inappropriate for the Panel to limit its analysis to the fact that the EU Seal Regime lays down intrinsic characteristics in the negative form, by providing that all products may not contain seal. The determination of whether the EU Seal Regime lays down product characteristics should also take into account the exceptions, because it is the permissive elements, *together* with the prohibition, that determine the situations where seal products may be placed on the European Union market.¹³⁶

¹²⁷ Canada's first written submission, para. 363; Norway's first written submission, para. 499; complainants' responses to Panel question No. 127.

¹²⁸ See, e.g. Norway's second written submission, para. 145.

¹²⁹ Canada's first written submission, paras. 364-365; Norway's first written submission,

paras. 502-503. ¹³⁰ Norway defines the term "processes" as "a systematic series of actions or operations directed to

some end". (Norway's second written submission, para. 161 (citing the Panel Report in *EC – Trademarks and Geographical Indications (Australia)*, para. 7.510)). ¹³¹ Norway's opening statement at the first meeting of the Panel, para. 64; second written submission,

¹³¹ Norway's opening statement at the first meeting of the Panel, para. 64; second written submission, para. 161.

¹³² Norway's second written submission, para. 162.

¹³³ Canada's response to Panel question No. 21, para. 109.

¹³⁴ European Union's first written submission, para. 199.

¹³⁵ European Union's first written submission, para. 213.

¹³⁶ European Union's first written submission, para. 216.

- 42 -

7.94. According to the European Union, what is decisive for the characterization of the EU Seal Regime is that none of the three exceptions lays down product characteristics. The IC exception concerns the type of hunters, the traditions of their communities, and the purpose of the hunt, but not the intrinsic or related features of the products, such as their composition or presentation.¹³⁷ The MRM exception imposes requirements relating to the size of the hunt, the intensity and purpose of the hunt, and the marketing conditions of the products. In the European Union's view, none of these conditions sets out intrinsic or related features of the products.¹³⁸ The European Union argues that the EU Seal Regime differs in that sense from the measure in EC - Asbestos, where the exceptions themselves referred to particular characteristics intrinsic to the product.¹³⁹

7.95. With respect to the question whether the EU Seal Regime prescribes applicable administrative provisions, the European Union contends that Annex 1.1 addresses only those administrative provisions that apply to "product characteristics or their related PPMs". Given that the procedural requirements set out in the Implementing Regulation are not related to any product characteristics or their related PPMs, such provisions, according to the European Union, do not constitute "applicable administrative provisions" within the meaning of Annex 1.1.¹⁴⁰

7.96. Finally, the European Union argues that the EU Seal Regime does not regulate any PPMs.¹⁴¹ The European Union contends that the measure does not set out methods for the production of seal products, compliance with which would allow their placing on the market. The ban, together with the exceptions, allows the placing on the market of seal products depending on the purpose of the hunt, which has nothing to do with methods for the production of seal products.¹⁴² According to the European Union, to include the purpose of production within the meaning of "PPM" would improperly stretch the limit of the concept of "product characteristics and related PPMs".¹⁴³

7.3.1.1.2 Analysis by the Panel

7.3.1.1.2.1 Aspects of the EU Seal Regime to be examined

7.97. The parties agree in principle that in determining whether the EU Seal Regime qualifies as a technical regulation, the Panel should consider the measure "as a whole".¹⁴⁴ However, the parties disagree on whether both the prohibition and the exceptions under the Regime must individually lay down product characteristics or their related PPMs in order for the measure to qualify as a technical regulation.

7.98. The complainants contend that the exceptions *per se* do not have to lay down product characteristics or their related PPMs, as long as the measure as a whole, i.e. through one of its components, meets the criterion. The European Union submits that a measure cannot be characterized as laying down product characteristics on the basis of its prohibitive element alone. If the measure contains not only a ban but also exceptions, these permissive elements should also be taken into account in determining whether the measure as a whole qualifies as a technical regulation.¹⁴⁵

¹³⁷ European Union's first written submission, para. 220; response to Panel question No. 127.

¹³⁸ European Union's first written submission, para. 221.

¹³⁹ European Union's first written submission, para. 224.

¹⁴⁰ European Union's first written submission, paras. 229-234.

¹⁴¹ The European Union observes that the European Commission's proposal would have allowed the placing on the market of seal products upon proof that these were obtained under conditions which ensured that the seals were killed and skinned "without causing avoidable pain, distress and any other form of suffering", which is not the case under the current Regime. (European Union's first written submission, paras. 226-227).

¹⁴² European Union's second written submission, para. 190.

¹⁴³ European Union's response to Panel question No. 21.

¹⁴⁴ See, e.g. Canada's first written submission, para. 351; Norway's second written submission,

para. 145; and European Union's first written submission, para. 208. See also Appellate Body Report, *EC - Asbestos*, para. 64.

¹⁴⁵ Norway's second written submission, para. 145; and European Union's first written submission, para. 208.

7.99. We recall that, in *EC* – *Asbestos*, the Appellate Body emphasized that the measure should be examined as a whole "taking into account, as appropriate, the prohibitive and the permissive elements that are part of it".¹⁴⁶ Based on that premise, the Appellate Body examined both the prohibitive and permissive aspects of the measure in that dispute and found:

Viewing the measure as an integrated whole, we see that it lays down characteristics for all products that might contain asbestos, and we see also that it lays down the 'applicable administrative provisions' for certain products containing chrysotile asbestos fibres which are excluded from the prohibitions in the measure. For these reasons, we conclude that the measure constitutes a technical regulation under the *TBT Agreement*.¹⁴⁷

7.100. In our view, the Appellate Body's analysis of the measure at issue in EC – Asbestos does not suggest that for a measure consisting of a ban and certain exceptions to qualify as a technical regulation, both the prohibition and the exceptions must individually lay down product characteristics or their related PPMs.

7.101. A panel may have to examine different components of a measure separately in order to make a holistic analysis of the measure's legal character. However, the final decision on the character of the measure must be based on the measure as a whole, "taking into account, as appropriate, the prohibitive and permissive elements that are part of it".¹⁴⁸

7.102. With these considerations in mind, we proceed to examine the prohibitive and permissive aspects of the EU Seal Regime with a view to determining whether the EU Seal Regime, taken as a whole, lays down product characteristics or their related PPMs within the meaning of Annex 1.1.

7.3.1.1.2.2 Whether the EU Seal Regime lays down product characteristics or their related PPMs, including applicable administrative provisions

7.103. The Appellate Body defined the term "characteristics" in EC – Asbestos as "any objectively definable 'features', 'qualities', 'attributes', or other 'distinguishing mark' of a product".¹⁴⁹ Such "characteristics" include not only features and qualities that are intrinsic to the product itself, but also related characteristics, "such as the means of identification, the presentation and the appearance of a product".¹⁵⁰ The meaning of the phrase "their related PPMs" has not yet been examined in a WTO dispute.

7.104. In *EC* – *Asbestos*, the Appellate Body found that the prohibition on asbestos fibres as such did not, in itself, lay down any "characteristics" because it simply banned asbestos fibres in their natural state.¹⁵¹ The prohibition on asbestos-containing products, however, was found to lay down a product characteristic in the negative form by requiring that all products must not contain asbestos.

7.105. As noted above, the EU Seal Regime prohibits all seal products, whether they are made exclusively of seal or contain seal as an input. The Regime makes an exception with regard to the import and/or placing on the market of seal products in three situations, namely when they result from IC hunts, MRM hunts, or in the case of Travellers imports.¹⁵² The Implementing Regulation sets out the specific requirements that seal products must fulfil in each of these three situations.

¹⁴⁶ Appellate Body Report, *EC* – *Asbestos*, para. 64. In reaching this conclusion, the Appellate Body took into account the content of Canada's request for the establishment of a panel (i.e. Canada's identification of the Decree concerned as "the measure at issue") as well as the content of the measure itself (consisting of prohibitions and limited exceptions). The Appellate Body then examined each component (i.e. prohibitions and exceptions) of the measure before making an overall assessment of whether the measure, viewed as an integrated whole, was a "technical regulation" within the meaning of Annex 1.1.

¹⁴⁷ Appellate Body Report, *EC – Asbestos*, para. 75.

¹⁴⁸ Canada's response to Panel question No. 126, paras. 77 and 83.

¹⁴⁹ Appellate Body Report, *EC – Asbestos*, para. 67.

¹⁵⁰ Appellate Body Report, *EC – Asbestos*, para. 67.

¹⁵¹ Appellate Body Report, *EC – Asbestos*, para. 71.

¹⁵² For instance, Article 3(1) of the Basic Regulation provides that "[t]he placing on the market shall be allowed only where ..."; Article 3(2)(b) states that "the placing on the market shall also be allowed where ..."; and Article 3(2)(a) provides that "the import of seal products shall be allowed where ...".

7.106. Based on the text of the Regulations, and in light of the reasoning of the Appellate Body in EC – Asbestos, we believe that the prohibition on seal-containing products under the EU Seal Regime lays down a product characteristic in the negative form by requiring that all products not contain seal.153

7.107. Further, the Appellate Body considered that, through its exceptions, the measure in *EC* - *Asbestos* set out the "applicable administrative provisions, with which compliance is mandatory' for products with certain objective 'characteristics'".¹⁵⁴ On this question, the Appellate Body relied on the panel's finding that the marketing criteria applying to products falling under the exceptions "relate to the characteristics of one or more given products or processes or production methods relating to them".¹⁵⁵ According to these criteria, products containing chrysotile asbestos could be marketed provided that there was no substitute fibre available (i) that presented less occupational health risk to workers than chrysotile fibre; and (ii) that met all technical guarantees of safety appropriate to the use. The panel noted that such requirements had to be read in conjunction with administrative provisions requiring a statement and supporting documents to attest that the criteria of the exceptions were satisfied.

7.108. Similarly, we find that the EU Seal Regime sets out, through its exceptions, the "applicable administrative provisions with which compliance is mandatory" for products with certain objective "characteristics". First, the exceptions define the scope of the prohibition in the EU Seal Regime, albeit implicitly. Second, the nature of the exceptions is to allow products containing seal on the EU market, subject to compliance with strict administrative requirements. Finally, the scope of the exceptions is determined under the Regime based on a set of criteria.

7.109. Specifically, in order to fall under the IC or MRM exceptions, products containing seal must meet *inter alia* the following criteria relating to seal hunts from which the seals used as their input are derived: the identity of the hunter (Inuit or indigenous); the type of hunt (traditional Inuit hunts); the purpose of the hunt (subsistence or marine resource management); and the way in which the products are marketed (non-systematically and on a non-profit basis).¹⁵⁶ Any person wishing to import and/or place seal products on the market under these exceptions must have such products certified by a recognized body as meeting the necessary criteria under each exception. Furthermore, the products must be accompanied by an attesting document at the time of placing on the market.¹⁵⁷ In addition, with respect to the particular unit of seal products for which it is issued, the attesting document indicates whether the products result from hunts conducted by Inuit or other indigenous communities, or from hunts for the sustainable management of marine resources.¹⁵⁸

7.110. The criteria under the exceptions thus identify the seal products that are allowed to be placed on the European Union market. They do so by defining the categories of seal that can be used as an input for such products; only seals obtained from the specific type of hunter and/or the

¹⁵³ We note that such conclusion is not affected by the fact that the prohibition of seals "in their natural state" might not, in itself, prescribe or impose any "characteristics". In this regard, Norway argues that the appropriate analogues to the "raw mineral form" of asbestos in the context of the EU Seal Regime would be live seals or unprocessed seal carcasses. In Norway's view, the majority of seal products are in fact "mixed" products, i.e. they must be combined with other products derived from other sources. (Norway's second written submission, paras. 154-155).

¹⁵⁴ Appellate Body Report, *EC* – *Asbestos*, para. 74 (quoting Panel Report, para. 8.69). The panel in that dispute found that:

[[]A]rticle 2 of the Decree sets out the criteria for marketing the products identified in the Decree and not solely the criteria for excluding products from the market. The second sentence in Article 3.I of the Decree completes these criteria. In our view, the marketing criteria in Article 2.I of the Decree relate to the characteristics of one or more given products or processes or production methods relating to them. This is particularly true of the second subparagraph on the technical guarantees of safety appropriate to use ... We also note that Article 2.II and Article 3 in particular cover the administrative provisions applicable to the technical regulations. (Panel Report, EC – Asbestos, paras. 8.68-8.69, (cross-referencing para. 8.1 of the Panel Report)). ¹⁵⁵ Panel Report, *EC – Asbestos*, para. 8.69.

¹⁵⁶ The conditions for a seal product to qualify under the IC and the MRM exceptions are described above in section 7.2.1.

¹⁵⁷ See Implementing Regulation, Articles 3(2), 5(2), and 7(1). A model form for the attesting document is annexed to the Implementing Regulation. ¹⁵⁸ European Union's response to Panel question No. 160, para. 245. See Annex of the Implementing

Regulation, item 7.

qualifying hunts may be used in making final products. These criteria in our view constitute "objectively definable features" of the seal products that are allowed to be placed on the EU market and consequently lay down particular "characteristics" of the final products. Therefore, as was the case in EC – *Asbestos*, the exceptions under the EU Seal Regime identify a group of products with particular "characteristics" through a narrowly defined set of criteria.

7.111. In sum, the EU Seal Regime considered as a whole lays down characteristics for all products that might contain seal. The Regime also lays down the applicable administrative provisions for certain products containing seal inputs that are exempted from the prohibition under the measure.

7.112. We recall that in order to meet this criterion of the definition of technical regulation under Annex 1.1 of the TBT Agreement, the complainants must prove that the document lays down either "product characteristics" or "their related PPMs". Since we have found that the measure as a whole lays down product characteristics within the meaning of Annex 1.1 of the TBT Agreement, we do not find it necessary to examine whether the EU Seal Regime also lays down PPMs.

7.3.1.2 Whether the EU Seal Regime applies to an identifiable product or group of products

7.113. The parties do not contest that the EU Seal Regime applies to an identifiable group of products. The Regime determines whether products may or may not contain seal, depending on whether they meet the conditions of the IC, MRM, or Travellers imports exceptions.¹⁵⁹ The range of products covered by the European Union's measure is identifiable by virtue of the presence or absence of the characteristic of being derived or manufactured from, or of containing, seal.

7.114. The EU Seal Regime establishes rules concerning the placing on the market of seal products.¹⁶⁰ The term "seal products" is defined in the Basic Regulation as "all products, either processed or unprocessed, deriving or obtained from seals, including meat, oil, blubber, organs, raw fur skins and fur skins, tanned or dressed, including fur skins assembled in plates, crosses and similar forms, and articles made from fur skins".¹⁶¹ As discussed in section 7.2.2.2 above, in order to be imported and/or placed on the European Union market, products may not contain seal, unless they meet the conditions under the IC, MRM, or Travellers' imports exceptions.

7.115. We note that in *EC* – *Asbestos*, the measure at issue also prescribed a characteristic that effectively applied to all products, namely that they must not contain asbestos. Although the prohibition applied to a large group of products which could not be determined from the terms of the measure itself, the Appellate Body found that the measure applied to an "identifiable" group of products. Like the measure in *EC* – *Asbestos*, the prohibition under the EU Seal Regime applies to an identifiable group of products by prescribing that all products may not contain seal.¹⁶²

7.116. In addition, we note that numerous product categories to which the EU Seal Regime applies were identified in the European Commission's Technical Guidance Note.¹⁶³ In our view, this list of products, albeit indicative, is further evidence that the EU Seal Regime applies to an identifiable group of products.¹⁶⁴

¹⁵⁹ Canada's first written submission, para. 356; Norway's first written submission, para. 496.

¹⁶⁰ Basic Regulation, Article 1.

¹⁶¹ Basic Regulation, Article 2(2).

¹⁶² The subject matter of the Basic Regulation is to establish harmonized rules concerning the placing on the market of seal products. (See Basic Regulation, Article 1).

¹⁶³ Canada's first written submission, para. 358; Norway's first written submission, para. 497 (referring to Exhibit JE-3). The document submitted in Exhibit JE-3 was issued pursuant to Article 3(3) of the Basic Regulation. According to recital (18) of the preamble of the Basic Regulation, such technical guidance notes were to be issued by the Commission "[w]ith the aim of facilitating enforcement operations carried out by the relevant national authorities".

¹⁶⁴ As stated in the "Foreword" section of the document, the tariff codes identified in the list are those that have the "greatest likelihood of covering products subject to the prohibition". The European Commission also indicates in the document that more products than those covered by the tariff codes identified in the list are likely to be affected by the prohibition. (Technical Guidance Note, (Exhibit JE-3), p. 44).

7.117. In light of the above, the Panel considers that the EU Seal Regime applies to an "identifiable group of products" in accordance with Annex 1.1 to the TBT Agreement.¹⁶⁵

7.3.1.3 Whether compliance with the EU Seal Regime is mandatory

7.118. The Appellate Body in *EC* – *Asbestos* clarified the concept of "mandatory" under Annex 1.1 as follows:

A "technical regulation" must ... regulate the "characteristics" of products in a binding or compulsory fashion. It follows that, with respect to products, a "technical regulation" has the effect of *prescribing* or *imposing* one or more "characteristics" – "features", "qualities", "attributes", or other "distinguishing mark".¹⁶⁶ (emphasis original)

7.119. The Appellate Body also found that enforceability through the application of sanctions indicated mandatory compliance. $^{\rm 167}$

7.120. As we stated above in the section on Preliminary Matters, the combined effect of the Basic and Implementing Regulations is to prohibit seal products from the European Union market, except in cases where the products meet the conditions prescribed in Article 3 of the Basic Regulation and Articles 3 and 5 of the Implementing Regulation. These conditions are compulsory from the point of view of seal products being placed on the market; unless these conditions are met, seal products are denied access to the EU market.

7.121. We further note that the Basic Regulation contains language of a mandatory nature. For example, Article 3(1) provides that "[t]he placing on the market of seal products shall be allowed *only* where ...".¹⁶⁸ By way of derogation from Article 3(1), Article 3(2) also sets out circumstances where the placing on the market of seal products "shall be allowed". The use of the words "shall" and "shall only" in the above-mentioned provisions indicate that the terms of the provisions are obligatory. Several provisions of the Implementing Regulation contain similar wording of a mandatory nature.¹⁶⁹

7.122. The European Union's Regime is also supported by enforcement measures, as penalties may apply in case of infringement of the regulation. In particular, under Article 6 of the Basic Regulation, "[EU] Member States shall lay down the rules on penalties applicable to infringements of [the] Regulation".¹⁷⁰

7.123. Finally, we note that both the Basic and Implementing Regulations state that "[the] Regulation shall be binding in its entirety and directly applicable in all Member States." The Appellate Body in EC – Sardines interpreted similar wording contained in the measure at issue in that case as meaning that compliance with the regulation was mandatory.¹⁷¹

7.124. In light of these considerations, the Panel is of the view that the EU Seal Regime is mandatory within the meaning of the definition in Annex 1.1 of the TBT Agreement.

7.125. Based on our analysis of the three criteria set out in Annex 1.1 of the TBT Agreement, we find that the EU Seal Regime is a document which "lays down product characteristics

¹⁶⁵ Appellate Body Report, *EC – Asbestos*, para. 70. See also Appellate Body Report, *EC – Sardines*, para. 180.

¹⁶⁶ Appellate Body Report, *EC – Asbestos*, para. 68.

¹⁶⁷ Appellate Body Report, *EC – Asbestos*, para. 72.

¹⁶⁸ Basic Regulation, Article 3(1). (emphasis added)

¹⁶⁹ For example, Articles 3(1) and 5(1) of the Implementing Regulation provide that "[s]eal products... may only be placed on the market where ...", and Articles 3(2) and 5(2) provide that "the seal product shall be accompanied by the attesting document ...". Article 4 also states that seal products for the personal use of European Union residents and their families "may only be imported where one of the following requirements is fulfilled". The use of the terms "may only" and "shall" in the above-mentioned provisions indicates that compliance with such provisions is obligatory rather than voluntary.

 ¹⁷⁰ According to this provision and as emphasized in recital (19) of the preamble of the Basic Regulation, it is for the European Union member States to lay down rules on penalties and to ensure their implementation.
 ¹⁷¹ Canada's first written submission, para. 372; Norway's first written submission, para. 508 (referring

^{1/1} Canada's first written submission, para. 372; Norway's first written submission, para. 508 (referring to Appellate Body Report, *EC* – *Sardines*, para. 194).

- 47 -

including the applicable administrative provisions, with which compliance is mandatory". Accordingly, the EU Seal Regime constitutes a technical regulation within the meaning of Annex 1.1 of the TBT Agreement.

7.3.2 Canada's claim under Article 2.1¹⁷²

7.126. We recall our finding in the previous section that the EU Seal Regime is a technical regulation within the meaning of Annex 1.1 of the TBT Agreement. As such, the EU Seal Regime is subject to the obligations set forth in Article 2.1 of the TBT Agreement.

7.127. Article 2.1 of the TBT Agreement provides that:

With respect to their central government bodies ... Members shall ensure that in respect of technical regulations, products imported from the territory of any Member shall be accorded treatment no less favourable than that accorded to like products of national origin and to like products originating in any other country.

7.128. We note that Article 2.1 contains a most-favoured-nation (MFN) and a national treatment obligation. In the present dispute, Canada makes claims in respect of both obligations. With respect to the MFN treatment obligation, Canada contends that the EU Seal Regime gives less favourable treatment to Canadian imports of seal products than to like seal products originating from Greenland. Regarding the national treatment obligation, Canada argues that the EU Seal Regime gives less favourable treatment to its imports of seal products as compared to the treatment accorded to like domestic products.

7.129. In order to establish that the EU Seal Regime is inconsistent with Article 2.1 of the TBT Agreement, Canada must demonstrate the following: (a) the imported and domestic/other foreign products at issue are like products; and (b) the treatment accorded to imported products is less favourable than that accorded to like domestic and/or other foreign products (less favourable treatment).¹⁷³

7.130. The Appellate Body explained the meaning of the term "less favourable treatment" under Article 2.1 of the TBT Agreement as follows: "[A] panel examining a claim of violation under Article 2.1 should seek to ascertain whether the technical regulation at issue modifies the conditions of competition in the market of the regulating Member to the detriment of the group of imported products *vis-à-vis* the group of like domestic products".¹⁷⁴ The Appellate Body added that "[h]owever ... the context and object and purpose of the TBT Agreement weigh in favour of interpreting the "treatment no less favourable" requirement of Article 2.1 as not prohibiting detrimental impact on imports that stems exclusively from a legitimate regulatory distinction".¹⁷⁵

7.131. Accordingly, once imported and domestic/other foreign products are found to be like, two elements must be examined to determine whether the measure at issue accords imported products less favourable treatment than that accorded to like domestic/other foreign products: (a) whether the measure causes a detrimental impact on competitive opportunities for the group of imported products *vis-à-vis* the group of domestic/other foreign products; and (b) whether the

¹⁷² Although Norway did not put forward a claim under Article 2.1 of the TBT Agreement, to the extent that Norway holds the same view as Canada with respect to Canada's claim under Article 2.1 and presented its arguments and evidence in other parts of its claims, we also refer to them, as necessary and as appropriate, in this section. (See, for example, Norway's opening statement at the first substantive meeting of the Panel, paras. 103-105). In the context of its claim under Article 2.2 of the TBT Agreement (legitimacy of objective), Norway presents its view on the alleged legitimacy of the distinction between commercial and non-commercial seal hunting under the measure.

 $^{^{173}}$ Appellate Body Report, *US* – *Clove Cigarettes*, para. 87. The other element mentioned by the Appellate Body, i.e. that the measure at issue must be a technical regulation, was addressed in section 7.3.1 above.

¹⁷⁴ Appellate Body Report, *US – Clove Cigarettes*, para. 180.

¹⁷⁵ Appellate Body Report, US – Clove Cigarettes, para. 181.

detrimental impact on imports, if found to exist, stems exclusively from a legitimate regulatory distinction rather than reflecting discrimination against the group of imported products.¹⁷⁶

7.132. In sum, to determine whether the EU Seal Regime violates the MFN and national treatment obligations under Article 2.1 of the TBT Agreement, we need to examine the following three elements:

- a. whether the imported and domestic/other foreign seal products are like products;
- b. whether the EU Seal Regime causes a detrimental impact on competitive opportunities for the group of imported seal products *vis-à-vis* the group of domestic/other foreign seal products; and
- c. whether the detrimental impact on imports, if found to exist, stems exclusively from a legitimate regulatory distinction rather than reflecting discrimination against the group of imported products.
- 7.133. We examine these elements in turn.

7.3.2.1 Like products

7.134. The first element of Article 2.1 of the TBT Agreement that we examine is whether imported seal products are like domestic and/or other foreign seal products at issue.¹⁷⁷

7.135. According to the Appellate Body, the interpretation of the concept of "likeness" in Article 2.1 of the TBT Agreement entails the following considerations:

The interpretation of the concept of "likeness" in Article 2.1 has to be based on the text of that provision as read in the context of the *TBT Agreement* and of Article III:4 of the GATT 1994, which also contains a similarly worded national treatment obligation that applies to laws, regulations, and requirements including technical regulations. ... [T]he determination of likeness under Article 2.1 of the TBT Agreement, as well as Article III:4 of the GATT 1994, is a determination about the nature and extent of a *competitive relationship between and among the products at issue*. To the extent that they are relevant to the examination of certain "likeness" criteria and are reflected in the products' competitive relationship, regulatory concerns underlying

¹⁷⁶ In setting out this analytical framework, the Appellate Body in *US* – *Clove Cigarettes* specifically referred to the situation "where the technical regulation at issue does not *de jure* discriminate against imports". (Appellate Body Report, *US* – *Clove Cigarettes*, para. 182). ¹⁷⁷ Canada submits that conforming and non-conforming seal products are like because the products

¹⁷⁷ Canada submits that conforming and non-conforming seal products are like because the products falling within both categories are in a competitive relationship and thus substitutable on the EU market. (Canada's first written submission, paras. 311-321). Conforming and non-conforming seal oil or seal skins, for instance, have the same physical characteristics, end-uses, and tariff classification. According to Canada, the evidence points to the fact that consumers (e.g. producers of seal oil, tanners, or manufacturers of garments and accessories made of seal fur skins) valued the quality of the seal input rather than the fact that products were derived from Inuit, marine resource management, or commercial hunts. (Canada's first written submission, paras. 312-315 and 319-320, cross-referencing Norway's first written submission, paras. 301, 305, and 319-320 (referring to Statement of Ms Linn Elice Kanestrøm on behalf of Fortuna Oils AS (31 October 2012), (Exhibit NOR-53); and Statement of Mr Anders Arnesen on behalf of GC Rieber Skinn AS (31 October 2012), (Exhibit NOR-53); Canada's response to Panel question No. 38, para. 148, and Exhibits CDA-88-90. See also Norway's first written submission, paras. 301-302, 305-307, and 319-320).

Furthermore, Canada notes that prior to the entry into force of the EU Seal Regime, consumers did not distinguish between seal products on the basis of whether they were derived from commercial, IC, or MRM hunts. (Canada's first written submission, para. 316 (referring to Seals and Sealing Network website information on Denmark (Greenland), (Exhibit CDA-69)). In support of its argument, Canada observes that seal skins derived from its commercial (non-Inuit) hunts were in fact imported by Greenland for the purpose of rendering them into various seal products.

The European Union acknowledges that all seal products are like regardless of the type and purpose of the hunt. (European Union's first written submission, para. 254).

technical regulations may play a role in the determination of likeness.¹⁷⁸ (emphasis added)

7.136. We thus assess the likeness of products based on *inter alia* the following criteria: (a) the properties, nature, and quality of the products; (b) the end-uses of the products; (c) consumers' tastes and habits; and (d) the tariff classification of the products.¹⁷⁹ As emphasized by the Appellate Body in *EC* – *Asbestos*, these four criteria provide a framework for analysing the likeness of particular products on a case-by-case basis and are meant to serve as tools to assist in the task of sorting and examining the relevant evidence.

7.137. The complainants' claims in this dispute relate to the treatment of seal products in general. As noted above, the EU Seal Regime covers a broad range of products falling under different chapters of the Combined Nomenclature.¹⁸⁰ The dispute between the parties is based on the distinction made between seal products that are prohibited under the EU Seal Regime (non-conforming) and seal products that are allowed because they meet the specific requirements under the exceptions (conforming).

7.138. We recall that the complainants argue that conforming and non-conforming seal products are like. The European Union does not contest that all seal products are like products, irrespective of the distinction drawn in the measure between non-conforming and conforming products.

7.139. The Panel shares the parties' view¹⁸¹ that the type or purpose of the seal hunt does not affect in any way the final product's physical characteristics, end-use, or tariff classification. As regards the criterion of consumers' tastes and habits, the complainants presented evidence to demonstrate that, prior to the EU Seal Regime, consumers did not make any distinction between seal products based on the type or purpose of the hunt.¹⁸² This evidence consists of statements by manufacturers and producers of seal products who maintain that the quality of the product, rather than the type or purpose of the hunt, was the main factor for consumers' choice. We note that the European Union has not contested this evidence.

7.140. Based on the above, we conclude that conforming and non-conforming seal products are like products within the meaning of Article 2.1 of the TBT Agreement.

7.3.2.2 Whether the EU Seal Regime causes a detrimental impact on imported products

7.3.2.2.1 Main argument of the parties

7.3.2.2.1.1 Canada

7.141. Canada claims that the EU Seal Regime *de facto* discriminates against the group of Canadian imports of seal products. Canada argues that a determination as to whether there has been less favourable treatment entails comparing the entire universe of like products (including both conforming and non-conforming seal products), as opposed to making a "category-to-category" comparison within the group of like products (i.e. comparing conforming to conforming, and non-conforming to non-conforming seal products), as suggested by the European Union.¹⁸³

7.142. According to Canada, the purpose and scale of the hunt or the ethnic identity of the hunter are irrelevant for the purpose of establishing whether a measure accords less favourable treatment to certain products.¹⁸⁴ In Canada's view, to be able to justify differences in treatment between

¹⁷⁸ Appellate Body Report, US - Clove Cigarettes, para. 120. The Appellate Body also noted that "the concept of like products serves to define the scope of products that should be compared to establish whether less favourable treatment is being accorded to imported products." (Appellate Body Report, US - Clove *Cigarettes*, para. 116).

¹⁷⁹ Appellate Body Report, *EC – Asbestos*, para. 102.

¹⁸⁰ See section 2.2 above.

¹⁸¹ See footnote 177 above.

 ¹⁸² See Canada's first written submission, para. 316 and response to Panel question No. 38, para. 148.
 ¹⁸³ Canada's second written submission, para. 231; comments on the European Union's response to

Panel question No. 124. ¹⁸⁴ Canada's opening statement at the first meeting of the Panel, para. 60.

sub-groups of like products on such a basis would eliminate the possibility of finding de facto discrimination.¹⁸⁵ The main element that needs to be taken into account is whether the EU Seal Regime affects the conditions of competition to the detriment of Canadian seal products.¹⁸⁶

7.143. Canada maintains that in the case of the IC exception, the conditions that must be satisfied effectively permit all seal products from Greenland to be placed on the market and to circulate freely between European Union member States.¹⁸⁷ At the same time, the conditions exclude the "vast majority" of Canadian seal products from the EU market because the commercial harvest from which the products are derived does not meet the requirements under the IC exception.¹⁸⁸ Canada argues that the "design, structure and expected operation" of the IC exception indicate that the Regime will have a detrimental impact on the competitive opportunities of Canadian seal products as compared to their like counterparts from Greenland.¹⁸⁹ The fact that there is equal treatment granted to Inuit seal products from Canada and Greenland does not change the fact that there is discrimination against nearly all Canadian seal products.¹⁹⁰

7.144. Similarly, regarding the MRM exception, Canada submits that the Regime effectively allows domestic seal products from Sweden, Finland, and possibly the United Kingdom to be placed on the EU market while excluding virtually all Canadian seal products.¹⁹¹ Through its requirements, the MRM exception conditions market access on the basis of whether seal products are derived or manufactured from seals culled under specific types of marine management programs. Canada asserts that the requirement to adopt an "ecosystem-based approach" for the management plan will likely operate to exclude Canadian seal products from the market because Canadian hunts are based on the sustainability of seal populations and not the eco-system.¹⁹² Furthermore, the "non-systematic" and "non-profit" requirements will prevent Canadian imports of seal products from being placed on the Community market because the east coast harvest in Canada is conducted based on a "fixed plan or system", on a yearly basis and for the specific purpose of commercial gain.¹⁹³ In contrast, according to Canada, "the design, structure, and expected operation of the category indicate that EU seal products are likely to meet those conditions".¹⁹⁴

7.3.2.2.1.2 European Union

7.145. The European Union does not consider that conforming and non-conforming seal products should be compared for the purpose of assessing whether the EU Seal Regime has a detrimental impact on Canadian imports. For the European Union, conforming and non-conforming seal products are in "different situations". Therefore, the two groups of products cannot be compared to establish whether the measure has detrimentally affected the conditions of competition of Canadian seal products.¹⁹⁵ According to the European Union, the analysis must be undertaken within each category of the group of like products, namely by comparing conforming to conforming

¹⁸⁸ Canada's first written submission, para. 323.

¹⁹³ Canada's first written submission, para. 339.

¹⁸⁵ Canada's second written submission, para. 233.

¹⁸⁶ Canada's opening statement at the first meeting of the Panel, para. 60; second written submission,

para. 237. ¹⁸⁷ Canada's first written submission, para. 407. Canada notes that Greenland is a self-regulating part of the Kingdom of Denmark. As such, Greenland is not part of the European Union but rather a country or territory associated with the European Union in accordance with the provisions of Part Four of the Treaty on the Functioning of the European Union, Article 355 of the Treaty on the Functioning of the European Union, and Protocol No. 34 on Special Arrangements for Denmark (Greenland). As regards the status of Greenland in the WTO, Canada notes that Greenland was notified in 1951 by Denmark as a territory to which the GATT 1947 applied. Denmark makes WTO notifications and statements on behalf of Greenland and has stated that it represents Greenland (which is not part of the European Union) in the WTO. (Canada's first written submission, paras. 124-125; Norway's first written submission, paras. 71-72).

⁽See, for instance, The Territorial Application of the General Agreement: A Provisional List of Territories to Which the Agreement Is Applied, (Exhibit NOR-20) and Preparations for the 1999 Ministerial Conference, Communication from Denmark, WT/GC/W/384 (8 November 1999)).

¹⁸⁹ Canada's first written submission, para. 324; opening statement at the first meeting of the Panel, para. 66.

¹⁹⁰ Canada's opening statement at the first meeting of the Panel, para. 61; response to Panel question No. 23, paras. 120-121. ¹⁹¹ Canada's first written submission, para. 341.

¹⁹² Canada's first written submission, para. 344.

¹⁹⁴ Canada's first written submission, para. 337; opening statement at the first meeting of the Panel, para. 65. ¹⁹⁵ European Union's first written submission, paras. 295 and 325.

and non-conforming to non-conforming seal products. Under the EU Seal Regime, products of all origin falling within the same category are treated equally in terms of their access to, or prohibition to access the EU market.¹⁹⁶

7.146. For the European Union, the fact that most of Canada's seal products cannot be placed on its market, while most of the like domestic or other foreign products can be placed on the EU market, is insufficient to establish the existence of a detrimental impact on Canadian seal products.¹⁹⁷ When the treatment granted to the group of Canadian imports of seal products as a whole is compared to the treatment granted to the entire group of like products from domestic/other origin, there is no alteration of the aggregate competitive opportunities of Canadian seal products as compared to seal products of domestic or other origin.¹⁹⁸

7.147. With regard to the IC exception in particular, the European Union argues that non-Inuit seal products from Canada and Inuit seal products from Greenland are in "different situations" because the two types of hunts differ in respect of their purpose (subsistence on the one hand, profit on the other hand), their intensity, and the moral perception of the EU public.¹⁹⁹ The European Union further argues that Canadian Inuit seal products are treated in the same manner under the EU Seal Regime as Greenlandic Inuit seal products.²⁰⁰ The fact that the portion of Greenlandic seal products falling under the IC exception is greater than the portion of Canadian seal products qualifying under the same exception does not make the EU Seal Regime discriminatory *per se.*²⁰¹ Moreover, Canada's allegation that the IC exception benefits Greenlandic seal products is unfounded, as not all seal products originating in Greenland will automatically be covered under the IC exception.²⁰²

7.148. Regarding the MRM exception, the European Union argues that seal products derived from MRM hunts and seal products derived from the Canadian commercial hunts cannot be compared for the purpose of determining whether Canadian imports have been treated less favourably.²⁰³ According to the European Union, the two groups of products essentially differ in terms of their scale and their commercial or non-commercial motivations.²⁰⁴ The European Union asserts that Canada could, in principle, carry out MRM hunts and place the by-products of such hunts on the EU market.²⁰⁵ However, the European Union notes that Canada has not requested to be included in the list of recognized bodies authorized to issue attesting documents for placing on the market under the MRM exception.²⁰⁶

¹⁹⁶ European Union's first written submission, para. 324; second written submission, paras. 205 and 552. ¹⁹⁷ European Union's opening statement at the second meeting of the Panel, para. 55.

¹⁹⁸ European Union's first written submission, paras. 293 and 324; second written submission, para. 204; response to Panel question No. 124. The European Union notes that it does not produce any nonconforming seal products. However, the European Union asserts that the fact that there are no domestic like products falling within the category of non-conforming seal products is not an obstacle to making a "categoryto-category" comparison of the treatment granted by the EU Seal Regime, i.e. examining the potential treatment that those domestic like products would receive under the EU Seal Regime.

¹⁹⁹ European Union's first written submission, paras. 295-300.

²⁰⁰ European Union's first written submission, para. 294.

The European Union recalls that 5 per cent of Canadian seal products could potentially fall under the IC exception (a fact that is not denied by Canada) and that Norway could also qualify under the IC exception. However, neither Canada nor Norway has submitted any request to the European Commission to be included in the list of recognized bodies authorised to issue the necessary attesting documents for their products to be placed on the European Union market. (European Union's first written submission, paras. 551 and 562).

²⁰¹ European Union's first written submission, paras. 289 and 561.

²⁰² European Union's first written submission, para. 295 citing COWI 2010 Report, (Exhibit JE-21), Annex 5, p. 17. ²⁰³ European Union's first written submission, para. 323.

²⁰⁴ European Union's first written submission, paras. 325-329.

²⁰⁵ European Union's second written submission, paras. 247 and 250. ²⁰⁶ European Union's first written submission, footnote 446; response to Panel question No. 123,

paras. 91-92; second written submission, para. 250 (citing Canada's Marine Mammal Regulations, (Exhibit CDA-21), Article 26.1). The European Union notes that according to Canada's own regulations, licences (which are mandatory for hunting seals) may be granted for commercial use, for personal use, and for "nuisance seal" hunts. Seal products derived from such nuisance seal hunts would in principle qualify under the MRM exception.

7.3.2.2.2 Analysis by the Panel

7.149. Before we engage in our analysis of whether the EU Seal Regime causes a detrimental impact on competitive opportunities for the group of Canadian imported products vis-a-vis the group of domestic and/or other imported products, we must first determine the groups of products to be compared.²⁰⁷

7.150. The "universe" of products covered in this dispute, as agreed by the parties, is reflected in Table 1 below.

Distinction	Domestic Seal Products	Norwegian Seal Products	Canadian Seal Products	Other Foreign Seal Products	
				Greenland	Other
Non-conforming	А	В	С	D	E
Conforming (IC and MRM hunts)	F	G	Н	Ι	J

Table 1: Group of like seal products

7.151. We recall that while the parties agree in principle that all seal products are like, they disagree on the groups of like products to compare for the purpose of determining whether the EU Seal Regime is consistent with Article 2.1 of the TBT Agreement.

7.152. Canada is of the view that the entire group of imported seal products (including conforming and non-conforming products) should be compared to the entire group of domestic and/or Greenlandic seal products. The European Union considers that the treatment granted under the EU Seal Regime to conforming and non-conforming seal products cannot be compared because these products are in "different situations".

7.153. We note that the question of the groups of products to be compared was also addressed in US – *Clove Cigarettes*. In analysing Indonesia's claim under Article 2.1 of the TBT Agreement, the Appellate Body compared clove cigarettes (the imported product subject to the ban) and menthol cigarettes (the domestic product exempted from the ban). The fact that certain non-clove cigarettes from Indonesia were exempted from the ban was not considered relevant to Indonesia's claim that the competitive opportunities for its clove cigarettes, comprising the "vast majority" of Indonesia's exports to the United States, were being negatively affected *vis-à-vis* menthol cigarettes in the United States, comprising the "vast majority" of like domestic cigarettes in the United States.²⁰⁸

7.154. Thus, contrary to the European Union's position, the Appellate Body's approach suggests that the group of imported products should be compared with the group of domestic or other origin products. Thus, for the purpose of considering Canada's claim under Article 2.1 and with reference to Table 1 above, Canada's seal products (cells C+H), the vast majority of which are non-conforming products, are to be compared to domestic seal products (cells A+F) and to Greenlandic seal products (cells D+I) respectively. That is so even if a small percentage of seal products from Canada may still be eligible to qualify for placement on the EU market under one of the exceptions.

7.155. With this in mind, we turn to the question of whether the EU Seal Regime causes a detrimental impact on competitive opportunities for the group of Canadian imported products *vis-à-vis* the group of other imported or domestic like seal products.

7.156. The Appellate Body confirmed that this question requires consideration of the totality of the facts and circumstances before the panel, and an assessment of the implications for competitive conditions discernible from the *design*, *structure*, *and expected operation* of the

²⁰⁷ See Appellate Body Report, *US – Clove Cigarettes*, paras. 190 and 192.

²⁰⁸ Appellate Body Report, US – Clove Cigarettes, paras. 197-200.

measure. Furthermore, the examination of the measure's impact on the market need not be based on the *actual effects* of the contested measure in the market place.²⁰⁹

7.157. For the purpose of our analysis, we must therefore assess the "design, structure, and expected operation" of the EU Seal Regime, as well as any other relevant features of the market, which may include the particular characteristics of the industry at issue, the relative market shares in a given industry, consumer preferences, as well as historical trade patterns.²¹⁰

7.158. We recall that under the EU Seal Regime, only seal products that meet the requirements under the IC or MRM exceptions may be placed on the EU market. The prohibition against placing on the market applies to all seal products other than those that satisfy the IC and MRM requirements.

7.159. Considered in light of the specific requirements of the IC and MRM categories, the majority of seals hunted in Canada would not qualify under the exceptions. Canada argues that the prohibition under the EU Seal Regime was specifically targeted at the Canadian non-conforming seal hunt, from which some 95 per cent of all Canadian seal products derive.²¹¹ The evidence referenced by Canada suggests that the EU legislation on seals was in fact primarily aimed at excluding seal products resulting from the non-conforming seal hunt in Canada.²¹² This is not disputed by the European Union.

7.160. We note that Canada relies on a study conducted at the request of the European Commission in 2010 by COWI, a Danish Consulting Group (COWI 2010 Report), which concludes that only a minority of Canadian seal products are expected to qualify under the IC exception.²¹³ In contrast, the Report finds that the Greenlandic hunt is likely to meet the IC requirements.²¹⁴

7.161. Relevant data before us also demonstrate that most if not all of Greenlandic seal products are expected to conform to the requirements under the IC exception, as compared to roughly 5% in Canada, where only a small portion of the overall seal harvest is hunted by Inuit communities.²¹⁵ Therefore, the share of the total production that would *not* be eligible to be placed

²¹⁰ Appellate Body Report, US – COOL, para. 269 (footnotes omitted) (referring to Appellate Body Report, Canada – Autos, paras. 81 and 85-86; Appellate Body Report, US – Clove Cigarettes, para. 206; Appellate Body Report, Thailand – Cigarettes (Philippines), para. 130; Appellate Body Report, US – Tuna II (Mexico), paras. 233-234; Panel Report, Mexico – Taxes on Soft Drinks, para. 8.119; and Appellate Body Report, Korea - Various Measures on Beef, para. 145).
²¹¹ Canada notes that "[r]oughly 95 per cent of Canada's seal harvest in the last five years that has

²¹¹ Canada notes that "[r]oughly 95 per cent of Canada's seal harvest in the last five years that has been placed into commerce has come from the east coast commercial hunt." Canada explains that this figure is derived from the average proportion of Canada's total seal harvest attributable to the commercial harvest in the years 2006-2011. (See Canada's first written submission, para. 286 and footnote 391).

²¹² See, e.g. European Parliament Debates – Item A6-0118/2009 (4 May 2009), (European Parliament Debates) (Exhibit JE-12). See also Parliament Declaration, (Exhibit JE-19), point A (reference by the European Parliament to the hunting of harp seal pups in the "North West Atlantic").

²¹³ The COWI 2010 Report stresses that:

Inuit products make up only a very small share of Canadian seal trade, and the amount of products currently ending on the EU market is negligible. Whether products from the commercial hunt now directed to the EU will be replaced by Inuit products depend on a number of factors, but several stakeholders have already indicated that it will not be possible in any case – nor desirable as far as the Inuit communities are concerned – to increase significantly the scale of the Inuit hunt. (COWI 2010 Report, (Exhibit JE-21), pp. 64-65; see also European Union's first written submission, para. 292).

²¹⁴ Canada's first written submission, para. 277; opening statement at the first meeting of the Panel, para. 66 (citing COWI 2010 Report, p. 30). The COWI Report indicates that in Greenland about 90% of the total population is Inuit; there is a long tradition of seal hunting which has been part of the cultural heritage of the communities; and seals are not hunted for the sole purpose of placing products on the market, but are consumed and used in the local community and contribute to the local economy. We note that these facts are also corroborated by the report of the Greenlandic Government on the management and utilization of seals in Greenland. (Government of Greenland: Ministry of Fisheries, Hunting & Agriculture, *Management and Utilization of Seals in Greenland* (April 2012), (Exhibit JE-26), pp. 15-16).

²¹⁵ See Canada's first written submission, para. 286 and footnote 391; Norway's first written submission, para. 391 and Table 1 on "Indigenous Communities Requirements" compiling relevant data from COWI 2010 Report; Nunavut Department of Environment, Fisheries and Sealing Division, *Report on the Impacts of the European Union Seal Ban, (EC) No. 1007/2009, in Nunavut* (2012), (Nunavut Report (2012)), (Exhibit JE-30); Management and Utilization of Seals in Greenland, (Exhibit JE-26); Icelandic Marine Research Institute, *Summary of State of Marine Stocks in Icelandic Waters 2011/2012: Prospects for the Quota Year 2012/2013* (2012) (Exhibit NOR-21); Joint Norwegian/Russian Fisheries Commission, *Report of the Working*

²⁰⁹ Appellate Body Report, US – FSC (Article 21.5 – EC), para. 215.

on the market under the IC exception is relatively high (i.e. some 95%) for Canada, whereas most if not all of Greenland's seal products are eligible.²¹⁶

7.162. The European Union asserts that seal products from Canada could be eligible for placement on the market under the IC exception, although no entity in Canada has yet made any request to be added to the list of recognized bodies.²¹⁷ The European Union further contends that it has engaged in "multiple efforts" to assist the Inuit in Canada to benefit from the IC exception.²¹⁸ Canada does not deny the fact that some of its Inuit seal products could in principle qualify under the IC exception. Canada argues, however, that placing these products directly on the market may be difficult for the Inuit because they have limited access to the distribution networks, processing facilities, and marketing opportunities needed to export their seal products to the European Union.²¹⁹

7.163. However, as observed above, the Appellate Body in US - Clove Cigarettes clarified that the fact that a small group of imported products was exempted from the ban in question was not considered relevant when assessing the ban's overall impact on the vast majority of imported products *vis-à-vis* the majority of like domestic products. Likewise, the possibility that some of Canada's Inuit seal products *could* enter the EU market does not change the fact that the vast majority of Canada's seal products are *in fact* excluded from the same market on the basis that they derive from a "non-conforming" seal hunt.

7.164. We note that Greenland's Department of Fisheries, Hunting and Agriculture was recently recognized for the purpose of Article 6 of the Implementing Regulation.²²⁰ As such, Greenland is

Group on Seals to the 40th Session – Appendix 8, (Exhibit NOR-22); United States Department of Commerce, Marine Mammals; Subsistence Taking of Northern Fur Seals – Harvest Estimates, Federal Register, Vol. 77, No. 27 (9 February 2012), (Exhibit NOR-23); and Norwegian Ministry of Fisheries and Coastal Affairs, Facts about Fisheries and Aquaculture 2010 (Exhibit NOR-63).

²¹⁶ COWI observes in its 2010 Report that "[i]t is unlikely that all of the Greenland harvest is eligible under Article 3.1 [of the Basic Regulation]." (See COWI 2010 Report, Annex 5, p. 17). However, COWI does not elaborate on the reasons for its assertion. Given the recent authorization of a Greenlandic entity to act as a recognized body pursuant to Article 6 of the Implementing Regulation, it is possible that all seal products from Greenland will be allowed for placement on the EU market under the IC exception.

²¹⁷ Canada's response to Panel question Nos. 84, 85, and 117.

²¹⁸ European Union's response to Panel question No. 116. The European Union states that "based on the requirements of the IC exception and based on the best information available, seal products derived from hunts conducted by the Inuit in Canada would qualify under the exception." (See also NunatsiaqOnLine, dated 23 April 2013, "European Commission representative visits Iqaluit on good-will trip, Christian Leffler says commission wants to give effect to Inuit exemption" (Exhibit EU-145)).

²¹⁹ For instance, Inuit from Nunavut (Canada) have an annual production of less than 8,000 pelts of ringed seals, which they view as insufficient to generate market interest alone on an international scale. The ringed seal pelt industry in Nunavut benefits from the harp seal industry by "piggybacking" on markets that are created and maintained by the much larger harp seal industry. Any variation in the market demand for harp seal products deriving from Canada's commercial hunts would also be felt by Nunavut's ringed seal industry. Commenting on the European Commission's Proposal, the Nunavut Government noted that "[t]his market reality is one of the major factors contributing to the ineffectiveness of the Inuit exemption to the EU Seal ban". (Nunavut Report (2012), (Exhibit JE-30), p. 9).

Canada notes that it considered options to address the negative impact of the EU Seal Regime on the ability of Canadian Inuit to market their seal products internationally. (Canada's response to Panel question Nos. 116 and 117). The COWI 2010 Report draws the following conclusions on the likely impact of the EU Seal Regime:

As [the Canadian commercial] hunt is unlikely to fulfil the conditions of [the Basic Regulation], some of the sourcing for blubber may shift towards notably Greenland, which is likely to fall within the conditions under Article 3. This may even result in additional investments in Greenland in collection and/or processing facilities. ... It is likely that only Greenland will be able to make the investments needed to make use of the exemptions, as the scale of the Canadian hunt is too small and not as centrally organized as that in Greenland (Canadian Inuit hunt essentially uses the sales and marketing chains of the commercial hunt, implying it would need to invest heavily in separating its Inuit products from the rest). For buyers and producers alike, the investments are not likely to outweigh the benefits due to the limited amount of products concerned. (COWI 2010 Report, pp. 62 and 72)

²²⁰ See European Union's response to Panel question No. 156; Commission decision of 25 April 2013 recognising the Greenland Department of Fisheries, Hunting and Agriculture (APNN), (Exhibit EU-149), p. 3. The European Union confirmed that prior to the Greenlandic entity obtaining recognized body status, the Danish customs authorities had processed imports based on certificates issued by the Greenlandic authorities. The European Union notes that this was based on an interpretation of the Implementing Regulation, whereby

entitled to deliver attesting documents for the placing on the EU market of Greenlandic seal products. In light of this fact, and taking into account the arguments of the parties, we believe that all, or virtually all, seal products from Greenland are eligible to access the EU market under the IC exception, while the majority of like products produced by Canada do not conform to the requirements of the IC exception and thus are ineligible to benefit under the EU Seal Regime.

7.165. With regard to the MRM exception, Canada further argues that the requirements *a priori* exclude virtually all of its seal products. In particular, Canadian seal products would not be eligible under the MRM exception because they derive from seal hunts that take place on a "systematic" and organised basis. Furthermore, sealing is a commercial activity in Canada and therefore the hunt could not qualify under the non-profit requirement. In addition, although the seal hunt in Canada is based on sustainability principles, it does not follow an "ecosystem-based approach" as required under the MRM exception.

7.166. The European Union notes that, currently, seal products from Sweden accompanied by the relevant document in accordance with Article 5(2) of the Implementing Regulation can be placed on the market under the MRM exception.²²¹

7.167. The evidence submitted by the complainants suggests that while European Union seal products are likely to benefit from the MRM exception, Canadian seal products are not expected to benefit from the same market access opportunities under the EU Seal Regime.²²² A study conducted by COWI in 2008 (COWI 2008 Report) found that a complete prohibition on the placing on the market of seal skins and products derived therefrom would have only a minor economic impact on EU member States.²²³ This finding was based on the assumption that the transit of seal skins and other products would continue to take place under the EU Seal Regime.²²⁴ Conversely, the COWI 2008 Report concludes that the economic impact of the measure would be more significant for non-EU sealing states, such as Canada and Norway²²⁵, based on the importance of the EU market and the fact that the size of the hunt is much larger in these countries.²²⁶

7.168. We observe that the volume of seal products derived from seal hunts covered or potentially covered by the MRM exception is limited.²²⁷ Currently, only entities from Sweden are certified as recognized bodies entitled to deliver attesting documents for placing seal products on the EU market.²²⁸ In the Panel's view, however, the limited impact of the exception is not relevant to assessing whether the MRM exception negatively affects the competitive opportunities for imported products *vis-à-vis* like domestic products on the EU market. Even if the MRM exception concerns only a small number of seal products, most of the European Union's seal products are potentially eligible for placement on the EU market under this exception, while virtually all Canadian seal products are not. In light of the above, the Panel considers that the requirements under the IC and MRM exceptions were designed, structured, and expected to operate so as to

the issuance of attesting documents complying with the Implementing Regulation would also be allowed during the application process for recognized body status and not only once the process had been completed. (European Union's response to Panel question No. 161).

²²¹ European Union's first written submission, para. 519. See also List of recognized bodies in accordance with Article 6 of the Implementing Regulation, (Exhibit EU-77), and Commission decision of 18 December 2012 recognising the Swedish County Administrative Boards, (Exhibit EU-159). The European Union observes that small-scale hunts or hunts for the purpose of managing marine resources also take place in other countries within the European Union, namely in Finland. (European Union's first written submission, para. 320, footnote 429).

²²² Canada's first written submission, paras. 99-102; response to Panel question No. 167, para. 207 (where Canada notes that the products of nuisance seal hunts in Canada "may not be commercialized"); European Union's first written submission, paras. 320, , 519, 521, and footnote 429; COWI 2010 Report, p. 66 and Annex 4.

²²³ COWI 2008 Report, p. 117.

²²⁴ COWI 2008 Report, p. 117.

²²⁵ Canada's first written submission, para. 276; COWI 2008 Report, p. 118.

²²⁶ We note that the COWI 2008 study also considers the impact of a total prohibition of trade in seal products, i.e. one that would extend also to the transit of seal products. COWI concludes that the economic impact would be more important for EU member States (in particular Finland and Germany) than a prohibition limited to the placing on the market. (COWI 2008 Report, p. 120).

²²⁷ The European Union asserts that the number of seals covered by the MRM exception in 2011 in
 Sweden was 86. (European Union's response to Panel question No. 122, para. 77).
 ²²⁸ The European Union notes that no other entity, in Finland or elsewhere in the European Union, has

²²⁸ The European Union notes that no other entity, in Finland or elsewhere in the European Union, has requested authorization to be a recognized body in accordance with the Implementing Regulation. (European Union's second written submission, para. 236).

exclude seal products deriving from the *majority* of Canadian seal hunts, which are not IC or MRM hunts, from being placed on the EU market. In other words, by virtue of its design, the measure excludes all but a very small percentage of potential products from Canada, while at the same time permitting the majority or all of like products from certain EU members.

7.169. As a final observation, we address the European Union's position that the treatment granted under the EU Seal Regime to conforming and non-conforming seal products cannot be compared, because these products are in "different situations" with regard to the type of hunt from which each category of products are derived. We note that despite its position on this particular point, the European Union considers that conforming and non-conforming seal products are "like". Based on the examination of the "nature and extent of the competitive relationship between the products in the [EU] market", we found that Canada's seal products are "like" seal products of Greenlandic and EU origin.²²⁹ In our view, because the two groups of products were found to be "like", such products can be compared for the purpose of determining the implications of the measure on their competitive relationship on the EU market. We are not persuaded by the European Union's assertion that products found to be "like" may not be compared for the purpose of determining whether one group of products are negatively affected in terms of their competitiveness on the market against another group. In our view, the European Union's argument that conforming and non-conforming seal products are in "different situations" is relevant to the justification of the regulatory distinction under the EU Seal Regime. As such, this argument can be more appropriately assessed in the context of our subsequent analysis of whether any detrimental impact caused by the measure to the imported products reflects discrimination against such products.

7.170. On the basis of our examination of the design, structure, and expected operation of the EU Seal Regime, as well as evidence relating to other relevant features of the market, the Panel finds that the Regime has a detrimental impact on the competitive opportunities of Canadian imported products *vis-à-vis* Greenlandic imported and EU domestic products. Next, we turn to the question of whether such detrimental impact caused by the EU Seal Regime results in according less favourable treatment to the imported seal products in violation of Article 2.1 of the TBT Agreement.

7.3.2.3 Whether the detrimental impact caused by the EU Seal Regime "stems exclusively from legitimate regulatory distinctions"

7.171. We recall the Appellate Body's explanation that the "treatment no less favourable" requirement of Article 2.1 should not be interpreted as prohibiting detrimental impacts on competitive opportunities for imports in cases where such detrimental impact on imports stems exclusively from legitimate regulatory distinctions rather than reflecting discrimination against imported products.²³⁰

7.172. Regarding how to assess whether a detrimental impact on imports stems exclusively from legitimate regulatory distinctions, the Appellate Body stated:

[S]ome technical regulations that have a *de facto* detrimental impact on imports may not be inconsistent with Article 2.1 when such impact stems exclusively from a legitimate regulatory distinction. In contrast, where a regulatory distinction is not designed and applied in an even-handed manner — because, for example, it is designed or applied in a manner that constitutes a means of arbitrary or unjustifiable discrimination — that distinction cannot be considered "legitimate", and thus the detrimental impact will reflect discrimination prohibited under Article 2.1. In assessing even-handedness, a panel must "carefully scrutinize the particular circumstances of the case, that is, the design, architecture, revealing structure, operation, and application of the technical regulation at issue.²³¹

²²⁹ See Appellate Body Report, US – Clove Cigarettes, para. 191.

 ²³⁰ See Appellate Body Reports, US – Clove Cigarettes, paras. 169, 174, 182, and 194; US – Tuna II (Mexico), para. 215; US – COOL, para. 271.
 ²³¹ Appellate Body Report, US – COOL, para. 271. On this point the Appellate Body was referring in this

 $^{^{231}}$ Appellate Body Report, US - COOL, para. 271. On this point the Appellate Body was referring in this respect to its earlier report in US - Clove Cigarettes, para. 182.

7.173. We recall that we have found that the IC and MRM exceptions cause a detrimental impact on competitive conditions for Canada's seal products imported on the EU market. In light of the Appellate Body's guidance on the obligations under Article 2.1 of the TBT Agreement, we proceed to examine whether the European Union has established that such detrimental impact stems exclusively from legitimate regulatory distinctions.²³² As part of that analysis, we will also evaluate whether the regulatory distinctions are designed and applied in an even-handed manner and thus do not reflect discrimination against Canadian seal products.

7.174. Our examination of this question entails an analysis of two main questions: (a) first, what are the relevant regulatory distinctions under the EU Seal Regime; and (b) second, are such regulatory distinctions "legitimate".²³³ We address these questions in turn.

7.3.2.3.1 "Regulatory distinctions" drawn under the EU Seal Regime

7.175. The EU Seal Regime distinguishes between seal products that conform to the IC or the MRM requirements under the exceptions (conforming products)²³⁴, on the one hand, and those that do not conform to these requirements (non-conforming products), on the other hand.²³⁵ Specific requirements for the exceptions are set out in Article 3 of the Basic Regulation and Articles 3 and 5 of the Implementing Regulation; only those products satisfying the IC or MRM requirements under these provisions are allowed on the EU market.²³⁶

7.176. As indicated in the text of the concerned provisions, and as observed in the section on the definition of a technical regulation²³⁷, the distinction between conforming and non-conforming products is based on specific criteria relating to seal hunts from which seals are derived and used as inputs in the final products. These criteria include the identity of the hunter (Inuit or indigenous); the type of hunt (traditional Inuit hunts²³⁸); the purpose of the hunt (subsistence or marine resource management); and the way in which the products are marketed (non-systematically and on a non-profit basis). The criteria at issue thus do not contain any requirements concerning specific hunting methods.

7.177. Accordingly, the regulatory distinction drawn by the measure is linked to seal hunts; a particular category of the hunt from which a seal is derived determines whether a certain product containing seal is conforming or non-conforming under the measure. Put simply, products with seal inputs derived from IC or MRM hunts as defined under the measure are allowed, whereas products with seal inputs derived from any other hunts are prohibited. The regulatory distinction that the European Union must justify is therefore that between IC and MRM hunts and hunts that are not IC or MRM hunts.

²³⁷ See section 7.3.1 above.

²³² See Appellate Body Report, *US* – *Tuna II (Mexico)*, para. 216. With respect to the allocation of the burden of proof, we note the Appellate Body's statement in *US* – *Tuna II (Mexico)* that "[a]lthough the burden of proof to show that the US 'dolphin-safe' labelling provisions are inconsistent with Article 2.1 of the TBT Agreement is on Mexico as the complainant, it was for the United States to support its assertion that the US 'dolphin-safe' labelling provisions are 'calibrated' to the risks to dolphins arising from different fishing methods in different areas of the ocean." (Appellate Body Report, *US* – *Tuna II (Mexico)*, para. 283 (referring to Appellate Body Report, *Japan – Apples*, para. 157)).

²³³ See, for example, Appellate Body Report, *US – COOL*, para. 341.

²³⁴ We recall that seal products qualifying under the Travellers exception are also allowed to be imported (as that term is defined in the Regulation) under Article 3(2)(a) of the Basic Regulation. For its claim under Article 2.1 of the TBT Agreement, however, Canada has not taken issue with that particular exception.

 $^{^{235}}$ The parties do not contest that the distinction drawn by the measure is between non-conforming and conforming seal products. (See Canada's second written submission, para. 245; Norway's second written submission, para. 259; European Union's response to Panel question No. 28. See also Appellate Body Report, US - Tuna II (Mexico), para. 284).

²³⁶ See sections 7.2.1.and 7.2.2 above.

²³⁸ The criteria of the EU Seal Regime pertain to "hunts traditionally conducted by Inuit and other indigenous communities", which, for clarification, does not refer to the *methods* of hunting but rather to the requirement that a community have a tradition of seal hunting "in the geographical region". (See Basic Regulation, Article 3(1); Implementing Regulation, Article 3(a); European Union's response to Panel question No. 30).

7.178. The European Union asserts that this distinction is based on the purpose of the hunt from which seal inputs used in a given product are derived.²³⁹ Products from hunts allegedly conducted for "non-commercial" purposes, namely IC and MRM hunts, are allowed, and products from hunts that are "commercial" in nature are prohibited. The European Union contends that these two types of hunts (non-commercial and commercial) present different moral considerations and different levels of animal welfare risks in seal hunting.²⁴⁰ Canada disagrees with the European Union, arguing that the distinction drawn by the European Union between "commercial" and "non-commercial" hunts is not legitimate. According to Canada, seal welfare concerns exist equally in all seal hunts, irrespective of the type and purpose of the hunt. Further, the purported distinction between commercial and non-commercial seal hunting is illusory because all seal hunts have commercial dimensions.²⁴¹

7.179. Given the parties' positions, we must determine whether the distinction between IC and MRM hunts, on the one hand, and commercial hunts, on the other hand, is legitimate and does not reflect discrimination against imported seal products derived from non-IC and non-MRM hunts. In this regard, we are mindful that the parties contest whether the purposes of these hunts can be characterized as "non-commercial" and "commercial" as such. For ease of reference in these Reports, and without prejudice to our ultimate view on the question, we will use the term "commercial hunts" for hunts other than IC or MRM hunts. For the so-called "non-commercial hunts" as referenced by the European Union to indicate IC and MRM hunts, we will use the specific terms "IC hunts" and "MRM hunts".

7.180. We will begin our analysis with factual aspects of commercial hunts as it is the main distinction drawn by the measure against both IC and MRM hunts. We will then evaluate the specific distinction between commercial hunts and IC hunts and that between commercial seal hunts and MRM hunts to determine whether the respective distinctions are "legitimate" within the meaning of Article 2.1 of the TBT Agreement.

7.3.2.3.2 Preliminary question on commercial seal hunts

7.181. Throughout the proceedings, the parties debated extensively the characteristics of commercial seal hunts. $^{\rm 242}$

7.182. In essence, the European Union asserts that seal hunting is inherently inhumane and raises moral concerns especially when the hunting is conducted for commercial purposes. Further, the profit-oriented nature of the hunt increases the risk that seals may be killed inhumanely. Based on *inter alia* scientific opinions, the European Union takes the position that a humane killing method cannot be applied effectively and consistently in the circumstances of commercial seal hunts, which constitute the majority of seal hunts in Canada and Norway.²⁴³

7.183. The complainants emphasize the equal presence of a commercial component in all types of seal hunting. On that premise, they assert that the distinction between commercial and other types of hunts has no relevance to animal welfare outcomes in seal hunting. Further, they contest

²⁴² As explained in footnote 172, we also address Norway's arguments in this section.

²³⁹ The European Union explains that the "purpose" of the hunt is not different from the "type" of the hunt in the sense that there are three "types" of hunts in view of their "purpose", namely commercial, IC, and MRM hunts. (European Union's response to Panel question No. 30). We note the confirmation from the European Union that the non-commercial purpose of subsistence to which the IC exception corresponds is tied to the *identity* of the hunter as well. Specifically, the EU Seal Regime distinguishes hunts conducted "for subsistence purposes, where seals are killed primarily in order to contribute to the subsistence of Inuit and other indigenous communities". (European Union's response to Panel question No. 29, para. 100).

²⁴⁰ European Union's response to Panel question Nos. 122 and 133.

²⁴¹ See Canada's response to Panel question No. 28, para. 130 and second written submission, paras. 262-264. Norway takes a similar view as Canada concerning the European Union's position based on the distinction between commercial and non-commercial hunts, particularly the European Union's factual arguments regarding so-called "commercial" hunts. (See complainants' comments on the European Union's response to Panel question No. 133).

²⁴³ See, e.g. European Union's first written submission, paras. 37 and 122. The European Union submits that the question it is asking is whether its view that a humane killing method cannot be applied effectively and consistently in the circumstances of Canada's and Norway's commercial seal hunts finds adequate support from qualified scientific evidence. (European Union's opening statement at the first substantive meeting of the Panel, para. 11).

whether there are "inherent obstacles" in seal hunts to the application of humane killing methods and monitoring and enforcement of regulations.²⁴⁴

7.184. In this section, to make an objective assessment of factual assertions advanced by the parties regarding commercial seal hunts, we examine all factual evidence, including scientific opinions and video recordings, submitted by the parties regarding seal hunting.²⁴⁵ In our review of

²⁴⁴ See, e.g. Canada's opening statement at the first substantive meeting of the Panel, paras. 12-34; response to Panel question No. 55, para. 239; second written submission, para. 264; Norway's opening statement at the first substantive meeting of the Panel, paras. 147-202; response to Panel question No. 55, paras. 301-313; second written submission, paras. 217-219.

²⁴⁵ We note that the parties have made various contentions as to the credibility and weight of certain evidence, particularly with respect to scientific reports and video recordings.

With respect to video evidence, we note that the complainants have cited the EFSA Scientific Opinion in support of their argument that video evidence is of limited value. (See, e.g. Canada's opening statement at the first substantive meeting of the Panel, para. 23; Norway's opening statement at the first substantive meeting of the Panel, para. 23; Norway's opening statement at the first substantive meeting of the Panel, para. 23; Norway's opening statement at the first substantive meeting of the Panel, para. 169). The cited passage of the EFSA Scientific Opinion, however, refers to "the difficulties in evaluating whether or not a seal has been rendered unconscious by a blow to the head or by a bullet *at a distance or on videotape*". (EFSA Scientific Opinion, p. 54) (emphasis added) We do not read this statement as denying the probative value and reliability of video evidence. Viewed in its context, EFSA's statement, which is applied equally to first-hand observation at a distance, serves to explain divergence in interpretation among different studies. Although this suggests reason for caution in the interpretation of video evidence (as it also does for recorded first-hand observations), we consider that video recordings may be usefully consulted as part of the totality of the evidence. Indeed, the complainants have not argued that video evidence should be disregarded, and we find acknowledgement from both Canada and Norway of the utility of video technology in the monitoring of seal hunts and enforcement of regulations. (See Canada's response to Panel question Nos. 155 (para. 199) and 174 (para. 230); Second Statement by Dr Knudsen, (Exhibit NOR-162), para. 12).

Further, Canada called into question the value of several exhibits relied upon by the European Union. In particular, Canada characterizes Butterworth (2012) as a "re-packaged version of the same data used" in Butterworth (2007) and states that "it is framed as an advocacy document rather than an objective scientific paper". Canada further criticizes Butterworth (2012) for its reliance on "the type of video footage that EFSA dismissed as being unreliable as evidence". (Canada's opening statement at the first substantive meeting of the Panel, para. 23). Canada also states that the authors of Burdon (2001), Butterworth (2007), Richardson (2007), and Butterworth (2012) do not have extensive experience or expertise in seals. Further, Burdon (2001) and Butterworth (2007) were organized and funded by NGOs opposed to the Canadian commercial seal hunt. (Canada's opening statement at the first substantive meeting of the IVWG Report (2005) (which included veterinarians with specific seal expertise) and Daoust (2012) (veterinary experts with direct observation). (Canada's opening statement at the first substantive meeting of the Panel, paras. 24 and 30. See also Canada's second written submission, para. 21). Finally, Canada objects to Richardson (2007) and Butterworth (2012) for lacking original empirical research and the lack of peer review for Richardson (2007). (Canada's second written submission, paras. 28-29. See also Canada's opening statement at the second substantive meeting of the Panel, paras. 10-17).

In Norway's opening statement at the first substantive meeting of the Panel, it pointed out that Daoust (2002) and Daoust (2012) are the only veterinary studies on the methods employed in seal hunting that are published in peer-reviewed scientific journals, whereas Burdon (2001) and Butterworth (2007) are "unpublished, non-peer reviewed reports, that base all their conclusions on analysis of extracted sequences of video clips and/or examination of abandoned carcasses". (Norway's opening statement at the first substantive meeting of the Panel, para. 169, comments of Dr Knudsen). Norway additionally contends that "all of the studies relied on by the European Union (with the exception of papers by Daoust), lack scientific methodology and are the basis for erroneous conclusions by the European Union." (Norway's second written submission, footnote 393 (referring to Second Statement by Dr Knudsen, (Exhibit NOR-162), paras. 4-32)). Norway specifically asserts that studies relied on by the European Union, i.e. Burdon (2001), Butterworth (2007), and Richardson (2007), "are all unpublished reports or statements, made by individuals or NGOs, that have not been subjected to peer-review". (Second Statement by Dr Knudsen, (Exhibit NOR-162), para. 5). Further, Butterworth (2012) "does not report any new studies or new research results", and refers to other unpublished reports relied on by the European Union (Burdon (2001) and Butterworth (2007)). (Second Statement by Dr Knudsen, (Exhibit NOR-162), para. 6). In Norway's view, scientific articles published in merited scientific journals cannot be compared to unpublished reports that have not been subjected to peer-review, and Norway questions the value of analysis of extracted sequences of video clips and/or post-mortem examinations of carcasses. (Norway's second written submission, footnote 393; Second Statement by Dr Knudsen, (Exhibit NOR-162), paras. 8-32).

The European Union responds that formal peer-review is just one of many ways that a study may be reviewed, and, in particular, refers to Burdon (2001) and Butterworth (2007) as having undergone some form of expert review in subsequent studies (including the EFSA Scientific Opinion and peer-reviewed publications such as Daoust (2002) and Butterworth (2012)). The European Union also draws a comparison between Burdon (2001) and the IVWG Report (2005) as having been designed to produce recommendations to the Canadian government for regulatory review. (European Union's second written submission, paras. 5-13). The European Union defends the relevant expertise of the authors of several reports and other experts relied upon, including: Burdon (2001) (six veterinarians, three with experience in seals and wildlife and another part of

the evidence before us, we have given due consideration to the arguments of the parties regarding the reliability and credibility of various sources. Specifically, in assessing the evidence in its entirety, we have taken into account *inter alia* analytical and empirical rigor; relevant expertise of the authors; and the purpose and/or mandate of the studies, statements, and reports submitted to the Panel. The majority of this evidence concerns hunts conducted in Canada and Norway; relatively little scientific or empirical information is provided regarding the actual animal welfare outcomes in hunting conducted in other sealing countries.²⁴⁶ We also note that in support of their respective positions, the parties have extensively cited the findings and conclusions of the Scientific Opinion of the European Food Safety Authority (EFSA Scientific Opinion) on the animal welfare aspects of the killing and skinning of seals.²⁴⁷ The reliability and accuracy of the EFSA Scientific Opinion has not been challenged by any party. Based on the entirety of such evidence, therefore, we will assess the characteristics of seal hunting in general and subsequently the alleged characteristics of commercial seal hunting in particular.

7.3.2.3.2.1 Characteristics of seal hunting in general

7.185. The alleged unique conditions in seal hunting include the physical environment of the hunts, characteristics of seals, and the application of killing methods in seal hunting. We examine these conditions in turn.

Physical environment

7.186. The parties do not dispute that the physical environments in which seals live and are hunted can, in certain respects, be distinguished from those existing in the hunting of other wildlife

IVWG); Richardson (2007) (experts in zoology, marine mammal veterinary science and humane slaughter with experience of direct observation of hunt, post-mortem examinations, and review of video evidence); and Professor Broom as a "world leading authority in the field of animal welfare". (European Union's second written submission, paras. 14-19). The European Union also contends that scientific research is not unreliable merely because it has been commissioned or facilitated by NGOs with a non-commercial interest. In the case of the Canadian commercial seal hunt, observation can practically only occur with the facilitation of the Canadian government or an NGO with government permission. (European Union's second written submission, paras. 20-23). The European Union highlights links of Dr Daoust to the Canadian fur industry, adding that had these links been disclosed to EFSA he would not have been permitted to be a member of the Working Group that prepared its Scientific Opinion. (European Union's second written submission, paras. 24-28). The European Union defends the reliability of video evidence, and, pointing out that Canada claims to use such evidence for its monitoring of the hunt, specifically contends that it is in several ways more accurate than first-hand observation/memory and that it is obtained in a random fashion. (European Union's second written submission, para. 35-53). Lastly, the European Union states of Mr Danielsson, who has provided multiple statements to the Panel as to practices and animal welfare in seal hunts (Exhibits NOR-4, NOR-128, and NOR-163), that he "participated in the hunts as a government employee and cannot be regarded as an independent party", and "the statements consist entirely of bare assertions, unsupported by evidence and have been prepared expressly for the purpose of this dispute under the control of the Norwegian authorities." (European Union's second written submission, para. 56).

²⁴⁶ See EFSA Scientific Opinion, p. 24: "Seal hunts have occurred in various parts of the world throughout history, and the different stunning and killing methods used have been documented in various ways. Very little robust information is available, however, on the efficacy of each of these methods and their respective advantages and disadvantages in relation to animal welfare."

With respect to Namibia, which is one of the principal sealing countries, the European Union introduced video evidence and commentary on the harvesting of Cape fur seals during the first meeting of the Panel.

²⁴⁷ EFSA Panel on Animal Health and Welfare, Scientific Opinion on the Animal Welfare aspects of the killing and skinning of seals (2007).

The EFSA Scientific Opinion addresses the biology of various species of seals; different killing and skinning methods and how they should be used in theory; the use of the killing methods in practice; the neurophysiological aspects of the determination of death; and the training and competence of sealers. (EFSA Scientific Opinion, (Exhibit JE-22), p. 12).

In this connection, as part of the stakeholders' consultation of the EFSA process, the Norwegian Scientific Committee for Food Safety ("VKM") provided an opinion on the subject with an *ad hoc* group of national experts to prepare the necessary scientific documents. (VKM Panel on Animal Health and Welfare, *Scientific Opinion on Animal Welfare Aspects of the Killing and Skinning in the Norwegian Seal Hunt* (2007), (VKM Scientific Opinion), (Exhibit JE-31)).

or in the commercial slaughter of farmed animals.²⁴⁸ The parties' disagreement concerns whether the prevalence of such conditions amount to inherent obstacles to humane killing.²⁴⁹

7.187. The Panel notes that various seal species are found throughout the world along the coasts of polar, sub-polar, and temperate regions.²⁵⁰ The seal hunts to which the specific physical conditions at issue may be ascribed are those occurring within or near Arctic and sub-Arctic regions, particularly in the Arctic and northern Atlantic Oceans as well as the Barents, White, and Greenland Seas.²⁵¹ In these regions, seals must be hunted in their marine habitats among varying ice formations, which can create attendant conditions of seal hunting such as variable winds, ocean swells and waves, and low/freezing temperatures.²⁵² This therefore distinguishes the physical environment of seal hunts from that of terrestrial wildlife hunts²⁵³ or commercial slaughterhouses.²⁵⁴ Further, we observe that deterioration of ice conditions in sealing regions has been observed in recent years²⁵⁵ and that the volatility of ice conditions may impact the working environment for sealers.²⁵⁶

²⁵¹ EFSA Scientific Opinion, pp. 13-23; Canada's Department of Fisheries and Oceans, *Overview of the Atlantic Seals Hunt, 2006-2010*, (DFO Overview of the Atlantic Seals Hunt 2006-2010), (Exhibit EU-40), pp. 6-7; VKM Scientific Opinion, pp. 10-18; Norwegian Ministry of Fisheries and Coastal Affairs, *Geographic Distribution of Seal Hunt* (September 2009), (Exhibit NOR-14). This would therefore exclude practices for harvesting of Cape fur seals in Namibia.

²⁵² Butterworth (2012), pp. 7-8 and Table 3 (providing data on wind speeds and wave heights on the opening days of the commercial seal hunt from 2007-2011); IVWG Report (2005), p. 5 (recognition of "the specific challenges presented by weather, sea and ice conditions" in seal hunts); EFSA Scientific Opinion Annex A, footnote 1 (explanation of risk characterization parameters that "[g]ood weather refers e.g. to fine and ideal weather and bad weather refers e.g. to poor visibility, heavy swells and gusty winds."); Video presented by the European Union at the first meeting of the Panel, (Exhibit EU-79).

²⁵³ We note that complainants' references to the hunts of wild animals have consisted of reference to *terrestrial* animals.
 ²⁵⁴ The EFSA Scientific Opinion notes, for example, that in contrast to "an abattoir where the floor

²⁵⁴ The EFSA Scientific Opinion notes, for example, that in contrast to "an abattoir where the floor should be stable, even and not slippery, seals are killed on different substrates e.g. on land, in the water, on solid ice, loose pack ice, moving ice floes, in environmental conditions that may rapidly alter the position of both sealer and seal, and in weather conditions that may affect visibility". (EFSA Scientific Opinion, p. 88; see also Ibid. p. 35 ("Seals are wild animals and so it is valid to compare the criteria used and the controls in how we kill other wild animals, as well as how domesticated animals are killed in abattoirs ... Care should be taken when comparing the efficacy of these different methods of killing because of the great variation in environmental conditions involved."); VKM Scientific Opinion, p. 38; NOAH Report (2012), p. 19 and Appendix D, p. 4 (statement by Norwegian sealing inspector)).

²⁵⁵ See DFO Overview of the Atlantic Seals Hunt 2006-2010, (Exhibit EU-40), p. 6 ("Although variable ice conditions have been observed historically, there has been an increased frequency of poor ice cover in recent years."); EFSA Scientific Opinion, p. 25 (commenting that "poor ice conditions" in the southern Gulf of St. Lawrence had resulted in a shift of the Canadian hunt towards the northern Gulf). See also Canada's Department of Fisheries and Oceans, *2011-2015 Integrated Fisheries Management Plan for Atlantic Seals*, (DFO Integrated Seals Management Plan 2011-2015), (Exhibit EU-42), pp. 6 ("Although variable ice conditions have been observed historically, there has been a dramatic decline in harp seal-friendly ice cover in recent years. 2010 saw the lowest ice cover ever observed in the Gulf, and suitable ice occurred much further north than is normal at the Front."), 14, and 16-17; NOAH Report (2012), pp. 21-22, Appendix G, p. 5, and Appendix P, pp. 4-5; Canadian Department of Fisheries and Oceans, *Canadian Commercial Seal Harvest Overview 2011* (October 2012), (DFO Commercial Seal Harvest Overview 2011), (Exhibit JE-27), p. 3.

²⁵⁶ The European Union argues that two factors affect the working environment for sealers by causing more broken and unstable surfaces: first, seasonal effects of hunting seals a few weeks older than in past hunts; and second, the alleged impacts of climate change. (European Union's first written submission,

²⁴⁸ See European Union's first written submission, paras. 127, 412; response to Panel question No. 105, paras. 28-32; Canada's response to Panel question No. 55, para. 239; Norway's response to Panel question No. 55, paras. 301-313.

²⁴⁹ See, e.g. Norway's opening statement at the first substantive meeting of the Panel, paras. 177-192 (comments by experts); Canada's opening statement at the first substantive meeting of the Panel, paras. 12-34.

²⁵⁰ EFSA Scientific Opinion, pp. 13-23 and 87.

We note that Cape fur seals are distributed along the southern and western coasts of Southern Africa. In Namibia, which has a history of harvesting seals dating from the 17th century, Cape fur seals occupy colonies located either on the mainland or on small, rocky islands, and Namibia is the only major sealing nation in the southern hemisphere. Harvesting in Namibia is limited to three mainland colonies, and is conducted on dry land by driving selected seals away from the sea. Namibia has submitted to the Panel information about its legislative framework on seal harvesting, including with respect to animal welfare and sustainable ecosystem management. (See Namibia's third-party submission, pp. 4-14). In addition, Namibia is of the view that the EU Seal Regime is discriminatory and fails to achieve the objective of protecting animal welfare. (Ibid. pp. 14-34).

- 62 -

7.188. By contrast, certain conditions such as the visibility afforded by the wide open habitat and the manoeuvrability of boats in open ice formations have been suggested to present possible advantages in seal hunting.²⁵⁷ The evidence taken as a whole, however, indicates that the physical conditions of seal hunting are distinct from those present in the hunting of other wildlife or in the commercial slaughter of farmed animals and pose certain additional challenges in seal hunting. Further, the parties agree that the environmental conditions of the Canadian and Norwegian seal hunts are similar in respect of factors influencing the conduct and humaneness of the hunt.²⁵⁸

Characteristics of seals

7.189. The European Union argues that seals have certain unique features enabling them to stay under water for long periods and that, as a result, seals may experience suffering "peculiar to that species".²⁵⁹

7.190. The Panel observes that seals have special anatomical and physiological adaptations as compared to other animals, such as the ability to withstand poor levels of oxygenation over extended periods of time.²⁶⁰ This has been understood by some to create potential for prolonged life and comparatively prolonged suffering, raising concerns as to the application of 'conventional' slaughter processes to seals.²⁶¹ As a result of these adaptations seals may continue to prolonged display activity while unconscious and even after death, though this characteristic is not limited to seals and is observed in other animals following acute trauma to the brain.²⁶² At the same time, it has been pointed out that the adaptations of seals do not have any effect on killing times for tools causing extensive brain damage (even if they may affect post-stunning or post-mortem reactions).²⁶³

7.191. Although "seals conform to the general mammalian [anatomical] pattern" in terms of their skeleton and internal organs, it has been noted that "compared to terrestrial animals of the same

para. 125; opening statement at the first meeting of the Panel (comments of Professor Broom)). The complainants acknowledge variability in ice conditions and do not dispute the observed reduction of ice cover in recent years. (Canada's response to Panel question No. 65, paras. 285-288; Norway's response to Panel question No. 65, paras. 356-362).

With respect to the seasonal element, we note that some present day hunts do target older seals (in Canada in Norway, for instance, where it is no longer permitted to hunt whitecoat pups) with the result that the hunt takes place later in the spring when the melting and breaking up of ice floes may be more advanced. (Butterworth (2012), p. 2; Richardson (2007), p. 38). Female harp seals give birth in March, and young harp seals are weaned after approximately 12 days, following which they are typically hunted in the "beater" stage of development (i.e. harp seals approximately 3-4 weeks old) as the main target of the hunt. (See Daoust (2002), pp. 1-2; Daoust (2012), p. 445; EFSA Scientific Opinion, pp. 20, 24; see also Johnston et al., "Variation in sea ice cover on the east coast of Canada from 1969 to 2002: climate variability and implications for harp and hooded seals", *Climate Research*, Vol. 29:209-222 (2005), (Exhibit EU-41) (research paper explaining scientists' observation on variation in sea ice over, its potential impact on certain seal populations, and the possible implications of observed changes in climate)).

²⁵⁷ Daoust (2002), p. 7; VKM Scientific Opinion, p. 27.

²⁵⁸ See parties' responses to Panel question No. 72.

²⁵⁹ European Union's first written submission, para. 407. We note that the complainants have not directly challenged the European Union's assertion that seals' physiological capacities may have implications for the killing process. However, Canada has called into question the evidence relied upon by the European Union as "speculative" and of questionable relevance in the context of killing methods that involve the destruction of the brain. (Canada's second written submission, para. 32 and comments on the European Union's response to Panel question No. 155, para. 127). Norway has also advanced arguments regarding "stun/kill" methods causing destruction of the brain. (See, e.g. Norway's opening statement at the first meeting of the Panel, paras. 157-158 and 164-165).

²⁶⁰ See e.g. EFSA Scientific Opinion, p. 69 (noting that seals have a "relatively large total oxygen storing capacity"); VKM Scientific Opinion, p. 25 (stating that seals "are endowed with an enhanced tolerance to lack of oxygen (hypoxia)" and "have a large and impressive capacity to stay submerged for extended periods of time"); NAMMCO Report (2009), p. 10 (describing the "diving physiology of seals" based on certain "morphological and physiological adaptations, including the "ability to store substantial amounts of oxygen in a large blood volume" and economize oxygen stores, as well as "enhanced tissue hypoxia tolerance at the cellular level").

²⁶¹ Butterworth (2012), pp. 6-7.

²⁶² See e.g. VKM Scientific Opinion, p. 25; First Statement by Dr Knudsen, (Exhibit NOR-5), para. 41; E. O. Øen, *The Norwegian Sealing and the Concept of 'Humane Hunting'* (2006), (Exhibit NOR-36), p. 5 ("reflex movements can continue for several minutes after the seal is unconscious and brain dead" and "other organs like the seal's heart may be beating for a long time after the seal is dead").

²⁶³ NAMMCO Report (2009), pp. 10-11; VKM Scientific Opinion, p. 25.

- 63 -

size the seal's body is characteristically torpedo-shaped, the limbs are short and there is a lack of external landmarks on its torso".²⁶⁴ In terms of behaviour, seals may move when or even after they are shot or hit by a hakapik posing a challenge to sealers.²⁶⁵ However, seals have been regarded by some as relatively docile compared to other hunting targets.²⁶⁶

Methods for hunting seals

Humane killing methods in seal hunting

7.192. The parties acknowledge generally accepted principles of humane killing, described by EFSA as "the act of killing an animal that reduces as much as possible unnecessary pain, distress and suffering i.e. that causes no avoidable pain, distress, fear or other suffering".²⁶⁷ Based on such principles, the parties agree that a three-step killing method is the commonly recognized benchmark for the humane killing of seals.²⁶⁸ This method consists of (a) effective stunning that results in loss of consciousness; (b) checking to ensure loss of consciousness²⁶⁹; and (c) effective bleeding out. These three steps are described below.

7.193. EFSA and other scientific studies explain that the most commonly applied methods of stunning involve targeting the seal's head with either firearm or clubbing instrument to render it irreversibly unconscious.²⁷⁰ The principal tools used for this purpose are: hand-held striking instruments such as a hakapik (consisting of a wooden handle with a metal ferrule at one end that has a slightly bent spike on one side and a blunt projection on the other)²⁷¹ or club²⁷²; and

²⁶⁴ VKM Scientific Opinion, p. 23. The VKM Scientific Opinion goes on to conclude that "[t]herefore, unlike most other forms of wildlife hunts, the head (and brain) is the preferred target rather than the thorax as in terrestrial large game."

²⁶⁵ EFSA Scientific Opinion, p. 41; NOAH Report (2012), pp. 10 and 25 (statements by a Norwegian sealer and a sealing inspector); VKM Scientific Opinion, p. 35. ²⁶⁶ Daoust (2002), p. 7; First Statement by Mr Danielsson, (Exhibit NOR-4), para. 30. Note that younger

seals (predominantly hunted in Norway and Canada) are regarded as more docile, whereas adult seals have been noted to be more responsive to human presence and likely to attempt to escape. (European Union's opening statement at the first meeting of the Panel (statement of Professor Broom)). ²⁶⁷ EFSA Scientific Opinion, p. 117 (definition of "Humane killing"). See also VKM Scientific Opinion,

p. 10; IVWG Report (2005), p. 2; Butterworth (2012), p. 4; E. O. Øen, The Norwegian Sealing and the Concept of 'Humane Hunting' (2006), (Exhibit NOR-36), p. 1; American Veterinary Medical Association ("AVMA"), Guidelines on Euthanasia (June 2007), (Exhibit NOR-91); AVMA Guidelines for the Euthanasia of Animals: 2013 Edition, (Exhibit NOR-133); Canada's first written submission, para. 91; Norway's first written submission, para. 172; European Union's first written submission, para. 79.

²⁶⁸ Canada's first written submission, paras. 93-98 (describing its Marine Mammal Regulations as patterned on the three-step method); response to Panel question No. 62 (considering that "the three-step process represents the best available scientific knowledge for best practices to ensure a good animal welfare outcome"); Norway's first written submission, paras. 172-180 (describing what is formally a two-step process of stunning and bleeding, but referring to the first step as an outcome that must be assured before bleeding); response to Panel question No. 62 (describing its second step as "second stunning" with a hakapik spike and explaining the stun/kill concept of its approach); European Union's first written submission, paras. 98-105 (recognizing the three-step method as a "recommended killing method"); response to Panel question No. 63 (explaining that a "genuinely humane method for killing seals [could be] based on the three-step method"). See also DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), p. 7; Statement of Mr Landmark, (Exhibit NOR-8), para. 38.

²⁶⁹ See footnote 309 below with respect to the practice of "second stunning" used in Norway to ensure irreversible unconsciousness as the second step of the killing process.

²⁷⁰ See, e.g. EFSA Scientific Opinion, p. 68; VKM Scientific Opinion, p. 28; NAMMCO Report (2009), p. 11; IVWG Report (2005), pp. 7-8; Burdon (2001), p. 4.

²⁷¹ We note also the existence in Norway of a modified version of the hakapik called a "slagkrok". This is an iron club with a sharp hooked spike opposite the club. (VKM Scientific Opinion, pp. 7, 33-34). EFSA observed that "slagkroks are now rarely used" and therefore limited reference to the slagkrok "to when its use is different from the hakapik". (EFSA Scientific Opinion, p. 38). Other studies have included reference to the slagkrok in connection with discussion of the hakapik. (See VKM Scientific Opinion, pp. 32-35; NAMMCO Report (2009), p. 17 and Appendix 1, Section 2.2). ²⁷² Generally speaking, clubs are distinguished from the hakapik by the absence of a spiked metal

ferrule at one end. (See EFSA Scientific Opinion, pp. 41-42).

firearms.²⁷³ It may entail multiple shots/blows either to ensure that the seal has been effectively stunned or to cure any previously unsuccessful attempt.²⁷⁴

7.194. After stunning, the next step requires checking to ensure that the seal is in fact unconscious and insensible to pain. Two generally recognized methods for checking loss of consciousness are "blinking tests" to check corneal reflex and skull palpation to assess physical brain damage. There are certain challenges associated with this step, such as those posed by the physical conditions of seal hunts and neurophysiological features of seals.²⁷⁵

7.195. Bleeding (or exsanguination) is to be performed on an effectively stunned and checked seal so as to terminate blood flow to the brain and ensure death at the time the seal is skinned.²⁷⁶ In some instances where the seal is killed by the application of stunning, bleeding can be a precautionary step rather than the primary means of killing.²⁷⁷

7.196. Furthermore, there is considerable evidence to show that the effectiveness of the method used to kill seals is at least partially dependent on the abilities and competence of sealers. For instance, EFSA explained that for any given killing method "the best practice for that method" would involve a competent person with well-maintained equipment²⁷⁸ and that differing observations between studies of seal hunts might be accounted for by "individual differences in sealer behaviour and competence".²⁷⁹

Application of humane killing methods in seal hunting

7.197. The European Union argues that, although it could be possible in theory to prescribe a humane method for killing seals, in practice the unique conditions in which seal hunting takes place make it impossible to *apply* and enforce any such method in an effective and consistent manner.²⁸⁰ The complainants submit that the conditions in which seal hunts occur do not create

²⁷⁹ EFSA Scientific Opinion. p. 54. See also ibid. p. 43 ("Besides the power of the rifle and ammunition used, a rifle is only as effective in its ability to stun/kill an animal as the marksmanship of the hunter and the conditions under which the hunt is conducted").

Indications from both participants in the hunts and veterinary experts recognize the heavy demands and difficult conditions of seal hunts. (See e.g. NOAH Report (2012), p. 12 and Appendix K, pp. 6, 9 (noting indications of sealers' fatigue over the course of a hunt); p. 24 and Appendix P, p. 4 (comment by the Norwegian Fisheries Directorate); IVWG Report (2005), pp. 5, 8-9 (noting of a specific location, the Canadian Front, that "[b]ecause of ice, sea and weather conditions there are greater challenges for hunters to carry out" the recommended killing method)).

²⁸⁰ European Union's first written submission, para. 373.

²⁷³ See EFSA Scientific Opinion, pp. 37-45.

²⁷⁴ See para. 7.205 below regarding the application of humane killing methods as to the competing views about the humaneness of applying multiple shots/blows for stunning. Although Norway does not require "checking" in the sense of monitoring the animal, its requirement of "second stunning" is intended to serve the similar function of confirming irreversible unconsciousness.

²⁷⁵ See paras. 7.208-7.210 discussing the application of checking seals and paras. 7.217-7.218 regarding hooking/gaffing conscious seals.

²⁷⁶ See EFSA Scientific Opinion, p. 49.

²⁷⁷ See e.g. EFSA Scientific Opinion, p. 94, General Recommendation 2 ("Seals should be killed and skinned in a way that meets the three steps of effective stunning or killing, effective monitoring and effective bleeding out, before being skinned".); Burdon (2001), p. 1; IVWG Report (2005), p. 2; Daoust (2012), p. 446 (illustration: "Poster provided to Canadian sealers illustrating a three-step process for killing young harp seals, with the purpose of inflicting minimal or no pain to the animal".).

Note that some suggest that there may be a fourth step in a humane killing, namely ensuring that unconsciousness persists throughout the bleeding process. (See e.g. Richardson (2007), p. 17 and *amicus curiae* submission by Anima et al., pp. 61-62 (Exhibit EU-81), pp. 12-13). Richardson notes that "[t]he potential for tests of unconsciousness to be inaccurately performed in the context of the seal hunt, paired with the reality that neither test [corneal reflex and cranial palpation] confirms irreversible unconsciousness necessitates a fourth step in humane killing at the commercial seal hunt – re-stunning the seal if needed during exsanguination." Richardson then goes on to add, "[t]here is good reason to believe this four step killing process would not be effectively and consistently carried out in the physical environment in which the Canadian seal hunt takes place." (Richardson (2007), p. 17).

²⁷⁸ EFSA Scientific Opinion, p. 37.

inherent obstacles to humane killing and that, in fact, seals are killed humanely in their respective hunts.²⁸¹

7.198. The Panel examines, based on the evidence before it, the parties' arguments concerning the application of generally recognized humane killing methods at each step of the seal killing process.

Stunning (clubbing²⁸² or shooting)

7.199. The parties contest whether the physical environment and conditions in which seal hunting occurs form an inherent obstacle to accurate and effective stunning.²⁸³

7.200. The Panel first recalls that the most commonly applied stunning methods in seal hunts involve targeting the seal's head with either firearm or clubbing instrument to render it irreversibly unconscious. The evidence before us shows that physical conditions can affect the choice of instrument as well as the manner in which the stunning takes place.²⁸⁴ EFSA, for example, notes that more solid ice accumulation can facilitate the application of short-range tools such as hakapiks, whereas more unstable and sparse ice formations will favour the use of stunning from longer ranges with firearms.²⁸⁵ In light of the observed deterioration of ice conditions in recent hunts, EFSA concludes that the use of rifles is likely to continue to dominate and even to increase if the poor ice conditions persist.²⁸⁶

7.201. Against this backdrop, we proceed to examine the degree to which stunning methods in seal hunts can be effectively and consistently applied.

²⁸³ European Union's first written submission, paras. 127-138; Norway's opening statement at the first meeting of the Panel, paras. 177-192 (comments by experts); Canada's opening statement at the first meeting of the Panel, paras. 12-34.

²⁸⁴ In this connection, various exhibits explain the unsuitability of other tools and methods of stunning in light of the physical environment and conditions of seal hunting. (See VKM Scientific Opinion, p. 42 (noting that "neither the use of electricity, gas, nor captive bolts would be practical nor realistic to try in the environmental conditions in which the seal hunting is carried out" and that captive bolt methods" would not function well under arctic conditions" and direct placement against the seal's head may not be practicable for unrestrained seals); EFSA Scientific Opinion, p. 48 (commenting on unsuccessful research using pistols and that "proper maintenance of such weapons under cold and wet conditions on the ice floes would be difficult"; "captive bolts may not be appropriate [because] they may not function properly under harsh weather conditions" and use on "unrestrained animals on the ice would pose additional limitations"; and that poisoning/anaesthetic drugs had not been found to be practical; see also NAMMCO Report (2009), p. 13; E. O. Øen, *The Norwegian Sealing and the Concept of 'Humane Hunting'* (2006), (Exhibit NOR-36), p. 4 ("[p]istols and revolvers would be too inaccurate", "penetrative or non-penetrative captive bolt pistols used in slaughterhouses ... would probably be even poorer than pistols", and "[t]he use of shotguns is unthinkable")).

Note, however, that other factors of a given seal hunt may also be relevant to the choice of instrument. (See, e.g. VKM Scientific Opinion, p. 27 (explaining the skull thickness of younger seals making hakapiks appropriate)).

²⁸⁵ EFSA Scientific Opinion, pp. 25 (depending on ice conditions and the size of ice floes, "hunters either shoot the seal from their boats or go down on the ice and strike the animals with a hakapik") and 27 (regarding the Norwegian hunt "[t]he percentage of young seals which are shot (as opposed to the hakapik being used as the primary killing method) may vary somewhat depending on the time of year and the conditions of the hunt (i.e. stability and thickness of ice)"); Daoust (2012), p. 445 ("[d]epending on ice conditions, these seals are hunted mainly with one of two regulation tools ...: the hakapik ... in years of good ice, when the sealers can get down on the ice and approach the animals; or a rifle ... where ice floes are typically much smaller and more spread out"); Butterworth (2007), p. 8.
²⁸⁶ EFSA Scientific Opinion, pp. 45 and 89, point 5. See also Daoust (2002), p. 2 ("Ice conditions, which,

²⁸⁶ EFSA Scientific Opinion, pp. 45 and 89, point 5. See also Daoust (2002), p. 2 ("Ice conditions, which, in recent past, have varied considerably from year to year, also influence the nature of the hunt. Years of poor ice formation, with predominance of small ice floes, have led to a larger proportion of the animals being shot rather than struck with a hakapik."); NAMMCO Report (2009), p. 17 ("The environmental conditions and the nature of the hunt determine to a large extent whether the hakapik is being used as the primary tool or not. For example the use of the hakapik as the primary tool has been reduced due to poor ice conditions and changes in targeted age classes").

²⁸¹ Canada's opening statement at the first meeting of the Panel, paras. 12-14; Norway's opening statement at the first meeting of the Panel, paras. 140-143.

²⁸² We note that, unless otherwise indicated, the use of the term "clubbing" in these Reports refers generally to the striking of seals and is not limited to the use of a specific instrument. The general term "clubbing" may be understood as corresponding to EFSA's reference to "physical methods (e.g. hakapik or club)", and therefore is not confined to clubs (as distinguished from hakapiks). (See EFSA Scientific Opinion, p. 68).

7.202. Regarding the effectiveness of firearms for stunning, the main risk is "the targeted animal being hit with insufficient force and/or accuracy to cause instantaneous death or unconsciousness, and possibly escaping wounded".²⁸⁷ Contributing factors to this risk include firing from excessive distance and unstable platforms (e.g. by the relative motion of the boat and of the ice floe on which the seal is resting). Multiple scientific reports have explicitly correlated the accuracy (and thus effectiveness) of firearms with such environmental factors and the small size of the target (head and upper neck).²⁸⁸

7.203. At the same time, there is evidence of means employed to enable effective stunning with firearms and to mitigate the risks of inaccurate shooting. For example, the use of optical sights with magnifying lenses on modern rifles can serve to enhance accuracy at the shooting distances relevant for seal hunting.²⁸⁹ Additionally, several sources stress the destructive power of various firearms and ammunition used for stunning that can help ensure that struck animals will be rendered unconscious.²⁹⁰ Finally, there are also suggestions that hunters may exercise their judgment to refrain from attempting to stun seals when poor conditions pose a risk of ineffective stunning.²⁹¹

7.204. With respect to what EFSA terms "physical methods" of stunning, namely by striking with a hakapik or other clubbing instrument, there is also evidence detailing the circumstances in which such methods may result in ineffective stunning. As with firearms, an effective stun will largely depend on the accuracy and force of the blow, which in turn can be affected by the stability of the platform, the balance of the hunter, and the hunter's position relative to a potentially moving

²⁹⁰ See e.g. VKM Scientific Opinion, pp. 30-31; NAMMCO Report (2009), p. 14; First Statement by Mr Danielsson, (Exhibit NOR-4), paras. 41-46; First Statement by Dr Knudsen, (Exhibit NOR-5), para. 22 (stunning involving "a massive impact to the seal's brain" and in the case of rifles potentially causing "complete destruction of the brain after impact of the high-velocity expanding bullet").

²⁸⁷ EFSA Scientific Opinion, p. 44.

²⁸⁸ EFSA Scientific Opinion, pp. 43-44 and 89; VKM Scientific Opinion, p. 32 ("There are disadvantages from the use of firearms in the hunt. The risk for bad shots (i.e. strayshots or hits outside vital areas) is present and increase with longer shooting ranges (>50m) and/or if the boat (platform) is moving heavily or the ice floes are in motion in bad and windy weather conditions. The seal might also suddenly move its head. In such situations, the shot might either miss completely, or wound and injure the animal. Seals shot and wounded might escape before they are re-shot."); Butterworth (2012), p. 5. See also NOAH Report (2012), p. 19 and Appendix D, p. 4 (description by sealing inspector of relevant factors to accurate shooting); Letter from Norwegian Fishing Vessel Owners Association, (Exhibit EU-44) (noting that a sealing vessel will be "in motion due to swell, waves and its own progress").

²⁸⁹ EFSA Scientific Opinion, p. 44 ("Modern rifles with optical sights, possibly combined with rangefinder are very accurate weapons at shooting distances relevant for seal hunting, and shots fired at the brain will usually be grossly destructive with severe bleeding and tissue damage".); First Statement by Mr Danielsson, (Exhibit NOR-4), para. 29 (stating that shooting ranges are usually from 30-40m, but may be between 10-70m, which can be "effectively very close" with magnification lens); Third Statement by Mr Danielsson, (Exhibit NOR-163), para. 5.

²⁹¹ EFSA Scientific Opinion, pp. 40, 45; First Statement by Mr Danielsson, (Exhibit NOR-4), para. 38; Second Statement by Mr Danielsson, (Exhibit NOR-128), paras. 4-5 (recognizing that hunting is an outdoor activity with conditions that "could affect the accuracy of a shot", but that hunters may adjust for this factor) and para. 22 (stating that hunting may not proceed under inclement conditions). But see also NOAH Report (2012), p. 10 (comments of a penalized captain of a vessel describing the difficulty of exercising judgment, stating "[t]here will always be an evaluation from the hunter if he dares to shoot. You seldom get a perfect situation in hunting. In our line of business many wrong decisions are made ...").

seal.²⁹² On the other hand, there are indications of the suitability and effectiveness of the hakapik on younger seals due to their relatively weaker skulls and lesser reaction to human intrusion.²⁹³

7.205. We note that the parties have made specific arguments as to the practice of inflicting multiple blows and/or firing multiple shots on a targeted seal during the stunning process.²⁹⁴ On this matter, we do not consider that the infliction of multiple blows or repeated shots is *per se* unacceptable in terms of animal welfare, given that this practice may be precautionary and not necessarily a consequence of ineffective stunning.²⁹⁵ As a remedial action for an ineffective first stunning attempt, multiple blows or shots could be consistent with humane killing principles to the degree they were performed accurately and rapidly. At the same time, we observe that attempts at re-stunning (particularly with firearms) may pose the same challenges to accurate and effective stunning as in the original instance, but with the possibility of additional obstacles from the motions of a struck seal and the difficulty of determining whether re-application of stunning is required.²⁹⁶

7.206. The European Union has described the difficulties of effective stunning as part of the inherent obstacles to humane killing in seal hunts. Although the complainants also accept the possibility of ineffective stunning in seal hunts, they dispute the prevalence of its occurrence and implications for the general characterization of the humaneness of methods applied in seal hunts.²⁹⁷ In this regard, EFSA observes that there is a generally a limited amount of data available on animal welfare in seal hunts, including with respect to effective stunning²⁹⁸, and that existing data may give rise to conflicting conclusions, "i.e. uncertainty is high in data interpretation".²⁹⁹

²⁹³ EFSA Scientific Opinion, p. 39; VKM Scientific Opinion, p. 45. See also Norway's opening statement at the first meeting of the Panel, paras. 163-170 (comments of Dr Knudsen regarding the effectiveness of the hakapik when properly used and consistency with AVMA guidelines); Daoust (2012) (regarding use of hakapiks on seals at the 'beater' stage of development).

²⁹⁴ See European Union's first written submission, paras. 114 and 172; Canada's second written submission, paras. 49, 60-61; Norway's opening statement at the first meeting of the Panel, para. 167. There is some conflicting evidence as to the animal welfare implications of such practice, as some experts have considered the practice unacceptable (See e.g. Burdon (2001), pp. 1 and 13; Richardson (2007), p. 13; Butterworth (2012), p. 6) whereas other have highlighted its effect of ensuring effective stunning (See e.g. EFSA Scientific Opinion, p. 38; IVWG Report (2005), p. 8). Various sources point to the option of firing follow-up shots in the case of any doubt that a seal has been effectively stunned with the first shot. (See e.g. Daoust (2002), p. 6; First Statement by Mr Danielsson, (Exhibit NOR-4), para. 29).

²⁹⁵ See EFSA Scientific Opinion, p. 38 (explaining that observations of sealers striking seals multiple times can be pursuant to regulatory standards or "in order to for the sealer to ensure that the animal is indeed stunned or killed, not necessarily because the extra blows are needed").

²⁹⁶ In the case of follow-up shots with a rifle, even assuming that the seal's state of consciousness was clearly ascertained, "the erratic movements of the wounded animal, coupled with the bobbing movements of the vessel, would not guarantee the success of the 2nd shot." (Daoust (2002), p. 6; see also paras. 7.208-7.210 regarding the difficulties of checking consciousness).

²⁹⁷ See Canada's and Norway's responses to Panel question No. 69 (both referring to the EFSA finding that no killing method will work at all time and under all circumstances).

²⁹⁸ See, e.g. EFSA Scientific Opinion, pp. 12 and 76.

²⁹⁹ EFSA Scientific Opinion, p. 91.

The data and findings of various scientific reports submitted to the Panel draw upon eyewitness accounts and inspections; review of video of the Canadian hunt in different locations and years; and post-mortem skull and carcass examinations. (See EFSA Scientific Opinion, pp. 53-61; NOAH Report (2012), pp. 12, 18-19, and 25-27 (comments of Norwegian sealing inspectors and sealers); Butterworth (2007), pp. 3, 36, and 41; Butterworth (2012), pp. 2-3; VKM Scientific Opinion, p. 32; Transcript of statements to the radio station CBH-FM (28 December 2008), (Exhibit EU-105), pp. 10-11 (comments of Canadian sealer)). This evidence is predominantly taken from the Canadian commercial seal hunt, and certain limitations have been noted in the availability of data on the Norwegian hunt. (See Norway's response to Panel question Nos. 56 (para. 322)

²⁹² EFSA Scientific Opinion, pp. 39 (noting possibility that the "first blow from the hakapik may not stun the animal immediately if it does not hit the calvarium but hits, for example, the jaw or snout or other part of the body"), 40 ("The hunter has to be close to a seal, and if it moves its head or moves away, the accuracy of the strike may be compromised. ... The hunter should withhold striking a blow if it is unlikely to be of sufficient force or accuracy e.g. if he is off-balance or if the animal is in a position to escape into the water."), 41 (notes as disadvantage of clubbing that "the accuracy of blows is compromised by the fact that the targets are nearly always in motion"), 70 ("The angle of striking the skull of a seal with a rounded club or hakapik may influence the impact energy delivered to skull, the direction of travel for the spike of the hakapik and, as a consequence, the efficacy of the strike".), and 88, point 4 ("The accuracy of the strike may be compromised if the seal moves its head, or moves away and this depends very much on the behaviour of the seal species, the skill of the hunter and the environmental conditions in which the hunt takes place".); NAMMCO Report (2009), p. 17 ("When using the blunt projection of the hakapik the hunter's relative position to the animal is less important than a stable platform"); Butterworth (2007), p. 41.

- 68 -

Having due regard for these limitations, our review of available evidence confirms that inaccurate and ineffective stunning by both hakapik and firearms does occur and that seal hunting poses inevitable risks that some seals will not be instantly and effectively stunned.³⁰⁰

7.207. There is some evidence in the form of data samples and sealing inspector observations that indicate potentially high rates of accurate stunning.³⁰¹ Though this may serve as empirical

("Statistics on the incidence of poor animal welfare outcomes have not historically been maintained for the Norwegian seal hunt.") and 61 (para. 345) (regarding independent observers). See also EFSA Scientific Opinion, pp. 28, 66; VKM Scientific Opinion, p. 3 ("scientific, peer-reviewed studies and scientific data on the actual performance of the Norwegian hunt are very limited")).

³⁰⁰ We note that the parties have devoted considerable argument to the comparability of stunning in seal hunts and commercial abattoirs. At the same time, all parties have recognized that there are significant differences in the physical environments of the two settings. (See e.g. Canada's response to Panel question No. 56; second written submission, footnote 66; Norway's response to Panel question No. 56; second written submission, paras. 290-294). Also, as we have noted above, stunning tools commonly employed in slaughterhouses (such as captive bolts) have been generally deemed unsuitable for use in seal hunts on account of basic environmental factors and the challenge of hunting unrestrained wild animals approached from a distance in their marine habitat. In our view, these basic differences limit the value and relevance of a comparison to the stunning in commercial slaughterhouses when determining the risks of inhumane stunning and killing in seal hunts in particular. (See e.g. VKM Scientific Opinion, p. 39; NAMMCO Report (2009), p. 16). For similar reasons, we find comparisons to terrestrial wildlife hunts to be of limited value in the determination of such risks based on *inter alia* differences in the hunting tools, areas targeted on the animal, and physical settings of the hunts. (See EFSA Scientific Opinion, p. 36; VKM Scientific Opinion, p. 36-38; E. O. Øen, *The Norwegian Sealing and the Concept of 'Humane Hunting'* (2006), (Exhibit NOR-36), p. 2; NAMMCO Report (2009), p. 15).

In any event, the evidence before does not clearly establish that effective stun rates in the seal hunt are comparable to those in a commercial abattoir, as argued by the complainants. EFSA has estimated that the rate of mis-stunning in an abattoir is 4 to 6.6% for captive bolt shooting, and notes of a field study of cattle stunning with captive bolt that "[a]II catle where the first shot missed were immediately restunned". (EFSA Scientific Opinion, p. 36; see also Daoust (2012), p. 453 (comparing findings on the Canadian seal hunt to hunts of deer, waterfowl, and noting that "the success rate (ie no return to sensibility) for stunning cattle in the best slaughter plants under the best conditions averages 97-98%", though pointing out lower rates in other samples); Tables summarising the evidence available on the rates of mis-tuns and delays in commercial sealing and in slaughterhouses, (Exhibit EU-128)). Canada has compared its struck and lost rate of 5% to the figures given by EFSA, but we note that this comparison does not account for welfare outcomes of inhumanely stunned seals that were ultimately recovered rather than lost. (See Canada's response to Panel question No. 56, para. 252). It is also difficult to draw any conclusion with respect to the Norwegian hunt due to the lack of representative data on stunning rates. Further, it is unclear that the animal welfare consequences of misstunning in a slaughterhouse are the same as those in a seal hunt. For example, the circumstances and challenges of *re-stunning* in a commercial abattoir are distinguishable from those in a seal hunt based on *inter* alia the proximity to and restraint of the animal, though this may present distinct challenges for re-stunning in a slaughterhouse that are not present in a seal hunt. (See e.g. NAMMCO Report (2009), p. 15; EFSA Scientific Opinion related to welfare aspects of the main systems of stunning and killing the main commercial species of animals, EFSA Journal (2004), (Exhibit CDA-47)).

In this connection, we note the complainants' arguments that seal hunts do not pose the pre-killing stress experienced in commercial slaughters. While it is true that seals are not kept in captivity, there is some evidence indicating that seals can experience distress before stunning. Behavioural change during pain and fear in seals (including escape attempts, vocalisations, freezing, rearing up in a defensive posture, opening the mouth, and violent body movements) have been clearly observed in seals immediately prior to being clubbed, though they were generally noted to be relatively short-term events. (See Butterworth (2007), p. 38 and Video footage cited in Butterworth (2012), (Exhibit EU-38), "Fear and Distress"). Ultimately, however, we do not consider the pre-killing condition of the seal to be relevant to the question of humane *killing* and the related risks to seals of poor animal welfare in seal hunts.

³⁰¹ For instance, Daoust (2012), a 2012 study of the Canadian hunt, compiled results from four sealing seasons in the Canadian Front and Gulf. The most empirically robust of the data from this study are taken from the 2009 hunt at the Canadian Front east of Newfoundland and Labrador, from which observations were made from a sealing vessel of the complete killing process for 280 seals (only two of which were killed with a hakapik) with the collection of detailed information on several parameters associated with the process. Of particular note, the study records the original site of injury for 245 seals, of which 218 seals (89%) were struck in the head, 25 seals (10.2%) in the neck, and two seals (0.8%) in the "cranial thoracic region". Fourteen (5%) out of the 278 seals that were shot were considered to have had a poor animal welfare outcome as they were not killed immediately with the first shot and were not shot again before being retrieved. This study also recorded the skull damage to over 200 seals stunned by hakapik in the 2005 hunt in the Gulf of St. Lawrence (the majority of which received multiple blows), revealing a completely crushed skulled in all but 12 seals that had varying severity of fractures. (Daoust (2012), p. 449-450). In addition, records maintained by an inspector of the Norwegian hunt show very low struck and lost rates for 5,647 seals hunted in two years, suggesting some degree of accuracy and effectiveness in the use of the rifle. (Second Statement by Mr Danielsson, (Exhibit NOR-128), Annex; see also VKM Scientific Opinion, pp. 29-30; E. O. Øen, *The Norwegian Sealing and*

- 69 -

confirmation that humane stunning can be carried out in some instances, it does not contradict the existence of risks of ineffective stunning and thus poor animal welfare.³⁰² It is also not clear that the studies claimed to reflect high levels of animal welfare can be conclusively generalized, as they may lack "adequate sampling that is representative of the entire hunt with respect to sample size and sampling design".³⁰³ However, even if such data could be extrapolated to the entire hunt, a poor welfare rate of 5% (as found in the most robust data sample of Daoust (2012) with the greatest continuity of evidence) would still reflect a risk of inhumane killing in seal hunts that, depending on the scale of the hunt, could represent the suffering of a large number of seals. Furthermore, there are studies showing potentially less accurate and effective applications of stunning methods³⁰⁴, and accounts given by seal hunt participants likewise demonstrate a certain level of risk and inaccuracy in stunning.³⁰⁵

Checking

7.208. There is disagreement among experts about the most appropriate method for checking seals to ensure irreversible loss of consciousness. The two principal methods are a corneal reflex "blink test"³⁰⁶ and palpation of the seal's skull.³⁰⁷ While the loss of a blink reflex can indicate loss of consciousness, there have been concerns about its reliability and the difficulty of interpreting seals' reflexes. Skull palpation directly examines the physical damage to the seal's skull and brain,

be alive)). ³⁰² As stated in Daoust (2012) of the hakapik and rifle, and affirmed by its empirical findings above, "neither tool offers a complete guarantee of instant death". (Daoust (2012), p. 453). ³⁰³ EFSA Scientific Opinion, p. 51. See also Canada's response to Panel question No. 172, para. 220;

³⁰³ EFSA Scientific Opinion, p. 51. See also Canada's response to Panel question No. 172, para. 220;
 Norway's response to Panel question No. 172, para. 333.
 ³⁰⁴ An earlier study by Dr Daoust et al., Daoust (2002), combined first-hand observations in different

locations and years of the Canadian hunt along with post-mortem cranial examination of seven killed seals. The recorded observations in the different locations were of a different nature according to the type of examination performed, the number of seals examined, and the instrument used to stun the seal. Thus, from the 1999 hunt at the Canadian Front examining 47 carcasses of shot seals: 35 (75%) were shot in the head; six (13%) were shot in the neck with complete transection; three (6%) were shot in the ventral region of the neck with tissue damage but no bone fracture; the remaining three (6%) were shot in thorax or abdomen, one of which was found alive by itself on an ice floe. In the 1999 hunt in the Gulf, "a minimum of 225 carcasses" were examined and showed only four (1.8%) skulls without multiple calvarial fractures (highlighted by some as evidence of high stunning rates, despite the limitations noted in the EFSA Scientific Opinion, p. 57). For the 2001 hunt at the Canadian Front, a total of 167 seals shot or stuck on the head were observed, recording those brought on board (158) and lost (9), and considering three of seals to be alive after even after they were brought on board. The specific skull damage caused by a hakapik was recorded for 100 seals with the following results: "86 had a completely crushed calvarium with complete destruction of both cerebral hemispheres" and the remaining 14 had partial fractures (e.g. only one side or frontal portion of the calvarium). Other studies involving post-mortem cranial examinations show findings that were interpreted to indicate inaccurate stunning attempts with a hakapik. (See e.g. Burdon (2001) (76 cranial evaluations showing 17% without any apparent skull fractures, leading to the conclusion that it was highly unlikely that the seals would have been unconscious, and 25% with "minimal" or "moderate" fractures) and Butterworth (2007) (17 post-mortem examinations of seals killed in 2007, all of which had been clubbed and one that had also been shot, noting that 47% had been clubbed on the face or neck, and 82% had ocular damage)).

³⁰⁵ See e.g. Transcript of statements to the radio station CBH-FM (28 December 2008), (Exhibit EU-105), pp. 10-11 (comments of Canadian fisherman and sealer that not "every shot is a clean one, and no matter what, how good the people are at controlling the hunt is, there's going to be some animals that's [sic] going to have to be given the second shot"); NOAH Report (2012), Appendix L, p. 6 ("Shooting whilst moving, in sea swell and/or on moving ice is highly demanding. Different degrees of wounding of animals will therefore arise, but my impression was that the proportion of animals wounded by gunshot lies within the framework of what one would necessarily expect with this kind of hunting.") and Appendix I, p. 4 (instances of wounded but not killed seals due "to a large sea swell, which during parts of the hunt made accurate shooting very difficult").

³⁰⁶ See EFSA Scientific Opinion, p. 72 (explaining that "absence of a corneal reflex" can be "used as an indicator of brain damage or brain failure"); Burdon (2001), p. 4 (stating that "bilateral loss of corneal reflexes ('blinking eye' reflexes) is generally accepted as the most accurate means of confirming a loss of consciousness").

³⁰⁷ IVWG Report (2005), p. 7 ("Checking (palpation of the skull) is the process of manually depressing the skull to ensure that the crushing process has been thorough (including both hemispheres of the brain) and has resulted in the desired irreversible loss of consciousness or death.").

the Concept of 'Humane Hunting' (2006), (Exhibit NOR-36) (Norwegian study carried out on 349 weaned harp seal pups shot with rifles showing 418 fired shots of which 384 struck seals, with extra shots fired at seals shot outside the neck and head area; 343 (98.3%) of the seals were considered to have been rendered instantly unconscious or dead, while two of the 338 seals hit in the head and four of the other 11 seals were judged to be alive)).

but has drawn concern that consciousness (and therefore sensibility to pain) may persist despite severe damage.³⁰⁸

7.209. The parties specifically dispute the feasibility of determining the consciousness of seals from a distance when hunted with firearms.³⁰⁹ There is evidence showing that a *successful* stun can be associated with complete immobility as well as with seizures of varying intensity resulting in a "swimming reflex".³¹⁰ Furthermore, there is evidence that an *unsuccessful* stun can be indicated by both continued movement (especially coordinated or "directed" movements) as well as immobility from "fear-induced paralysis".³¹¹ Observations have been made that some potentially distinguishing features between these various states would make checking unconsciousness in seals plausible.³¹² Nevertheless, there is a general recognition that the neurophysiological

³⁰⁹ European Union's first written submission, para. 119 and response to Panel question No. 57; Norway's opening statement at the first meeting of the Panel, para. 156; Canada's second written submission para. 50; Parties' responses to Panel question No. 57.

³¹¹ EFSA Scientific Opinion, p. 71, footnote 15: "A behaviour known as 'fear-induced paralysis' (the equivalent of tonic immobility or freezing behavior shown in poultry and rabbits) and characterized by tonic contraction of the whole body, has been described in a young harp seals and is shown by some animals that feel threatened. Such immobile seals might be interpreted as dead but would still be conscious." (citations omitted) See also C. Lydersen and K.M. Kovacs, "Paralysis as a defence response to threatening stimuli in harp seals (*Phoca groenlandica*)", *Canadian Journal of Zoology*, Vol. 73 (1995), (Exhibit CDA-113), p. 486 (finding that 328 of 382 harp seal pups tested (86%) and 26 of 46 adult seals tested (57%) exhibited the passive defence response of paralysis).

³¹² See EFSA Scientific Opinion, p. 71:

Despite these strong movements the animal moves non-directionally, as opposed to the directional flight response of a threatened animal. These reactions have been described as the 'swimming reflex' and may be ... viewed as the movements of a successfully stunned animal ... When complete relaxation of the body is seen, it is likely to be associated with complete destruction of the brain and brain stem, and with no breathing. On the other hand if an animal is immobile with some muscle tone (head raised off the ice, flippers load bearing) or it exhibits a state of fear-induced paralysis, and remains immobile, with its head retracted and its front flippers flat against its body, it may still be conscious. (footnote omitted)

See also Norway's opening statement at the first meeting of the Panel, paras. 154-170 (comments of Dr Knudsen); IVWG Report (2005), p. 17; Butterworth (2007), p. 18; Canada's and Norway's responses to Panel question No. 57 (emphasizing the relaxation, contraction, or motion of the seal's head as a factor to distinguish

³⁰⁸ EFSA Scientific Opinion, pp. 72 (noting the divergence in some studies as to the indicative value of the corneal reflex test for loss of consciousness, and that skull palpation would need to be accompanied by criteria "before any particular degree or extent of damage felt during palpation could be reliably interpreted for field use"), 40, 50-51, and 60; Burdon (2001), p. 4 ("skull palpation is not the most reliable as a means of interpreting death or level of consciousness. The location and severity of crush injuries involving the CNS will affect the possible outcome; it is therefore open to misinterpretation. The bilateral loss of corneal reflexes ('blinking eye' reflexes) is the most accurate means of confirming a loss of consciousness."); Daoust (2002), p. 6 (nothing that "[c]omplete collapse of the calvarium can be verified quickly and reliably by palpation through the skin and blubber" and "[d]isappearance of the corneal reflex in general reflex is an accepted method to determine deep unconsciousness. However, the process for checking the corneal reflex is not simple, and can be very difficult to perform by a sealer on the ice. The nature of some head injuries may lead to the eyes being fixed and staring, despite the seal being conscious and perceiving pain."); Richardson (2007), p. 15; Butterworth (2012), p. 6; Daoust (2012), p. 452.

In this connection, the complainants emphasize the possibility of re-shooting of any conscious seal. (See e.g. Canada's second written submission, para. 67; Norway's opening statement at the first meeting of the Panel, para. 156). We simply note that this would require an assessment of the seal's consciousness from a distance on a moving boat followed and an accurate shot of a seal that, if conscious, may be a moving target that is capable of exhibiting directed and/or escape movements.

We also recognize that in Norway, "second stunning" is used in place of checking and serves as a precaution to ensure irreversible unconsciousness. (Norway's response to Panel question No. 62; VKM Scientific Opinion, pp. 27, 34). The parties dispute whether this is a faster and more reliable way of guaranteeing the success of stunning attempts. (See Third Statement by Mr Danielsson, (Exhibit NOR-163), paras. 11-12, 17-18; European Union's first written submission, para. 172). We note that although this may avoid the challenges of applying a blink test or skull palpation, its capacity to ensure irreversible unconsciousness will depend on delivery of an accurate strike of sufficient force and may require some interval of time before performance of the second step at close proximity.

³¹⁰ See EFSA Scientific Opinion, p. 71: "When stunned or killed by acute trauma to the brain, harp seals, like other animals, may undergo a period of tonic and clonic seizures. These consist of tonic contraction and lateral movements which can be very strong in some animals, and tended to be accompanied by contralateral movements of the cranial portion of the body."

responses of seals to physical trauma do pose a challenge to the assessment of consciousness, particularly since forms of both movement and immobility can be consistent with either effective or ineffective stunning.³¹³

7.210. In the event of a gunshot powerful enough to destroy a seal's head, it would be possible to verify death by visual inspection. 314 However, even following extensive brain damage seals may "on rare occasions display some coordinated activity, if those parts of the brainstem that are responsible for basal control of breathing and/or motor activity remain intact".³¹⁵

Bleeding

7.211. There does not seem to be any dispute that, as an isolated step, bleeding out can be performed effectively in seal hunts. However, the main concerns regarding bleeding out relate to it being performed expeditiously after successful completion of the preceding steps, i.e. on a dead or irreversibly unconscious seal.³¹⁶

Areas of concern in the application of humane killing methods in seal hunting

Delay

7.212. The European Union has argued that when seals are shot from a distance it may take extended periods of time for sealers to manoeuvre their vessels into place to retrieve the animals.³¹⁷ The complainants have argued *inter alia* that the issue of delay is often irrelevant as shooting seals serves as a combined "stun/kill" method for a majority of seals such that there is no subsequent suffering.318

successful stunning from the freezing reflex and directed/coordinated movements from unconscious movements such as the swimming reflex).

At the same time, there is evidence that reflex motions may not be easily distinguished from directed movements of the upper-half of the body. For example, Daoust (2012), while defining the swimming reflex as "lateral motions ... of the caudal portion of the body", goes on to note that "[c]lose observation of post mortem reflex movements ... showed that they were frequent and could also involve front flippers." (Daoust (2012), p. 452). The same study also recounts the poor animal welfare outcome of seven seals shot on the ice "[m]ost of [which] did not move at first, having presumably been stunned by the shot. By the time they showed some evidence of consciousness, mainly through head movements, the vessel was already too close to the ice floe, thus preventing the hunter from taking another shot for safety reasons." (Daoust (2012), p. 450; see also NOAH Report (2012), p. 4 and Annex R, p. 3 (of 250-300 seals hooked aboard the vessel, "[t]wo animals showed signs of life after being taken on board with an extended fishhook. It was not obvious that these animals were conscious prior to their being hooked. This shows that it is difficult to assess whether an animal is dead from the deck of the ship."))

³¹³ EFSA Scientific Opinion, p. 71 ("These 'swimming' reflex movements can last for a considerable period of time and such seals have been verified as dead by veterinarians. Nevertheless, there is a concern, in the absence of other indicators such as skull palpation, that some seals showing a swimming reflex may not be unconscious ... Some of these conscious responses may resemble swimming reflexes, and it is not always be [sic] easy to distinguish between conscious and unconscious reactions from a distance, and particularly, when it is not possible to examine the animal clinically ... There is a potential paradox between a successful stun being evidenced by seizures and also by relaxation."); Burdon (2001), pp. 1, 4; Daoust (2002), pp. 2, 6 ("The frequent occurrence of strong swimming actions in seals killed by trauma complicates the determination of their death from a distance" and "the pattern of this reflex activity can be erratic and does not necessarily decrease gradually in intensity from the time of death"; further, "[c]omplete immobility immediately following a blow to the head should actually alert the sealer to the possibility that the animal is still conscious"); IVWG Report (2005), p. 16; Daoust (2012), pp. 449, 452.

The physiological features of seals discussed in paras. 7.189-7.191 have been cited in explaining the prolonged reflex reactions of seals (including muscle contraction and cardiac activity). (See NAMMCO Report (2009), pp. 10-11; Daoust (2002), p. 6). ³¹⁴ See VKM Scientific Opinion, p. 30.

³¹⁵ NAMMCO Report (2009), p. 11 (further that a "brainless (i.e. decapitated and consequently pain free) animal might actually continue to display breathing activity for several minutes"). ³¹⁶ European Union's first written submission, paras. 120-121; EFSA Scientific Opinion, p. 49; IVWG

Report (2005), pp. 9-10; Butterworth (2007), p. 5; Daoust (2012), p. 449 and Table 2.

⁷ See European Union's first written submission, paras. 139-144.

³¹⁸ Norway's second written submission, para. 289; Canada's second written submission paras. 38 and 51. The complainants further contend that the proximity required for use of a hakapik means there will not be a delay if this instrument is used. (Norway's opening statement, paras. 154-170 (comments by Dr Knudsen); Canada's second written submission para. 44 (although Canada also acknowledges delay is

- 72 -

7.213. Several sources emphasize that for the three-step method to be an effective and humane way of killing seals, each step must be completed and pursued in immediate (or at least rapid) succession for any given seal.³¹⁹ We note that there is some divergence as to what degree of delay between steps can still be consistent with humane killing.³²⁰ While not being in a position to decide based on the evidence presented to us a discrete minimum lapse of time between completion of the steps of the three-step method for a kill to be regarded as humane, we observe that delay between steps in the killing method can lead to prolonged suffering in seals and enhance the magnitude of poor welfare outcomes.³²¹

7.214. Regarding actual delays in the killing process in seal hunts, evidence indicates that delays between steps in the killing process are an occurrence in seal hunts and that such delays can be attributable to pervasive characteristics of the hunts, including the physical conditions and the instruments used.³²² Evidence specifically confirms that the likelihood of delay is greater when a firearm is the stunning instrument used because of the distance between the sealer and the seal.³²³

³²⁰ For example, EFSA recommends that "[t]he time between shooting and monitoring of the state of the shot animal should be short" and that "[s]eals should be bled-out as soon as possible and, preferably immediately, after they have been successfully stunned and checked to ensure that they are irreversibly unconscious or dead". (EFSA Scientific Opinion, pp. 89-90; see also IVWG Report (2005), p. 2 ("The three steps in the humane killing process -- stunning, checking that the skull is crushed (to ensure irreversible loss of consciousness or death), and bleeding -- should be carried out in sequence as rapidly as possible".); Daoust (2002), p. 6 ("No interval between an animal being shot and losing consciousness will ever be acceptable to some people.")).

We note that some reports, including EFSA, have provided recommendations and conclusions within a limited mandate to provide an opinion on the *minimization* of pain and suffering in seal hunts, taking account of practical limitations, which may affect the allowance for delay. (See EFSA Scientific Opinion, p. 49 ("Considering the safety issues associated with the difficult working conditions often encountered during certain seal hunts (e. g. the small size of some of the ice floes on which seals may be stunned), and that animals may be shot from a distance, a regulation requiring the animal to be bled immediately after stunning may not always be practicable, depending on the hunt.")).

³²¹ See EFSA Scientific Opinion, pp. 75-82. We note that EFSA's risk assessment to identify hazards to seal welfare relied upon "a qualitative evaluation of the nature of the adverse effect associated with the hazard in terms of intensity and *duration*." Ibid. p. 75 (emphasis added) Consequently, the duration of an adverse effect was one of the key determinants (along with intensity and probability) of EFSA's evaluation of the risks to animal welfare in seal hunting.

³²² There are certain empirical measurements of delays in seal hunts. Daoust (2002) examined IFAW video recordings of six seals and found that "the average interval between the seal being shot and being struck with a hakapik or, in one instance, being shot and hooked (in order to be brought on deck) and then bled and skinned without being struck, was 45.2 s (range 12-111 s)." (Daoust (2002), p. 4). Daoust (2012) measured the interval between steps one and two for 254 seals with a mean duration of 63.5 seconds. This increased to 91.95 seconds for seals shot in water (20 seals) and to 76.5 seconds for seals retrieved with a gaff (90 seals). (Daoust (2012), Table 2).

³²³ See e.g. Daoust (2012), p. 447; NAMMCO Report (2009), pp. 18, 21, Appendix N, pp. 6-7, and Appendix G, p. 5; EFSA Scientific Opinion, p. 44 (of the use of a rifle, the "distance between hunter and seal implies a necessary delay in verifying the results of the shot" absent clear demonstration that the seal is only wounded); IVWG Report (2005), p. 9 ("The Group understands that at the Front, where seals are shot at distances of approximately 40-50 meters, there is often a delay in sealers being able to check for effective stunning."). There are also indications that the time required to reach a seal and complete the stun with a hakapik could be reduced by shooting again from the vessel (re-stunning). However, as we have found above, this again poses the risks of inaccurate shooting combined with the difficulties of assessing a seal's condition from a distance as well as the potentially erratic movements of the struck and still conscious seal. (See Daoust (2002), p. 6). The complainants have argued in this regard that what constitutes an "acceptable lapse of time" between steps in the killing process depends on the killing method used. (See Canada's and Norway's responses to Panel question No. 60).

[&]quot;endemic to all wild animal hunts" and that "generally accepted animal welfare standards recognize the legitimacy of a time lag"); see also Canada's and Norway's responses to Panel question No. 60).

However, given the trend of firearms being the predominant stunning instrument and the attendant risks of inaccurate shooting, we consider that the risks of delay persist for any seal that is not rendered unconscious by an effective "stun/kill" application.

³¹⁹ See, e.g. Burdon (2001), p. 1 ("We recommend that a process of rapid stunning (resulting in a rapid loss of consciousness), followed immediately by a bilateral corneal reflex check to assess loss of consciousness, followed immediately by bleeding out to ensure death occurs, are followed in order to reduce these levels of suffering."); Richardson (2007), p. 52; Butterworth (2012), p. 4; Norway's Fisheries Directorate, Proposal to amend the rules on seal hunting (2010), (Exhibit EU-45), pp. 3-4.

- 73 -

Struck and lost

7.215. Seals that are wounded and escape beneath the surface of the water are known as "struck and lost". Seals that are struck and lost can die shortly after escaping or survive with injuries that can profoundly affect their continued survival in the wild. There is evidence (including video recordings) showing that instances of "struck and lost" do take place as a part of seal hunting and that shooting seals in open water can contribute to its occurrence.³²⁴ Moreover, as concluded by EFSA, "struck and lost rates will also vary with the skill of the hunter and other variables, such as weather conditions."³²⁵

7.216. We note that the available information from different sealing countries, while confirming struck and lost seals, suggests that the actual rates may not be the same. The Canadian government currently estimates a 5% struck and lost rate in the commercial seal hunt, which in many years may be tens of thousands of seals.³²⁶ Empirical data from Norway are generally scarce³²⁷, and there are varying indications from sealing inspectors of the extent to which struck and lost is a problem in the Norwegian hunt.³²⁸ In Greenland, authorities have explained that the hunting of harp seals occurs "exclusively" from small boats with rifles and that the "shooting of seals at substantial distances is the cause of most hunting losses".³²⁹ However, because the hunt is conducted year-round, the loss rate varies according to the fat content of seals during different seasons and the salinity of different hunt localities.³³⁰ The government of Greenland has reported results from a questionnaire survey showing that 34 per cent of hunters report struck and lost as an ordinary catch when reporting their annual harvest.³³¹

Hooking/gaffing conscious seals

7.217. Physical conditions and concerns for sealers' safety may demand that the seal be hooked onto the boat if it cannot be checked on the ice.³³² This may have potentially severe negative

We note that shooting seals in water is prohibited under Norwegian regulations. (See Norwegian Ministry of Fisheries and Coastal Affairs, Regulation Relating to the Conduct of the Seal Hunt in the West Ice and East Ice, (Exhibit NOR-15), section 6).

³²⁵ EFSA Scientific Opinion, p. 58.

³²⁶ EFSA Scientific Opinion, p. 58; Butterworth (2012), p. 4. In its oral presentation at the first substantive meeting, the European Union noted that more than 136,000 seals would have been "struck and lost" at sea in the past decade alone. In a more isolated example, one study recorded that of 278 seals shot all were retrieved, i.e. none were struck and lost. Although a portion (8.6%) of these seals were shot in the water, the authors of the study record that all seals shot in the water floated and thus were able to be retrieved. (Daoust (2012), p 449).

With respect to Nunavut, "the largest Inuit territory where approximately 50 per cent of all Canadian Inuit live", COWI commented that the figure for seal catch levels "excludes hunting loss", which would require an "adjustment of at least 10%-20%" to ring seal catch figures. (COWI 2010 Report, p. 27, footnote 22).

³²⁷ We note that "[t]here are no official statistics on numbers or percentages of seals struck and lost, either alive or dead during Norwegian hunts". (VKM Scientific Opinion, p. 32).

³²⁸ See, e.g. Second Statement by Mr Danielsson, (Exhibit NOR-128), Annex (records kept by a sealing inspector showing extremely low levels of struck and lost rates for 5,647 seals hunted in two years); NOAH Report (2012), p. 27 and Appendix K, (2009 comments of a sealing inspector):

Of the last 20 animals shot ... 9 took wounding shots and were lost – almost a 50% rate of wounding shots over more than 2 hours. Many of these animals were then shot in the sea and 2 took 6 shots to the body on the ice before taking the last 2 shots in the sea ... A total of 58 animals took wounding shots and were lost in the sea, and around 10 were "lost" because of breaking ice. Then there were animals that took wounding shots on the ice and took more than 2 non-fatal shots to the body ... but there were more than 200 of them. ³²⁹ Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 19 (Box 4). See also

para. 7.268 regarding the Greenlandic use of rifles from boats in "open water hunting" as a hunting method.

³³⁰ Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 19 (Box 4).

³³¹ Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 28. See also COWI 2008 Report, p. 46; EFSA Scientific Opinion, p. 66 (providing the results of a questionnaire survey of seal hunters in Greenland according to which part-time hunters reported a mean struck and lost rate of 26% whereas full-time hunters reported a mean rate of 21%).

³³² See IVWG Report (2005), p. 10 ("Some members of the Working Group feel that bleeding should be a requirement of the Marine Mammal Regulations (2003), making it an offence not to bleed a seal before

³²⁴ See EFSA Scientific Opinion, pp. 25, 58, and 89; NOAH Report (2012), p. 27 and Appendix K; IVWG Report (2005), p. 2; Daoust (2012), p. 451; Daoust (2002), p. 4; B. Sjare and G.B. Stenson, "Estimating Struck and Loss Rates for Harp Seals (Pagophilus Groenlandicus) in the Northwest Atlantic", *Marine Mammal Science*, Vol. 18 (2002), (Exhibit CDA-115), pp. 710-720; A Review of Animal Welfare Implications of the Canadian Commercial Seal Hunt, cited footage, (Exhibit EU-38); Video presented by the European Union at the first meeting of the Panel, (Exhibit EU-79).

impacts on animal welfare if a seal is conscious and sensible to pain during this process. In recognition of this, EFSA has recommended that "[u]nless they are in the water, [seals] should not be moved, i.e. gaffed, hauled or moved from the position they have come to rest, until it has been confirmed that they are dead or irreversibly unconscious, or have been bled-out."³³³ Nonetheless, there are many reports and recorded instances of hooking/gaffing of potentially conscious seals in the evidence.³³⁴

7.218. The parties dispute the extent to which hooking/gaffing of conscious seals presents a problem in terms of animal welfare.³³⁵ The parties agree, however, that in actual hunts seals are recovered onto vessels with the use of hooks and gaffs³³⁶, which is consistent with the possibility of unstable ice conditions (so as to preclude performance of the three steps on ice) and the concomitant increase in the use of firearms. We find compelling evidence to show that the possibility of retrieving seals by hook/gaff is important to the feasibility of commercial seal hunting in Canada³³⁷ and Norway³³⁸. Furthermore, given the difficulties of assessing the consciousness of

³³³ EFSA Scientific Opinion, p. 90.

³³⁴ See e.g. Daoust (2012), pp. 449-450 (documenting instances of gaffing); Daoust (2002), p. 4; Burdon (2001), Appendices 3-5 (results of review of video footage showing many instances of hooking alive); Butterworth (2012), p. 4; A Review of Animal Welfare Implications of the Canadian Commercial Seal Hunt, cited footage, (Exhibit EU-38); Video presented by the European Union at the first meeting of the Panel, (Exhibit EU-79).

³³⁵ For example, in Norway, gaffing of unchecked seals is permitted by regulation only under specified circumstances that Norway argues serve to obviate welfare concerns, namely with seals that are judged to be dead. (VKM Scientific Opinion, p. 27; First Statement by Mr Danielsson, (Exhibit NOR-4), para. 28; see also footnote 338 below). Norway also contends that the European Union exaggerates the problem of hooking/gaffing conscious seals. (Norway's second written submission, para. 301).

³³⁶ See Canada's and Norway's responses to Panel question No. 153.

³³⁷ See Canada's response to Panel question No. 153, para. 183 ("Not all seals can be bled before being recovered, so a total prohibition on retrieval prior to bleeding is likely not feasible in practice."). See also IVWG Report (2005), p. 10.

³³⁸ Under Norwegian regulations, "[i]t is prohibited to use a hook to lift seals that have not been bled out on board the vessel", however "[seals less than one year old] that have been shot may be lifted on board using a hook if there is no doubt that they are dead and the ice conditions make it unadvisable to walk on the ice". (Norwegian Ministry of Fisheries and Coastal Affairs, Regulation Relating to the Conduct of the Seal Hunt in the West Ice and East Ice, (Exhibit NOR-15), Section 10).

In connection with this rule, we note that the Norwegian Fisheries Directorate issued a proposal to amend the regulations based on "several reported violations" relating to requirements "that the animals must be bled on the ice immediately after they have been shot, and provisions on the use of hooks". Specifically, the Fisheries Directorate identified the practice of hooking seals aboard as occurring due to a regulatory exception for "sporadic hunting" being "redefined in practice" by sealers in addition to "worsening ice conditions". While recognizing that "banning the use of hooks on seal pups could in certain respects make hunting difficult", the Fisheries Directorate proposed revoking the exception for their use. (Norway's Fisheries Directorate, Proposal to amend the rules on seal hunting (2010), (Exhibit EU-45), p. 5; see also Consultation of Norway's Fisheries Directorate on the proposed amendment, (Exhibit EU-44), p. 5 (assuming that, due to poor ice conditions, "the extent of hooking and bloodletting on board the vessel can be expected to increase"); NOAH Report (2012), Appendix Q, pp. 4-5 and Appendix R, p. 3 (inspectors' statements of hooking making the hunt more efficient and uncertainty whether it was being done in accordance with regulations)). The proposal was opposed by the Norwegian Fishermen's Association (which stated that the "changes will substantially reduce the efficiency of seal hunting") and the Fishing Vessel Owners Association (which considered that "[i]ce conditions will usually not permit hunting if hooking is prohibited" and "the proposed tightening of the regulations could affect the economic viability of hunting"). (Consultation of Norway's Fisheries Directorate on the proposed amendment, (Exhibit EU-44)). The Ministry of Fisheries and Coastal Affairs ultimately "concluded that amending the regulation was not required". (See Norway's response to Panel question No. 170, para. 326).

hooking or skinning. Other members of the Group feel that worker safety and the difficulties presented by the natural environment in which the hunt takes place were considerations that could make such a regulation difficult to apply, specifically in relation to hooking a seal."); Exhibits EU-44 and EU-45 (assessments of the Norwegian Fisheries Directorate of the safety of sealers and the difficulty of complying with narrow conditions to allow bleeding to take place on board).

As noted in the context of delay, we observe that some sources that have addressed the issue of hooking and gaffing before checking have given explicit consideration to accommodation of the practical demands of seal hunting. (See IVWG Report (2005), p. 7 ("The Group recognizes that part of contributing to improved animal welfare and reduced suffering is to produce recommendations that are realistic in the context of the hunt, so that sealers will accept and implement them. There needs to be a realistic balance between ideal procedure and methodology, and what is practical and achievable".)).

the seal and the challenges of re-stunning by firearm, there is a possibility that some seals will be conscious when hooked or gaffed leading to severe negative consequences for animal welfare.³³⁹

Monitoring and enforcement of the application of humane killing methods in seal hunting

7.219. Both Canada and Norway have adopted regulations pertaining to the method for killing seals and for monitoring and enforcing such regulations, and provide for certain monitoring resources and activities for the purpose of achieving compliance with seal hunting regulations.³⁴⁰

7.220. There is evidence indicating that monitoring and enforcement of humane killing standards in seal hunts can be beneficial from an animal welfare perspective³⁴¹ but is challenging due to a variety of factors. In particular, the scale of a seal hunt and the large territory over which it can be dispersed contribute to the difficulties of monitoring and enforcing requirements related to humane killing.³⁴² We take note of the fact that there may be different situations with respect to monitoring and enforcement in the seal hunts of different countries. For example, the Canadian hunt is carried out by a considerably greater number of vessels operating in different locations, and many of the regulatory resources are either land-based or confined to a limited number of DFO vessels and helicopter(s).³⁴³ The Norwegian hunt is typically conducted with a smaller number of vessels and requires the presence of a government inspector to be on board.³⁴⁴ By way of comparison, in the

³⁴⁰ See para. 7.237.below on "organization and control" of commercial hunts for greater detail on the regulations of Canada and Norway.

³⁴¹ See EFSA Scientific Opinion, p. 95 ("Independent monitoring of hunts (without commercial/industry and NGO links) to provide certain critical information on seal killing and stunning from a welfare perspective should be instigated."); Daoust (2002), p. 5 (qualifying the recording of a relatively high proportion of completely crushed skulls with the possibility that the presence of an observer "may have incited sealers to hit the seals' skulls more vigorously").

³⁴² IVWG Report (2005), p. 11 (of competitive factors and the scale of the Canadian commercial hunt "notes that these conditions can make it difficult for DFO to undertake effective monitoring and enforcement") and 12 ("The physical realities of the Canadian harp seal hunt present a significant set of challenges for observation, supervision, monitoring and enforcement" and, with regards to the Canadian Front, which was supposed to account for two-thirds of the hunt, "[b]ecause of its remoteness and difficult environmental conditions, it is generally considered not to be well observed or monitored.").

³⁴³ See Canada's first written submission, paras. 113-114; response to Panel question Nos. 89 and 90. Canada has clarified that in recent years its monitoring, control, and surveillance resources have included a Canadian Coast Guard ice breaker and two to three Coast Guard helicopters, and various authorities and officers aboard the ice breaker. Land-based fisheries officers are also available for either coastal or aerial patrols and/or accompanying sealing vessels (though the extent and regularity of these monitoring activities has not been made clear). We note that Canada has provided detailed inspection data from the 2011 and 2012 hunts providing observations from these various sources, and has claimed that the data show a rate of "compliance with the three-step method" that exceeds 95 per cent. (DFO, Compliance Statistics for Three-step Method, 2011 and 2012, (Exhibit CDA-96); Canada's second written submission, para. 90; response to Panel question No. 173). We first note, however, that 1,636 and 2,998 seals were observed in 2011 and 2012, respectively, which may only constitute a relatively small portion of seals actually hunted (for example, more than 40,000 harp seals were killed in the 2011 hunt). Even within this sample, although the data record the number of seals monitored "with 3-step process issues", it is not clear what kind of "anomalies indicating a possible lack of compliance" were considered to fall into this category. (See Canada's response to Panel question No. 173, para. 226). Thus, on the basis of what we have been provided, there appear to be limitations on the amount of hunting actually monitored as well as on the interpretive value of "compliance with the [Marine Mammal Regulations'] three-step method" for conclusive assessments of overall animal welfare standards. (See European Union's comments on Canada's response to Panel question No. 173 (highlighting inter alia the imbalance between the number of sealers and vessels compared to surveillance resources)).

³⁴⁴ See Norway's first written submission, paras. 40, 252-257; Norwegian Ministry of Fisheries and Coastal Affairs, Regulation relating to regulatory measures and the right to participate in hunting of seals in the West Ice and East Ice in 2012, (Exhibit NOR-13), Section 10; COWI 2008 Report, p. 67; VKM Scientific

³³⁹ Evidence shows that the practice of hooking unchecked and potentially conscious seals aboard vessels can be a consequence of the cumulative difficulties of stunning, checking, and the distinct environmental factors of seal hunts. See Daoust (2012), p. 453:

Because of the variable ice conditions at the hunt, seals killed with a rifle shot may be in the water or on icefloes too small to allow a sealer to stand on. In such cases, a gaff must be used to retrieve the animal, and this may raise welfare concerns since verification of the animal's death or state of irreversible unconsciousness has not yet been performed with step two of the three-step process and, therefore, conscious or partly conscious animals may be gaffed. Whereas shooting seals in the water can be avoided, it is not always evident to a gunner shooting a seal on the ice from a long distance whether the ice floe on which the seal rests is thick enough to support the weight of a sealer and thus allow him to retrieve the animal manually.

Greenlandic hunt, conducted by a combination of full-time and part-time hunters year-round and in many locations along the Greenlandic coast, there is some provision for monitoring by wildlife officers.³⁴⁵

7.221. Nonetheless, we consider there to be difficulties of monitoring and enforcement (as commented upon in the context of each hunt) that exist in seal hunting generally notwithstanding specific differences in the manner of hunting and monitoring. Apart from the scale and large territory of the hunt, additional factors include the constraints of surveillance resources/personnel as well as overall difficulties in sealers' application of regulatory requirements under the actual circumstances of seal hunting.³⁴⁶

Overall assessment

7.222. Based on the examination of all available evidence in the record, the Panel finds that the circumstances and conditions of seal hunts present certain specific challenges to the humane killing of seals. Such challenges result in a risk in any given seal hunt that the targeted animals may suffer poor animal welfare outcomes of varying intensity and duration.

Although Norway requires the presence of an inspector on board every sealing vessel, EFSA has commented that the number of vessels in Canada would make it "much more difficult to institute a programme in Canada similar to that in Norway". (EFSA Scientific Opinion, p. 74).

³⁴⁵ See Management and Utilization of Seals in Greenland, (Exhibit JE-26); COWI 2008 Report, pp. 49-52 (noting that wildlife officers are employed by the Fisheries and Licence control, though the dispersed and opportunistic characteristics of the hunt pose a challenge to control); Government of Greenland reply to the Commission of 29 January 2013, (Exhibit EU-154) (providing explanation as to the monitoring of legal requirements regarding seal hunting, particularly the control of hunting licences).

³⁴⁶ See IVWG Report (2005), pp. 11-12; Richardson (2007), p. 45; DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42) (referring to the difficulty of tracking unregistered vessels, "especially in such a widespread activity as the seal harvest"); Consultation of Norway's Fisheries Directorate on the proposed amendment, (Exhibit EU-44), p. 5 ("The Fisheries Directorate would like to point out once again that since the hunting season 2005 infringements have been recorded of the regulations governing the practice of sealing, and that this mainly applies to provisions that were introduced in response to the industry's request for catchenhancing measures. *Experience suggests, therefore, that all catches are conditional on hunters complying in good faith with the rules that are applicable at any given time, and that the regulations themselves cannot in principle prevent infringements occurring.")* (emphasis added); Norway's Fisheries Directorate, Proposal to amend the rules on seal hunting (2010), (Exhibit EU-45), p. 2 (desire "to propose rules that did not leave much scope for exercising discretion, since this would have placed high demands on both hunters and inspectors and would have had the potential to lead to unfounded disputes").

A report prepared by NOAH (a Norwegian animal rights NGO) compiles interviews with multiple past sealing inspectors describing the challenges of overseeing the seal hunt on account of practical difficulties and potential social pressures. (See NOAH Report (2012), p. 3). For example, a former sealer stated in a 2000 interview that "[i]t is practically impossible for one person to control everything that is going on during the hunt; then you would need one inspector for each sealer! All inspectors I have talked to agree that it is impossible to manage a proper control." A former sealing inspector stated in a 2010 interview that "[o]ne man can of course not see all that is happening. ... there is a lot that you do not see. You have one inspector and several hunting teams, often the hunting teams go far away from the vessel on the ice. ... The system is no quarantee that regulations are followed." (See also NOAH Report (2012), Appendix A, p. 1; Appendix B, p. 2; Appendix K, pp. 7, 12 (inspectors' accounts of the observational limitations faced by an inspector); pp. 5-6, 11-12 (inspectors' statements of compromised objectivity of inspectors as participants in the hunt and colleagues of hunters, as well as possible unwillingness of sealers to follow inspectors' guidance); Appendix K, p. 9 (sealing inspector account of tension on board over reporting witnessed infractions)). We also note that another Norwegian sealing inspector has opined that "the inspector can effectively monitor the hunt" when hunting takes place from the main boat, and can either join small boats when used or observe them with binoculars from the main boat. This same inspector added that "[i]n order to monitor effectively, it is not necessary to see every kill at close distance. An inspector can have a very good control on what is going on by keeping a more general overview." (Third Statement of Mr Danielsson, (Exhibit NOR-163), paras. 24-27).

With regard to monitoring and enforcement, we also take into consideration references to the challenges of conducting empirical research and the obstacles that seal hunts pose to gathering concrete data. EFSA, for example, points out that "there are logistical difficulties inherent in assessing objectively the processes involved when these hunts are conducted under very different, remote, uncontrolled and unverifiable conditions". (EFSA Scientific Opinion, p. 24; see also Daoust (2012), pp. 447-448 (indications that the opportunistic and sequential observations of researchers could only give detailed information on a portion of the total observed samples, which themselves were of limited size)).

Opinion, p. 41. See also European Union's response to Panel question No. 64, para. 205 ("Norway's commercial hunt is a smaller operation and, *a priori*, easier to monitor.").

- 77 -

7.223. Specifically, there are characteristics of the physical environment of seal hunting that affect the way seals are stunned and that can impact the degree of effectiveness of stunning attempts. We have also noted that attempts to strike or shoot a targeted seal more than once may not ameliorate the risks of ineffective stunning. Combined with the difficulties of assessing the consciousness of seals, seal hunting can present delays in carrying out the killing process and may pose specific animal welfare problems for seals that are struck and lost as well as for seals that are gaffed and hauled onto a sealing vessel while conscious.

7.224. The challenge of reconciling the requirements of humane killing with the practical risks and difficulties of seal hunting, together with the potentially large territory of the hunt, poses an obstacle to monitoring and enforcement of the application of humane killing methods. Our assessment of the evidence taken together indicates that these risks to seal welfare are present in seal hunts in general.

7.3.2.3.2.2 Characteristics of commercial seal hunting

7.225. The European Union refers to commercial seal hunts as hunts conducted "for commercial purposes, where seals are killed primarily or exclusively in order to make a profit out of the skins, oil and other products from the hunted seals".³⁴⁷ Apart from the motive to make profits, according to the European Union, this commercial purpose is reflected in other characteristics of the hunt, such as its size (usually large-scale involving tens or hundreds of thousands of seals); intensity (systematic, competitive, and over a short timeframe); and the end-use of the derived products.³⁴⁸ The European Union argues that these conditions characterizing commercial seal hunts therefore distinguish commercial seal hunts from IC and MRM hunts.³⁴⁹

7.226. Canada and Norway do not contest that the majority of seal hunting conducted in Canada and Norway are commercial hunts with the motive to make profits. Canada and Norway contend however that their commercial hunts are strictly regulated, conducted in a humane manner, and sustainable.³⁵⁰ They further contest that their hunts can be distinguishable from IC and MRM hunts based on the purpose of the hunt as asserted by the European Union: Canada and Norway emphasize the equal presence of a commercial component in all types of seal hunting. We address the parties' arguments on this question in the subsequent section.

Factual aspects of commercial seal hunting

Identity of the hunter

7.227. Most commercial sealers in Canada are fisherman for whom the seal hunt supplements the income from fisheries.³⁵¹ There is some conflicting evidence as to the economic significance of sealing to the Canadian coastal communities where the majority of the hunt occurs.³⁵²

³⁴⁷ European Union's response to Panel question No. 29, para. 100.

³⁴⁸ European Union's response to Panel question Nos. 8, 29, and 30.

³⁴⁹ European Union's response to Panel question No. 8, para. 19.

³⁵⁰ See e.g. Canada's first written submission, paras. 85, 91-121; opening statement at the first meeting of the Panel, paras 13-20; Norway's first written submission, paras. 47-55, 231-266.

³⁵¹ COWI 2008 Report, p. 24. See also COWI 2008, p. 22, (providing the history of Canadian commercial seal hunting dating back to the 18th century); COWI 2010 Report, Annex 2 (pp. 1-2).

³⁵² See DFO website, Sealing in Canada – Frequently Asked Questions, (Exhibit JE-28), p. 8 (stating based on estimates from the DFO and province of Newfoundland and Labrador find that "between 5,000 and 6,000 individuals derive some income from sealing", i.e. approximately 1% of the total provincial population and 2% of the labor force, which is "a substantial number of individuals in the context of small rural communities"); DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), pp. 3-4 ("The harvest provides important seasonal income and food to residents of small coastal communities where there have been fishery closures and employment opportunities are limited.") and 13 ("Canada's seal harvest is also an economic mainstay for numerous rural communities in Atlantic Canada, Quebec and the North. It supports many coastal families that can derive as much as 35% of their annual income from this practice."); DFO Empirical Base for Canada's Seal and Seal Products Industry, (Exhibit CDA-17), pp. 4 (noting that the volume and value of landings are not recorded against licence numbers, making it difficult to estimate individual incomes from seals) and 9 ("the seal hunt is an economic mainstay for numerous rural communities as a some coastal families who can derive as much as 35% of their annual income from this practice."); DFO Empirical Base for Canada's seal hunt is an economic mainstay for numerous rural communities in Atlantic Canada, Quebec and the North. Canada's seal hunt supports many coastal families who can derive as much as 35% of their annual income from this practice."); DFO website, Canadian Seal Harvest – Myths and Realities, (Exhibit CDA-38), p. 1.

- 78 -

7.228. The participants in the Norwegian commercial seal hunt mainly come from communities in northern Norway.³⁵³ Norway states that seal hunting contributes to the sustainability of the settlements and workplaces of coastal communities.³⁵⁴

Purpose of the hunt

7.229. Commercial gain is recognized as one of the main reasons for which seals are killed.³⁵⁵ As noted by EFSA, the rationale for a particular hunt can however include one or more purposes. For example, a commercial seal hunt may generate some products that are consumed or used within the sealers' community while products sold commercially provide income for the livelihood of sealers.³⁵⁶ While noting the possible overlap of purposes, EFSA maintains a distinction between "subsistence hunts" and "commercial hunts", for example when referring to data availability "as the vast majority of available data is from commercial hunts".³⁵⁷

7.230. Nevertheless, information before us confirms that a "commercial hunt" has commercial profit (rather than direct use or consumption of seal products) as its sole or primary objective.³⁵⁸

Scale of the hunt

7.231. The size of commercial seal hunts appears to be characterized by a large number of seals killed. There is, however, evidence of differences between commercial hunts in the exact amount of seals killed as well as potentially large fluctuations of seals killed over time for a given commercial hunt. For example, the Canadian harp seal hunt over the past decade has harvested a peak of roughly 365,000 seals, which has declined to just over 40,000 seals in 2011. The

See also IFAW, Economics of Canada's Commercial Seal Hunt (2011 update), (Exhibit CDA-5), p. 1 (comparing the "minor economic importance to Newfoundland and Labrador" of commercial sealing as compared to other fisheries).

³⁵³ COWI 2010 Report, p. 31.

³⁵⁴ Norway's first written submission, paras. 267-268. See also O. Vollan, *The Seal Hunt in the Nordic Countries* (Forlaget Nordvest, 1985), (Exhibit NOR-10), p. 35 (noting that Norwegian vessels began leaving for the Arctic on a regular basis starting in the 19th century and that "sealing became more important to provide income in both the northernmost and the southernmost parts of the country").

³⁵⁵ For example, EFSA distinguishes this from the other two principal reasons for killing seals, namely "subsistence and cultural purposes" or "because seals are perceived as pests or competitors with humans and their activities ... or as threats to other species of concern". (EFSA Scientific Opinion, p. 12; see also para. 7.225 above).

³⁵⁶ EFSA Scientific Opinion, pp. 12-13. See also Canada's response to Panel question No. 132 (recognizing a "spectrum" between a "pure commercial hunt" and a "pure subsistence harvest", and that certain hunts may fall closer to one end of the spectrum than the other depending on the proportion of seal products sold commercially); Norway's opening statement at the first meeting of the Panel, paras. 92-93 and 103-105; response to Panel Question No. 28, paras. 183-185; second written submission, paras. 217-224; response to Panel question No. 132 (arguing that seal hunting will typically have a range of purposes and that it is not possible to distinguish between "commercial" and "non-commercial" hunts); European Union's comments on Canada's and Norway's responses to Panel question No. 132.

³⁵⁷ See also COWI 2008 Report, pp. 22-24 and 61 (distinguishing a "commercial hunt" in both Canada and Norway).

³⁵⁸ We note that there are multiple exhibits, including government documents from sealing countries, which recognize commercial sealing as a distinct activity, notwithstanding the contribution of sealing to the income and livelihood of those engaged in the seal hunt. (See DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42) (generally referring to the "commercial" harvesting of seals as distinct from personal use and aboriginal subsistence hunting) and pp. 21-23 (explaining allocation and sharing arrangements of TAC quotas with a specific "commercial allocation" and quota allocations "set aside for special projects outside the developed commercial structure", as well as distinct rules in Canada applied to hunts by aboriginal peoples); DFO Empirical Base for Canada's Seal and Seal Products Industry, (Exhibit CDA-17), pp. 1 ("Seals are hunted commercially on the Atlantic coast of Canada, and a subsistence hunt is carried out by Aboriginal peoples in the Canadian Arctic.") and 3 ("Although [the subsistence hunt] is not a commercial hunt, cash is generated from the sale of sealskins in order to help finance the hunt, which has become more and more expensive due to higher capital and operating costs, as well as the need to travel greater distances to hunt."); DFO Overview of the Atlantic Seals Hunt 2006-2010, (Exhibit EU-40), pp. 7 and 17-20.

See also COWI 2008, pp. 22 (identifying one of the "main types" of seal hunting in Canada as being "organised commercial seal hunting" and identifying its focus as "the commercial large scale hunt"), 61 (distinguishing the "Norwegian commercial seal hunt" from the coastal hunt that is "much smaller than the commercial hunt" and a separate hunt "carried out by a limited number of hunters" in Spitsbergen and Jan Mayen), and 64 ("According to Norwegian law, the hunt is considered commercial if it contributes to the [income] of the hunter, separately or combined with other incomes."). Norwegian hunt involved between 15,000 and 20,000 seals prior to the adoption of the EU Seal Regime, but has since fluctuated between approximately 1,000 and 10,000 seals harvested.³⁵⁹

7.232. Canada has been reported to annually issue around 15,000 seal hunting licenses, though a significantly smaller portion may actually be used.³⁶⁰ In previous years, between 1,500 and 2,200 boats participated in the hunt annually³⁶¹, although this number has been reduced more recently.362

7.233. In Norway, the commercial seal hunt is typically carried out by a relatively small number of larger vessels (approximately 60 meters long).³⁶³ In recent years, usually two to four ships have participated in the annual hunt in the West Ice, each with a crew of 13 to 15 people.³⁶⁴

Seal hunting period

7.234. Under Canadian regulations, the season for the commercial hunt of harp and hooded seals in the Front is generally from November 15 to May 15. The specific timing of the hunt can depend on the movement and condition of ice floes, and the majority of the hunt occurs between late-March to mid-May³⁶⁵ beginning around the third week of March in the southern Gulf of St.Lawrence and around the second week of April in the Front.³⁶⁶

7.235. The Norwegian commercial seal hunt is divided between the "East Ice" and "West Ice" hunts with seasons from April 10 to June 30 and March 23 to May 15 respectively.³⁶⁷ Further, the period of the hunt is determined on the basis of breeding and moulting times of harp seals so as to ensure compliance with the ban on hunting un-weaned pups. 368

Section 3.2 ³⁶² DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), pp. 13-14 (stating that "the top from E40 active vessels the previous vear"); DFO Empiric number of active vessels in 2010 dropped to 106, from 540 active vessels the previous year"); DFO Empirical Base for Canada's Seal and Seal Products Industry, (Exhibit CDA-17), p. 2 ("Participation varies from year to year, and depends upon ice conditions, price of pelts, etc.").

³⁶³ EFSA Scientific Opinion, p. 26; COWI 2008 Report, pp. 61-62 (less than ten vessels participating in the commercial hunt over a range of years); Norwegian Ministry of Fisheries and Coastal Affairs, English Summary of White Paper No. 27 (2003-2004) on Norway's Policy on Marine Mammals, (Exhibit NOR-9), p. 4. ³⁶⁴ Norway's first written submission, para. 51.

³⁶⁵ See EFSA Scientific Opinion, pp. 24-26; COWI 2008 Report, pp. 23-24.

Further, there are indications that the majority of Canadian commercial hunt can occur within a narrower timeframe (of even a few days) during the designated hunting period. DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), p. 23; DFO website, Sealing in Canada - Frequently Asked Questions, (Exhibit JE-28), p. 7 (indicating that the first day of the harvest is the most lucrative); DFO Overview of the Atlantic Seals Hunt 2006-2010, (Exhibit EU-40), Section 3.2 and JE-29, p. 23; Canada's first written submission, para. 55 ("In practice, the harvest season is fairly short, lasting only a few weeks in the spring, beginning on March 15 and ending some time in mid-April to mid-May, depending on the ice conditions."); Richardson (2007), p. 43 (that as many as 145,000 seals have been killed in less than two days); Butterworth (2012), p. 1 (that most seals are killed in "just a few days").

³⁶⁷ VKM Scientific Opinion, p. 9 (referring to the opening and closing dates of the seasons for 2007); EFSA Scientific Opinion, p. 26. ³⁶⁸ Statement of Mr Landmark, (Exhibit NOR-8), para. 33.

³⁵⁹ See Table 3.

³⁶⁰ EFSA Scientific Opinion, p. 25; COWI 2008 Report, p. 24 (stating that only between 5,000 and 6,000 licences are actually used); DFO Overview of the Atlantic Seals Hunt 2006-2010, (Exhibit EU-40), Section 3.2 ("In 2005 the number of participants (active licence holders) was 7,000, representing 50% of the commercial sealing licences issued."); DFO Commercial Seal Harvest Overview 2011, (Exhibit JE-27), p. 2 ("Approximately 225 active professional seal licence holders participated in the Atlantic Canada Seal harvest in 2011" and 117 participated in 2010); DFO website, Sealing in Canada - Frequently Asked Questions, (Exhibit JE-28), p. 6 (stating that in 2011 there were approximately 14,000 commercial licences issued to sealers in 2011, but only an estimated 5,000 to 7,000 of those were active); DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), p. 14 (stating that there are approximately 6,400 active commercial licences in Atlantic Canada and "In 2010, approximately 390 people participated in the Atlantic Canada Seal harvest. This number is down significantly from 2009 which reported 1,755 active participants.").

³⁶¹ EFSA Scientific Opinion, p. 25; DFO Overview of the Atlantic Seals Hunt 2006-2010, (Exhibit EU-40),

³⁶⁶ DFO website, Sealing in Canada – Frequently Asked Questions, (Exhibit JE-28), pp. 4-5; DFO Overview of the Atlantic Seals Hunt 2006-2010, (Exhibit EU-40), Section 3.2; DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), p. 23 (also stating that the "peak commercial harvest" in the Gulf is in early April).

- 80 -

Hunting methods

7.236. According to sealing regulations in each country, the permitted hunting tools include hakapiks in both Canada and Norway (as well as clubs in Canada) of specified dimensions as well as firearms of specified power and ammunition. The use of nets is not allowed in the Canadian and Norwegian commercial seal hunts.³⁶⁹

Organization and control of the hunt

7.237. Both Canada and Norway maintain a licensing system for seal hunting that determines conditions of participation in the hunt.³⁷⁰ In addition, both Canada and Norway establish annual total allowable catch (TAC) quotas, both of which administer these quotas through regional allocations for the various geographic areas of the commercial seal hunt.³⁷¹ Finally, there are regulations in Canada and Norway imposing requirements on the manner in which the seal hunt is conducted and dealing with the qualifications and training of sealers.³⁷²

³⁷² See Marine Mammal Regulations, (Exhibit CDA-21) and 2011-2012 Seal License Conditions for Newfoundland and Labrador, (Exhibit EU-39); Norwegian Ministry of Fisheries and Coastal Affairs, Regulation relating to regulatory measures and the right to participate in hunting of seals in the West Ice and East Ice in 2012, (Exhibit NOR-13) and Norwegian Ministry of Fisheries and Coastal Affairs, Regulation Relating to the Conduct of the Seal Hunt in the West Ice and East Ice, (Exhibit NOR-15).

The parties have provided a great deal of information and arguments pertaining to the regulations of the Canadian and Norwegian hunt, including the development of the current regulatory schemes. While the relevant issue before us is not the adequacy of the complainants' sealing regulations, we have considered this evidence to the extent it provides insight into the conduct of seal hunts, especially regarding the challenges of applying humane killing methods and the risks of poor animal welfare. (See Norway's opening statement at the first meeting of the Panel, para. 143 ("the issue is not whether Norway's sealing regulation or practices achieve an appropriate level of protection. The issue is whether it is *possible* for the EU, as part of an alternative measure, to legislate market access requirements that would ensure humane killing of seals."); Canada's second written submission, paras. 55-69 (defending various aspects of its sealing regulatory framework, such as that steps be carried out "as soon as possible" when rifles are used and that checking occur "immediately" after use of a hakapik or club).

Canada explains that the most recent amendments to its Marine Mammal Regulations came into force on 12 February 2009, reflecting in particular the recommendations of the IVWG Report (2005). These amendments consisted of: the prohibition of the use of the hakapik or club as the primary killing instrument for any seal over one year of age; licence conditions in certain regions imposing "a daily harvest limit based on the size of the vessel with the objective of slowing the pace of the harvest"; and, for 2014, mandatory training in the three-step process for all commercial sealers, which is presently mandatory for all personal use sealers. (See Canada's response to Panel question No. 54 and first written submission, paras. 91-98). Regarding the training of sealers, for many years the basic qualification for becoming a professional sealer in Canada was to serve two years as an apprentice to an existing licensed sealer. More recently a training program was developed through the cooperation of several agencies covering aspects of the Marine Mammal Regulations, approved weapons, and the three-step process, which, although currently voluntary, will be mandatory as of 2014. (Canada's response to Panel question No. 59 and first written submission, paras. 105-112).

³⁶⁹ See Marine Mammal Regulations, (Exhibit CDA-21) and 2011-2012 Seal License Conditions for Newfoundland and Labrador, (Exhibit EU-39); Norwegian Ministry of Fisheries and Coastal Affairs, Regulation relating to regulatory measures and the right to participate in hunting of seals in the West Ice and East Ice in 2012, (Exhibit NOR-13) and Norwegian Ministry of Fisheries and Coastal Affairs, Regulation Relating to the Conduct of the Seal Hunt in the West Ice and East Ice, (Exhibit NOR-15). See also EFSA Scientific Opinion, p. 26; COWI 2008 Report, pp. 28-29; DFO website, Sealing in Canada – Frequently Asked Questions, (Exhibit JE-28), p. 6. See paras. 7.192-7.196 above regarding seal hunting methods.

³⁷⁰ See COWI 2008 Report, p. 24; 2011-2012 Seal License Conditions for Newfoundland and Labrador, (Exhibit EU-39); Norwegian Ministry of Fisheries and Coastal Affairs, Regulation relating to regulatory measures and the right to participate in hunting of seals in the West Ice and East Ice in 2012, (Exhibit NOR-13); Norway's first written submission, para. 51; Canada's first written submission, paras. 105-112.

³⁷¹ EFSA Scientific Opinion, pp. 25 and 27; COWI 2008 Report, p. 27, 62-63, and 66-67; VKM Scientific Opinion, pp. 14-15; DFO website, Sealing in Canada – Frequently Asked Questions, (Exhibit JE-28), p. 5; DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), pp. 17, 21-26; DFO Empirical Base for Canada's Seal and Seal Products Industry, (Exhibit CDA-17), p. 1; Norwegian Ministry of Fisheries and Coastal Affairs, *English Summary of White Paper No. 27 (2003-2004) on Norway's Policy on Marine Mammals*, (Exhibit NOR-9); Norwegian Ministry of Fisheries and Coastal Affairs, Regulation relating to regulatory measures and the right to participate in hunting of seals in the West Ice and East Ice in 2012, (Exhibit NOR-13); Joint Norwegian/Russian Fisheries Commission, *Report of the Working Group on Seals to the 40th Session – Appendix 8*, (Exhibit NOR-22); Norway's first written submission, para. 52; Statement of Mr Landmark, (Exhibit NOR-8), para. 22.

7.238. The Canadian commercial seal hunt is carried out by small vessels (less than 35') and "longliners" (35'-65'), and larger vessels may only participate as collector vessels. Smaller vessels may have a crew of 2-5 sealers and generally land daily to offload their catches, while longliner vessels carry larger crews and tend to stay out for a few days at a time.³⁷³

7.239. The Norwegian commercial seal hunt is carried out by registered ocean-going vessels found suitable and equipped for seal hunting.³⁷⁴ The particular manner in which the Norwegian commercial seal hunt is conducted requires the use of large vessels that can operate with equipment and provisions for several weeks at a time.³⁷⁵

Use of products derived from the hunt

7.240. Evidence shows that commercial hunts are primarily directed toward the sale of seal products such as skins, blubber or oil, and meat. Historically, seal skins and furs have been the primary commodity sold commercially.³⁷⁶

7.241. Seal skins are used to make a wide variety of garments and accessory items, including jackets, hats, boots, slippers, mittens, purses, wallets, and novelty items.³⁷⁷ Seal oil and blubber is refined and processed to make Omega-3 products, the commercial trade of which has been noted to have surpassed seal skin products in recent years.³⁷⁸ The international trade in seal meat is a relatively smaller part of the commercial sale of seal products.³⁷⁹

Norway has described the regulatory history of instituting the three-step method; requirements regarding qualifications and training of those participating in the hunt; rules regarding the first stunning weapon; strengthened rules on the three-step process, for example through assignment of one person per marksman to second and third steps; a prohibition against shooting seals if the conditions are such that they cannot be struck with a hakapik or slagkrok afterwards and be bled out on the ice; and a penal provision. (Norway's response to Panel question No. 54 and first written submission, paras. 231-257; EFSA Scientific Opinion, pp. 27-28). As to training, all participants in Norway, including inspectors, must attend courses held by the Directorate of Fisheries. Hunters must pass a test on the use of a hakapik and marksmen must pass a government approved shooting proficiency test prior to every seal hunting season. (Norway's response to Panel question No. 59 and first written submission, paras. 248-251).

The European Union has taken issue with the Canadian amendments as being inadequate in several respects from the perspective of animal welfare, and cites the most recent amendments of Norway's sealing regulations in 2003 as "a major step backwards in terms of animal welfare", namely the exception allowing seals to be hooked on board before the final two steps are carried out. (European Union's second written submission, paras. 75-85 and first written submission, paras. 112-121, 171-175).

³⁷³ DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), pp. 3-4; DFO Overview of the Atlantic Seals Hunt 2006-2010, (Exhibit EU-40), Section 2. See also DFO Empirical Base for Canada's Seal and Seal Products Industry, (Exhibit CDA-17), p. 5.

³⁷⁴ See EFSA Scientific Opinion, p. 27; Norwegian Ministry of Fisheries and Coastal Affairs, Regulation relating to regulatory measures and the right to participate in hunting of seals in the West Ice and East Ice in 2012, (Exhibit NOR-13), section 3; Statement by Mr Landmark, (Exhibit NOR-8), para. 29; Norway's first written submission, para. 51.
 ³⁷⁵ See Norwegian Ministry of Fisheries and Coastal Affairs, *English Summary of White Paper No. 27*

³⁷⁵ See Norwegian Ministry of Fisheries and Coastal Affairs, *English Summary of White Paper No. 27* (2003-2004) on Norway's Policy on Marine Mammals (19 March 2004), (Exhibit NOR-9), p. 8.
 ³⁷⁶ See Canada's first written submission, paras. 61-70; Norway's first written submission,

³⁷⁰ See Canada's first written submission, paras. 61-70; Norway's first written submission, paras. 85-102.
 ³⁷⁷ See, e.g. DFO website, Sealing in Canada – Frequently Asked Questions, (Exhibit JE-28), p. 8; DFO

³⁷⁷ See, e.g. DFO website, Sealing in Canada – Frequently Asked Questions, (Exhibit JE-28), p. 8; DFO Empirical Base for Canada's Seal and Seal Products Industry, (Exhibit CDA-17), pp. 8-9. The European Union has characterized products derived from commercial hunts (such as clothing and accessories) as "inessential" items. (See, e.g. European Union's first written submission, para. 39).

³⁷⁸ See Canada's first written submission, paras. 79-80; DFO Commercial Seal Harvest Overview 2011, (Exhibit JE-27), table 12.

³⁷⁹ Canada's first written submission, paras. 61-70; Norway's first written submission, paras. 86-102; COWI 2010 Report, pp. 37-38; DFO Commercial Seal Harvest Overview 2011, (Exhibit JE-27) (providing a recent update and outlook on trade in seal skins, seal oil, and seal meat); DFO website, Sealing in Canada – Frequently Asked Questions, (Exhibit JE-28), p. 8 ("Seal pelts are transformed into a wide range of final products including coats, vests, hats, boots, mittens, trims, seal leather items, and novelty items. Seal oil is used in Omega 3 health products, in paints and for fuel in Northern/Inuit communities. Seal meat is sold in a variety of raw and prepared forms for both human and animal consumption"); DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), p. 13 ("Traditionally, seals have been harvested for meat and for pelts, both to use locally and to sell. Pelts have been historically the most commonly-sold commercial product although prices have been highly volatile over the years, resulting in large fluctuations in the economic value of the industry."); DFO Empirical Base for Canada's Seal and Seal Products Industry, (Exhibit CDA-17), pp. 7-8; Government of Newfoundland, *Commercial Utilization*, (Exhibit CDA-23). 7.242. With respect to the commercial purpose of the hunt and its relation to the seals killed, Canada states that the pelts of beaters (a weaned harp seal of 3 weeks to 3 months old that has moulted its white lanugo fur) are of higher quality and have greater value in the marketplace.³⁸⁰

Application of humane killing methods in commercial seal hunting

7.243. We recall our conclusion above that seal hunts in general pose various risks to the welfare of seals, including the risks of ineffective stunning, delays in the killing process, struck and lost seals, and the hooking of conscious seals.

7.244. In this section, we have examined evidence of the competitive nature of commercial hunts, including how the competitive pressures in the seal hunt may have changed over time. In particular, the allocation of quotas, in combination with the specific time window for hunts³⁸¹, has been noted to place pressures on sealers to the potential detriment of animal welfare.³⁸² In a similar vein, there is evidence that hunts with commercial profit as their sole or primary objective operate with the incentive to kill more seals in order to maximize profit.³⁸³ By contrast, commercial considerations have also been asserted to promote humane practices in seal hunting on the grounds that sealers may want to kill seals efficiently and thus preserve pelt quality by only targeting the seal's head.³⁸⁴ As noted, commercial hunts are also conducted in accordance with licensing schemes and sealing regulations which determine *inter alia* who may participate in the hunt and when it occurs.

³⁸² See EFSA Scientific Opinion, pp. 24-27 (describing the compressed sealing season and quota system in both Canada and Norway); DFO Overview of the Atlantic Seals Hunt 2006-2010, (Exhibit EU-40), pp. 17-18 (quota overrun by 10,000 seals in 2005 "largely due to the competitive nature of the hunt", also commenting on "a competitive race for seals in the Gulf of St. Lawrence in 2004 and 2005"); NOAH Report (2012), p. 10 and Appendix H, pp. 11, 15 (identifying certain "negligent infringements" of the regulations on hooking and gaffing seals "committed because hunters were excessively focused on capturing as many animals as possible, which must be considered in the company's interest."). See also NOAH Report (2012), pp. 10-13 and 21-22; Consultation of Norway's Fisheries Directorate on the proposed amendment, (Exhibit EU-44) (describing aspects of tension between commercial interest and animal welfare); Butterworth (2012), p. 8; DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), pp. 17, 22, and 25; IVWG Report (2005), p. 13; Butterworth (2007), pp. 4 and 13.

We note that Canada states that it has taken steps to limit the number of seals that individual vessels can take per day through seal licensing conditions. (Canada's opening statement at the first meeting of the Panel, para. 19). Canada notes, however, that such limits are not imposed in some regions "because the different nature of the hunt [in the Gulf] makes such limits unnecessary". (Canada's second written submission, para. 78). Canada submits that "the implementation of daily limits has slowed down the hunt considerably", and has clarified that limits are not applied elsewhere "because the number of sealing vessels is considerably smaller and thus there is less competition". (Canada's response to Panel question No. 171, para. 217). The European Union responds citing "public data made available to IFAW by DFO" to argue that "quota overruns" have occurred during years of high pelt prices in the region where daily limits are not currently applied by the seal licence conditions. (European Union's comments on Canada's response to Panel question No. 171). Norway has submitted comments denying a "race between sealers" and pointing out that catches have been below quota levels in recent years. (Norway's opening statement at the first meeting of the Panel, para. 188; Third Statement by Mr Danielsson, (Exhibit NOR-163), paras. 6-8).

³⁸³ See Butterworth (2007), pp. 12-13; Richardson (2007), pp. 43-44; Butterworth (2012), p. 8. See also Norway's comments on the European Union's response to Panel question No. 118 (commenting in the context of the Greenlandic hunt that "[p]rofessional seal hunters have an incentive to maximize their income by hunting more seals.")

³⁸⁴ Canada's second written submission, para. 59; Second Statement by Mr Danielsson, (Exhibit NOR-128), paras. 34-37; EFSA Scientific Opinion, p. 49; DFO Empirical Base for Canada's Seal and Seal Products Industry, (Exhibit CDA-17), pp. 12-13 (outlining Canada's pelt grading and classification system that indicates lower quality grading for pelts with holes). See also Canada's and Norway's responses to Panel question No. 70.

At the same time, it has been suggested that preserving the commercial value of the pelt can provide a disincentive to attempt to re-shoot seals that may still be conscious, thus causing delay in the killing process and prolonging suffering. European Union's opening statement at the first meeting of the Panel; Richardson (2007), p. 39.

³⁸⁰ Canada's response to Panel question No. 91.

³⁸¹ DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), p. 23; DFO website, Sealing in Canada – Frequently Asked Questions, (Exhibit JE-28), p. 7 (indicating that the first day of the harvest is the most lucrative); DFO Overview of the Atlantic Seals Hunt 2006-2010, (Exhibit EU-40), Section 3.2; Richardson (2007), p. 43 (that as many as 145,000 seals have been killed in less than two days and that in 2005, 78 per cent of the harp seals killed in the commercial seal hunt were slaughtered in just six days).

7.245. Based on the evidence presented before us, therefore, we find that to the extent that commercial motives lead to killing a greater number of seals in hunts conducted within a limited

period of time, this may additionally contribute to subjecting seals to the animal welfare risks

7.3.2.3.3 Whether the distinction between commercial hunts and IC hunts is legitimate

7.3.2.3.3.1 Main arguments of the parties

identified above with respect to seal hunts in general.

Complainant (Canada)³⁸⁵

7.246. Canada argues that the detrimental impact of the EU Seal Regime on Canadian seal products through the IC exception is not related exclusively to a "legitimate regulatory distinction".³⁸⁶ To determine whether the detrimental effect on Canadian imports stems from a legitimate regulatory distinction, Canada suggests that the Panel take into account the following factors: first, whether the regulatory distinction at issue is necessary in order to achieve the objectives of the measure; second, whether there is a rational connection between the regulatory distinction is arbitrary and therefore not legitimate; and, third, whether the evidence shows that the distinction is intended to discriminate against imports as this would undermine the "legitimacy" of the regulatory distinction.³⁸⁷

7.247. Canada argues that the regulatory distinction between seal products derived from Inuit hunts and those derived from non-Inuit hunts in no way contributes to the advancement of the EU Seal Regime's animal welfare objective.³⁸⁸ The cultural heritage or ethnicity of the hunters is not a legitimate regulatory distinction because it is unrelated to the central objective of the EU Seal Regime of responding to concerns about animal welfare.³⁸⁹ Canada points out that the European Union is not imposing any animal welfare requirements on Inuit from Greenland, and refers to evidence suggesting that a significant number of seals in Greenland are killed inhumanely, i.e. in a manner likely to lead to avoidable pain or suffering.³⁹⁰

7.248. Furthermore, Canada argues that, to the extent that seal products derived from hunts in Canada and in Greenland exhibit the same characteristics, they should be afforded the same treatment from a regulatory standpoint.³⁹¹ Canada contends that there are significant similarities between the historical and socio-economic contexts of seal hunting in Canada and Greenland. According to Canada, in both cases, the practice of sealing is deeply rooted in the culture and tradition of the communities where the hunt takes place; the by-products of seal hunts are not only marketed outside of the country or territory but are also consumed and used in the local economy; seal hunting provides much needed employment in areas where there are otherwise not many opportunities for employment; and seal hunting is a vital and essential source of income for the community. Canada posits further that the Greenlandic seal hunt has also a significant commercial aspect and is "very sophisticated, extensive, well-organized, well-marketed, and international in scope".³⁹² In this respect, Canada argues, the Greenlandic seal hunt is very similar to the Canadian east coast seal harvest.³⁹³

7.249. For Canada, given these similarities between the hunt in Canada and Greenland, the regulatory distinction under the EU Seal Regime is not even-handed, and therefore not "legitimate".³⁹⁴ Canada asserts that the regulatory distinction arbitrarily and unjustifiably

³⁸⁵ As mentioned in footnote 172 above, as relevant and as appropriate, we will refer to Norway's arguments in this section. (See, for example, Norway's second written submission, paras. 256-266).

³⁸⁶ Canada's first written submission, paras. 335-346.

³⁸⁷ Canada's first written submission, para. 387

³⁸⁸ Canada's first written submission, para. 402.

³⁸⁹ Canada's first written submission, para. 406.

³⁹⁰ Canada's first written submission, para. 402; response to Panel question No. 8; second written submission, para. 248. Canada refers to the COWI 2008 Report, where it is stated that 16 per cent of seals in Greenland are caught using nets, and that causing death by suffocation as a result of trapping seals under water is considered as "inherently inhumane". (COWI 2008 Report, p. 52).

³⁹¹ Canada's first written submission, para. 403.

³⁹² Canada's first written submission, para. 406.

³⁹³ Canada's first written submission, para. 406.

³⁹⁴ Canada's first written submission, para. 405.

discriminates against the vast majority of Canadian seal products³⁹⁵ and observes that except for the condition relating to the 'indigenous' status of the hunter, the Canadian seal hunt meets all of the conditions under the IC exception. Canada maintains that the distinction in the IC category is thus fundamentally one between permitted and prohibited seal products based on the "indigenous" status of the harvester.396

7.250. Finally, Canada observes that since the adoption of the EU Seal Regime in 2009, Greenland has now surpassed Canada to have the world's largest seal harvest.³⁹

Respondent (European Union)

7.251. The European Union submits that the "regulatory distinction" under the EU Seal Regime between IC hunts and commercial hunts is "legitimate" because (a) it is based on a legitimate objective, and (b) it is designed and applied in an even-handed manner.³⁹⁸

7.252. The European Union asserts that if the objective of the IC exception is found to be legitimate, then a fortiori, the regulatory distinction should also be considered "legitimate".³⁹⁹ On this basis, the European Union highlights the importance of seal hunting for the subsistence, cultural identity, and social cohesion of Inuit and indigenous communities.⁴⁰⁰ Furthermore, the European Union notes that the sale of seal skins, an important by-product of the hunts, serves to cover the hunting expenses incurred by Inuit and indigenous communities.⁴⁰¹

7.253. Further, the European Union submits that the IC distinction is neither "rationally disconnected" from nor does it "undermine" the objective pursued by the EU Seal Regime. 402 For the European Union, when assessing the moral implications of seal hunting it is both legitimate and appropriate to take into account the purpose of each type of hunt. The European Union contends that traditional hunts conducted for subsistence purposes do not raise the same moral concerns as commercial hunts conducted solely for the purpose of obtaining products, such as fur, to be used in manufacturing inessential goods.

7.254. The European Union argues that, in light of the "unique" situation in which Inuit and indigenous communities find themselves, it would have been "morally wrong" for the EU legislator to prohibit the placing on the market of seal products resulting from the hunts traditionally conducted by those communities.⁴⁰³ In essence, for the European Union, seal hunts conducted for the subsistence of Inuit and indigenous communities benefit from an "inherent legitimacy" that "overrides the general concerns over the killing methods for purely commercial motives".⁴⁰⁴ The European Union stresses that its regulatory approach on seal products, in particular regarding the IC exception, is in line with a consistent body of international law echoing the legitimacy of protecting the interests of Inuit and indigenous communities, and that the European Union is bound by these international legal instruments.405

7.255. According to the European Union, the IC exception is designed and applied in an even-handed manner⁴⁰⁶; it is "calibrated" and does not go beyond what it is necessary to achieve its purpose.⁴⁰⁷ Moreover, the European Union maintains that the IC exception is not discriminatory

³⁹⁵ Canada's first written submission, para, 405.

³⁹⁶ Canada's first written submission, para. 405; second written submission, para. 247.

³⁹⁷ Canada's first written submission, para. 406.

³⁹⁸ European Union's first written submission, para. 259; second written submission, paras. 219-234.

³⁹⁹ European Union's first written submission, para. 261; second written submission, paras. 220-227.

⁴⁰⁰ European Union's first written submission, paras. 263-266.

⁴⁰¹ European Union's first written submission, para. 266.

⁴⁰² European Union's opening statement at the first substantive meeting of the Panel, paras. 12-17.

⁴⁰³ European Union's first written submission, para. 268.

⁴⁰⁴ European Union's first written submission, para. 268; second written submission, para. 221.

⁴⁰⁵ European Union's first written submission, paras. 270-272; second written submission,

paras. 223-224 and footnote 245 (citing UN Department of Economic and Social Affairs, "State of the World's Indigenous People", ST/ESA/328 (2009), p. 10, available at

http://www.un.org/esa/socdev/unpfii/documents/SOWIP_web.pdf)). ⁴⁰⁶ European Union's first written submission, para. 301; second written submission, paras. 228-233.

⁴⁰⁷ European Union's second written submission (citing Appellate Body Report, US – Tuna II (Mexico), para. 297).

because it is equally available with regard to all hunts conducted by indigenous communities, including the Canadian Inuit.408

7.3.2.3.3.2 Analysis by the Panel

7.256. In this section, we address the question of whether the distinction drawn by the EU Seal Regime between commercial hunts and IC hunts, and consequently between products derived from each category of hunts, is legitimate within the meaning of Article 2.1 of the TBT Agreement.⁴⁰⁹

7.257. We recall the Appellate Body's explanation that the "legitimacy" of the regulatory distinctions drawn by a measure must be analysed in light of the objective of the measure and based on inter alia the particular circumstances of the dispute, including the measure's design, architecture, structure, operation, and application of the measure.⁴¹⁰ The Appellate Body further explained that where a regulatory distinction is not designed and applied in an even-handed manner — because, for example, it is designed or applied in a manner that constitutes a means of arbitrary or unjustifiable discrimination — that distinction cannot be considered "legitimate".⁴¹¹

7.258. Given the close relationship between the TBT Agreement and the GATT 1994⁴¹², including the similarities in their texts⁴¹³, we find it useful, in examining the measure's consistency with the TBT Agreement, to recall the Appellate Body's guidance in previous disputes concerning the obligations under the chapeau of Article XX of the GATT 1994. According to the Appellate Body, analysing whether discrimination is "arbitrary or unjustifiable" under the chapeau would entail an analysis that relates primarily to the "cause" or the "rationale" of the discrimination "put forward [by a regulating Member] to explain its existence".⁴¹⁴

7.259. The guidance provided by the Appellate Body regarding an analysis of the requirements under Article 2.1 of the TBT Agreement and the chapeau of Article XX of the GATT 1994 therefore suggests that the legitimacy of the regulatory distinction between commercial hunts and IC hunts should be determined by examining the following questions: first, is the distinction rationally connected to the objective of the EU Seal Regime; second, if not, is there any cause or rationale that can justify the distinction (i.e. "explain the existence of the distinction") despite the absence of the connection to the objective of the Regime⁴¹⁵, taking into account the particular

⁴⁰⁹ See paras. 7.130-7.131 above.

⁴¹² See Appellate Body Report, US – Clove Cigarettes, paras. 91-101. The Appellate Body observed that "the two agreements [the TBT Agreement and the GATT 1994] overlap in scope and have similar objectives." ⁴¹³ The chapeau of Article XX of the GATT 1994 provides:

Subject to the requirement that such measures are not applied in a manner which would constitute a means of arbitrary or unjustifiable discrimination between countries where the same conditions prevail, or a disguised restriction on international trade ... (emphasis added)

The fifth recital of the preamble of the TBT Agreement provides:

Recognizing that no country should be prevented from taking measures necessary to ensure the quality of its exports, or for the protection of human, animal or plant life or health, of the environment, or for the prevention of deceptive practices, at the levels it considers appropriate, subject to the requirement that they are not applied in a manner which would constitute a means of arbitrary or unjustifiable discrimination between countries where the same conditions prevail or a disguised restriction on international trade, and are otherwise in accordance with the provisions of this Agreement. (emphasis added)

⁴¹⁴ Appellate Body Report, Brazil – Retreaded Tyres, paras. 225-226 (citing Appellate Body Reports, US – Gasoline, pp. 25-26 and 28-29; US – Shrimp, paras. 166 and 172; US – Shrimp (Article 21.5 – Malaysia), paras. 144 and 147).

The Appellate Body in Brazil - Retreaded Tyres observed that the Appellate Body's analysis of the measures under the chapeau of Article XX of the GATT 1994 in previous disputes focused on whether discrimination that might result from the application of those measures had a legitimate cause or rationale in the light of the objectives listed in the paragraphs of Article XX of the GATT 1994.

⁴¹⁵ See Appellate Body Report, *Brazil – Retreaded Tyres*, paras. 226-234.

The Appellate Body stated, "we have difficulty understanding how discrimination might be viewed as complying with the chapeau of Article XX when the alleged rationale for discriminating does not relate to the pursuit of or would go against the objective that was provisionally found to justify a measure under a paragraph of Article XX." (emphasis added)

⁴⁰⁸ See, e.g. European Union's second written submission, para. 207.

⁴¹⁰ Appellate Body Report, US – COOL, para. 271 (referring to Appellate Body Report, US – Clove Cigarettes, para. 182).

⁴¹¹ Appellate Body Report, US – COOL, para. 271 (referring to Appellate Body Report, US – Clove Cigarettes, para. 182).

- 86 -

circumstances of the current dispute; and, third, is the distinction concerned, as reflected in the measure, "designed or applied in a manner that constitutes arbitrary or unjustifiable discrimination" such that it lacks "even-handedness". 416

7.260. We examine these questions in turn.

Whether the IC distinction is connected to the objective of the EU Seal Regime

Characteristics of IC hunts

Identity of the hunter

7.261. The EU Seal Regime defines "Inuit" as "indigenous members of the Inuit homeland, namely those arctic and subarctic areas where, presently or traditionally, Inuit have aboriginal rights and interests, recognised by Inuit as being members of their people and includes Inupiat, Yupik (Alaska), Inuit, Inuvialuit (Canada), Kalaallit (Greenland⁴¹⁷) and Yupik (Russia)"⁴¹⁸; and "other indigenous communities" as "communities in independent countries who are regarded as indigenous on account of their descent from the populations which inhabited the country, or a geographical region to which the country belongs, at the time of conquest or colonisation or the establishment of present State boundaries and who, irrespective of their legal status, retain some or all of their own social, economic, cultural and political institutions".⁴¹⁹

7.262. The complainants do not contest the definition of "Inuit" or "other indigenous communities" as stipulated in the EU Seal Regime.

Purpose of the hunt

7.263. Seal hunting by Inuit or other indigenous communities appears to largely serve two purposes: first, for their own use and consumption as part of their culture and tradition⁴²⁰; and

In our view, the Appellate Body's reasoning in US - Clove Cigarettes also supports our approach here. In that dispute, regarding the legitimacy of the regulatory distinction drawn by the measure in question (i.e. distinction between clove and menthol cigarettes), the Appellate Body examined the following questions: first, whether the distinction was connected to the objective of the measure that justified the prohibition of clove cigarettes; and, second; in the negative, whether the United States provided any reasons independent of the objective of the measure that could justify the distinction ("the reasons presented by the United States for the exemption of menthol cigarettes from the ban"). (Appellate Body Report, US - Clove Cigarettes, para. 225). ⁴¹⁶ See Appellate Body Report, US - COOL, para. 340.

In this connection, we take note of Canada's point that the conditions for the IC and MRM exceptions are not the distinctions that must be assessed under the legitimate regulatory distinction test but can be evidence indicating whether the distinction between conforming and non-conforming products is administered in an even-handed manner. (Canada's response to Panel question No. 28, paras. 127-129; second written submission, para. 246).

⁴¹⁷ Ninety per cent of the total population (56,600) in Greenland is Inuit. (COWI 2010 Report, p. 28; Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 11).

⁴¹⁸ Basic Regulation, Article 2(4).

⁴¹⁹ Implementing Regulation, Article 2(1).

According to the COWI 2010 Report, Inuit or indigenous communities with a tradition of seal hunting that are likely to meet the definitions of such status under the EU Seal Regime are located in Alaska, Canada, Greenland, Norway, Russia, and Sweden. (COWI 2010 Report, pp. v, 23-33; see also Figure 3-1 in the COWI 2010 Report for an overview of the geographical spread of Inuit and other indigenous communities. Ibid. p. 23).

Further, the COWI 2010 Report explains that a hunter's status as belonging to an Inuit or indigenous community is based on self-determination. (See Annex 3, p. 1 (referring to United Nations Declaration on the Rights of Indigenous Peoples, Resolution of the General Assembly 61/295, September 2007, (UN Declaration), Article 3; ILO Convention 169, Indigenous and Tribal Peoples Convention, 1989, (ILO Convention), Art. 1; and Charter of the Inuit Circumpolar Council (Article 6 of which defines Inuit as "indigenous members of the Inuit homeland recognized by Inuit as being members of their people and shall include the Inupiat, Yupik (Alaska), Inuit, Inuvialuit (Canada), Kalaalit (Greenland) and Yupik (Russia)").

⁴²⁰ See COWI 2010 Report, pp. 24, 26, 29, and 32. For example, in describing Inuit and indigenous communities in Northern and North-Eastern Canada, the Report notes that "seal meat is a traditional staple component in the Inuit diet and Inuit continue to hunt predominantly ring seals for their meat and skins ... Sharing of seal meat fosters relationships throughout the community, while the hunt is a means for learning across generations, not just about the hunt itself, but about the environment in which Inuit live and survive. Clothing made of sealskin is still regularly worn both for practical and cultural reasons. Finally, selling some of

second, albeit to varying degrees depending on the Inuit or indigenous community concerned, for the exchange of by-products of seals such as seal skins either through barter for other goods, or sale on the market to generate income.⁴²¹ In some cases, and particularly in the case of Greenland, seal products such as skins obtained from Inuit hunters may also be traded in international markets.⁴²²

7.264. We address below the parties' specific arguments concerning the purpose of IC hunts in the context of our analysis of the European Union's justification for the distinction between commercial hunts and IC hunts.

Scale of the hunt

7.265. Inuit or members of other indigenous communities hunt seals mostly on an individual basis using small boats or using sledge dogs and catching a few seals at a time.⁴²³

7.266. Although relatively little information on the number of seals hunted by Inuit or indigenous communities other than Greenland has been submitted to us, available data suggest potentially wide variation in the scale of different Inuit hunts. For example, the annual average catch of harp, ringed, and hooded seals in Greenland is reported to be approximately 164,000 seals.⁴²⁴ COWI provides various data showing a "harvest of as many as 1,600 animals" by Alaskan Aleuts and indigenous populations and 35,000 (predominantly ring) seals annually hunted in Nunavut in Canada.⁴²⁵

Seal hunting period

7.267. In contrast to commercial hunts, which were noted to occur during limited time periods within established seasons, IC hunts are typically conducted throughout the year.⁴²⁶

See also Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 11 ("The hunting of seals is a vital component of everyday life and culture in Greenland. It provides a significant amount of nutritious food and income to families living in remote coastal communities."); Nunavut Report (2012), (Exhibit JE-30), pp. 1 and 8.

⁴²¹ See Nunavut Report (2012), (Exhibit JE-30), p. 2; Management and Utilization of Seals in Greenland, (Exhibit JE-26), pp. 25-28; COWI 2008 Report, pp. 26 (describing both the sale of seal skins by Inuit hunters in Canada to generate income as well as economic contribution from "production of meat and skins for garments and arts and crafts") and 45 (distinguishing the "formal economy" of Greenlandic sealing that refers to economic transactions and the "informal economy" covering "the use of the catch for own consumption, barter or [unreported] sales").

⁴²² See Management and Utilization of Seals in Greenland, (Exhibit JE-26), pp. 25-28; Nunavut Report (2012), (Exhibit JE-30), p. 2; COWI 2010 Report, p. 29; COWI 2008 Report, pp. 26 and 45-46.
 ⁴²³ COWI 2010 Report, p. 27; Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 11.

⁴²³ COWI 2010 Report, p. 27; Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 11.
 ⁴²⁴ See Management and Utilization of Seals in Greenland, (Exhibit JE-26), pp. 1-2 (indicating

approximate annual catch levels of 82,000 harp seals; 78,000 ringed seals; and 4,600 hooded seals). Full time hunters constitute almost 7% of the work force (approximately 32,000) in Greenland. In Canada, by comparison, 38,018 harp seals were taken in 2011, and less than 100 hooded seals have been taken in recent years.

The COWI 2010 Report describes seal hunts in Greenland as "large scale" stating that "the nature and scale of the hunt vary considerably across the sealing countries. From large scale commercial hunt in Canada, Greenland, Namibia, Russia to small scale hunting in Sweden and Finland with a few hundred seals killed on an annual basis ... All countries with the exception of Greenland have seal management plans in place and/or quotas for a yearly total allowance catch (TAC)." (COWI 2010 Report, pp. iv-v).

⁴²⁵ COWI 2010 Report, pp. 23 and 27.

⁴²⁶ See COWI 2010 Report, pp. 27-28; Nunavut Report (2012), (Exhibit JE-30), p. 1; Management and Utilization of Seals in Greenland, (Exhibit JE-26), pp. 18-20.

the sealskin to markets provides additional income for a population group that has an average income far below the Canadian average." With respect to an Inuit community in Alaska (Aleut), it describes that "the current hunt in Alaska by Aleut takes place purely for subsistence, with most products consumed locally, or shipped to Aleut communities outside Alaska. It acts as a supplement to the Aleut diet and is still seen as contributing to social and cultural traditions." (Ibid. pp. 26-27). Concerning Inuit or indigenous communities in Russia, the Report states that "the majority of seals that are hunted by Inuit or indigenous communities are not industrialised, but consist of small-scale hunts serving as input to the daily life of these communities ..." (Ibid. p. 32).

- 88 -

Hunting methods

7.268. Inuit communities use both traditional methods/tools (e.g. harpoons, kayaks, dog sleds) as well as more modern equipment (e.g. rifles, boats, snowmobiles).⁴²⁷ Evidence also shows that methods such as "trapping and netting" are used in IC hunts.⁴²⁸ The COWI 2010 Report explains in this regard that the notion of "hunts traditionally conducted" referred to in Article 3.1 of the Basic Regulation can be defined as "hunts that are part of the cultural tradition of a given community located in a specific geographical area"⁴²⁹, and does not indicate "hunts conducted traditionally, i.e. in a traditional manner". Based on information concerning Greenland, the use of rifles from boats in "open water hunting" or trapping and netting appear to be the main hunting methods for Greenlandic Inuit.430

Organization and control of the hunt

7.269. IC hunts take place mostly on an individual basis and Inuit hunters are not usually subject to monitoring or enforcement of sealing regulations in their hunts.⁴³¹

7.270. However, we observe that Greenland requires a full time hunter to have a licence to qualify for selling the skins to the tannery Great Greenland A/S.⁴³² According to a document published by the Government of Greenland, a large number of hunters use "the possibility to sell skins to the tannery in total a couple of months a vear".433

Use of products derived from the hunt

7.271. Inuit and other indigenous communities use all parts of the hunted seals. They consume seal meat as an essential part of their diet and use seal skins and other parts of seals for a variety of purposes as part of their culture and tradition.⁴³⁴ As noted above, Inuit and other indigenous communities also sell by-products of the hunted seals, mostly seal skins, to markets.⁴³⁵

> Connection between the IC distinction and the objective of the EU Seal Regime

7.272. Based on our examination of the evidence pertaining to IC hunts described above, we have determined that there are certain characteristics that are unique to IC hunts⁴³⁶, namely: they are

Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 9.

⁴²⁷ See Parties' responses to Panel guestion No. 67.

⁴²⁸ Netting and underwater trapping as a killing method are intended to "restrain the seal in a submerged position long enough for it to exhaust its oxygen supply and to die from asphyxiation" (EFSA Scientific Opinion, p. 46). Nets are currently employed in Artic regions where seasonal and environmental factors make other hunting methods unviable. Whereas humane killing methods, and the three-step method in particular, are designed to minimize the suffering experienced by the targeted animal, netting has raised severe animal welfare concerns for subjecting seals to prolonged durations and intense magnitudes of suffering. (See EFSA Scientific Opinion, pp. 46-48; NAMMCO Report (2009), p. 11).

⁴³⁰ Management and Utilization of Seals in Greenland, (Exhibit JE-26), pp. 18-20; see also Nunavut Report (2012), (Exhibit JE-30), p. 1.

⁴³¹ See EFSA Scientific Opinion, p. 13; COWI 2008 Report, pp. 25 and 44.

⁴³² Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 21; see also COWI 2008

Report, p. 45. ⁴³³ Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 21 (stating that selling sealskins is for many hunters a very important secondary income, and approximately 100 hunters make more than EUR 10,000 yearly on sealskins).

⁴³⁴ COWI 2010 Report, pp. 24, 26; Management and Utilization of Seals in Greenland, (Exhibit JE-26), pp. 2 and 11; Parties' responses to Panel question No. 66. Canada submits that in east coast sealing communities, seal meat is consumed and very popular. Norway submits that meat from seal hunted in Norway (non-Inuit) is sold to restaurants, at local markets, or directly to consumers who come to the ships to buy.

In Greenland, seal meat is also used as food for sledge dogs, which power the sledges from which icefishing takes place. (Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 11).

⁵⁵ In Greenland, for instance, ringed seal was the most important species in relation to food supply and income, but the demand from the fur industry made it more attractive to hunt harp seals as Greenlandic hunters in some years offered a slightly better price for sealskins from harp seals compared to those from ringed seals. (Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 22)

⁴³⁶ The European Union highlights the uniqueness of IC hunts based on the following elements: (a) identity of the hunter: hunts conducted by Inuit or other indigenous communities characterized by its high dependence on seal hunting and a tradition of seal hunting in the geographical region; (b) end-use of the

conducted by Inuit and indigenous communities with a tradition of seal hunting dating back thousands of years⁴³⁷; they are normally carried out on an individual basis using small boats; and they take place throughout the year. In addition, by-products of the hunted seals are usually used and consumed by the community and, depending on the Inuit or indigenous community concerned, also sold on the market to generate income.⁴³⁸

7.273. We recall our assessment above that the circumstances and conditions of seal hunts present certain challenges to effecting humane killing of seals and that there is a risk in any given seal hunt that the targeted animals may suffer poor animal welfare outcomes of varying intensity and duration.⁴³⁹ IC hunts are no different; they are conducted in a similar physical environment often using similar hunting methods as described above. Thus, similar challenges to effecting humane killing of seals exist in IC hunts. Further, evidence shows that hunting methods used by Inuit or indigenous communities such as "trapping and netting" are not consistent with humane killing methods.⁴⁴⁰

7.274. As discussed in detail in section 7.3.3.1 below, the objective of the EU Seal Regime is to address the moral concerns of the EU public with regard to the welfare of seals. Specifically, the EU public moral concerns as described by the European Union are two-fold. They include: (a) the incidence of inhumane killing of seals; and (b) EU citizens' individual and collective participation as consumers in, and their exposure to, the economic activity which sustains the market for seal products derived from inhumane hunts. As part of our analysis in section 7.3.3.1, we also found that the EU public concerns on seal welfare relate to seal hunting in general and are not confined to any particular type of hunts.

7.275. Given that the same animal welfare concerns as those arising from seal hunting in general also exist in IC hunts, and considering the evidence showing the use by Inuit hunters of methods such as "trapping and netting", we find that IC hunts can cause the very pain and suffering for seals that the EU public is concerned about. Accordingly, the IC distinction does not bear a rational relationship to the objective of addressing the moral concerns of the EU public on seal welfare.⁴⁴¹

7.276. Canada submits that this rational disconnection between, on the one hand, the regulatory distinction between IC hunts and other hunts, and, on the other hand, the objective of protecting the welfare of seals or the public morals to which they relate, indicates that the distinction in question is not justifiable, and hence discriminatory contrary to Article 2.1 of the TBT Agreement.⁴⁴² The European Union does not contest that seal products potentially qualifying

⁴³⁹ See paras. 7.222-7.224 above.

⁴⁴⁰ Some of the Greenlandic harvesting practices have in fact been recognized as creating poor animal welfare outcomes. (EFSA Scientific Opinion, p. 47). The "trapping and netting" method which is used in Greenland precisely because of the environmental factors that prevail during the winter months (limited sunlight), is "problematic" from an animal welfare perspective. EFSA has concluded that netting of seals is "inhumane". (EFSA Scientific Opinion, p. 89; see also Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 19). Norway claims that hunts that the European Union labels "non-commercial" pose severe animal welfare problems. (See, e.g. Norway's first written submission, paras. 680-684; response to Panel question No. 73, para. 408).

⁴⁴¹ We note our consideration in section 7.3.3.1 that the aim, purpose, and target of the EU Seal Regime could not be considered as protecting the economic and cultural interests of Inuit or indigenous communities.
 ⁴⁴² Canada's second written submission, para. 251. Canada argues that although the Inuit hunt may not

⁴⁴² Canada's second written submission, para. 251. Canada argues that although the Inuit hunt may not be as widespread, organized, systematic, intensive, or competitive as commercial hunts, the European Union has adduced no evidence to show that the existence of these factors produces a different animal welfare

by-products of the hunt: partial or entire use, consumption or processing of the by-products of the hunt within the communities according to their traditions; and (c) subsistence purpose of the hunt: contribution of the hunt to the subsistence of the community. (European Union's Response to Question 66; see also COWI 2010 Report, p. 27 (explaining that the majority of seal products are consumed locally by Canadian Inuit and where only one third of sealskins end up on the market) and pp. 29-30 (explaining local consumption by Inuit in Greenland); Canada's response to Panel question No. 74, para. 320; Norway's response to Panel question No. 41, para. 217 (confirming that half of the skins are traded in and exported from Greenland, while the other half are consumed locally)).

⁴³⁷ The European Union explains that a "tradition of seal hunting" does not relate to the methods of hunting but rather means that the community in question must have a tradition of seal hunting in the geographical region. (European Union's opening statement at the first substantive meeting of the Panel, para. 14 (also referring to Canada's response to Panel question No. 67, para. 292 ("the hunt itself is traditional and a fundamental element of the Inuit culture and society")).

⁴³⁸ See para. 7.263.

under the IC exception do not conform to the objective of protecting animal welfare and can in fact compromise it.⁴⁴³ The European Union explains, however, that the application of certain hunting methods such as "trapping and netting" is indispensable for the subsistence of the Inuit, who otherwise would not be able to hunt during almost half of the year, and this therefore overrides the animal welfare concerns.⁴⁴⁴

7.277. We turn to examine the rationale submitted by the European Union as a justification for the distinction between commercial and IC hunts.

Whether the cause or rationale put forward by the European Union for the distinction between commercial and IC hunts is justifiable

7.278. We understand the European Union's justification of the distinction between commercial and IC hunts to rest on two premises. First, if the objective of the IC exception is found to be legitimate, then, *a fortiori*, the regulatory distinction should also be considered "legitimate".⁴⁴⁵ Second, highlighting the alleged uniqueness of IC hunts, the European Union argues that IC hunts, which are conducted for the "subsistence" of Inuit and indigenous communities, benefit from an "inherent legitimacy" that "overrides the general concerns over the killing methods for purely commercial motives".⁴⁴⁶ According to the European Union, therefore, the *purpose* of the hunt distinguishes IC hunts from commercial hunts and justifies any risk of suffering inflicted upon seals as a result of the hunts conducted by those communities.⁴⁴⁷ The European Union explains further that, because the subsistence of the Inuit and other indigenous communities and the preservation of their cultural identity provide benefits to humans, from a moral point of view, this outweighs the risk of suffering inflicted upon seals as a result of the hunts conducted by those communities.

7.279. First, we are not persuaded by the European Union's premise that a distinction in treatment is justified on the basis of the legitimacy of the objective of the distinction itself, in this case the IC exception. We do not read the Appellate Body guidance on Article 2.1 to support this interpretation. Under Article 2.1 of the TBT Agreement, the inquiry, according to the Appellate Body, is whether the detrimental impact caused by a measure stems from a legitimate regulatory distinction drawn in the measure. If it does, then the detrimental impact is justified and will not offend the non-discrimination obligation under Article 2.1. The analysis of determining the legitimacy of a regulatory distinction is not, as the European Union suggests, simply whether there is a legitimate objective will not automatically imbue the discrimination under Article 2.1 with legitimacy; were that to be the case, one would simply need to assess whether the detrimental impact stems from a "legitimate" objective. Even if the objective of the IC exception were separately examined and found to be a "legitimate" policy objective within the meaning of Article 2.2, that alone would not necessarily lead to establishing the legitimacy of drawing the *distinction* – as opposed to the legitimacy of a certain policy objective *per se* – between

⁴⁴³ European Union's response to Panel question No. 10, para. 44.

⁴⁴⁴ European Union's response to Panel question No. 8, paras. 22-23; second written submission, para. 232 (citing Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 19 ("[F]rom October to the end of March, netting is the prevailing method since it is impossible to use any other technique during the dark winter months")).

⁴⁴⁵ European Union's first written submission, para. 261; second written submission, paras. 220-227.

In using the term "legitimate" in this context, the European Union did not clarify whether the objective of the IC exception is "legitimate" within the meaning of Article 2.2 of the TBT Agreement. Further, the European Union took the position in the context of Article 2.2 that the protection of the interests of Inuit or indigenous communities is *not* an independent objective of the EU Seal Regime as a whole. (See, below section 7.3.3.1).

⁴⁴⁶ European Union's first written submission, para. 268; second written submission, para. 221.

⁴⁴⁷ European Union's second written submission, paras. 230, 232. See also European Union's response to Panel question No. 10.

outcome. (Canada's second written submission, para. 249 (referring to European Union's response to Panel question No. 8, para. 20)).

See also Norway's first written submission, paras. 698-703. According to Norway, the same animal welfare conditions prevail in all countries where seals are hunted because all seals are equally vulnerable to hunting that does not respect animal welfare. Norway also makes reference to the EFSA Scientific Opinion, which states that traditional or "subsistence" hunts have "few, if any, regulations and are poorly monitored". Norway infers from this conclusion that "some traditional methods used in 'subsistence' hunts may be detrimental to animal welfare". (See Norway's first written submission, para. 680 and footnote 980 (citing EFSA Scientific Opinion, p.13)).

commercial and IC hunts through the IC exception within the meaning of Article 2.1. The objective of the IC exception is an element that may be examined as part of the "cause" or "rationale" put forward by the European Union to seek to justify the IC distinction. But it is not determinative of the issue of the legitimacy of the regulatory distinction.

7.280. Next, based on the alleged uniqueness of IC hunts, in particular the "subsistence" purpose of IC hunts, the European Union argues that IC hunts are justifiably distinguishable from commercial hunts conducted primarily or exclusively for commercial purposes.⁴⁴⁸ According to the European Union, the regulatory distinction made by the EU Seal Regime between conforming and non-conforming seal products is primarily based on the "purpose" of the hunt from which a given product was derived; the term "purpose" is used to refer to the principal aim of the hunt in question, i.e. the primary reason why the seal in question is killed. The purpose of the hunt is also reflected in other characteristics of the hunt, such as its size, intensity or end-use of the products, which together constitute a "type".449

7.281. Canada does not contest the unique characteristics of IC hunts. In fact, Canada acknowledges that, regardless of hunting methods, the Inuit hunt itself is traditional and a fundamental element of Inuit culture and society.⁴⁵⁰ For example, Canada states that the purpose of the Canadian Inuit hunt today is not materially different from the hunt 1,000 years ago, although the emergence of a monetized society and new technologies has caused the Canadian Inuit to commercialize some output to generate income.⁴⁵¹ Canada however disagrees with the European Union on whether the purpose of IC hunts ("subsistence" purpose) and the purpose of commercial hunts ("primarily or exclusively commercial reasons") can strictly be distinguishable as asserted by the European Union. Canada argues that the "subsistence" purpose of IC hunts can equally be used to describe the Canadian east coast seal hunt (commercial hunts).452

7.282. To assess the issue of whether the alleged difference in the purpose of the hunt constitutes a justifiable rationale or cause for the distinction in question, despite its disconnection from the objective of the measure, we must examine two questions: first, whether, and, if so, how, the purpose of IC hunts differs from the purpose of commercial hunts; and, second, whether any distinction found in the purpose of the hunt justifies the distinction drawn under the measure between commercial and IC hunts.

Whether the purpose of IC hunts differs from the purpose of commercial hunts

7.283. The term "subsistence"⁴⁵³ is not defined in the EU Seal Regime. A dictionary definition of the term provides inter alia the following: "the action or condition of subsisting or of supporting life, the provision of food etc", "means of supporting life; livelihood", "a bare or minimal level of existence; an income providing this", or "food supply, provisions".⁴⁵⁴ Dictionary definitions thus

⁴⁴⁸ European Union's response to Panel question No. 30; see also footnote 60 (referring to debates within European Parliament in submitting amendments to the Commission Proposal. See Report on the Proposal for a Regulation of the European Parliament and of the Council concerning Trade in Seals Products (5 March 2009), (Parliament Report), (Exhibit JE-4), p. 64).

For the term "commercial" in this context, the European Union explains that it means the following: "seals are killed with a view to making profit out of the sale on commercial markets of products such as skins or oil". Further, according to the European Union, commercial seal hunts are "organised hunting, on a wide scale with reference to the hunting area and/or the number of animals killed, by people paid to do this in order to supply seal product processing enterprises on a regular and continuous basis for commercial purposes".

¹⁴⁹ European Union's response to Panel question No. 29.

⁴⁵⁰ Canada's response to Panel question No. 67.

⁴⁵¹ Canada's response to Panel question No. 67.

⁴⁵² Canada's comments on the European Union's response to Panel question No. 121. Canada submits that seal hunting has a long tradition in the Atlantic region, dating back to the 15th century, and plays a role in the social and economic well-being of these communities.

 $^{^{453}}$ According to the preamble of the Basic Regulation, seal products deriving from hunts traditionally conducted by Inuit communities and which contribute to their subsistence should not be covered by the prohibitions provided for by this Regulation. COWI explains that this recital indicates that the intention of the regulation is to protect the given communities by avoiding negative impacts on the community, hence taking a broad interpretation of the term subsistence. (COWI 2010 Report, p. 9).

See also Basic Regulation, Article 3.1(c). ⁴⁵⁴ Shorter Oxford English Dictionary, 6th edn, A Stevenson (ed.) (Oxford University Press, 2007), Vol. 2, p. 3087. ("7. The action or condition of subsisting or of supporting life, the provision of food etc. ... 8. (A)

suggest that subsistence is closely linked to the notion of providing food or income to support life or livelihood.455

7.284. We observe that EFSA also correlates the subsistence purpose of the hunt to the *identity* of the hunter: "the term 'subsistence hunt' is often used to describe a hunt where the seal is killed by an aboriginal for personal consumption".⁴⁵⁶ Indeed, the European Union acknowledges that the distinction for hunts conducted for "subsistence purposes" relates specifically to hunts "where seals are killed primarily in order to contribute to the subsistence of Inuit and other indigenous communities".457

7.285. Information submitted to the Panel confirms that certain Inuit or indigenous communities also sell by-products, mostly seal skins, of their hunts on the market. The extent of such commercial transactions seems to vary, however, depending on the particular Inuit or indigenous community concerned (e.g. bartering with other goods, placing meat or skins on the local market, or selling skins for international markets).⁴⁵⁸ For example, based on the information in the COWI 2010 Report, the current hunt in Alaska by Aleut takes place purely for subsistence, with most products consumed locally, or shipped to Aleut communities outside Alaska. The same is true for Inuit in Russia; "the majority of seals that are hunted by Inuit or indigenous communities are not industrialised, but consist of small-scale hunts serving as input to the daily life of these communities ...".⁴⁵⁹ As regards Canadian Inuit, most of the seal products are consumed locally by Canadian Inuit themselves and only one third of sealskins end up on the market.⁴⁶⁰ In Greenland, where 90 per cent of the population are Inuit, half of the skins are consumed locally, and the other half are traded in and exported from Greenland.⁴⁶¹

7.286. The European Union agrees that gualifying IC hunts may "have a commercial dimension".462 According to the European Union, if this were not the case, the IC exception would have served no purpose under the EU Seal Regime. The European Union explains that part of the cultural heritage of seal hunting involves bartering the outputs for necessary goods. It also maintains that, now that bartering is rarely practised, that cultural heritage is continued through placing the products on the market and then using the proceeds to buy necessary goods and finance the cost of conducting seal hunting.

7.287. To us, the commercial aspect of IC hunts resembles the purpose of commercial hunts, which is to earn income (and make profits) by selling by-products of the hunted seals. Further, in our view, this commercial aspect of IC hunts is related more to their need to adjust to modern society rather than to continuing their cultural heritage of bartering. The European Union has not explained their position that the commercial aspect of IC hunts is merely a modern version of bartering.

7.288. Nevertheless, based on the definition of the term "subsistence" as well as the evidence concerning Inuit and indigenous communities with a tradition of seal hunting, we consider that the subsistence purpose of IC hunts encompasses not only direct use and consumption of by-products of the hunted seals as part of their culture and tradition, but also a commercial component, to the extent that Inuit or indigenous communities also exchange some by-products of the hunted seals

⁴⁵⁷ European Union's response to Panel question No. 29, para. 100. (emphasis added)

⁴⁵⁸ COWI 2010 Report, pp. 27, 29-30; Canada's response to Panel question No.74, para. 320; Norway's response to Panel question No. 41, para. 217. 459 COWI 2010 Report, p. 32.

⁴⁶⁰ Evidence indicates that "less than 2% of aboriginal people in Canada are involved in commercial trapping of animals for fur." (Andrew Linzey, Public Morality and the Canadian Hunt (2005), (Exhibit EU-35), p.13)

⁴⁶¹ See Management and Utilization of Seals in Greenland, (Exhibit JE-26), pp. 11 and 25.

⁴⁶² European Union's response to Panel question No. 32, para. 112; closing statement at the second substantive meeting of the Panel, p. 4.

means of supporting life; (a) livelihood. Now spec., a bare or minimal level of existence; an income providing this. B Food supply, provisions.").

⁴⁵⁵ We also note that "subsistence" can be defined as "using wildlife locally for food, clothing, and shelter, and for making tools, rather than putting wildlife products into trade." (Andrew Linzey, Public Morality and the Canadian Hunt (2005), (Exhibit EU-35), p. 35 (referencing inter alia D. Lavigne, V. Scheffer, and S. Kellert, "The evolution of North American attitudes toward marine mammals"' in J.R. Twiss Jr. and R. R. Reeves (eds), Conservation and Management of Marine Mammals (Washington and London: Smithsonian Institution Press, 1999), p. 37).

⁴⁵⁶ EFSA Scientific Opinion, p. 13.

- 93 -

for economic gain. As observed by EFSA, a particular hunt may have one or several purposes.⁴⁶³ Unlike commercial hunts, however, most Inuit and indigenous communities do not appear to hunt seals for the sole or primary purpose of selling them on the market. Rather, seal hunting is a manifestation of a way of living for Inuit and indigenous communities and is an activity that defines them as Inuit.⁴⁶⁴ The commercial aspect of IC hunts is thus not the same in its extent as that associated with commercial hunts.

7.289. In conclusion, while IC hunts may also have a commercial aspect, we are persuaded that the subsistence aspect of IC hunts, combined with the identity of the hunter as Inuit, has significance for their culture and tradition as well as for their livelihood.⁴⁶⁵ To that extent, the primary purpose of IC hunts is distinguishable from that of commercial hunts.⁴⁶⁶

Whether the difference in purpose between commercial and IC hunts justifies the distinction drawn under the measure between these two hunts

7.290. Having determined that there is no rational connection between the objective of the EU Seal Regime as a whole and the distinction between commercial and IC hunts, in essence because the IC hunts pose at least the same risks to the animal welfare of seals as the commercial hunts, we then examined whether the distinction could nevertheless be justified. The first element of this analysis involved a consideration of whether the primary purpose of the IC hunts could be distinguished from the primary purpose of the commercial hunts. In this we determined that the purpose of the two hunts were distinguishable. The second element of the examination involves an analysis of whether this difference in purpose justifies the IC distinction.

7.291. In this regard, we recall the Appellate Body's statement that:

the task of interpreting [the chapeau of Article XX of the GATT 1994] is essentially the "delicate one of locating and marking out a line of equilibrium between the right of a Member to invoke an exception under Article XX and the right of other Members under varying substantive provisions ... of the GATT 1994. ... *This line of equilibrium is not fixed and unchanging and moves "as the kind and the shape of the measures at stake vary and as the facts making up specific cases differ.*"⁴⁶⁷ (emphasis added)

7.292. The European Union points out that the protection of the economic and social interests of Inuit or indigenous communities is recognized at the international level as illustrated, for example, in the United Nations Declaration on the Rights of Indigenous Peoples (UN Declaration)⁴⁶⁸ and in

⁴⁶⁸ The UN Declaration is a Resolution of the UN General Assembly (General Assembly Resolution A/RES/61/295 of 13 September 2007) affirming indigenous peoples' right to self-determination (Articles 3 and 4) and "to maintain and strengthen their distinct political, legal, economic, social and cultural institutions" (Article 5). States are called on to "provide effective mechanisms for prevention of, and redress for ... [a]ny action which has the aim or effect of dispossessing them of their ... resources" (Article 8(2)(b)).

In this vein, further recognition of various social and economic interests, including the preservation of cultural heritage and control over resources, is reiterated throughout the UN Declaration. See e.g. Article 20(1) ("Indigenous peoples have the right to maintain and develop their political, economic and social systems or institutions, to be secure in the enjoyment of their own means of subsistence and development, and to engage freely in all their traditional and other economic activities".); Article 26(2) ("Indigenous peoples have the right to own, use, develop and control the lands, territories and resources that they possess by reason of traditional ownership or other traditional occupation or use, as well as those which they have otherwise acquired".); Article 29(1) ("Indigenous peoples have the right to the conservation and protection of the environment and the productive capacity of their lands or territories and resources".); Article 32 ("Indigenous peoples have the right to determine and develop priorities and strategies for the development or use of their lands or territories and other resources".)

⁴⁶³ EFSA Scientific Opinion, pp. 12-13.

⁴⁶⁴ See European Union's responses to Panel questions Nos. 8, para. 20; and 30, para. 103.

⁴⁶⁵ See for instance paragraph 7.272 above.

⁴⁶⁶ European Union's second written submission, para. 231 (referring to Canada's response to Panel question No. 67, para. 292).

⁴⁶⁷ Appellate Body Report, *US – Shrimp*, para. 159. See also Appellate Body Report, *Brazil – Retreaded Tyres*, para. 224.

See also the Appellate Body's approach in *US – Clove Cigarettes* as noted above in footnote 415.

the ILO Convention concerning Indigenous and Tribal Peoples in Independent Countries (ILO Convention).469

7.293. Further, the recognition of the interests of Inuit or indigenous communities is also reflected in the legislative history of the EU Seal Regime as well as in Canadian sealing regulations. The legislative history of the EU Seal Regime and other measures with respect to seal hunting show that the interests of the Inuit have consistently been addressed and/or taken into account in the form of exceptions. For example, the 1983 Directive banning imports of skins of whitecoat pups of harp seals and of pups of hooded seals (blue-backs) was limited to "only apply to products not resulting from traditional hunting by the Inuit people". 470 In addition, the 2006 Declaration of the European Parliament prompting the legislative process of the EU Seal Regime contemplated that the final regulation "should not have an impact on traditional Inuit seal hunting".⁴⁷¹

7.294. Canada also exempts Inuit from certain provisions of its sealing regulations.⁴⁷² Further, Canada acknowledges the conflicting interests at issue between seal welfare and the interests of Inuit and other indigenous communities engaged in seal hunting. For example, Canada notes that "the onus is not on the complainants to offer solutions to enable Greenlandic sealers to improve animal welfare standards without putting at risk the subsistence of the Inuit and the preservation of their cultural identity."473 We also observe Inuit exceptions in similar measures adopted by other WTO Members on trade in products derived from marine mammals.⁴⁷⁴

7.295. In our view, these sources, taken in their entirety as factual evidence⁴⁷⁵, demonstrate the recognized interests of Inuit and indigenous peoples in preserving their traditions and cultures.

We note that the definition of "other indigenous communities" in the Implementing Regulation mirrors in identical language provisions from the ILO Convention on its scope of application. See Article 1(b) of the ILO Convention and Article 2(1) of the Implementing Regulation.

⁴⁷⁰ Council Directive No. 83/129 of 28 March 1983 concerning the importation into Member States of skins of certain seal pups and products derived therefrom, Official Journal of the European Communities, L Series, No. 91 (9 April 1983), p. 30, (Seal Pups Directive), (Exhibit CDA-12), Article 3. We note that, although the Directive was predicated on concerns about the population status of harp and hooded seals, preamble recitals highlighted that "the exploitation of seals ... is a natural and legitimate occupation and in certain areas of the world forms an important part of the traditional way of life and economy".

⁴⁷¹ Parliament Declaration, (Exhibit JE-19), para. 2.

⁴⁷² Canada argues that the European Union is incorrect when it states that Canada itself exempts the Inuit from the animal welfare requirements provided in Canada's hunting regulations. According to Canada, the coverage of the MMR is determined by a complex set of constitutionally determined parameters in Canada, with some Aboriginal hunters subject to them while others are not. Inuit hunters in Nunavut - where most Inuit seal hunting occurs - are also subject to their own regulatory regime, which is determined by a land claims agreement negotiated between the federal government and the Inuit. It is therefore also not correct to say that Canada "allows" the Inuit to hunt seals by netting them, as this would imply that federal regulations include a permissive provision in this regard.

Based on Canada's explanation on this issue and the European Union's comments on Canada's explanation, the Panel understands that certain Inuit in Canada are subject to parts of the sealing regulations, but not all of the regulations. We observe in this regard the European Union's statement that as the regulations apply only to the designated seal hunting areas in Canada and exclude the areas where Inuit sealers operate, Inuit are exempted from, for example, the "prohibition against the use of nets or other weapons otherwise deemed to be unacceptably cruel". (See Canada's response to Panel question No. 114 (referring to the European Union's response to Panel question No. 8, para. 24) and the European Union's comments on Canada's response to Panel question No. 114). ⁴⁷³ See Canada's second written submission, para. 251.

⁴⁷⁴ See, e.g. United States Marine Mammal Protection Act of 1972, (Exhibit JE-15), section 101(a) (laying down a "moratorium on the taking and importation of marine mammal products") and section 101(b) exempting "the taking of any marine mammal" by Alaskan natives for "subsistence purposes or for making "native articles of handicrafts and clothing"; Ley General de Vida Silvestre (amendment of 26 January 2006), (Exhibit EU-29), Artículo 55bis (prohibiting the importation, exportation and reexportation of any species of "mamífero marino y primate") and Capítulo II "Aprovechameiento para fines de subsistencia".

⁴⁷⁵ In taking into account the recognition given by international instruments in the context of the United Nations and the ILO to the interests of Inuit and indigenous communities, the Panel is mindful that these

⁴⁶⁹ The ILO Convention (Convention No. 169 of 27 June 1989) similarly exhorts governments to account for and protect the interests of indigenous peoples through inter alia "promoting the full realisation of the social, economic and cultural rights of these peoples with respect to their social and cultural identity (Article 2(2)(b)). The ILO Convention recognizes that "the integrity of the values, practices and institutions of [indigenous] peoples shall be respected" (Article 5(b)). Most relevantly, the ILO Convention states that the "rights of [indigenous peoples] to the natural resources pertaining to their lands shall be specially safeguarded. These rights include the right of these peoples to participate in the use, management and conservation of these resources"(Article 15(1)).

More specifically, in the case of seal hunts, the evidence before us shows that seal hunting represents a vital element of the tradition, culture, and livelihood of Inuit and indigenous communities.

7.296. Although we agree with Canada that a cause or rationale for a certain distinction may not be justifiable if such cause or rationale is not connected to the main objective of the measure, we are mindful that the justifiability of a specific cause or rationale provided for a given distinction must be examined on a case-by-case basis.⁴⁷⁶ In the circumstances of this dispute, the interests to be balanced against the objective of the measure at issue are grounded in the importance, recognized broadly in national and international instruments, of the need to preserve Inuit culture and tradition and to sustain their livelihood, particularly in relation to the significance of seal hunting in Inuit communities.

7.297. Further, the factual circumstances of this dispute can be differentiated from those of a previous dispute where a rationale or cause for a certain exception or regulatory distinction was not found justifiable. In US – Clove Cigarettes, the United States explained the distinction at issue (i.e. allowing menthol cigarettes while prohibiting clove cigarettes) was based on the alleged risks (namely health care costs and black market smuggling) arising from withdrawal symptoms that would afflict menthol smokers. The Appellate Body did not find this reason persuasive enough to justify the distinction between prohibited (clove cigarettes) and permitted (menthol cigarettes) products which were found to be "like" and presented the same health risks for smokers. We also note that in Brazil - Retreaded Tyres, certain imports of retreaded tyres were excluded from the scope of the ban on the grounds that the MERCOSUR arbitral tribunal made a ruling to that effect (i.e. to respect trade rules under the MERCOSUR). The Appellate Body found the exception arbitrary and unjustifiable due to the lack of any rational connection to the objective of the ban (i.e. environmental purposes), and the rationale for the exception, namely the MERCOSUR ruling was not considered by the Appellate Body sufficient to justify the exception in the face of the rational disconnection to environmental purposes.

7.298. Unlike the situations in US – Clove Cigarettes or Brazil – Retreaded Tyres, the cause or rationale for the exception granted under the EU Seal Regime to products derived from IC hunts is justifiable despite the rational disconnection to protecting seal welfare⁴⁷⁷, because it is founded on the unique interests of Inuit and indigenous communities, which are and have been recognized broadly, as discussed above. Additionally, as noted above, evidence shows that Inuit interests have always been raised as an important consideration when adopting a regulation relating to seal products, including the current measure.⁴⁷⁸ Under these circumstances, we are persuaded that the protection of Inuit interests justifies the distinction between commercial and IC hunts. We thus consider that the European Union has explained sufficiently the basis for distinguishing IC hunts from commercial hunts through the IC exception.

7.299. Before turning to our conclusion on the justifiability of the rationale given by the European Union for the distinction between IC and commercial hunts, we recall the European Union's reference to the alleged moral concerns of the EU public concerning the economic and social interests of Inuit and indigenous communities. According to the European Union, the "standard of the EU public's morality" requires examining in each case whether the suffering inflicted upon animals is outweighed by the benefits to humans (such as Inuit and other indigenous communities) or to other animals.⁴⁷⁹ Although we found based on available evidence that the EU public had moral concerns on seal welfare in general, we did not consider that the evidence before us supports the European Union's position that the EU public attributes a higher moral value to the protection of Inuit interests as compared to seal welfare.480

 $^{\rm 480}$ See para. 7.410 and footnote 676.

instruments are not WTO instruments and they do not set out WTO obligations per se. We are considering the content of these instruments as part of the evidence submitted by the European Union to support its position concerning the interests of Inuit and indigenous communities, not as legal obligations of Members.

⁴⁷⁶ See para. 7.172 above.

⁴⁷⁷ By definition, the term "exception" refers to "a particular case or individual that does not follow some general rule or to which a generalization is not applicable". (Shorter Oxford English Dictionary, 6th edn,

A. Stevenson (ed.) (Oxford University Press, 2007), Vol. 1, p. 885). ⁴⁷⁸ In contrast, the exceptions in *Brazil – Retreaded Tyres*, for instance, were introduced during the application of the ban. ⁴⁷⁹ European Union's response to Panel question No. 31.

Nor are we presented with evidence establishing the precise scope of the "basic morality" of EU citizens as claimed by the European Union. 481

7.300. In conclusion, based on the factors considered above, we are persuaded by the European Union's explanation that the primary purpose of IC hunts, namely to preserve the tradition and culture of Inuit and to sustain their livelihood, is distinguishable from that of commercial hunts, and justifies the IC distinction, which protects IC interests. We do not find, however, in the evidence presented to us that the rationale or the cause of the distinction can be linked to the alleged "standard of the EU public's morality" in general.⁴⁸²

7.301. We next proceed to examine whether this distinction between commercial and IC hunts, as reflected in the EU Seal Regime through the IC exception, is designed and applied in an even-handed manner.⁴⁸³

Whether the distinction between commercial and IC hunts, as reflected in the IC exception of the EU Seal Regime, is designed and applied in an even-handed manner

7.302. The IC exception is embodied in Article 3(1) of the Basic Regulation. It provides that "the placing on the market of seal products shall be allowed only where the seal products result from hunts traditionally conducted by Inuit and other indigenous communities and contribute to their subsistence. These conditions shall apply at the time or point of import for imported products".

7.303. To implement this provision, Article 3(1) of the Implementing Regulation sets out that, to fall under the *IC hunts* category, seal products must originate from seal hunts that satisfy the following three conditions:

- a. seal hunts conducted by Inuit or other indigenous communities which have a tradition of seal hunting in the community and in the geographical region;
- b. seal hunts the products of which are at least partly used, consumed or processed within the communities according to their traditions; and
- c. seal hunts which contribute to the subsistence of the community.

7.304. Canada submits that with respect to Greenland in particular and its qualification under the IC exception, Greenland's commercialization of output is considerably more extensive and organized than Inuit elsewhere, with a large-scale commercial enterprise (Great Greenland A/S), significant capital investment, such as processing and manufacturing facilities, and sophisticated distribution channels. Greenland's production volume and value dwarfs the commercialized output

The guidance provided by previous panels, as referenced by the European Union to support its position in this regard, pertained to the scope of "public morals" with respect to the policy objective pursued by the regulating Member through the *main measure* at issue. Therefore, the context in which such reasoning was developed in the previous disputes is not the same as in the current context where the European Union must justify the existence of the regulatory distinction under the measure through an exception (i.e. IC exception).

⁴⁸² See paras. 7.401-7.402.

⁴⁸³ We recall the Appellate Body's statement in *US – Gasoline*:

The chapeau by its express terms addresses, not so much the questioned measure or its specific contents as such, but rather *the manner in which that measure is applied* ... The provisions of the chapeau cannot logically refer to the same standard(s) by which a violation of a substantive rule has been determined to have occurred. To proceed down that path would be both to empty the chapeau of its contents and to deprive the exceptions in paragraphs (a) to (j) of meaning. (Appellate Body Report, *US* – *Gasoline*, pp. 22-23)(emphasis added, original footnote omitted)

⁴⁸¹ Referring to the panel reports in *US* – *Gambling* and *China* – *Audiovisual Products*, the European Union states:

The European Union considers that, once it is established that the basic standard of conduct which the EU Seal Regime seeks to uphold is part of the European Union's "public morals", it is not necessary to prove that each of the individual outcomes from the application of that rule in specific situations is regarded by the EU public as a separate rule of public morality on its own. Instead, the mere fact that the EU legislator has made a proper application of the basic rule of morality would be sufficient to confer upon each of those outcomes the status of "public morals". (European Union's response to Panel question No. 31)

of Canada's Inuit; whereas commercial sale of sealskins by Nunavut Inuit is clearly a mere adjunct to the more central purpose of the Canadian Inuit hunt, which is subsistence, Greenlandic Inuit are major commercial operators and conduct the largest commercial seal hunt in the world since 2009.⁴⁸⁴ In that sense, according to Canada, the hunts occurring in Greenland and in Canada both have strong commercial elements. Therefore, the artificial distinction created by the European Union by virtue of IC hunts has no basis and is simply unjustified.⁴⁸⁵

7.305. At the outset, we observe that Canada does not contest the status of Greenland as Inuit. Rather, Canada's argument is focused on the fact that compared to other Inuit and indigenous communities practising seal hunts, the Greenlandic seal hunts practically have a commercial aspect that closely resembles that of commercial hunts. This, according to Canada, demonstrates the arbitrariness in the design and application of the distinction between commercial and IC hunts as reflected in the IC exception under the measure.

7.306. We observe that since the introduction of the EU Seal Regime in 2010, Greenland has been the only Inuit community that has applied for and obtained the benefits of the IC exception under the measure. Although such fact alone is not sufficient to establish arbitrariness in the design or application of the IC exception, it may be an indication that a certain *inherent* flaw in the design and structure of the IC exception prevents other potentially qualifying Inuit and indigenous communities from benefiting from the exception. Against this background, we examine whether the IC exception is designed or applied such that only Greenland can *de facto* benefit from the exception.

7.307. Based on a variety of considerations, we considered above that the "subsistence" purpose of IC hunts includes the need to preserve the culture and tradition of Inuit and indigenous communities and to sustain their livelihood. We found that this purpose of IC hunts, combined with the identity of the hunter as Inuit, distinguishes IC hunts from that of commercial hunts.

7.308. Based on the text, we consider that the requirements of the IC exception are generally linked to the characteristics of IC hunts as discussed above, particularly in terms of the identity of the hunter with a tradition of seal hunting, the use of by-products from the hunted seals, and the contribution of the hunts to the subsistence of the community. The scope and meaning of the "subsistence" criterion under the requirements, however, is not defined under the measure.⁴⁸⁶ Regardless, the parties do not dispute that all of the communities mentioned in the illustrative list of Inuit and indigenous communities under the Basic Regulation, including from Canada's Nunavut and from Greenland, would potentially qualify under the measure.⁴⁸⁷

7.309. Having regard to the actual application of the IC requirements, particularly the commercial aspect of the subsistence criterion, the information before us indicates that, compared to IC hunts in Canada, Alaska, or Russia where most of the hunted seals are used for personal use, over 50 per cent of the hunted seals in Greenland are sold to the tannery of Great Greenland A/S. The tannery, Great Greenland A/S, is owned by the Government of Greenland and is equipped with a modern facility using state of the art technology in the processing of sealskins.⁴⁸⁸ In fact, the tannery of Great Greenland A/S is currently claimed to be "one of the world's leading in producing high quality furs and leather from sealskins".⁴⁸⁹ Seals in Greenland are hunted by paid, full time seal hunters (2,100 over the last five years) and paid, leisure time hunters (5,500).⁴⁹⁰ It is also

⁴⁸⁴ Canada's response to Panel question No. 67.

⁴⁸⁵ Norway also emphasizes the commercial aspect of Greenland's hunts whereby 53 per cent of seal hunting is described as a "commercial activity". (Norway's comments on the European Union's response to Panel question No. 121).

⁴⁸⁶ We note an observation in the COWI 2010 Report that for a hunt to qualify as a hunt for "subsistence" under the Basic Regulation, the following criteria must be met: (a) the hunt is not conducted for the sole purpose of placing on the market (i.e. the motivation behind the hunt is not purely commercial); (b) part of consumption is on the local market (i.e. the seal is not killed in order to export the products for a commercial profit); and (c) contribution to maintaining the community economically or socially. It also states that the hunt must not be organized on a large scale. (COWI 2010 Report, p. 13).

⁴⁸⁷ Canada agrees that seal products from Inuit and other indigenous communities located in Canada would qualify to be imported and placed on the market in the European Union based on the IC requirements under the measure. (Canada's response to Panel question No. 116)

⁴⁸⁸ Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 25.

 ⁴⁸⁹ Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 25. The tannery operates 46 trading stations all over the country, making it possible for hunters in small communities to sell their sealskins.
 ⁴⁹⁰ Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 21.

noted that since 2009 a hunter requires a licence as a full time hunter in order to qualify for selling the skins to the tannery Great Greenland A/S.

7.310. Based on available data, we further observe that the number of seals caught annually in Greenland has always been over 163,000 for the period of 1993-2009. Half of these skins are normally traded, and it is reported that Greenland has stored around 300,000 sealskins since the introduction of the EU Seal Regime.⁴⁹¹ By contrast, in the case of Nunavut in Canada, "in 2006, over 6,000 sealskins were exported". The volume of seals hunted and traded in Greenland is thus comparable to that of commercial (rather than IC) hunts in Canada⁴⁹², and much larger than other Inuit or indigenous communities that may potentially qualify under the IC exception. Although the scale of the hunt *per se* is not a determinative in distinguishing IC hunts from commercial hunts, we recall that the large scale of the hunt was highlighted as one of the factors characterizing commercial hunts.⁴⁹³

7.311. We also take insight from the following statement in "Management and Utilization of Seals in Greenland":

Previously, ringed seal was the most important species in relation to food supply and income, without any doubt. However, the demand from the fur industry has now made it more attractive to hunt harp seals since Greenlandic hunters in some years were offered a slightly better price for sealskins from harp seals compared to sealskins form ringed seals, as prizes [sic] are fixed while the skins are subsidized by the Government of Greenland. The increasing numbers of harp seals have also played an important role in the choice of hunting method.⁴⁹⁴

7.312. The processing of, and trade in, seal products are also integrated among Greenland, Canada, and Norway.⁴⁹⁵ For instance, when the supply of sealskins from local hunters were low due to weather conditions, the tannery of Great Greenland A/S found it necessary to import raw sealskins from Canada to make the best possible use of the capacity at the tannery, and thus also be able to continue to offer local Inuit hunters reasonable prices for their sealskins.

7.313. The factors considered above, namely the level of development in the commercial aspect of Greenlandic seal hunts; the volume of sealskins traded in Greenland; and the integrated nature of the seal product industries in Greenland, Canada, and Norway, indicate that the purpose of seal hunts in Greenland has characteristics that are closely related to that of commercial hunts. Although we recognize that about half of the hunted seals are also used for personal purposes in Greenland and form an important part of their culture and tradition as Inuit, the degree of the commercial aspect of their hunts is comparable to that of the commercial hunts. Greenlandic seal hunts are thus the most commercialized among any other Inuit or indigenous communities. The Government of Greenland itself acknowledges that "Greenland is a country of contrasts. We have culture and tradition that go 4000 years back in time. The Greenlandic society is also part of the

⁴⁹¹ See Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 22, table 2. See also Canada's first written submission, paras. 127-128; opening statement at the first meeting of the Panel, para. 84; response to Panel question No. 41.

We observe that Greenland is cited as one of the "large-scale commercial" sealing countries in certain exhibits. (See, e.g. Parliament Report, (Exhibit JE-4), p. 32).

We also recall the Appellate Body's guidance that the effects of discrimination [regulatory distinction] might be a relevant factor, among others, for determining whether the cause or rationale of the discrimination is acceptable or defensible and, ultimately, whether the discrimination is justifiable; however, it cannot be an exclusive factor. (Appellate Body Report, *Brazil – Retreaded Tyres*, paras. 225-226).

⁴⁹² See Table 2 and 3.

⁴⁹³ The COWI 2010 Report also states that the hunt must not be organized at large scale for a hunt to qualify as an IC hunt under the EU Seal Regime.

⁴⁹⁴ Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 22.

⁴⁹⁵ Parties' responses to Panel question No. 152; COWI 2010 Report, pp. 41-46 (illustrative scenarios for trade in seal products).

modern and developing world today".⁴⁹⁶ Hunting of seals for Greenland therefore is now a "mixed economy, with subsistence and monetary elements coexisting".⁴⁹⁷

7.314. As noted above, no other Inuit or indigenous communities, potentially eligible for the IC requirements, have applied for the IC exception since the introduction of the EU Seal Regime. For example, with respect to Inuit seal hunts in Canada, Canada explained that due to the reliance of the Canadian Inuit on the marketing channel provided by commercial hunts for the sale of their products, and given the limited volume of products derived from Canadian Inuit hunts, it is not cost effective under the current circumstances to segregate Inuit products from other products. We also observed that seals hunted by Inuit or indigenous communities in Russia or Alaska are almost entirely used for personal use and consumption. Therefore, based on available evidence, among a small number of Inuit and indigenous communities that may potentially satisfy the specific requirements of the IC exception, Inuit in Greenland and Canada are the most likely, if not the only, beneficiaries under the measure. Given the factual circumstances of the Inuit and indigenous communities in Canada as explained above, however, currently, Greenland, with the most commercialized of IC hunts, is in fact the only beneficiary of the IC exception.

7.315. The legislative history of the EU Seal Regime suggests that this is not merely an incidental effect of the application of the measure. We observe that prior to the crafting of the specific requirements of the IC exception in the Implementing Regulation, seal hunts in Greenland were considered to be the only Inuit hunts that could benefit from the IC exception. In fact, the COWI Reports anticipated that no other Inuit and indigenous communities would be able to benefit from the IC exception "as only Greenland will be able to make the investments needed to make use of exemptions" and "the scale of the Canadian hunt is too small and not as centrally organized as that in Greenland".⁴⁹⁸ Canada also explains that it is not economically feasible for Canadian Inuit to develop their own processing and distribution chains, given that the Inuit have relied on synergies with southern producers; as those networks may no longer be viable because of the EU Seal Regime, considerable investment would be needed to develop a new processing and distribution centre.⁴⁹⁹

7.316. Moreover, in the actual operation of the IC exception, Danish customs authorities processed imports based on certificates issued by the Greenlandic authorities prior to the Greenlandic entity obtaining recognized body status within the meaning of the Implementing Regulation.⁵⁰⁰ The European Union explains that Danish customs authorities proceeded in that manner "based on [their] interpretation of the Implementing Regulation whereby the issuance of attesting documents complying with the Implementing Regulation would also be allowed during the application process for recognized body status and not only once the process has been completed".

7.317. The considerations above, namely the text of the IC exception, its legislative history, and the actual application of the IC exception, cast serious doubt on the even-handedness of the design and application of the IC exception. Specifically, the rationale or cause of the exception (i.e. the distinction between the IC hunts and commercial based on their purpose) was the "subsistence" of Inuit and indigenous communities in terms of their culture and tradition as well as their livelihood. However, under the measure, the IC exception is available *de facto* exclusively to Greenland, where the Inuit hunt bears the greatest similarities to the commercial characteristics of commercial hunts. This suggests in our view that the IC exception was *not* designed or applied in

⁴⁹⁶ Management and Utilization of Seals in Greenland, (Exhibit JE-26), pp. 1 (foreword) and 24 ("Hunting of seals continues to be an important part of everyday life and culture in Greenland.").

⁴⁹⁷ Management and Utilization of Seals in Greenland, (Exhibit JE-26), pp. 25-26: "Today, hunting seals is not economically viable without a subsidy. However, harvesting provides the basic food supply for most communities."

⁴⁹⁸ COWI 2010 Report, p. 84. It further notes that Canadian Inuit hunt essentially uses the sales and marketing chains of the commercial hunt, implying it would need to invest heavily in separating its Inuit product from the rest. (See Parliament Report, (Exhibit JE-4), p. 34). We also find a similar assessment in the COWI 2008 Report, noting that the Canadian Inuit hunt is small and not centrally organized, and the infrastructure for commercial trade is already in place in Greenland. (COWI 2008 Report, p. 26).

⁴⁹⁹ Canada's first written submission, paras. 37-48; response to Panel question No. 84. Canada further submits that the Government of Nunavut has indicated that there has been no demand for Canadian Inuit products from European Union buyers since the coming-into-force of the ban. Therefore there has not been much incentive to pursue the marketing of seal products under the IC category.

⁵⁰⁰ European Union's response to Panel question No. 161. See also certificates issued by Greenlandic authorities prior to obtaining recognized body status, (Exhibit EU-162).

an even-handed manner so as to make the benefits of the exception available for all potential beneficiaries.

7.318. The European Union argues that any effects derived from the fact that operators in one country (like Canada) choose not apply for the IC exception cannot be attributed to the EU Seal Regime. In our view, this argument fails to take into account that the absence of the even-handedness in the design and application of the distinction between commercial and IC hunts is linked to the fact that the IC exception, as currently designed and applied under the measure, is not equally available to all Inuit or indigenous communities. Only those in Greenland have been able to benefit from it and this, in our view, is directly attributable to the regime itself and not to the actions of the operators in countries like Canada.

7.319. In light of the above, we conclude that although the distinction between commercial and IC hunts based on the purpose of the hunt is justifiable having regard to the explanations given by the European Union concerning the benefits to Inuit or indigenous communities, it is not designed and applied in an even-handed manner. Therefore, we find that the IC exception of the EU Seal Regime is inconsistent with the European Union's obligations under Article 2.1 of the TBT Agreement as the European Union has failed to demonstrate that the detrimental impact caused by the IC exception on Canadian seal products stems exclusively from a legitimate distinction.

7.3.2.3.4 Whether the distinction between commercial hunts and MRM hunts is legitimate

7.3.2.3.4.1 Main arguments by the parties

Complainant (Canada)

7.320. Canada argues that the EU Seal Regime draws an arbitrary distinction between commercial and MRM hunts by imposing conditions that are unrelated to the Regime's underlying policy objectives.⁵⁰¹ For Canada, the commercial purpose of the hunt has no bearing on whether seals are killed humanely.⁵⁰² Canada argues that the distinction between commercial and MRM hunts under the EU Seal Regime is "illusory" because MRM hunts are motivated primarily if not exclusively by commercial gain.⁵⁰³ In particular, Canada notes that the MRM exception only eliminates profit-making at the hunt level while allowing profit-making through the processing, manufacturing and retailing of seal products.⁵⁰⁴

7.321. Canada observes that the EU Seal Regime does not draw any distinction between seal products on the basis of animal welfare criteria.⁵⁰⁵ Therefore, seal products placed on the EU market under the MRM exception may still contain seal that suffered pain and distress at the time of killing.⁵⁰⁶ For Canada, such a result is counterproductive to achieving the objective of protecting seal welfare.⁵⁰⁷ In addition, Canada argues that the seal hunters' inability to sell seal products for profit may encourage hunting methods that run counter to positive animal welfare outcomes.⁵⁰⁸

⁵⁰⁷ Canada's second written submission, para. 257.

⁵⁰¹ Canada's first written submission, para. 393.

⁵⁰² Canada's first written submission, para. 391; second written submission, para. 252.

⁵⁰³ Canada's first written submission, para. 392; response to Panel question No. 28, para. 130; second

written submission, paras. 263-264; and opening statement at the second meeting of the Panel, para. 55. ⁵⁰⁴ Canada's first written submission, para. 392; response to Panel question No. 28, para. 130; second

written submission, paras. 263-264; and opening statement at the second meeting of the Panel, para. 55. ⁵⁰⁵ Canada's second written submission, paras. 222 and 265; response to Panel question No. 28.

Canada argues that the European Union could contribute to the protection of animal welfare by conditioning market access for seal products on compliance with animal welfare standards. In Canada's view, "[e]vidence of such compliance can be provided by the country from which the seal product originates which thus minimizes concerns that the EU Seal Regime is being applied extra-territorially". (See Canada's response to Panel question No. 135(a)).

⁵⁰⁶ Canada's second written submission, paras. 222 and 265.

⁵⁰⁸ Canada's second written submission, para. 257. See Canada's opening statement at the second meeting of the Panel, para. 58.

- 101 -

7.322. Moreover, Canada argues that the alleged moral basis for the MRM exception rests on an unfounded and speculative assumption that marine resource management hunters are more likely to meet animal welfare standards if they have a commercial incentive to recover their costs.⁵⁰⁹ According to Canada, the European Union has provided no evidence in support of this argument showing: (a) that hunters were complying with animal welfare standards prior to the EU Seal Regime⁵¹⁰; (b) that eliminating the MRM category would result in more suffering for seals culled in such circumstances⁵¹¹; or (c) that seal culls in EU member States are conducted in a manner consistent with the animal welfare standards that the European Union accuses Canada of failing to apply.⁵¹²

7.323. In addition, Canada claims that the MRM exception is not applied even-handedly, and that the conditions under the exception are arbitrary and unjustifiable.⁵¹³ In particular, the small-scale, non-systematic and non-profit requirements of the MRM category effectively prevent the placement on the EU market of seal products from countries such as Canada, where seals are also harvested in sustainable numbers in accordance with a marine resource management plan.⁵¹⁴ The conditions under the MRM exception are thus unrelated to the objective of sustainable marine resource management⁵¹⁵, and to the central objective of the EU Seal Regime of addressing concerns relating to seal welfare.⁵¹⁶ The regulatory distinction thus arbitrarily favours marine management programmes involving small populations of seals, such as those of Sweden, Finland and the United Kingdom.⁵¹⁷ Canada claims that the European Union's willingness to accommodate the interests of its member States while "completely ignor[ing]" the interests of other WTO Members is "plainly discriminatory" and "unjustifiable".⁵¹⁸

Respondent (European Union)

7.324. The European Union explains that the MRM exception was intended to exempt from the ban seal products deriving from small-scale, occasional hunts conducted with the purpose of managing marine resources.⁵¹⁹ According to the European Union, such hunts are conducted in several countries within and outside the European Union territory.⁵²⁰ The European Union argues that the conditions in which MRM hunts take place are in principle more favourable to the humane killing of seals than the commercial hunt.⁵²¹ For instance, the commercial nature of the hunt creates an incentive for hunters to kill as many seals as possible over a short period of time, thereby potentially disregarding the manner in which the seals are killed; this is not the case for MRM hunts which target specifically seals that pose a threat to fish stocks or fishing equipment.⁵²²

7.325. The European Union recognizes that there is also a commercial dimension present in the MRM hunts.⁵²³ However, the European Union explains that if hunters were not permitted to recoup

⁵²³ The European Union notes in this regard that "[t]he fact that ... hunts for managing marine resources have or may have a commercial dimension does not mean that those hunts are comparable to the hunts

⁵⁰⁹ Canada's opening statement at the second meeting of the Panel, para. 58.

⁵¹⁰ Canada's opening statement at the second meeting of the Panel, para. 58.

⁵¹¹ Canada's opening statement at the second meeting of the Panel, para. 58.

⁵¹² Canada's second written submission, para. 265 (referring to COWI 2008 Report, p. p. 38 and 86).

⁵¹³ Canada's first written submission, para. 393; second written submission, paras. 219-226.

⁵¹⁴ Canada's comments on the European Union's response to Panel question No. 123, paras. 59 and 61.

⁵¹⁵ Canada's first written submission, paras. 391, 510-523; second written submission, para. 252; and opening statement at the second meeting of the Panel, para. 54.

⁵¹⁶ Canada's first written submission, para. 391; second written submission, paras. 221 and 252.

⁵¹⁷ Canada's first written submission, para. 391; second written submission, para. 256; and opening statement at the second meeting of the Panel, para. 53.

⁵¹⁸ Canada further argues that where a measure is premised on an objection to conduct that takes place outside the territory of the regulating Member, there should be additional efforts to engage in negotiations to regulate the conduct in question. (See Canada's response to Panel question No. 135(a), paras. 118 and 119).

⁵¹⁹ European Union's first written submission, paras. 309-310 (citing the opinion of the European Parliament's Committee on Agriculture and Rural Development in the Parliament Report, (Exhibit JE-4), p. 57 and Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10), p. 16). The European Union explains that seals considered as pests, also referred to as "nuisance seals", pose a threat to fish stocks and can also damage fishing gear. (See European Union's response to Panel question No. 123(a)). ⁵²⁰ European Union's first written submission, para. 320. The European Union notes that seal products

⁵²⁰ European Union's first written submission, para. 320. The European Union notes that seal products derived from the hunts of nuisance seals in Canada could, in principle, fall under the MRM exception provided that all the relevant conditions were met. (European Union's second written submission, para. 250).

⁵²¹ European Union's response to Panel question No. 8, paras. 19 and 26.

⁵²² European Union's first written submission, paras. 326, 328, and 333.

- 102 -

their costs by placing on the market seal products derived from MRM hunts, they would be more likely to resort to inappropriate killing methods; such an outcome could compromise the objective of protecting seal welfare.⁵²⁴ In this regard, the European Union asserts that the MRM exception is rationally connected to the overall objective of the EU Seal Regime.⁵²⁵ Moreover, in the European Union's view, prohibiting the marketing of products derived from MRM hunts would not contribute to reducing the suffering of seals because these hunts would continue to take place in any event.526

7.326. The European Union notes that the MRM exception addresses the longstanding moral concerns of the EU public with regard to the presence on the EU market of seal products by permitting the placing on the market of certain "morally acceptable" seal products in view of the type and purpose of the hunt from which they derive.⁵²⁷ According to the European Union, although the MRM exception is not subject to compliance with animal welfare requirements, the benefits arising from the placing on the market of products deriving from these hunts, for humans and other animals outweigh the risk of suffering being inflicted upon seals.⁵²⁸ To the extent that the EU Seal Regime would permit the placing on the market of seal products deriving from seals hunted inhumanely, it would still be in accordance with the EU's standard of morality that the Regime seeks to uphold.529

7.327. Finally, the European Union argues that the MRM exception is designed and applied in an even-handed manner.⁵³⁰ The conditions set out under the MRM exception are essentially aimed at avoiding a potential circumvention of the ban on trade in seal products.⁵³¹ Also, the fact that the MRM exception allows profit-making at the downstream level does not show a lack of even-handedness; the exception aims at affecting the conduct of the hunter by eliminating the incentives to kill seals in an inhumane manner. The fact that other manufacturers or processors down the line can make a profit does not affect the hunter's behaviour when hunting seals.⁵³²

7.3.2.3.4.2 Analysis by the Panel

7.328. We turn to examine whether the distinction drawn by the measure between commercial and MRM hunts, and consequently between products derived from each category of hunt, is legitimate within the meaning of Article 2.1 of the TBT Agreement. As we did in connection with the IC exception, we assess the legitimacy of the regulatory distinction between commercial hunts and MRM hunts by examining the following questions: first, is the distinction rationally connected to the objective of the EU Seal Regime; second, if not, is there any cause or rationale that can justify the distinction (i.e. "explain the existence of the distinction") despite the absence of the rationale connection to the objective of the Regime⁵³³, taking into account the particular

conducted for commercial purposes, where the hunter is paid to kill as many seals as possible in a short period of time and with a view to the further processing and marketing of seal products through commercial channels." (European Union's closing statement at the second meeting of the Panel, p. 4).

⁵²⁴ The argument made by the European Union is that the possibility for the hunter to recover his costs (without necessarily making a profit) provides a motivation to retrieve the carcass of the seal hunted and thus to avoid "struck and lost" seals. It further provides a motivation for clean headshots rather than just shooting seals without regard to which part of the body is hit, or ensuring that the seal is dead. (European Union's response to Panel question No. 8, para. 27. See also European Union's first written submission, para. 316).

⁵²⁵ European Union's first written submission, para. 316.

⁵²⁶ European Union's first written submission, para. 315; response to Panel question No. 8, paras, 27-28.

⁵²⁷ European Union's first written submission, paras. 308, 317, and 329.

⁵²⁸ European Union's second written submission, para. 255.

⁵²⁹ European Union's opening statement at the second meeting of the Panel, para. 59. According to the European Union, the potential suffering of seals is outweighed by the benefits accruing to humans and other animals under the IC and MRM exceptions, respectively. (See the European Union's response to Panel question No. 136 and Canada's comments on the European Union's response to Panel question No. 136).

⁵³⁰ European Union's second written submission, paras. 257-263.

⁵³¹ European Union's second written submission, para. 260.

⁵³² European Union's first written submission, para. 334; second written submission, para. 261.

⁵³³ See Appellate Body Report, *Brazil – Retreaded Tyres*, paras. 226-234. The Appellate Body stated that it had "difficulty understanding how discrimination might be viewed as complying with the chapeau of Article XX when the alleged rationale for discriminating does not relate to the pursuit of or would go against the objective that was provisionally found to justify a measure under a paragraph of Article XX" (emphasis added)

In our view, the Appellate Body's reasoning in US - Clove Cigarettes also supports our approach here. In that dispute, regarding the legitimacy of the regulatory distinction drawn by the measure in question (i.e.

- 103 -

circumstances of the current dispute; and, third, is the distinction concerned, as reflected in the measure, "designed or applied in a manner that constitutes arbitrary or unjustifiable discrimination" such that it lacks "even-handedness". 534

Whether the MRM distinction is rationally connected to the objective of the EU Seal Regime

Characteristics of MRM hunts⁵³⁵

Identity of the hunter

7.329. MRM hunts are conducted essentially on a voluntary basis by fishermen whose fish stocks or fishing equipment are endangered by individual seals, or by the seal population in a particular area.⁵³⁶ To proceed with a marine resource management hunt, a special licence or permission from the local authorities is normally required.⁵³⁷

Purpose of the hunt

7.330. The European Union notes that while compliance with a resource management plan is one of the conditions under the MRM exception, the exception is not aimed at promoting a better management of marine resources; the European Union uses other instruments to achieve this purpose.⁵³⁸ Rather, the MRM exception takes into account that, alongside large-scale hunts carried out mainly for commercial purposes, there are also small-scale hunts conducted on an occasional basis for the purpose of ensuring that individual seals are eliminated for pest control ("nuisance seals"), or that seals are killed because according to scientific studies their population in a particular area poses a threat to fisheries and/or the ecosystem (seal culling).⁵³⁹

Scale of the hunt

7.331. All sealing countries with the exception of Greenland conduct their seal hunt on the basis of marine resource management plans based on scientifically established TAC.⁵⁴⁰ The complainants claim that their commercial seal hunt is fully consistent with the objective of sustainable marine

distinction between clove and menthol cigarettes), the Appellate Body examined the following questions: first, whether the distinction was connected to the objective of the measure which justified the prohibition of clove cigarettes; and, second; in the negative, whether the United States provided any reasons independent of the objective of the measure that could justify the distinction ("the reasons presented by the United States for the exemption of menthol cigarettes from the ban") (AB Report, *US – Clove Cigarettes*, para. 225).

⁵³⁴ See Appellate Body Report, US – COOL, para. 340.

⁵³⁵ This information is drawn mainly from COWI 2008 Report and COWI 2010 Report, which assess the potential impacts of the MRM exception on the different sealing states.

⁵³⁶ European Union's response to Panel question No. 123, para. 91. The European Union submits, with respect to the seal hunt in Sweden, that "[t]here are only a few hunters able to conduct seal hunt as it is not a very easy hunt and they also must have a permission [to hunt]." For example, in 2007, out of the 50 hunters who reported to have shot one or more seals, 20-30 of these hunters were commercial fishermen whose primary aim was to keep the seal away from their fishing equipment and reduce the damage created by seals to equipment and catch. (European Union's response to Panel question No. 123, paras. 86-87 (referring in part to Sweden's response of 6 October 2001 to the Commission deficiency letter, (Exhibit EU-158))).

⁵³⁷ European Union's response to Panel question No. 123, para. 86.

⁵³⁸ European Union's first written submission, para. 41.

⁵³⁹ European Union's response to Panel question No. 123(a), paras. 91-92. The European Union refers to the definition of "nuisance seals" in Canada's Marine Mammal Regulations, namely "a seal that represents danger to the fishing equipment despite deterrence efforts or, based on a scientific recommendation, to the conservation of fish stocks. (Marine Mammal Regulations, (Exhibit CDA-21), p. 2). According to EFSA, some hunts are conducted "because seals are perceived as pests or competitors with humans or their activities (e.g. direct or indirect impacts on fishing, aquaculture, or as vectors of fish parasites) or as threats to other species of concern (e.g. predation upon endangered species). In cases where individual animals (vs. population cull) are the focus, the animals are often referred to as 'nuisance seals'." EFSA Scientific Opinion, pp. 12-13. In Canada, the licences issued for the hunt of nuisance seals do not allow culling. (Canada's response to Panel question No. 167, para. 208). Data available on the number of licences granted for nuisance seal hunts in Canada and the total catch per region on an annual basis suggest that this type of hunt concerns only a small number of seals. (See National Nuisance Seal Licence Operational Guidelines and Procedures, (Exhibit CDA-143)).

⁵⁴⁰ COWI 2010 Report, pp. iv-v.

- 104 -

resource management and takes place within the limits of their respective TAC.⁵⁴¹ One of the main distinguishing factors between MRM hunts and other types of hunts is the size of the hunt.⁵⁴² The COWI 2010 Report indicates that small-scale hunts for marine resource management purposes are conducted in Sweden⁵⁴³, Finland⁵⁴⁴, and Scotland.⁵⁴⁵ Nuisance seal hunts are also conducted in Canada.546

Seal hunting period

7.332. The hunting period varies depending on the range country and the type of seals hunted. For instance, in Sweden and Finland, the Grey and Baltic ringed seals may be hunted with a licence during their respective hunting seasons. The hunting season in Sweden runs from 16 April to 31 December; in Finland, seal hunts of grey seals are carried out from 16 April to 31 December, while ringed seals are hunted from 16 April to 31 December and 1 September to 15 October.⁵⁴⁷ In the United Kingdom (Scotland), the season extends from 1 September to 31 December for grey seals and from 1 June to 31 August for harbour seals.⁵⁴⁸ There are annual "closed" seasons set for grey and common seals corresponding to the breeding period of both species; outside of the closed seasons, no licence is required to remove seals.⁵⁴⁹ In Canada, nuisance seal hunts generally take place during the open season pursuant to the Marine Mammal Regulations⁵⁵⁰; however, licences may also be delivered during the closed season.⁵⁵¹

Hunting methods

7.333. The hunting methods used for MRM hunts are generally similar to the methods used in commercial hunts. In Sweden, for instance, the seal hunt is conducted only with firearms.⁵⁵² In Finland and the United Kingdom (Scotland), the main weapons are firearms⁵⁵³ but other killing methods may also be used, such as harpoons, clubs, spears, traps, hooks or nets.⁵⁵⁴ In Canada and Norway, hakapiks (as well as clubs in Canada) may be used in addition to firearms.⁵⁵⁵

Organization and control of the hunt

7.334. While MRM hunts are not monitored per se, most range countries that conduct seal hunts on a small scale exercise control over the hunts through a licensing scheme.⁵⁵⁶ For instance, in the

Norway's first written submission, para. 437; second written submission, para. 81.

- ⁵⁴⁵ The number of seals killed in Scotland annually is estimated at 3,500 seals.
- ⁵⁴⁶ Marine Mammal Regulations, (Exhibit CDA-21), Article 26.

 ⁵⁴⁷ See EFSA Scientific Opinion, p. 32; COWI 2008 Report, pp. 82 and 88.
 ⁵⁴⁸ See EFSA Scientific Opinion, p. 32; COWI 2008 Report, pp. 82 and 88. EFSA reports that the Habitats Directive (Directive 92/43/EEC) protects five seal species regularly occurring within the European Union. EU member States are obliged to designate special areas of conservation and to monitor the conservation status of identified species. The Habitats Directive also lists the prohibited methods and means of capture and killing, and modes of transportation. (EFSA Scientific Opinion, p. 32).

⁵⁴⁹ EFSA Scientific Opinion, p. 33.

⁵⁵⁰ National Nuisance Seal Licence Operational Guidelines and Procedures, (Exhibit CDA-143).

⁵⁵¹ Canada's response to Panel question No.167 (referring to National Nuisance Seal Licence Operational Guidelines and Procedures, (Exhibit CDA-143)).

- ⁵⁵² EFSA Scientific Opinion, pp. 24 (Table 2), 31 and 33.
- ⁵⁵³ EFSA Scientific Opinion, pp. 24 (Table 2), 31 and 33.
- ⁵⁵⁴ EFSA Scientific Opinion, pp. 32.
- 555 COWI 2008 Report, p. 28.

⁵⁵⁶ This is the case, for instance, in Sweden, Finland, United Kingdom (Scotland), and Canada. For instance, in Sweden and Finland, samples of the seal must be returned by the hunter for analysis. (See COWI 2008 Report, pp. 35-44 and 78-87). In Canada, a licence is required for hunting "nuisance seals". (Marine Mammal Regulations, (Exhibit CDA-21), Article 26.1; see also National Nuisance Seal Licence Operational Guidelines and Procedures, (Exhibit CDA-143)).

⁵⁴¹ Canada's comments on the European Union's response to Panel question No. 123, para. 50;

⁴² See European Union's response to Panel question No. 123(a), para. 80; COWI 2010 Report, Annex 4, p. 5; COWI 2008 Report, pp. 36 and 87.

 $^{^{543}}$ The number of seals killed in 2010 was approximately 100-115 seals for a total quota of about 200 seals for the year.

⁵⁴⁴ COWI reports that the seal hunt in Finland is mainly recreational but that reducing the negative impact of seals on fisheries is also a consideration in the hunt. According to the Finnish Ministry of Agriculture and Forestry, between 800-1000 hunters of grey seals are licensed annually with a corresponding 400-500 grey seals hunted. While ringed seals are also found in the Baltic Sea region of Finland, no licences are issued for their hunt. COWI 2010 Report, pp. 27-28.

- 105 -

case of Sweden, quotas of seals to be felled are decided by the Swedish Environmental Protection Agency and set specifically on an annual basis for each county. There are areas within the counties where no hunting is allowed. To hunt seals in a particular area, the hunters will have to seek permission by applying to the County Administrative Board in the county where they intend to conduct the hunt. Permission will be granted provided the guota level has not been reached. For this purpose, the Swedish Coast Guard keeps a daily record. Hunters must contact the Swedish Coast Guard at the end of each day to report the result of their hunt. The County Administrative Board receives daily reports from the Swedish Coast Guards who, in addition to keeping track of daily catch, also patrol the waters.⁵⁵⁷ In Canada, nuisance seal hunts are also subject to strict conditions, including compliance with the animal welfare requirements imposed under the Marine Mammal Regulations.558

Use of products derived from the hunt

7.335. Seals killed in the context of MRM hunts are normally used on a private basis or sold in the local community.⁵⁵⁹ This is the case, for instance, for products derived from seal hunts in Sweden where the skin and meat are generally used by the hunter himself or sold on the local market.⁵⁶⁰ The European Union notes that the by-product of MRM hunts that may end up on commercial markets would be a small amount of fur skin.⁵⁶¹ In Canada, seals harvested under the authority of a Nuisance Seal Licence cannot not be sold, bartered or traded.

Connection between the MRM distinction and the objective of the EU Seal **Regime**

7.336. Based on the description above, MRM hunts are characterized by the fact that they are conducted occasionally on a small scale, primarily for sustainable marine resource management, particularly for controlling nuisance seals and seal culling.

7.337. The evidence shows that MRM hunts, though much smaller in scale than commercial hunts, also give rise to concerns regarding seal welfare that are present in seal hunting in general.⁵⁶² Although there is limited evidence on the animal welfare outcomes of the seal hunt in countries that engage in small-scale hunts, such as Sweden and Finland⁵⁶³, based on the evidence presented to the Panel, it would seem that seal hunts conducted in EU member States are not subject to onerous animal welfare requirements.

7.338. For instance, according to the COWI 2008 Report, seal hunting regulations in Sweden and Finland do not require hunters to apply the three-step method of humane killing. COWI reports that in Sweden, it is unclear how well monitored the hunt is due to the relative scarcity of inspectors⁵⁶⁴; in the case of Finland, hunters are largely self-regulated, and it is unclear whether there is any independent monitoring of the seal hunt. Therefore, there is no way of ensuring that MRM hunts are conducted in accordance with the objective of addressing the EU public concerns on seal welfare.⁵⁶⁵ In this connection, we find it speculative that the possibility for the hunters to recover the costs of the hunt through the placing on the market of seal products under the MRM

⁵⁵⁷ European Union's response to Panel question No. 123(a), para. 86 (citing COWI 2008 Report, pp. 81-86 and Sweden's response of 6 October 2001 to the Commission deficiency letter, (Exhibit EU-158)).

⁵⁵⁸ Canada's response to Panel question No. 167. See National Nuisance Seal Licence Operational Guidelines and Procedures, (Exhibit CDA-143) (outlining the conditions under which fishing licences are issued pursuant to Canada's Marine Mammal Regulations, (Exhibit CDA-21)).

⁵⁵⁹ European Union's first written submission, para. 327 (citing COWI 2010 Report, (Exhibit JE-21), p. 67). See also COWI 2010 Report, p. 35 and Annex 4, pp. 2 and 5; response to Panel question No. 123,

para. 87. ⁵⁶⁰ The European Union notes that the market price for a seal in Sweden would be in the order of 150 euros. (European Union's response to Panel question No. 123, para. 87).

⁵⁶¹ European Union's response to Panel question No. 123, para. 87.

⁵⁶² See for instance Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10), p. 16; European Commission's Impact Assessment on the ban of products derived from seal species (23 July 2008), (Commission Impact Assessment), (Exhibit JE-16), pp. 69 and 74.

⁵⁶³ The European Union notes that the regulation of seal hunting, including the animal welfare aspects of the hunt falls in principle within the competence of EU member States. 564 See COWI 2008 Report, pp. 78-87.

⁵⁶⁵ See COWI 2008 Report, pp. 35-44.

exception encourages more responsible behaviour on the part of the hunter with respect to the welfare of seals.

7.339. Finally, we do *not* consider that the limited scope of MRM hunts and the small volume of potential trade concerned by this exception as such are relevant factors in our assessment of whether the distinction in question is rationally connected to the objective of the measure.⁵⁶⁶ We find support for this view in the Appellate Body's consideration in *Brazil – Retreaded Tyres*: the fact that only a small amount of products were imported under the MERCOSUR exception did not affect the finding that the exception was rationally disconnected from the objective of the measure at issue.⁵⁶⁷

7.340. For the foregoing reasons, we find that the MRM distinction is not rationally connected to the objective of addressing the EU public moral concerns on seal welfare. According to the European Union, however, the purpose of the hunt, which distinguishes MRM hunts from commercial hunts, combined with its small scale and occasional occurrences, justifies any risk of suffering inflicted upon seals as a result of such hunts. We next examine whether the European Union's explanation justifies the MRM distinction.

Whether the cause or rationale put forward by the European Union for the distinction between commercial and MRM hunts is justifiable

7.341. In addressing the argument by the European Union, we first examine the purpose of MRM hunts, specifically controlling nuisance seals and seal culling, compared to that of commercial hunts. We then address the question of whether any distinction found between the purpose of the MRM hunts and the purpose of commercial hunts is justified despite the lack of a rational connection to the objective of the EU Seal Regime as a whole.

7.342. We note that the complainants do not challenge the objective of sustainable marine resource management as such.⁵⁶⁸ In fact, the complainants contend that their seal hunts are fully consistent with sustainable marine resource management principles and take place within the limits of their respective TAC.⁵⁶⁹ Moreover, the European Union has confirmed that the exception is not aimed at promoting a better management of marine resources as it has other instruments it uses for that purpose. The complainants argue however that the distinction between MRM and commercial hunts based on their purpose is illusory because MRM hunts also have a commercial purpose.⁵⁷⁰ Canada notes that MRM hunts are motivated primarily if not exclusively by commercial gain, for instance to support a thriving fishery or to prevent the destruction of fishing gear, all of

⁵⁶⁶ European Union's second written submission, para. 288; response to Panel question No. 122.

⁵⁶⁷ Appellate Body Report, *Brazil – Retreaded Tyres*, paras. 229 and 233.

⁵⁶⁸ See Canada's first written submission, para. 465 and Norway's first written submission,

paras. 723-724. Sustainable marine resource management seeks to ensure, among others, that the human exploitation of natural resources does not result in the long-term decline of the resources. As submitted by Norway, such principles have been recognized consistently by the international community. (See for instance the definition of the term "sustainable use" contained in Article 2 of the United Nations Convention on Biological Diversity (1992), (Exhibit NOR-66)).

⁵⁶⁹ COWI notes that Norway's seal hunt would probably meet the first two requirements of the MRM exception because it is conducted as part of a marine resource management plan that uses scientific population models of marine resources and applies the ecosystem-based approach, and because the hunt does not exceed the TAC established under the management plan. (See COWI 2010 Report, pp. 69-79 and Annex 4, pp. 3-4). Furthermore, Norway argues that the national quotas established for seal hunting are typically identical to those recommended by the International Council for the Exploration of the Seas (ICES), which provides scientific advice on the marine eco-system to all countries bordering the North Atlantic Ocean and the Baltic Sea. (Norway's first written submission, para. 639, second written submission, para. 245). Canada comments that its seal products are unlikely to meet this requirement under the MRM exception as its national management plan is not based on an "ecosystem-based approach". (Canada's first written submission, para. 344 (citing COWI 2010, p. 64 and Annex 4, p. 1)).

⁵⁷⁰ Canada's second written submission, paras. 262-263. Norway argues that fishermen derive a net economic benefit in the form of an improved fishing activity. Further, Norway argues that evidence on the legislative history of the measure indicates that the fishermen killing seals for MRM purposes are entitled to earn income compensating for the cost of their time.

- 107 -

which are used in a commercial venture. Canada further argues that the killing of seals during resource management culls is also a means to generate income. 571

7.343. As mentioned previously, the commercial seal hunt is characterized *inter alia* by the competitive pressure on hunters to kill as many seals as possible in a limited period of time. In the case of MRM hunts, the motivation of the hunter is not primarily linked to the exploitation of seals as a natural resource; rather, it is aimed at mitigating the damage caused by seals and is incidental to the conduct of another fishing activity. To that extent, we agree with the complainants that there is a commercial dimension to seal hunts conducted for the purpose of managing marine resources. The evidence before the Panel further shows that the costs associated with damage caused by seals can be significant in some cases.⁵⁷² Therefore, while the hunter cannot place seal products on the EU market for profit under the MRM exception, there is nevertheless an economic incentive for fishermen or seal hunters to conduct an MRM hunt. Further, while the MRM exception aims to eliminate profit at the hunt level, it still allows profitmaking at the downstream level.

7.344. While we recognize that MRM hunts take place on an occasional basis, and on a much smaller scale than commercial hunts, and that the primary means to generate income for those conducting MRM hunts is not seal hunting itself, in light of the considerations above, we are not convinced that the purpose of MRM hunts and the purpose of commercial hunts are of a different character or nature. Furthermore, the difference that might be found between the commercial aspects of an MRM and a commercial hunt is, in our view, not sufficient to justify the lack of a rational connection between the distinction in question and the objective of addressing the EU public moral concerns on seal welfare.

7.345. Finally, the European Union argues that the placing on the EU market of seal products derived from MRM hunts conforms with the "EU's standard of morality" because the potential suffering of seals is outweighed by the benefits accruing to other animals. However, as noted above, the evidence adduced by the European Union on the EU public's moral concerns regarding seal welfare does not clearly establish that the concerns of EU citizens vary according to the type of hunt.⁵⁷³

7.346. In conclusion, we do not find that the rationale put forward by the European Union based on the purpose of MRM hunts, combined with their small scale and occasional occurrences, justifies the MRM distinction in the absence of a rational connection to the objective of the EU Seal Regime concerning seal welfare.

7.347. Therefore, we conclude that the European Union has failed to establish that a detrimental impact caused by the MRM exception on Canadian seal products *vis-à-vis* the like EU domestic products stems exclusively from a legitimate regulatory distinction. Nevertheless, for completeness, we turn to consider the design and application of the regulatory distinction between MRM and commercial hunts under the EU Seal Regime.

⁵⁷¹ See Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10), p. 16 (stating that "[t]he seals are not hunted only as pests but they are used as a natural resource for livelihood and also as a means to generate income.").

⁵⁷² See Swedish Management Plan for Grey Seals, (Exhibit CDA-54), pp. 37-39; and Finnish Management Plan for Seals in the Baltic Sea, (Exhibit CDA-51), p. 3. The Finnish Management Plan for Seals in the Baltic Sea, for instance, states that:

On the basis of data on damage in the period 1997-1999, damage to professional fishing by seals in Finland was estimated to be roughly 1.68 million euros. Since then, the grey seal population has more than doubled. In the period 2000-2001, professional fishermen were compensated for 3.2 million euros for the damage they had sustained, although the claims sent in suggest that the overall damage amounted to 7.47 million euros. (Exhibit CDA-51, p. 40).

⁵⁷³ In this regard, we agree with Canada's comment that:

[[]t]he public opinion polls produced by the European Union ... do not specifically gauge whether the subjects of the poll view the nature of seal hunting as a matter of public morality. They also fail to solicit popular views about the acceptability of permitting seal products from ... resource management hunts, regardless of whether the seals from which those products were obtained were killed humanely. (Canada's response to Panel question No. 31).

- 108 -

Whether the distinction between commercial and MRM hunts, as reflected in the MRM exception of the EU Seal Regime, is designed and applied in an even-handed manner

7.348. We recall the specific requirements of the MRM exception as set out in Article 5(1) of the Implementing Regulation: (a) seal hunts conducted under a national or regional natural resources management plan which uses scientific population models of marine resources and applies the ecosystem-based approach; (b) seal hunts which do not exceed the total allowable catch quota established in accordance with the plan referred to in point (a); and (c) seal hunt by-products of which are placed on the market in a non-systematic way on a non-profit basis.⁵⁷⁴

7.349. We note that currently only Sweden has entities registered as recognized bodies entitled to deliver attesting documents permitting the placing on the market of seal products under the MRM exception. 575

7.350. According to Canada, the legislative history of the EU Seal Regime is proof that the MRM exception was designed to "fit the reality" of the seal hunt in EU member States.⁵⁷⁶ Canada refers to debates in the Committee on Agriculture and Rural Development of the European Parliament⁵⁷⁷, as well as comments by Sweden and Finland on the need to exempt from the ban seal products deriving from small-scale hunts conducted for marine resource management purposes.⁵⁷⁸ Canada argues that "the requirements that became the [MRM] category and its purpose as described by the EU in this dispute incorporate the exact elements set out by Sweden, i.e. allowing market access for seal products originating from states with 'small scale' 'statutory controlled hunting' with the 'purpose to reduce damages from fisheries', and which is 'done in accordance with a management plan'."⁵⁷⁹

7.351. In this regard, the Panel notes the conclusions of COWI (2010) that seal products from Sweden, Finland, and possibly the United Kingdom would likely qualify under the MRM exception, while seal products from Canada and Norway would not.⁵⁸⁰ In particular, the Report notes that seal hunts in Finland and Sweden do not take place on a "commercial basis" and seal products deriving from these hunts are not placed on the market "in a repetitive way".⁵⁸¹ In addition, most by-products resulting from the hunts are sold "on a private basis" in the local community.⁵⁸² Thus, the non-systematic and non-profit requirements of the MRM exception effectively rule out the eligibility of products from any type of sustainable marine management hunt other than the hunting of individual nuisance seals.⁵⁸³ The conclusions of COWI seem to be corroborated by

⁵⁷⁴ See section 7.2.1.

⁵⁷⁵ The European Union notes that at the time of the dispute, Finland had not submitted any request to have entities enlisted as recognized bodies under Article 6 of the Implementing Regulation.

 ⁵⁷⁶ Canada's first written submission, para. 341 and Section III. C.3.a; response to Panel question
 No. 123 (citing Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10), p. 20 and Parliament Report, (Exhibit JE-4), pp. 70-71 and 73).
 ⁵⁷⁷ Opinion of the European Parliament's Committee on Agriculture and Rural Development in Parliament

 ⁵⁷⁷ Opinion of the European Parliament's Committee on Agriculture and Rural Development in Parliament Report, (Exhibit JE-4), pp. 70-71 and 73.
 ⁵⁷⁸ Comments from EU Member States on Proposal for a Regulation of the European Parliament and of

 ⁵⁷⁸ Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10), pp. 16 and 19.
 ⁵⁷⁹ Canada's comments on the European Union's response to Panel question No. 123 (citing Comments)

⁵⁷⁹ Canada's comments on the European Union's response to Panel question No. 123 (citing Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10), p. 18).

⁵⁸⁰ COWI 2010 Report, pp. 67-68. See also European Union's response to Panel question No. 8; Canada's comments on the European Union's response to Panel question No. 123, para. 59. Norway highlights in its submissions the similarities between the Norwegian and Swedish seal hunts. The Norwegian hunt is statutorily controlled, under the direction of a government agency; quotas are set on an annual basis; few hunters take part in the hunt; permission for participation must be obtained; and the Norwegian authorities monitor the seal hunt closely to ensure conformity with sustainable resource management principles. In addition, the Norwegian hunt is better regulated from the perspective of animal welfare. On this basis, Norway concludes that "[t]he main difference between Sweden and Norway's seal hunt is, therefore, the scale of the Norwegian hunt, which as explained above, does not matter from the perspective of sustainable resource management, so long as the hunt is conducted in accordance with an SRM plan and a TAC established under that plan." (Norway's comments on the European Union's response to Panel question No. 123, paras. 116 -117).

⁵⁸¹ COWI 2010 Report, Annex 4, pp. 2-6.

⁵⁸² COWI 2010 Report, Annex 4, pp. 2-6.

⁵⁸³ Canada's comments on the European Union's response to Panel question No. 123.

the legislative history of the EU Seal Regime, which suggests that the MRM exception was designed with the situation of EU member States in mind. 584

7.352. Therefore, we conclude based on the considerations above that the MRM exception is not designed in an even-handed manner.

7.353. For the above reasons, we find that the MRM exception of the EU Seal Regime is inconsistent with the European Union's obligations under Article 2.1 of the TBT Agreement as the European Union has failed to demonstrate that the detrimental impact caused by the MRM exception under the EU Seal Regime on Canadian imports of seal products stems exclusively from a legitimate regulatory distinction within the meaning of Article 2.1 of the TBT Agreement.

7.3.3 Article 2.2

7.354. Article 2.2 of the TBT Agreement provides:

Members shall ensure that technical regulations are not prepared, adopted or applied with a view to or with the effect of creating unnecessary obstacles to international trade. For this purpose, technical regulations shall not be more trade-restrictive than necessary to fulfil a legitimate objective, taking account of the risks non-fulfilment would create. Such legitimate objectives are, *inter alia*: national security requirements; the prevention of deceptive practices; protection of human health or safety, animal or plant life or health, or the environment. In assessing such risks, relevant elements of consideration are, *inter alia*: available scientific and technical information, related processing technology or intended end-uses of products.

7.355. Article 2.2 can be parsed into several different elements: "legitimate objective"; "fulfilment"; "not ... more trade-restrictive than necessary"; and "taking account of the risks non-fulfilment would create".⁵⁸⁵ Based on these elements, the Appellate Body has described how a panel should assess a claim under Article 2.2 as follows:

[A] panel must assess what a Member seeks to achieve by means of a technical regulation. ... Subsequently, the analysis must turn to the question of whether a particular objective is legitimate ...

In sum, ... an assessment of whether a technical regulation is "more trade-restrictive than necessary" within the meaning of Article 2.2 of the TBT Agreement involves an evaluation of a number of factors. A panel should begin by considering factors that include: (i) the degree of contribution made by the measure to the legitimate objective at issue; (ii) the trade-restrictiveness of the measure; and (iii) the nature of the risks at issue and the gravity of consequences that would arise from non-fulfilment of the objective(s) pursued by the Member through the measure. In most cases, a comparison of the challenged measure and possible alternative measures should be undertaken ... 586

7.356. The Appellate Body stated that all these factors provide the basis for the determination of what is to be considered "necessary" in the sense of Article 2.2 of the TBT Agreement in a

⁵⁸⁴ Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10), pp. 16-19. For instance, Sweden indicated that its preference would be to introduce an exemption for seal products originating from states with "small scale, statutory controlled hunting with the main purpose to reduce damages from fisheries and which is done in accordance with a management plan". (Ibid. p. 18). Finland noted that about 500 seals were hunted yearly in its territory; seals in Finland are not only hunted as pests but are also used as a natural resource for livelihood and as a source of income. According to Finland, prohibiting the possibility for income at the local level would lead to a waste of resources as the hunting would continue without the possibility to make proper use of the seal. (Ibid. p. 16; see also European Parliament Debates, (Exhibit JE-12), pp. 65 and 72).

⁵⁸⁵ Appellate Body Report, US – Tuna II (Mexico), para. 312.

⁵⁸⁶ Appellate Body Report, US - Tuna II (*Mexico*), paras. 314-322. (footnotes omitted) See also Appellate Body Report, US - COOL, para. 374.

particular case.⁵⁸⁷ With this legal framework in mind, we begin our examination of the complainants' claim under Article 2.2 with an inquiry into the objective of the EU Seal Regime.

7.3.3.1 Identification of the objective(s) pursued through the EU Seal Regime

7.3.3.1.1 Main arguments of the parties

7.3.3.1.1.1 Complainants

7.357. The complainants claim that the EU Seal Regime pursues the following six objectives; (i) the protection of animal welfare, including public concerns regarding animal welfare in respect of seals; (ii) the protection of the economic and social interests of Inuit and other indigenous communities engaged in sealing (the IC interests); (iii) the encouragement of the sustainable management of marine resources (the MRM interests); (iv) the prevention of consumer confusion; (v) allowing EU consumers to choose to purchase seal products for their personal use from outside the European Union (the Travellers interests); and (iv) harmonizing the EU internal market with respect to member States' regulations on seal products.⁵⁸⁸

7.358. Furthermore, both complainants claim that the European Union has failed to show that addressing public moral concerns on seal welfare is the objective of the EU Seal Regime.⁵⁸⁹

7.359. Canada argues that the alleged public moral objective is not supported by the text or the design of the measure.⁵⁹⁰ It points out that the EU Seal Regime does not refer to public moral concerns and contends that the EU public concerns regarding animal welfare of seals mentioned in recital (5) of the Basic Regulation are not public moral concerns regarding seal welfare because they do not arise from perceptions as to the rightness or wrongness of specific conduct. Canada argues that concerns regarding seal welfare are gualitatively different from moral concerns on seal welfare.⁵⁹¹ Moreover, Canada maintains that, although the treatment of animals may, for some, be a matter of ethics, this does not amount to a shared view that the public's concern about seal welfare stems from an ethical perspective.⁵⁹² According to Canada, the variety of reasons why members of society oppose a particular activity cannot form the basis of an established community-wide standard of right and wrong conduct.593

7.360. Relying on US – Gambling, Canada submits that for a measure to fall within the scope of measures "necessary to protect public morals", it must be designed to protect the public moral in question by having a "sufficient nexus" with the interest (public moral) to be protected.⁵⁹⁴ In the case of the EU Seal Regime this nexus or rational connection does not exist. This is because, for Canada, the three categories of conditions in the operative parts of the measure are not rationally connected to either animal welfare or protecting EU citizens from the moral harm that would arise from the presence on the EU market of products derived from seals harvested inhumanely. This is due, according to Canada, to the combination of the absence of requirements relating to animal welfare and the market access granted to seal products derived from seals killed inhumanely as permitted under the three conditions set out in the measure.

7.361. Canada asserts, furthermore, that the idea that the EU Seal Regime addresses public moral concerns rests on a false premise that the commercial seal hunt is inherently inhumane.⁵⁹⁵ Any public concerns, whether moral in nature or reflecting some other value, are based on

paras. 268-269; Norway's first written submission, paras. 70-75; second written submission, paras. 167-171. ⁵⁸⁹ Canada's second written submission, paras. 137-165; Norway's opening statement at the first

⁵⁸⁷ See Appellate Body Report, US – Tuna II (Mexico), para. 318.

⁵⁸⁸ Canada's first written submission, paras. 449-450, 452-452*bis*; second written submission,

meeting of the Panel, paras. 76-99. Certain arguments of Canada on the identification of the measure's objectives are contained in its arguments on Article XX(a) of the GATT 1994. The Panel refers to them in this section to the extent that such arguments are also relevant to Canada's arguments under Article 2.2 of the TBT Agreement. ⁵⁹⁰ Canada's response to Panel question No. 31; second written submission, paras. 272-274, 290-295.

⁵⁹¹ Canada's response to Panel question No. 18; second written submission, paras. 13-45, 273.

⁵⁹² Canada's second written submission, para. 158.

⁵⁹³ Canada's second written submission, para. 158.

⁵⁹⁴ Canada's second written submission, paras. 290-295 (citing Panel Report, US – Gambling,

para. 6.455 and Appellate Body Report, *US* – *Gambling*, para. 292). ⁵⁹⁵ Canada's second written submission, para. 159.

misinformation about the seal hunt in Canada; indeed, Canada argues that it has shown that the Canadian seal hunt is not inherently inhumane.⁵⁹⁶

7.362. Canada submits further that the European Union has also failed to establish that there is a single coherent and consistent objective of public morality or even several objectives of public morality that are coherent and consistent.⁵⁹⁷ For Canada, accepting an ill-defined public morality objective effectively allows a Member to adjust *ex post facto* the content of the public morality objective to justify any measure.⁵⁹⁸

7.363. Although Norway does not dispute that, as a general principle, seal welfare could be a moral issue⁵⁹⁹, it argues that the European Union has shown "neither the *existence* of public morals whose protection is necessary through the various elements of [the] EU Seal Regime nor the *specific normative content* of any public morals that purportedly necessitate protection".⁶⁰⁰ The European Union has offered (a) the measure at issue; (b) surveys of the EU public opinion; and (c) scientific evidence⁶⁰¹ as evidence purporting to show the objective of addressing the public moral concerns on seal welfare.⁶⁰² However, according to Norway, the evidence does not support the European Union's assertion. Norway maintains, instead, that the legislative history shows that the particular choices made by the EU legislator were motivated by political expediency and not by public morals⁶⁰³; and the measure itself lays bare the absence of a standard of right and wrong conduct relating to the killing of seals and the sale in the EU market of products containing seal.⁶⁰⁴

7.364. Norway submits further that the public surveys submitted by the European Union do not support the existence of the "public morals" invoked by the European Union.⁶⁰⁵ Instead, the surveys: (a) highlight an extremely low level of knowledge about seal hunting; (b) did not use techniques that would provide information on the moral views of the respondents; and (c) do not even elicit information on the different public morals that the European Union invokes.⁶⁰⁶

7.365. Norway also claims that the scientific evidence invoked by the European Union as "grounds for the public moral concerns"⁶⁰⁷, including the evidence that was before the European Union during the legislative process, does not support the existence of the invoked public morals regarding animal welfare.⁶⁰⁸ To the contrary, the scientific evidence shows that the hunts the products of which are granted market access under the EU Seal Regime pose the greatest animal welfare problems.⁶⁰⁹

⁶⁰¹ Norway's opening statement at the first substantive meeting of the Panel, para. 81.

⁶⁰⁵ Norway's second written submission, paras. 188-200.

⁵⁹⁶ Canada's opening statement at the first substantive meeting of the Panel, para. 12-35; second written submission, paras. 62-82, 159.

⁵⁹⁷ Canada's second written submission, paras. 268-269; 275-289.

⁵⁹⁸ Canada's second written submission, para. 275.

⁵⁹⁹ See Norway's opening statement at the first substantive meeting of the Panel, para. 87; response to Panel question No. 9, para. 72.

⁶⁰⁰ Norway's response to Panel question No. 15, para. 109. (emphasis original) See also Norway's second written submission, paras. 172-208.

⁶⁰² Norway's second written submission, paras. 184-206.

⁶⁰³ Norway's first written submission, para. 616; second written submission, para. 204 (referring to European Parliament Debates, (Exhibit JE-12), p. 72; Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10), pp. 16-18; European Commission Proposal for a Regulation of the European Parliament and of the Council Concerning Trade in Seal Products (23 July 2008), (Commission Proposal), (Exhibit JE-9), Explanatory Memorandum ("Grounds for and objectives of the proposal"), p. 5). See also Norway's first written submission, para. 616.

⁶⁰⁴ Norway's opening statement at the first substantive meeting of the Panel, paras. 78-94; response to Panel question Nos. 14, 18, and 52; second written submission, para. 204.

⁶⁰⁶ Norway's opening statement at the first substantive meeting of the Panel, paras. 97-98; second written submission, paras. 188-200.

⁶⁰⁷ European Union's first written submission, Section 2.4.

⁶⁰⁸ Norway's second written submission, paras. 201-203.

⁶⁰⁹ See Norway's first written submission, paras. 679-684; opening statement at the first substantive meeting of the Panel, paras. 117-121; response to Panel question No. 14, paras. 101-102; response to Panel question No. 73, paras. 402-410. By contrast, COWI noted that "[s]eal hunting is comprehensively regulated in Norway and [Norway] has the most developed management system [for seal hunting]". (COWI 2008 Report, p. 133. See also Ibid. pp. 63 and 70).

7.366. According to Norway, the European Union attempts to place a moral gloss over the objectives it pursues (which, according to Norway, include animal welfare, indigenous communities, and sustainable resource management). In an attempt to present multifarious and competing objectives as a single, coherent objective, the European Union describes them as traits of an umbrella "public morals" objective.⁶¹⁰ Yet, the European Union has not succeeded in identifying any coherent and consistent standard of right and wrong conduct held within the European Union.⁶¹¹ Thus, for Norway, in order to find that EU public morals extend to concerns about the welfare of seals, the European Union would have to convince the Panel of the existence and normative content of such public morals.⁶¹²

7.3.3.1.1.2 Respondent

7.367. The European Union submits that the objective of the EU Seal Regime is "to address the moral concerns of the EU public with regard to the welfare of seals".⁶¹³ Those concerns arise from the fact that seal products may have been obtained from animals killed in a way that causes them excessive pain, distress, fear, or other forms of suffering. Further, the European Union asserts that "contributing to the welfare of seals by reducing the number of seals killed in an inhumane way" can be regarded as being simultaneously a legitimate objective on its own and one of the instruments to achieve the first, broader and overarching, objective of addressing public moral concerns on seal welfare.614

7.368. Further, the European Union contests that it pursues the objectives identified by the complainants. Contrary to the complainants' arguments, the EU Seal Regime does not pursue consumer choice and the prevention of consumer confusion.⁶¹⁵ The European Union asserts that the EU Seal Regime seeks to uphold a rule of public morality, equally applicable with regard to all members of the EU public, irrespective of their personal beliefs. The complainants have thus misunderstood the objective of the Travellers exception, which is not to "promote the personal choice" of the EU consumers, but to avoid inequitable results in the operation of the measure.⁶¹⁶ Further, the IC and MRM exceptions under the measure do not pursue independent objectives from those sought by the general ban; the concerns relating to those exceptions are articulations of the same "standard of morality".⁶¹⁷ The EU Seal Regime allows the placing on the market of seal products under the IC and MRM exceptions because products qualifying for those exceptions do not raise the same moral concerns as products from commercial seal hunts.⁶¹⁸

7.369. The European Union argues that in assessing the moral implications of seal hunting, therefore, it is essential to take into account, together with the welfare of the seals, the purpose of each type of hunt.⁶¹⁹ Specifically, some hunts are conducted primarily for commercial purposes, such as obtaining skins for manufacturing inessential clothing items.⁶²⁰ The EU public regards seal products from commercial hunts as morally objectionable and is repelled by their availability in the EU market.⁶²¹ In contrast, in the cases of other seal hunts conducted for non-commercial purposes, such as the subsistence of indigenous communities or the sustainable management of

⁶¹⁰ Norway's opening statement at the first substantive meeting of the Panel, paras. 69, 76-99; second

written submission, para. 170. ⁶¹¹ Norway's opening statement at the first substantive meeting of the Panel, paras. 76-99; Norway's response to Panel question Nos. 9, 15, 17, 18, and 48.

⁶¹² Norway's response to Panel question No. 109. Norway specifically contends that the European Union has failed to prove that its "umbrella moral norm" allows commerce in six distinct circumstances irrespective of animal welfare, namely where products are: (i) derived from IC hunts; (ii) derived from MRM hunts; (iii) imported for personal use by travellers; (iv) sold at auction in the European Union for re-export; (v) used in EU manufacturing for export under inward processing rules; or (vi) where they transit the European Union. See Norway's response to Panel question No. 18; second written submission, para. 172 and footnote 232; and opening statement at the second substantive meeting of the Panel, para. 54.

 $^{^{13}}$ European Union's first written submission, para. 33; response to Panel question No. 10.

 $^{^{\}rm 614}$ European Union's response to Panel question No. 10.

⁶¹⁵ European Union's first written submission, para. 43 (referring to Norway's first written submission, para, 600).

⁶¹⁶ European Union's first written submission, para. 42.

⁶¹⁷ European Union's response to Panel question No. 10; opening statement at the second meeting of the Panel, paras. 43-51.

⁶¹⁸ European Union's first written submission, para. 43.

⁶¹⁹ European Union's first written submission, para. 38.

⁶²⁰ European Union's first written submission, para. 39.

⁶²¹ European Union's first written submission, para. 36.

marine resources, it may be justified, or even required, from a moral point of view to tolerate a higher level of risk to the welfare of seals.⁶²²

7.370. To demonstrate further the existence of moral concerns on the welfare of seals among the EU public, the European Union points to a number of public opinion surveys. The European Union alleges that these survey results show the public's general feeling against seal hunting, particularly commercial hunting, and support for the ban.⁶²³

7.371. As for the objective of harmonizing the EU internal market, described by the European Union as the immediate objective of the EU Seal Regime, the European Union asserts that the elements challenged by the complainants are not necessary to achieve that objective but rather seek to address the moral concerns of the EU public with respect to the welfare of seals.⁶²⁴

7.3.3.1.2 Analysis by the Panel

7.372. As explained above, to assess a measure's consistency with the obligations under Article 2.2 of the TBT Agreement, the Panel must first identify the objective pursued by a regulating Member through the measure at issue.

7.373. The parties in this dispute disagree on what objectives the European Union aims to fulfil through the EU Seal Regime.

7.374. The European Union claims that the EU Seal Regime pursues two closely related objectives⁶²⁵:

- first, addressing the moral concerns of the EU population with regard to the welfare of seals ("addressing the public moral concerns on seal welfare"); and
- second, contributing to the welfare of seals by reducing the number of seals killed in an inhumane way.

7.375. Regarding the first objective, the European Union points to two aspects of the public's moral concerns: first, "the incidence of inhumane killing of seals"; and, second, EU citizens' "individual and collective participation as consumers in, and exposure to, the economic activity which sustains the market for commercially-hunted seal products".⁶²⁶ Further, according to the European Union, the second objective can also be regarded as one of the instruments to achieve the first, broader and overarching, objective of addressing the public moral concerns on seal welfare.

7.376. The complainants assert that the European Union failed to identify any coherent and consistent standard of right and wrong conduct held within the European Union with respect to seal welfare concerns. Rather, the European Union pursues a multiplicity of objectives through the measure such as protection of seal welfare, including the EU public's concerns on seal welfare; protection of the IC interests; and promotion of marine resource management.

⁶²² European Union's first written submission, para. 39.

⁶²³ European Union's first written submission, paras. 190-194 (referring to public opinion surveys of EU member States, (Exhibits EU-48-59). The European Union submits that the EU public moral concerns with regard to the killing of seals are deep and longstanding. Those concerns emerged in the 1950s, became widespread among the public by the 1960s, which in turn led to the introduction of various import restrictions by EU member States during the 1970s and 1980s. The European Union states that this culminated with the adoption of the 1983 Seal Pups Directive banning products made of whitecoat harp and blueback hooded seal pups. (Seal Pups Directive, (Exhibit CDA-12)).

⁶²⁴ See European Union's response to Panel questions No. 13 and 113, para. 53. The complainants have also addressed the objective of internal market harmonization. (See Canada's response to Panel question No. 13 ("Canada is of the view that the harmonization objective is qualitatively different from the other objectives, and should not be considered for the purposes of Article 2.2 of the TBT Agreement."); Norway's response to Panel question No. 13 ("Norway agrees that that the "harmonization of the EU internal market" is distinct from the policy objectives of the type covered by Article 2.2 of the TBT Agreement, and that the Panel should not consider this as a policy objective within the meaning of that provision.").

⁶²⁵ European Union's response to Panel question No. 9.

⁶²⁶ European Union's response to Panel question No. 9.

7.377. We consider that the disagreements between the parties on the objective of the EU Seal Regime come down, in essence, to two issues. First, while not disputing that the EU Seal Regime is aimed at addressing the public concerns on seal welfare, the parties disagree on whether the public concerns at issue are moral concerns for the EU public. Second, the parties also contest whether other interests addressed through the exceptions in the measure (i.e. IC, MRM, and Travellers exceptions) constitute objectives of the EU Seal Regime that are separate and independent from the objective of addressing seal welfare concerns.

7.378. In these circumstances, and based on guidance provided by the Appellate Body, we must assess the objectives of the EU Seal Regime on the basis of available evidence such as the texts of statutes, legislative history, and other evidence regarding the structure and operation of the measure at issue.⁶²⁷ We are also mindful that a panel is *not* bound by the objectives asserted by the regulating Member, and "may also find guidance" in contrary evidence proffered by the complainant in determining the objective pursued by the regulating Member.⁶²⁸

7.379. Before moving to examine these aspects, however, we find it useful to recall the meaning and scope of "public morals" as discussed in previous disputes.

7.380. In US – Gambling, the panel considered the term "public morals" as used in Article XIV of the GATS, which provides, "... measures: (a) necessary to protect public morals or to maintain public order". The panel observed that the term "public morals" denotes "standards of right and wrong conduct maintained by or on behalf of a community or a nation" and that the content of the concept of public morals "can vary in time and space, depending upon a range of factors, including prevailing social, cultural, ethical and religious values".⁶²⁹ For this reason, it considered that:

Members should be given some scope to define and apply for themselves the concepts of "public morals" and "public order" in their respective territories, according to their own systems and scales of values.630

7.381. The panel in *China – Audiovisual Products* agreed with the interpretation of the term "public morals" developed by the panel in US - Gambling for the purposes of its analysis under Article XX(a) of the GATT 1994.⁶³¹ In a similar vein to Article XIV of the GATS, Article XX of the GATT also refers to "measures ... (a) necessary to protect public morals". The assessment of the scope of "public morals" in the context of the general exception provisions of the GATT 1994 and the GATS referring to public morals therefore suggests that WTO Members are afforded a certain degree of discretion in defining the scope of "public morals" with respect to various values prevailing in their societies at a given time.

7.382. Although previous interpretations of the notion of "public morals" were developed in the context of examining general exceptions provisions under the GATT 1994 and the GATS, we consider that they are equally applicable in identifying a regulating Member's alleged moral objective in the context of Article 2.2 of the TBT Agreement. We observe in this regard the Appellate Body's statement in US - Clove Cigarettes that the second recital in the preamble of the TBT Agreement indicates that "the TBT Agreement expands on pre-existing GATT disciplines and

⁶²⁷ Appellate Body Report, US – Tuna II (Mexico), para. 314. See also Appellate Body Reports, EC – Sardines, paras. 276-282; US - COOL, para. 371; Panel Reports, US - Tuna II (Mexico), para. 7.405 (citing Appellate Body Report, US – Gambling, para. 304); and US – COOL, para 7.612 (citing an observation by the Appellate Body in Japan - Alcoholic Beverages that "Members of the WTO are free to pursue their own domestic goals through internal taxation or regulation so long as they do not do so in a way that violates ... any of the other commitments they have made in the WTO Agreement").

⁶²⁸ Appellate Body Report, US – Gambling, para. 304.

⁶²⁹ Panel Report, US - Gambling, paras. 6.465 and 6.461. ⁶³⁰ Panel Report, US - Gambling, paras. 6.461. In US - Gambling, for example, the panel assessed whether the gambling problem addressed by the measure falls within the scope of the "public morals" within the meaning of Article XIV(a) of the GATS. In doing so, the panel considered inter alia other members' trade measures with similar objectives as shown in trade policy review reports and secondary interpretative materials such as decisions by other jurisdictions on similar issue (e.g. the ECJ). Based on such resources, the panel found that gambling is a matter that could fall within the scope of public morals. Panel Report, US -*Gambling*, paras. 6.470-6.474. ⁶³¹ Panel Report, *China – Audiovisual Products*, para. 7.759. The public morals at issue in that dispute

concerned the protection of the public from potential negative impacts caused by cultural goods (e.g. reading materials and audiovisual products).

emphasizes that the two Agreements should be interpreted in a coherent and consistent manner". 632

7.383. Accordingly, the question of whether a measure aims to address public morals relating to a particular concern in the society of a regulating Member requires, in our view, an assessment of two issues: first, whether the concern in question indeed exists in that society; and, second, whether such concern falls within the scope of "public morals" as "defined and applied" by a regulating Member "in its territory, according to its own systems and scales of values".

7.384. In the factual circumstances of the present dispute, and bearing in mind the guidance of the Appellate Body referred to above as to how to assess the objective of a measure, we must therefore examine whether the evidence as a whole shows (a) the existence of the EU public's concerns on seal welfare and/or any other concerns or issues that the European Union seeks to address; and, (b) the connection between such concerns, if proven to exist, and the "public morals" (i.e. standards of right or wrong) as defined and applied within the European Union.

7.385. We begin our assessment with the text of the EU Seal Regime.

7.386. At the outset, we note that the EU Seal Regime does not have a specific section setting forth the objective of the EU Seal Regime. The preamble of the Basic Regulation, comprising 21 recitals, describes various concerns and observations on seal hunting and seal products. For example, the preamble refers to *inter alia* the following:

- legislative initiatives calling for a ban on trade in seal products and cruel seal hunting methods (recital 1);
- existing consumer confusion over products derived from seals and other products (recitals 3, 7);
- concerns by the public and governments regarding the pain, distress, fear, and other forms of suffering which the killing and skinning of seals cause to seals (recitals 4, 5, 10, 11);
- several EU member States' adoption or intention to adopt legislation regulating trade in seal products and differences between national provisions governing trade in seal products (recitals 5, 6);
- different trade regulations on seal products within the EU internal market and the need to harmonize such regulations (recitals 6, 7, 8, 10);
- obligation to "pay full regard to" the animal welfare requirements when formulating an EU internal market policy (recital 9);
- public concerns about the possible presence on the market of products obtained from animals killed and skinned in a way that causes pain, distress, fear and other forms of suffering (recitals 5, 10);
- design of the current measure in harmonizing the EU internal market and addressing animal welfare concerns (recitals 10, 12, 13);
- difficulty in consistent verification and control of hunters' compliance with animal welfare requirements in seal hunting (recital 11);
- fundamental economic and social interests of Inuit communities engaged in seal hunting (recital 14); and
- three conditions (IC, MRM, and personal use) according to which the placing and/or import of seal products on the EU market would be allowed (recital 17).

⁶³² Appellate Body Report, US – Clove Cigarettes, para. 91. See also Ibid. paras. 90, 96, and 101.

7.387. Based on our review of the preamble, the Basic Regulation appears to address three main considerations: first, the need to harmonize the regulations on seal products within the EU internal market (recitals 5, 6, 7, 8, 10, 15, 21); second, concerns about seal welfare issues (recitals 1, 4, 5, 10, 11); and, third, the need to preserve the economic and social interests of Inuit communities engaged in seal hunting and to define the conditions for IC, MRM, and Travellers exceptions (recitals 16 and 17).

7.388. The operative part of the Basic Regulation, discussed in section 7.2.1 above, reflects these main considerations by providing for the IC, MRM, and Travellers exceptions and prohibiting seal products derived from any other seal hunts. Specific rules for implementing the exceptions and other obligations under the Basic Regulations are contained in the Implementing Regulation.⁶³³

7.389. Overall, we can discern from the text of the EU Seal Regime that, in designing the Regime, the European Union sought to address the public concerns on seal welfare. In doing so, the European Union also took into account certain other interests (i.e. IC, MRM, and Travellers interests). Although we have also observed references to the need for harmonizing the EU internal market, we recall the acknowledgment of both the complainants and the respondent that, for the purpose of this dispute, it is not necessary for the Panel to address this particular consideration.⁶³⁴

7.390. Next, we proceed to examine the legislative history of the EU Seal Regime to assess whether it sheds further light on the measure's objective.

7.391. The initiative to introduce a measure governing trade in seal products in the European Union originates in the 2006 "Declaration of the European Parliament on banning seal products in the European Union" (Parliament Declaration).⁶³⁵ In its preamble, the Parliament Declaration references inter alia an observation that a certain proportion of seals hunted may have been skinned while still conscious.⁶³⁶ Although the Parliament Declaration does not explicitly elaborate on the specific reasons for the initiative for a ban on seal products, a list of points contained in the preamble suggest a connection between the Declaration and seal welfare.⁶³⁷ It also states that "this regulation should not have an impact on traditional Inuit seal hunting"⁶³⁸; no reference is made, however, to MRM hunts or Travellers' interests.

7.392. Subsequent to the Parliament Declaration, the Parliamentary Assembly of the Council of Europe made certain observations on seal hunting:

The Assembly welcomes the Declaration of 26 September 2006 by the European 5. Parliament on banning seal products in the European Union ...

...

The Assembly is aware that the international controversy surrounding seal 8. hunting is first and foremost a political debate, bringing different and sometimes

⁶³³ We note that while providing practical details necessary for the enforcement of the Basic Regulation, the Implementing Regulation does not in itself assist us in identifying the objective of the measure.

⁶³⁴ See para, 7,371 above.

⁶³⁵ Parliament Declaration, (Exhibit JE-19). We note that issues relating to sealing and seal products have been subject to discussions and debates within the European Union preceding the adoption of the EU Seal Regime. For instance, we take note of the adoption in 1983 of a ban on the importation of products derived from certain seal pups. (Seal Pups Directive, (Exhibit CDA-12)). The Council's Parliamentary Assembly also refers to its Recommendation 825 (1978) on the protection of wildlife and on seal hunting. (Council of Europe, Parliamentary Assembly, recommendation 1776 (2006) of 17 November 2006 on seal hunting, (Council of Europe Recommendation), (Exhibit EU-117), para. 1). In addition, the European Union refers to the adoption of several restrictions of seal products by various current EU member States dating back to 1970. (See European Union's first written submission, para. 615 (referring to certificates issued by Greenlandic authorities prior to obtaining recognized body status, (Exhibit EU-162)). 636 Parliament Declaration, (Exhibit JE-19), point D.

⁶³⁷ For instance, the reference to the skinning of conscious seals occurs after recitals noting the large number of harp seal pups "slaughtered" in the North West Atlantic (point A) and impacts on seal population from such a large scale of killing (point B). Moreover, the EU Parliament observes that the 1983 Directive banning seal pup products did not extend to products of older seals being targeted by sealers (point E).

³⁸ Parliament Declaration, (Exhibit JE-19), point H(2).

conflicting values, objectives and attitudes into play, and that public opinion is particularly sensitive to this matter.

9. The Assembly notes that during the last decade, the cruelty of seal hunting has been documented by videos from several authoritative television channels as well as by the personal observations of many members of national and European parliaments, scientists, celebrities and representatives of non-governmental organizations (NGOs). Such cruelty has generated a public morality debate in Europe ...

7.393. We note in particular that the observations guoted above refer to the existence of inter alia concerns on seal welfare and a "public morality debate" regarding seal hunting in Europe.

7.394. In 2008, the European Commission submitted a proposal for a regulation concerning trade in seal products (Commission Proposal). 640 This proposal refers to the public concerns about "the animal welfare aspects of the killing and skinning of seals and about trade occurring in products possibly derived from seals that have been killed and skinned with avoidable pain, distress and other forms of suffering ...".⁶⁴¹ The Impact Assessment accompanying the Commission Proposal⁶⁴² also describes the main overarching objectives of the initiative as "[p]rotect seals from acts that cause them avoidable pain, distress, fear and other forms of suffering during the killing and skinning process", and to "[a]ddress the concerns of the general public with regard to the killing and skinning of seals".⁶⁴³ Thus, these documents, which inform us of the legislative process leading to the adoption of the current measure, make explicit references to the public concerns on seal welfare.

7.395. Further, we observe that the Commission Proposal describes public concerns as relating to "ethical" considerations. For example, the following expressions are used in the Proposal: "out of ethical reasons"; "citizens' deep indignation and repulsion regarding the trade in seal products in such conditions"; "the public's growing awareness and sensitivity to ethical considerations in how seal products are obtained".⁶⁴⁴ More particularly, the Commission's 2008 proposal states:

For several years, many members of the public have been concerned about the animal welfare aspects of the killing and skinning of seals and about trade occurring in products possibly derived from seals that have been killed and skinned with avoidable pain, distress and other forms of suffering, which seals, as sentient mammals, are capable of experiencing. Those concerns have therefore been expressed by members of the public out of ethical reasons. The Commission received during the last years a massive number of letters and petitions on the issue expressing citizens' deep indignation and repulsion regarding the trade in seal products in such conditions.⁶⁴⁵

⁶³⁹ Council of Europe Recommendation, (Exhibit EU-117).

⁶⁴⁰ Commission Proposal, (Exhibit JE-9). The Commission Proposal was presented to the European Parliament and Council in 2008 following the Commission's undertaking to assess the animal welfare aspects of seal hunting and provide possible legislative proposals in response to the Parliament Declaration. (See Commission Impact Assessment, (Exhibit JE-16), p. 8).

⁶⁴¹ Commission Proposal, (Exhibit JE-9), pp. 2-3 ("grounds for and objectives of the proposal"). See also Ibid. p. 8 (stating that the Proposal "focuses on animal welfare considerations while [other existing Community legislation) addresses conservations issues") and p. 11 (referring to "the animal welfare concerns expressed by the public" in respect of suitable measures to address the issue).

Norway also acknowledges based on its own assessment of the Commission Proposal and Commission Impact Assessment that "protecting animal welfare and addressing public concerns relating to animal welfare" are described as "overarching objectives". (Norway's first written submission, paras. 612-617 (referring to Commission Impact Assessment, (Exhibit JE-16), p. 7). It nevertheless also points to the protection of IC interests as well as internal market harmonization as the objectives emerging from the European Commission documents.

⁶⁴² Commission Impact Assessment, (Exhibit JE-16). The Impact Assessment provides a more detailed description of the assessment process, including the subjects reviewed and legislative options canvassed by the EU Commission in the preparation of its Proposal.

⁶⁴³ Commission Impact Assessment, (Exhibit JE-16), p. 23. The document also establishes the specific objectives based on the main overarching objectives, namely (i) to "[e]nsure consistency and legal clarity in terms of the requirements for placing seal products on the EU market" and (ii) to "[p]romote and reward good sealing practices". ⁶⁴⁴ Commission Proposal, Explanatory memorandum, (Exhibit JE-9), pp. 2-3.

⁶⁴⁵ Commission Proposal, Explanatory memorandum, (Exhibit JE-9), p. 2.

- 118 -

...

The Treaty establishing the European Community does not provide for a specific legal basis allowing the Community to legislate *in the field of ethics as such*. However, where the Treaty empowers the Community to legislate in certain areas and that the specific conditions of those legal bases are met, the mere circumstance that the Community legislature relies on *ethical considerations* does not prevent it from adopting legislative measures.⁶⁴⁶ (emphasis added)

7.396. In our view, therefore, the references above to "ethical considerations" in the Commission Proposal, combined with the reference to a "public morality debate" in the Council of Europe Recommendation, provide evidence that the public concerns about seal welfare constitute a moral issue for EU citizens.⁶⁴⁷

7.397. The evidence also includes documents demonstrating various EU member States' views on the Commission Proposal as well as the measure in its current form.⁶⁴⁸ Based on our examination of these documents, we observe that certain EU member States expressed doubts as to various features of the proposed measure and that the "difficulties of balancing different views"⁶⁴⁹ had to be overcome in the legislative process.⁶⁵⁰ However, a majority of comments and statements documented in these exhibits show an overall support for the measure in light of the wishes of EU citizens to ban seal products from the EU market based on their concerns on seal welfare.⁶⁵¹

7.398. As with the text of the EU Seal Regime, the legislative history of the measure demonstrates the existence of the EU public's concerns about seal welfare. We observe in this

⁶⁵⁰ See Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10), p. 16 (view of Finland that certain welfare requirements in the Commission Proposal were "not necessary" for the specific manner of seal hunting in Finland) and p. 18 (preference of Sweden "to introduce a second exemption possibility for seal products originating from states with small scale, statutory controlled hunting with the main purpose to reduce damages from fisheries and which is done in accordance with a management plan"). See also Council of the European Union, Member States' Comments on the Proposal for a Regulation Concerning Trade in Seal Products (20 July 2009), (Exhibit JE-11), p. 1 (opinion of Denmark that "trade in seal products as a whole is a legitimate activity, which should not be unnecessarily hampered and stigmatised").

We also observe certain opposing views during the Parliamentary debates. (See European Parliament Debates, (Exhibit JE-12), p. 64 ("we are solving nothing at all in terms of seal hunting. We are simply relocating the problem ... you are not banning seal hunting. You are potentially relocating the problem to China or to other countries, which will be able to accept these products"); p. 68 ("This proposal destroys people's lives and their communities in remote regions. It destroys business opportunities on both sides of the Atlantic and seriously harms good relations ... "; and p. 69 (" ... on behalf of Greenland, which is part of the Kingdom of Denmark. There are a few tiny, remote settlements in the far north with a population of just 10-20 people who live from hunting seals. If we take away their livelihood, they will have no chance of economic survival")).

⁶⁵¹ Specifically, the Parliamentary debates indicate that European citizens have consistently expressed concerns on seal welfare and their wish to have seal products banned from the market through numerous letters and e-mails. (See, e.g. European Parliament Debates, (Exhibit JE-12), pp. 61-62 ("we will have respected the wishes of many citizens in many of our countries across the EU who tell us that they do not like what they see of the commercial seal hunt and that they wish to have no association with the trade that results from that hunt. We have respected that wish: we have dealt solely with what we can deal with within the confines of Europe's internal market: the circulation of goods in the market that arise from the commercial hunt ... our consumers should be assured that nothing from the commercial hunt will be sold on Europe's market"); pp. 62-63 ("seal hunting and the way it is carried out has resulted in the expression of serious reaction and concern by the public ..."); p. 65 ("The public have demonstrated in numerous polls across European countries that they want an end to the trade..."); p. 67 ("We are meeting the wishes of the many citizens who have asked us in countless letters and e-mails to take action in this area"); p. 67 ("many EU citizens do not, and they support a total ban on the import of seal fur skins ...")).

⁶⁴⁶ Commission Proposal, Explanatory memorandum, (Exhibit JE-9), p. 3.

⁶⁴⁷ We recall in this regard the view of the panel in US - Gambling that public morals can "depend[] on a range of factors, including prevailing social, cultural, *ethical* and religious values". (Panel Report, US - Gambling, para. 6.461) (emphasis added)

⁶⁴⁸ See, e.g. Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10); Council of the European Union, Member States' Comments on the Proposal for a Regulation Concerning Trade in Seal Products (20 July 2009), (Exhibit JE-11); and European Parliament Debates, (Exhibit JE-12).

⁶⁴⁹ European Parliament Debates, (Exhibit JE-12), p. 64. The Parliamentary debates contain repeated references to the "compromise" reflected in the text of the draft legislative proposal. (See Parliament Report, (Exhibit JE-4)).

regard that the EU public's concerns on seal welfare found in the evidence are related to seal hunting in general, and not to any particular type of hunting.652 The public survey results submitted by the European Union are also informative, although to a limited extent, in demonstrating the EU public concerns.653

7.399. Furthermore, on the basis of a plain reading of the text of the Basic Regulation, and in the light of the evidence mentioned above, we conclude that, in drawing up the measure, the European Union accommodated other interests or considerations, such as the Inuit communities engaged in seal hunting, seals hunted for marine management purposes, and seal products brought into the European Union for personal use. As mentioned above, however, the parties contest whether these interests should be considered as separate objectives of the EU Seal Regime independent from the objective of addressing seal welfare concerns.

7.400. The Appellate Body in US - Clove Cigarettes observed that "measures, such as technical regulations, may have more than one objective".⁶⁵⁴ Thus, it may not be uncommon for a measure to have "a multiplicity of objectives".⁶⁵⁵ Hence there is no reason in principle why the measure at issue could not have several objectives.656

7.401. However, based on its text and legislative history as well as its structure and design, we are not convinced that the "aim", "target", or "goal" of the EU Seal Regime was to protect the interests of IC, MRM, and Travellers.⁶⁵⁷ Based on the text of the EU Seal Regime, we found that the Regime consists of a ban, albeit formulated in positive terms, and exceptions. Next, the structure and design of the measure - a ban with exceptions - establish that it operates as a prohibition against seal products, unless they meet specific conditions prescribed in the measure. The legislative history described above further shows, in our view, that the principal objective of adopting a regulation on trade in seal products was to address public concerns on seal welfare. Specifically, we noticed references in certain evidence relating to the measure's legislative history

⁶⁵² See, for instance, COWI 2008 Report, pp. 124-127. A public consultation conducted regarding their view on regulation of seal hunting, which COWI cautions should be reviewed with the limited representativeness of respondents, shows that "62.1% [of the respondents] state that seals should not be hunted for any reason", whereas "17.6% state that hunting is most acceptable when the hunter belongs to a traditional seal hunting culture/community or depend on seal hunt for his main income."

⁶⁵³ See Public survey results, (Exhibits EU-48-59). The public survey results submitted by the European Union exhibit the existence of a certain level of public awareness and concern on seal welfare in the European Union and other countries. However, their probative value is limited as the surveys, considered alone, are insufficient to establish the existence of the EU public moral concerns on seal welfare as argued by the European Union.

Notably, some of these surveys were conducted with respect to citizens of countries other than those belonging to the European Union. Therefore, such surveys do not provide sufficient factual guidance in the assessment of public morals in the European Union.

We do note that the European Union has also presented evidence of surveys carried out in various countries which are member States of the European Union. (See Public survey results from Germany, the Netherlands, Portugal, Slovenia, Belgium, France, Austria, Sweden, and the Czech Republic, (Exhibits EU-50-57)). As argued by Canada, however, we observe that the public surveys (with the exception of the survey in Sweden) do not solicit views on whether it is acceptable to exempt seal products obtained from Inuit hunts and resource management-related culls from a prohibition on the sale of seal products. (See Canada's second written submission, para 93. See also Norway's second written submission, paras. 188 and 200). We further note that the survey companies' responses cited by the European Union do not rebut this specific claim put forth by both complainants. (See Exhibit EU-131 A, B, and C; European Union's opening statement at the second meeting of the Panel, para. 51).

 ⁶⁵⁴ Appellate Body Report, US – Clove Cigarettes, paras. 113 and 115.
 ⁶⁵⁵ Appellate Body Report, US – Clove Cigarettes, paras. 113 and 115.

⁶⁵⁶ Colombia has submitted that, although Members may try to address more than one policy concern in a technical regulation, "it seems rather difficult to accept that it would be legitimate to enact a measure with one main objective and then impose exceptions to that measure that ... undermine the principal policy objective". (Colombia's third-party submission, paras. 5-7, 32; third-party statement, paras. 6 and 15). Colombia thus maintains that the important legal issue in the alleged breach of provisions of the TBT Agreement is "the relationship that exists between [different policy objectives]" of a technical regulation. (Colombia's third-party submission, para. 23).

⁶⁵⁷ The Appellate Body describes the word "objective" as a "thing aimed at or sought; a target, a goal, an aim" (Appellate Body Report, US - Tuna II (Mexico), para. 313, (citing Shorter Oxford English Dictionary, 6th edn, A. Stevenson (ed.) (Oxford University Press, 2007), Vol. 2, p. 1970)).

that the interests of Inuit and indigenous communities engaged in seal hunting should be protected from possible trade regulations on seal products.⁶⁵⁸

7.402. For us, the interests that were accommodated in the measure through the exceptions must be distinguished from the main objective of the measure as a whole. Further, unlike the issue of seal welfare, we do not find in the evidence submitted that the interests covered by the IC, MRM, and Travellers exceptions are grounded in the concerns of EU citizens. Rather, the evidence suggests that they appear to have been included in the course of the legislative process. For all these reasons, we do not consider that the interests incorporated in the IC, MRM, and Travellers exceptions form independent policy objectives of the EU Seal Regime as a whole.⁶⁵⁹

7.403. In paragraph 7.383 above, we explained that to determine the objective of the EU Seal Regime, we must examine whether the evidence as a whole shows (a) the existence of the EU public's concerns on seal welfare and/or any other concerns or issues that the European Union seeks to address; and, (b) the connection between such concerns, if proven to exist, and the "public morals" (i.e. standards of right or wrong) as defined and applied within the European Union.

7.404. As regards the first question, we have concluded that the text and legislative history of the measure established the existence of the EU public's concerns on seal welfare. We therefore proceed to examine the second question, namely, whether the concerns at issue fall within the scope of "public morals" in the European Union. In this connection, given our determination that IC and MRM interests do not constitute objectives of the EU Seal Regime, we find it unnecessary to determine whether such interests are "articulations of the same standard of morality" governing the public concerns on seal welfare as claimed by the European Union.⁶⁶⁰ Thus, our task is confined to assessing whether the public concerns on seal welfare are anchored in the morality of European societies.

7.405. We found that the legislative history of the measure suggests a link between the public concerns on seal welfare and an ethical or moral consideration.⁶⁶¹ With that in mind, we now turn to other evidence before us. The European Union refers to various other sources, which in its view establish that the public concerns on seal welfare are indeed a moral issue within the European Union as a community.⁶⁶² Specifically, the European Union refers to the various actions

⁶⁶⁰ The European Union submits that "the rule of morality" invoked by it requires balancing in each case the welfare of the animals concerned and other interests. (European Union's opening statement at the second meeting of the Panel, para. 44; response to Panel question No. 104).

⁶⁶¹ See para. 7.396 above.

⁶⁶² The European Union argues that the way in which humans treat animals is a matter of public morals, and humans are not free to treat and use animals as they wish, but ought instead to conform to certain moral standards of right and wrong. (European Union's first written submission, para. 61). The European Union

⁶⁵⁸ For example, the 1983 ban on products of certain seal pups refers to "non-traditional hunting" and notes in its preamble that "the exploitation of seals ... and in certain areas of the world forms an important part of the traditional way of life and economy; whereas hunting, as traditionally practised by the Inuit people, leaves seal pups unharmed and it is therefore appropriate to see that the interests of the Inuit people are not affected". (Seal Pups Directive, (Exhibit CDA-12)). The legislative history of the EU Seal Regime shows that exceptions for Inuit and indigenous communities from the regulation on seal products, albeit varying in scope, were consistently a consideration. The European Commission's proposed regulation, subsequent comments on the regulation, and the parliament's proposal also indicate references to exempting imports of occasional nature and for personal use. (See Commission Proposal, (Exhibit JE-9), p. 5; Parliament Report, (Exhibit JE-4), p. 10 (Justification for Amendment 8).

An exception for "communities dependent on artisanal fishing and which contribute to their subsistence or to the regulated and controlled management of seal populations with a view to mitigating the damage occasioned to fish stocks" appears for the first time in the Committee on Agriculture and Rural Development's proposed amendments to the Commission's proposed regulation. (Parliament Report, (Exhibit JE-4), p. 66, Amendment 18). We observe that this suggestion was not included in the parliament's proposal.

⁶⁵⁹ We note that, in *Brazil – Retreaded Tyres*, while the measure concerned was a prohibition on the importation of retreaded tyres, certain imports were exceptionally allowed (e.g. imports from MERCOSUR partners or subject to domestic court ruling). Such derogations from the general ban in that dispute did not form part of the measure itself; rather they were made effective through the application of the measure. Under the circumstances of that dispute, Brazil identified the protection of environment as the objective of the measure (namely an import ban) under Article XX(b) of the GATT 1994, but never raised, for instance, the reasons behind such derogations as the "policy objective" of the measure *per se*. Moreover, the panel in that dispute addressed the situations involving the derogation from the ban under the chapeau of Article XX of the GATT 1994 (i.e. arbitrary or unjustifiable discrimination among imported products).

taken by the European Union as well as EU member States concerning animal protection in general; various pieces of legislation and Conventions on animal welfare within the European Union and in other countries, including Norway and Canada; and various international instruments. We examine these materials in turn.

7.406. "The Treaty of Lisbon amending the Treaty on European Union and the Treaty establishing the European Community (the Treaty of Lisbon)" stipulates that "the Member States shall, since animals are sentient beings, pay full regard to the welfare requirements of animals".⁶⁶³ The European Union asserts that, consistent with the mandate in the Treaty of Lisbon, the European Union adopted a comprehensive body of legislation on the welfare of farm animals within the framework of its Common Agricultural Policy.⁶⁶⁴ Further, the European Union submits that although the protection of the welfare of wild animals and pets falls, in principle, within the scope of the competence of the EU member States, the EU legislators have in certain cases deemed it necessary to take protective action also with regard to such animals, including the measure in question in this dispute.⁶⁶⁵ The European Union also points out that EU legislation on animal welfare is based to a large extent on a series of Conventions⁶⁶⁶ elaborated since the 1960s at the Council of Europe, which were the first international instruments laying down comprehensive ethical rules for the use of animals.

7.407. The European Union also refers to EU member States' animal protection laws based on public morals considerations, as well as evidence of the moral objectives of measures taken with respect to seal welfare. For example, the European Union highlights laws in Austria (Federal Act which in §1 expressly aims at "the protection of the life and well-being of animals based on man's special responsibility for the animal as a fellow creature") and the United Kingdom (Animal Welfare

claims that the first law on animal welfare was enacted in 1822 in the United Kingdom. (European Union's first written submission, para. 62 and Peter Sandøe and Stine B. Christiansen, *Ethics of Animal Use* (2008), (Exhibit EU-1); Austria's Federal Act on the Protection of Animals, (Exhibit EU-2); United Kingdom's Animal Welfare Act (2006), (Exhibit EU-3)). ⁶⁶³ Article 13 of the Treaty on the Functioning of the European Union provides: "In formulating and

⁶⁶³ Article 13 of the Treaty on the Functioning of the European Union provides: "In formulating and implementing the Union's agriculture, fisheries, transport, internal market, research and technological development and space policies, the Union and the Member States shall, since animals are sentient beings, pay full regard to the welfare requirements of animals, while respecting the legislative or administrative provisions and customs of the Member States relating in particular to religious rites, cultural traditions and regional heritage." (European Union's first written submission, para. 63, footnote 28)

The European Union explains that Article 13 of the Treaty on the Functioning of the European Union replaces and reproduces, with some slight changes, the content of the Protocol on Animal Welfare annexed to the Treaty Establishing the European Community by the Treaty of Amsterdam, which entered into force on 1 January 1999.

1 January 1999. ⁶⁶⁴ See the legislative references provided in the Parliament Report, (Exhibit JE-4). For instance, regarding the protection of animals at the time of slaughter and killing, the exhibit references the following EU regulations: Council Directive 93/119/EC of 22 December 1993 on the protection of animals at the time of slaughter or killing; Council Regulation (EC) No 1099/2009 of 24 September 2009 on the protection of animals at the time of killing; European Convention for the protection of animals for slaughter (Official Journal L 137, 02/06/1988 p. 0027-0038); and 88/306/EEC (Council Decision of 16 May 1988 on the conclusion of the European Convention for the Protection of Animals for Slaughter Official Journal L 137).

The European Union refers in particular to its amended regulation on the protection of animals at the time of killing, which asserts the status of animal welfare as "a Community value" and that "protection of animals at the time of slaughter or killing is a matter of public concern". (See European Union's first written submission, para. 64 (referring to Council Regulation (EC) No 1099/2009 of 24 September 2009 on the protection of animals at the time of killing, Official Journal of the European Union, L303/1, 18 November 2009, (Exhibit CDA-31))).

⁶⁶⁵ Such cases also include the 1983 Seal Pups Directive, (Exhibit CDA-12); Council Regulation (EEC) No 3254/91, of 4 November 1991, prohibiting the use of leghold traps, Official Journal of the European Union, L Series, No. 308 (9 November 1991) (Exhibit EU-5); and Regulation (EC) No 1523/2007 of the European Parliament and the Council, of 11 December 2007, banning the placing on the market and the import to, or export from, the Community of cat and dog fur, and products containing such fur, Official Journal of the European Union, L Series, No. 343 (27 December 2007), (Exhibit EU-6).

⁶⁶⁶ European Union's first written submission, para. 66, footnote 36. The Council of Europe has drawn up five conventions for the protection of animals: (a) European Convention for the protection of animals during transport, Paris, 13 December 1968, E.T.S. No 65; (b) European Convention for the protection of animals kept for farming purposes, Strasbourg, 10 March, 1976, E.T.S. No 87; (c) European Convention for the protection of animals kept animals for slaughter, Strasbourg, 10 May 1979, E.T.S. No 102; (d) European Convention for the protection of vertebrate animals used for experimental and other scientific purposes, Strasbourg, 18 March, 1986, E.T.S. No 123; and (e) European Convention for the protection of pet animals, Strasbourg, 13 November 1986, E.T.S. No 125. http://conventions.coe.int

Act of 2006 with provisions for the prevention of harm to and promotion of welfare of animals); and, references a moral aspect of the public concerns on seal welfare found in Belgian and Dutch material, such as "au nom de la morale publique"; "outrage"; and "an offense to public order and decency in this country".⁶⁶⁷

7.408. Further, the European Union refers to recommendations of the Office International des Epizooties (OIE) (Guiding Principles for Animal Welfare)⁶⁶⁸, certain other WTO Members' measures on seal products based on moral grounds (e.g. Chinese Taipei⁶⁶⁹; Russia⁶⁷⁰; and Switzerland⁶⁷¹), as well as the "philosophy of animal welfarism" and its connection to "a long-established tradition of moral thought".⁶⁷²

7.409. The evidence presented by the European Union, taken as a whole, illustrates in our view "standards of right and wrong conduct maintained by or on behalf of [the European Union]" concerning seal welfare. In reaching this conclusion, we observe that ascertaining the precise content and scope of morality in a given society may not be an easy task.⁶⁷³ As the panel in US - Gambling stated, we are mindful that Members should be given some scope to define and apply for themselves the concepts of "public morals" in their respective territories, according to their own systems and scales of values. Although not all evidence presented to us makes an explicit link between seal or animal welfare and the morals of the EU public, we are nevertheless persuaded that the evidence as a whole sufficiently demonstrates that animal welfare is an issue of ethical or moral nature in the European Union. International doctrines and measures of a similar nature in other WTO Members, while not necessarily relevant to identifying the European Union's chosen objective, illustrate that animal welfare is a matter of ethical responsibility for human beings in general.⁶⁷⁴

⁶⁷⁰ The European Union cites statements of Russian officials reported in the press and Russia's response to the Panel's question that the purpose of its hunting ban is for population concerns "and to protect wild animals and baby animals from hard treatment". (See European Union's second written submission, para. 165 (citing the Russian Federation's third-party response to Panel question No. 16)).

⁶⁷¹ The European Union cites portions of the Swiss legislative proposal characterizing the seal hunt as
 "extremely cruel". (European Union's second written submission, paras. 169-170).
 ⁶⁷² The European Union refers to the philosophy of "animal welfarism" and its connection to "a long-

⁶⁷² The European Union refers to the philosophy of "animal welfarism" and its connection to "a longestablished tradition of moral thought", which is discussed in an *amicus curiae* submission filed by Howse et al. (European Union's second written submission, para. 140; response to Panel question No. 10).

⁶⁷³ We take note of the following statements by third parties on this question.

Japan states that "while Members have the right to determine whether a specific objective forms part of 'public morals' on the basis of their own prevailing social, cultural, ethical and religious values, ... Members are not free to label any policy objective as forming part of public morality." (Japan's third-party submission, para. 11).

para. 11). ⁶⁷⁴ For example, we observe references to a link between animal welfare and ethical considerations in the complainants' government documents relating to seal hunting. (See Norway's Fisheries Directorate, Proposal to amend the rules on seal hunting (2010), (Exhibit EU-45), p. 1 (explaining regulatory initiatives in Norway for "measures that would be compatible with today's animal welfare requirements and ethical standards")).

⁶⁶⁷ European Union's first written submission, para. 62 (referring to the comments of one of the sponsors of the proposal leading to the Belgian ban on seal products and an explanatory memorandum accompanying the Dutch ban). See Loi relative a l'interdiction de fabriquer et de commercialiser des produits dérivés de phoques, 16 March 2007, (Exhibit EU-110); Chambre des députés de Belgique, minutes of the session of 25 January 2007, (Exhibit EU-111); Bulletin of Acts, Orders and Decrees of the Kingdom of the Netherlands, (Exhibit EU-112).

⁶⁶⁸ European Union's first written submission, para. 71; second written submission, para. 152-153; response to Panel question No. 46. One such "guiding principle" cited by the European Union is that "the use of *animals* carries with it an ethical responsibility to ensure the *welfare* of such *animals* to the greatest extent practicable". OIE's Terrestrial Animal Code, Chapter 7.1, (Exhibit EU-116) (emphasis original)

⁶⁶⁹ The European Union has submitted the proposal leading to the ban in Chinese Taipei, which states that "the hunting process has proven to be extremely brutal and inhumane" and alludes to the contribution of consumers to the slaughter of "innocent seals" (European Union's second written submission, paras. 161-164 (referring to Chinese Taipei Parliament, YZ-No 1749, Committee proposal of bill No. 13359, (Exhibit EU-126)).
⁶⁷⁰ The European Union cites statements of Russian officials reported in the press and Russia's response

Iceland states that the public morals exception "is bound to be based on more subjective and less tangible arguments", but public morals should not be equated with "broad political and public support for a measure" and that there must be "actual public moral concerns" to justify a trade restriction". (Iceland's third-party submission, paras. 14 and 16-17).

See also Norway's Ministry of Agriculture and Food, Parliamentary Report No. 12 Regarding animal husbandry and animal welfare, (Exhibit EU-114)). This report was the product of an initiative by the Norwegian Ministry of Agriculture published in 2002 as a white paper and "constitutes a broad review and evaluation of all

- 123 -

7.410. In sum, based on our examination of the text and legislative history of the EU Seal Regime, as well as other evidence pertaining to its design, structure, and operation, we conclude that the objective of the EU Seal Regime is "to address the moral concerns of the EU public with regard to the welfare of seals". Specifically, these concerns have two aspects as claimed by the European Union: (a) "the incidence of inhumane killing of seals"; and, (b) EU citizens' "individual and collective participation as consumers in, and exposure to ('abetting'), the economic activity which sustains the market for" seal products derived from inhumane hunts.⁶⁷⁵ Further, in light of the evidence before us, we note that the EU public concerns on seal welfare appear to be related to seal hunts in general, not any particular type of seal hunts.⁶⁷⁶ Therefore, the second aspect of the objective of the EU Seal Regime pertains to seal products derived from inhumane hunts rather than "commercially-hunted seal products" as submitted by the European Union.

7.411. In this connection, we note that contributing to the welfare of seals by reducing the number of seals killed in an inhumane way, which the European Union claims can constitute the second objective of the measure, is closely linked to the first aspect of the public moral concerns, namely the moral concerns on "the incidence of inhumane killing of seals". We will therefore consider this as one aspect of the moral concerns in question, rather than as a separate, second objective of the measure.

animal care in Norway from an *ethical* and animal welfare perspective. It also includes proposals for long-term goals and actions, hereunder an *ethical* platform". (Norway's Ministry of Agriculture and Food, "Norwegian Action Plan on Animal Welfare", (Exhibit EU-115), p. 1 (emphasis added) Notably, the white paper identifies a "widespread moral view in Norway today [that] takes elements from utilitarianism and rights philosophy", and specifically links such moral considerations to the treatment and slaughter of animals. (Norway's Ministry of Agriculture and Food, Parliamentary Report No. 12 Regarding animal husbandry and animal welfare, (Exhibit EU-114), p. 12. See also DFO Integrated Seals Management Plan 2011-2015, (Exhibit EU-42), p. 17 (describing "public attitudes toward the seal hunt" based on the results of a national survey)).

We also take note of the United States' comment that "while the focus must be on the responding Member's system and scale of values, what Members other than the responding Member consider to be public morals can offer confirmation of a panel's determination as to what constitutes a public moral within the system of the responding Member." (United States' third-party submission, para. 4).

⁶⁷⁵ For ease of reference, we will use in these Reports the phrase "addressing public moral concerns on seal welfare" as shorthand for the specific objective as identified and described in paras. 7.324-7.325 including the two specific aspects of addressing such public moral concerns.

⁶⁷⁶ We observe the European Union's reference to Canada's Royal Commission Report on sealing (Exhibit EU-48) to support its position that some of the opinion polls confirm that "the public do value very differently the various types of seal hunt." (European Union's response to Panel question No. 31). The Report shows that in two of five polls reviewed in the Report, the public showed a greater level of acceptance for killing animals if it is carried out for a person's survival or livelihood than if it is carried out for making a profit. (Report of the Royal Commission on Seals and Sealing (1986), Volume I, Chapter 11, "Public Opinion on Sealing", (Exhibit EU-48), p. 160). The Report, however, points out two uncertainties about this result: first, the question in the polls refers to "animals" and not "seals"; and, second, the term "livelihood" is not clearly defined and would mean something equivalent to subsistence (i.e. food or clothing) for some people and providing a cash income to others. It also mentions the much lower percentage of people willing to accept seal hunting by Inuit and other local communities to provide cash, and particularly to provide cash to enable them to undertake the hunting and fishing essential for their survival, than people willing to accept seal hunting by Inuit and other local communities to personally use the hunted seals. This suggests, according to the Report, that the public has very little understanding of the socio-economic realities by which these communities survive.

We consider that the results of the public opinion polls referenced in Canada's Royal Commission Report have limited probative value because the reliability of the results of these surveys, including the formulation of the questions asked and the target audience, has not been clearly demonstrated to us.

Further, we have taken note of an observation regarding the public perception of seal hunting in the COWI 2008 Report that the survey respondents showed some ambivalence on the issue of concerns over animal welfare versus concerns over the well-being of local communities (58% stating that animal welfare is equally important as well-being of local communities) and that the results of this question in relation to the question on the acceptability of different uses of seal products showed the complexity of the issue. We also note that 62% of the survey respondents in the COWI 2008 Report stated that seals should not be hunted for any reason, whereas 18% stated that hunting is most acceptable when the hunter belongs to a traditional seal hunting culture/community or depends on the seal hunt for his main income. (COWI 2008 Report, p. 126).

Overall, the results shown in two polls referenced in the Royal Commission Report, in our view, are not sufficient to establish a variance in the EU public's concerns on seal welfare depending on the type of hunts.

7.3.3.2 Legitimacy of the identified objective

7.3.3.2.1 Main arguments of the parties

7.412. Canada argues that the European Union's defined objective is *not* legitimate because it is based on an arbitrary and unjustifiable distinction between "commercial" and "non-commercial" hunts. It argues further that the EU Seal Regime, under the IC and the MRM categories, contains trade-restrictive requirements with no rational connection to the objectives of animal welfare and public concerns regarding animal welfare, which in fact undermine those objectives.⁶⁷⁷

7.413. Norway would accept that addressing a public moral relating to seal welfare (without the other peculiar contours of the "umbrella" public moral alleged by the European Union) is "legitimate". Norway argues, however, that a *distinction* based on the alleged "commercial" and "non-commercial" seal hunts is not legitimate. The objective of the IC exception is not legitimate either because it discriminates in favour of particular communities.⁶⁷⁸

7.414. The European Union claims that its stated objective of addressing the EU public moral concerns on seal welfare is legitimate because the public authorities must define and enforce certain moral standards to which humans must conform in treating and using animals.⁶⁷⁹

7.3.3.2.2 Analysis by the Panel

7.415. We now examine the question of whether the identified objective of the EU Seal Regime, i.e. "addressing EU public moral concerns on seal welfare", is legitimate within the meaning of Article 2.2 of the TBT Agreement. This policy objective is not included in the non-exhaustive list of exemplary legitimate objectives in Article 2.2. Therefore, its legitimacy must be assessed based on several reference points.

7.416. The Appellate Body considered that the following would be relevant factors in assessing the legitimacy of a non-listed objective: (a) objectives provided in Article 2.2 of the TBT Agreement; (b) objectives listed in the sixth and seventh recitals of the preamble of the TBT Agreement; and (c) objectives recognized in other provisions of the covered agreements.⁶⁸⁰

7.417. Article 2.2 of the TBT Agreement lays out the following objectives as examples of legitimate objectives: "national security requirements"; "the prevention of deceptive practices"; "protection of human health or safety, animal or plant life or health, or the environment". Further, the fifth and sixth recitals of the preamble recognize the following objectives: "to ensure the quality of its exports"; "for the protection of human, animal or plant life or health, of the environment"; "for the prevention of deceptive practices"; and "for the protection of its essential security interest". "Public morals" as such is thus not included in the lists provided in Article 2.2 and the fifth and sixth recitals of the preamble of the TBT Agreement.

7.418. With respect to policy objectives in other covered agreements, the objective of protecting public morals is recognized in both Article XX of the GATT 1994 and Article XIV of the GATS. The explicit inclusion of "public morals" as one of the general exceptions for a GATT- or GATS-inconsistent measure demonstrates that WTO Members considered this objective to be particularly significant. In light of this, and considered together with the objective of the TBT Agreement to further the objectives of the GATT 1994 as referenced in recital (2) of its preamble, we conclude that "public morals" falls within the scope of "legitimate" objectives under Article 2.2.

7.419. Proceeding on that basis, a subsequent question is whether addressing public moral concerns on a specific type of issue, seal welfare in this dispute, is a legitimate objective under Article 2.2 of the TBT Agreement. As discussed above, the concept of public morals is a relative term which needs to be defined based on the standard of right and wrong in a given society. Given that the European Union has established that the concerns of the EU public on animal welfare involve standards of right and wrong within the European Union as a community, we consider that

⁶⁷⁷ Canada's second written submission, paras. 296-303.

⁶⁷⁸ Norway's response to Panel question No. 109; opening statement at the first meeting of the Panel, paras. 100-111.

⁶⁷⁹ European Union's first written submission, paras. 61 and 354.

⁶⁸⁰ Appellate Body Reports, US – Tuna II (Mexico), para. 313; US – COOL, para. 370.

addressing the public moral concerns on seal welfare, identified as the objective of the measure at issue, is "legitimate" under Article 2.2 of the TBT Agreement.

7.420. In support of its position that the stated objective is legitimate, the European Union also refers to various measures on animal welfare and seal products adopted by other WTO Members as well as international instruments on animal welfare.⁶⁸¹ We need not determine whether these examples as such exhibit the existence of a global social norm ("a universal value" according to the European Union) on animal welfare in general or seal welfare in particular. Nevertheless, these various actions concerning animal welfare at the international as well as national levels suggest in our view that animal welfare is a globally recognized issue. This further supports our conclusion above that the objective of addressing the public moral concerns on seal welfare falls within the scope of legitimate objectives within the meaning of Article 2.2 of the TBT Agreement.

7.421. Finally, we observe that the complainants do not dispute that "addressing the public moral concerns on seal welfare" *per se* is legitimate within the meaning of Article 2.2 of the TBT Agreement. Both complainants dispute a specific aspect of the EU Seal Regime – a distinction between allegedly "commercial" and "non-commercial" seal hunts – which they claim renders the objective illegitimate. In our view, the complainants' position with respect to this particular distinction is not relevant to addressing the question of whether the identified objective is legitimate within the meaning of Article 2.2 of the TBT Agreement. We discussed the questions raised by the complainants regarding the distinction between "commercial" and "non-commercial" seal hunts in the context of our examination of Canada's claim under Article 2.1 of the TBT Agreement.⁶⁸²

7.3.3.3 Whether the EU Seal Regime is more trade restrictive than necessary to fulfil a legitimate objective, taking account of the risks non-fulfilment would create

7.422. As noted above, an examination of the obligations under Article 2.2 requires a relational analysis of all of the following elements: (a) trade-restrictiveness of the EU Seal Regime; (b) degree of the measure's contribution to the identified objective; and (c) availability of alternative measures. We have identified the objective of the EU Seal Regime as addressing EU public moral concerns on seal welfare. In this section, we examine, based on the elements highlighted above, whether the EU Seal Regime is more trade restrictive than necessary to fulfil the identified objective, taking account of the risks non-fulfilment would create.

7.3.3.3.1 Whether the EU Seal Regime is trade restrictive

7.3.3.3.1.1 Main arguments of the parties

7.423. The complainants submit that the EU Seal Regime is trade restrictive because it imposes limiting conditions or restrictions on imports into the European Union and placing on the EU market of seal products.⁶⁸³ If a given seal product does not satisfy the cumulative conditions of one of the sets of requirements, the seal product is prohibited on the EU market. The EU Seal Regime is therefore, by nature, trade restrictive because it lays down regulatory conditions limiting the importation and sale of seal products. Further, in practice, the mere expectation of the adoption of the EU Seal Regime hampered trade.⁶⁸⁴

⁶⁸¹ European Union's first written submission, paras. 67-76.

⁶⁸² See section 7.3.2.3 above.

⁶⁸³ Canada's first written submission, paras. 472-476; Norway's first written submission, paras. 662-672.

⁶⁸⁴ Norway's first written submission, paras. 673-674 (referring to a COWI briefing note of 2009): At the same time [as the financial crisis,] the current legislation has been in the pipeline and has created uncertainty about the EU market. Hence, trade numbers are down substantially since 2007 and so is the market price of raw skin (less than half) ... Some European markets [for seal oil], Sweden, Denmark, Finland and Germany were emerging but halted in recent years due to the Regulation. (COWI, Traceability systems for trade in seal products – Briefing note for workshop participants, 20 October 2009, pp. 16 and 19 in COWI 2010 Report, Annex 5)

Norway also refers to another COWI warning (before the Commission tabled its original proposal) that "[s]ince seal hunting mostly takes place outside the Community territory, any restrictions to market access in the Community will have trade impacts". (COWI 2008 Report, p. 102.)

7.424. The European Union does not dispute that the EU Seal Regime restricts trade to the extent that it imposes a prohibition on the placing on the EU market of all seal products. The ban aspect of the measure aims at being very trade-restrictive, consistently with the high level of fulfilment sought by the EU Seal Regime of its policy objective.⁶⁸⁵ The European Union contends however that unlike the ban, the three exceptions to that prohibition are not trade-restrictive.⁶⁸⁶ To the contrary, they allow trade which would otherwise be prohibited by the ban.

7.3.3.3.1.2 Analysis by the Panel

7.425. "Trade-restrictive" aspects of the measure mean the aspects of the measure that "hav[e] a limiting effect on trade"687, or that constitute "limiting condition[s]" on trade.688 The Appellate Body in US - Tuna II (Mexico) referred to this meaning as similar to that in the context of Article XI of the GATT 1994, under which the Appellate Body noted that the term "prohibition" is defined as "a legal ban on the trade or importation of a specified commodity".689

7.426. Given the broad scope of "trade restriction", we believe that the EU Seal Regime, considered in its entirety, is trade restrictive because it does "hav[e] a limiting effect on trade" by accessing the EU market. The European Union itself acknowledges that the measure "aims at being very trade-restrictive".⁶⁹⁰ prohibiting certain seal products, including those imported from Canada and Norway, from

7.427. In this connection, we disagree with the European Union's position that the exceptions under the Regime do not need to be "justified" under Article 2.2 because it is not trade restrictive. As explained in the context of assessing the measure's qualification as a technical regulation, both the ban and exceptions under the EU Seal Regime define the scope of products that are prohibited and allowed on the EU market. The proper assessment of the EU Seal Regime's operation as a technical regulation, including its trade-restrictiveness, is thus based on an examination of the Regime as a whole.

7.428. We next turn to examine the extent to which the measure fulfils the objective of addressing the EU public moral concerns on seal welfare.

7.3.3.3.2 Degree of the EU Seal Regime's fulfilment of the identified objective

7.3.3.3.2.1 Main arguments of the parties

Complainants

7.429. Canada argues that the absence of animal welfare requirements in the conditions of the IC or MRM categories demonstrates that the level of fulfilment was not a high level of fulfilment approaching complete fulfilment of the objective.⁶⁹¹ Rather the design, structure, and operation of the measure indicate that a low level of fulfilment of the animal welfare and alleged public moral concern regarding animal welfare is acceptable to the EU public. Consequently, the level of fulfilment sought, as articulated through the design, structure and expected operation of the measure is also low.

⁶⁸⁵ European Union's first written submission, paras. 357-358.

⁶⁸⁶ European Union's first written submission, paras. 357-358; second written submission,

paras. 269-270. The European Union submits that the three exceptions could only be regarded as being trade restrictive if they discriminated in favour of domestic seal products or between different sources of imports. However, all and each of the three exceptions are consistent with Article 2.1 of the TBT Agreement. Particularly, the Travellers exception benefits exclusively imports of all origins.

According to the European Union, as the exceptions are not trade-restrictive, they do not require justification under Article 2.2. The exceptions could be relevant for the analysis under Article 2.2 only to the extent that they detracted from the contribution made by the ban to the objectives of the EU Seal Regime. ⁶⁸⁷ Appellate Body Reports, US – Tuna II (Mexico), para. 319; US – COOL, para. 375.

⁶⁸⁸ Appellate Body Report, China – Raw Materials, para. 319; Panel Report, Colombia – Ports of Entry, paras. 7.232-7.241. ⁶⁸⁹ Appellate Body Report, US – Tuna II (Mexico), para. 319 (citing Appellate Body Report, China – Raw

Materials, para. 319). ⁶⁹⁰ European Union's first written submission , paras. 357 and 586.

⁶⁹¹ Canada's second written submission, paras. 310-316.

- 127 -

7.430. Norway submits that the EU Seal Regime does not contribute to the welfare of seals, whether it is considered as a distinct objective or as a component of the alleged "public morals".⁶⁹² First, the EU Seal Regime does not condition market access on compliance with animal welfare requirements.⁶⁹³ Second, the measure does not "contribut[e] to the welfare of seals by reducing the number of seals killed in an inhumane way"⁶⁹⁴ because, once the EU Seal Regime is fully implemented, eligible seal products from Greenland will match or exceed the total size of the EU market prior to the ban.⁶⁹⁵ Thus, rather than reducing the overall quantity of seal products available within the European Union, the EU Seal Regime simply reduces the list of countries from which such products may be sourced.

7.431. Third, according to Norway, the EU Seal Regime does not "shield ... the EU public from being confronted"⁶⁹⁶ with seal products, including seal products derived from "an immoral act (the killing of seals in an inhumane way)".⁶⁹⁷ Consumers are not even informed of the fact that the products in question contain seal, let alone of whether the seals were caught humanely.⁶⁹⁸ Fourth, the EU Seal Regime does not prohibit the "commercial exploitation" of seal products "within the EU territory"⁶⁹⁹: seal products may be placed on the EU market or imported regardless of compliance with animal welfare requirements; seal products eligible for market access under the IC or MRM requirements may be hunted for commercial purposes; and, the EU Seal Regime does not restrict transit across the European Union, processing for export in the European Union under an inward processing procedure, production for export, or sale at auction houses for export, for any seal product irrespective of the type of hunt. In other words, EU citizens are allowed to participate in, and earn money from, the commercial exploitation of seal resources.

7.432. Fifth, with regard to the alleged moral dimension of the objective pursued, the measure does not contribute to the preservation or safeguard ("protection") of morals, which Article XX(a) of the GATT 1994 (being relevant context where "public morals" are invoked as an objective for purposes of Article 2.2 of the TBT Agreement) shows is the relevant public morals objective.⁷⁰⁰ The relevant public moral – if it exists – will not be threatened by making seal products available on shop shelves. Hence, a ban is not needed to "protect" that moral – it would remain anyway. If a public moral will not be threatened by trade, a trade ban cannot be justified simply by the need to avoid certain negative "feelings" on the part of consumers. In that regard, Norway recalls that these "feelings" on the part of EU consumers could already be engendered under the EU Seal Regime for consumers who feel strongly about seal welfare irrespective of the type of hunt, because seal products can be sold on the EU market.⁷⁰¹

⁶⁹² Norway's second written submission, paras. 231-238.

⁶⁹³ Norway's responses to Panel questions Nos. 14, 69, and 72.

⁶⁹⁴ European Union's response to Panel question No. 10 (describing the alleged objectives of the neasure).

measure). ⁶⁹⁵ Norway's response to Panel question No. 41. As explained in that response, based on Eurostat data analysed by COWI, the total size of the EU market in 2006 was approximately 110,000 skins. (COWI 2008 Report, p. 106, Table 5.2.3, total of the figures under "Import to EU-27" (including intra-community trade)). In 2006, for example, 109,201 seal skins were traded in Greenland. (Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 27, Table 3). Export figures for 2004 are even higher, with 115,723 skins sold, of which 71% were sold in the European Union, even in the presence of competition from supplies from non-Greenlandic sources. (Ibid. p. 28, Table 4)

See also Norway's first written submission, para. 690 (citing P. Fitzgerald, "'Morality' May Not Be Enough to Justify the EU Seal Products Ban: Animal Welfare Meets International Trade Law", *Journal of International Wildlife Law and Policy* (2011), Vol. 14, pp. 85-136, (Exhibit NOR-86), p. 128).

⁶⁹⁶ European Union's response to Panel question No. 10.

⁶⁹⁷ European Union's response to Panel question No. 10.

⁶⁹⁸ Norway's first written submission, paras. 705-716; Norway's opening statement at the first meeting of the Panel, paras. 88, 135.

⁶⁹⁹ European Union's response to Panel question No. 10, para. 42.

⁷⁰⁰ Norway refers to Oxford English Dictionary, OED Online, Oxford University Press, accessed 27 March 2013, http://www.oed.com/view/Entry/153127?redirectedFrom=protect&, *protect*.

⁷⁰¹ Norway notes that, under the EU Seal Regime, many consumers may not experience such "feelings" simply on the basis that they do not know whether a product they are consuming contains seal, since no labelling is required under the EU Seal Regime.

- 128 -

Respondent

7.433. The European Union submits that through the ban, the EU Seal Regime makes a very substantial contribution to its policy objective in two ways.⁷⁰²

7.434. First, the ban provides a direct response to the moral concerns of the EU population by prohibiting, as a general rule, the marketing within the EU territory of the products which the EU population regards as morally abhorrent.

7.435. Second, by limiting the global demand for seal products, the ban reduces the number of seals which are killed every year in a manner that may cause them excessive suffering, thereby contributing to the welfare of seals. There are clear indications that the EU Seal Regime, and the bans of the EU member States which preceded it, have had a significant impact on the global demand for products resulting from commercial hunts and, consequently, on the number of seals killed inhumanely every year. The volume of catches declined considerably in both Canada and Norway after 2006, coinciding with the introduction of the first EU member States' bans.⁷⁰³ Exports from Canada declined even more drastically after 2006.⁷⁰⁴ While they have recovered slightly in the last two years, they remain far below the levels reached during the last decade.⁷⁰⁵

7.436. Further, as regards the exceptions under the measure, the European Union asserts that they are not trade restrictive and thus do not have to be justified under Article 2.2. Instead, it is the restrictive effects of the ban which need to be justified under that provision. By focusing exclusively on the conditions attached to the exceptions, the complaining parties seek to draw the Panel's attention away from the obvious fact that the ban does make a major contribution to the achievement of the objective pursued by the EU Seal Regime.

7.437. In any event, the exceptions do not undermine the objective of the EU Seal Regime because they are based on moral grounds; the products falling within the scope of the IC and the MRM exceptions do not raise the same moral concerns as other types of seal products because the suffering inflicted upon seals is outweighed by the benefits to humans (or other animals). Hence the marketing of products complying with those exceptions is allowed. Thus, the fact that the ban is subject to exceptions does not prevent it from making a substantial contribution to its public moral objective in the first of the two ways described above. While the contribution of the EU Seal Regime to the welfare of seals could have been even greater in the absence of any exceptions, this does not mean that the ban aspect of the measure makes no contribution at all to the welfare of seals.

7.438. Further, both the Travellers exception and the MRM exception have been very narrowly defined and apply to a very small volume of trade: as recognized by the complainants, the trade impact of the Travellers exception is "minuscule"⁷⁰⁶; and, the scope of the MRM exception is very limited and thus the trade potentially concerned very small.⁷⁰⁷

7.439. Although the IC exception has, potentially, a broader scope of application than the other two exceptions, the complainants' allegations that, as a result of the IC exception, exports from Greenland will replace exports from Canada and Norway (hence the global demand for seal products and thus the number of seals killed inhumanely will remain unaffected), are speculative and implausible.⁷⁰⁸ The IC exception does not seek to promote exports from Greenland, but

⁷⁰² European Union's first written submission, paras. 359-366; second written submission,

paras. 277-300. See also European Union's first written submission, paras. 360-365 (referring to Canada's first written submission, paras. 480-496, and Norway's first written submission, paras. 677-704).

⁷⁰³ DFO Commercial Seal Harvest Overview 2011, (Exhibit JE-27), tables 2, 8, 9, and 10. See also *amicus curiae* submission by Anima et al., pp. 61-62 (Exhibit EU-81); complainants' responses to the Panel guestion Nos. 96 and 99.

⁷⁰⁴ DFO Commercial Seal Harvest Overview 2011, (Exhibit JE-27), tables 3-7 and 12-15.

⁷⁰⁵ The European Union submits that the Canadian Government has recognized that the EU Seal Regime has had "significant negative impacts on Canada's sealing industry". (Canada's first written submission, paras. 80-81).

⁷⁰⁶ Canada's first written submission, para. 286.

⁷⁰⁷ Currently the MRM exception is available only with regard to seals hunted in Sweden. In 2011 only 86 seals were hunted in Sweden. (European Union's response to the Panel's question No. 96).

⁷⁰⁸ See Canada's first written submission, paras. 487-488; Norway's first written submission, para. 683. The European Union argues that such allegation relies on little else than an unsupported assertion by an

instead to mitigate the necessarily adverse impact of the EU Seal Regime on the Inuit and other indigenous populations to the extent compatible with the animal welfare objectives of the EU Seal Regime.⁷⁰⁹ The complainants also assume that any products obtained from seals hunted by any member of an indigenous community will necessarily qualify for the IC exception, which is not the case because there are specific conditions to be met.⁷¹⁰

7.440. Finally, even if the Panel were to conclude that the EU Seal Regime makes no contribution to its public moral objective in the first of the ways described above, it is beyond question that the EU Seal Regime would still make a material contribution to its public moral objective in the second way.

7.3.3.3.2.2 Analysis by the Panel

7.441. The Appellate Body stated that the question of whether a technical regulation "fulfils a legitimate objective" in Article 2.2 is concerned with the *degree* of contribution that the technical regulation makes toward the achievement of the legitimate objective. The "degree of contribution of the measure" to the fulfilment of the legitimate objective is in turn revealed through the measure itself⁷¹¹ and "may be discerned from the design, structure, and operation of thetechnical regulation, as well as from evidence relating to the application of the measure".⁷¹² The relevant question is thus the degree of *actual* contribution that the technical regulation, as written and applied, makes to the fulfilment of a legitimate objective(s).⁷¹³

7.442. We must therefore assess the degree of the EU Seal Regime's actual contribution to the fulfilment of its stated objective. In this regard, we recall that the EU Seal Regime as a whole, consisting of both the prohibitive and permissive aspects, was found to constitute a technical regulation within the meaning of the TBT Agreement.⁷¹⁴ It is thus the EU Seal Regime in its entirety that must be assessed for its contribution to the objective, and not just a particular element of it.

7.443. The objective of the EU Seal Regime is to address the public moral concerns on seal welfare in respect of two aspects: first, the EU public's participation as consumers in and exposure to products derived from seals killed inhumanely; and, second, the overall number of seals killed

individual member of European Parliament (who was the rapporteur for the IMCO Committee during Parliamentary deliberations). As explained above, her views were overwhelmingly rejected by the IMCO Committee and, eventually, by the European Parliament.

⁷⁰⁹ The European Union argues that the complainants also overlook that a large part of the seals hunted in Greenland (in some years more than 50%) is used for subsistence purposes and not traded. (See Management and Utilization of Seals in Greenland, (Exhibit JE-26), p. 27).

⁷¹⁰ The European Union argues that the complainants base this assumption on an opinion casually expressed in the COWI 2010 Report. (See COWI 2010 Report, Annex 5, p. 17). According to the European Union, COWI had neither the authority, nor the qualifications nor the mandate to engage in the legal interpretation of the Basic Regulation. The European Union further submits that the Implementing Regulation, which specifies the conditions for qualifying for the IC exception, had not even been adopted at the time when the COWI Report was issued.

We however note that the purpose of the COWI Report was to assist the European Commission in preparing the specific regulations implementing the rules set out in the Basic Regulation. The COWI 2010 Report states in its Executive Summary:

"COWI A/S has been contracted by the European Commission, DG Environment, to undertake a Study on implementing measures for trade in seal products, which provides input to the Commission's process of developing implementing measures for [the Basic] Regulation. Therefore the results of this present study are providing input to the development of a suitable traceability scheme that can ensure that the conditions stipulated in the Regulations are met while defining the implementing rules. The overall purpose of the study is to provide the Commission with additional information in order to draft implementing measures in terms of traceability schemes in accordance with the Regulation on trade in seal products." (COWI 2010 Report, p.iii).

⁷¹¹ Appellate Body Report, US – Tuna II (Mexico), para. 317.

⁷¹² Appellate Body Reports, *US – Tuna II (Mexico)*, para. 317; *US – COOL*, para. 373. See also Appellate Body Reports, *Brazil – Retreaded Tyres*, para. 151 (indicating that contribution to the objective of a measure "could consist of quantitative projections in the future, or qualitative reasoning based on a set of hypotheses that are tested and supported by sufficient evidence"); *EC – Asbestos*, para. 167.

 713 Appellate Body Report, US - Tuna II (*Mexico*), paras. 317-318. The Appellate Body recalled that as in other situations, such as, for instance, when determining the contribution of a measure to the achievement of a particular objective in the context of Article XX of the GATT 1994, a panel must assess the contribution to the legitimate objective *actually achieved* by the measure at issue.

⁷¹⁴ See section 7.3.1 above.

inhumanely. We assess the degree of the measure's actual contribution to these two aspects in turn. We then make an overall assessment of the measure's contribution to the objective as a whole.

7.444. Regarding the first aspect of the objective, the European Union argues that the measure makes a contribution to its objective in the following manner: the ban ensures that "the EU population does not render itself accomplice to the inhumane killing of seals in the commercial hunts and is not confronted with the products resulting from such immoral killing".⁷¹⁵

7.445. Based on the evidence before us, we found that the EU public concerns on seal welfare pertained to seal hunting in general; available evidence did not establish that the EU public concerns were linked exclusively to commercial seal hunts or that the EU public found poor animal welfare outcomes in IC and MRM hunts morally justifiable as suggested by the European Union. Therefore, the degree of the measure's actual achievement of the first aspect of the objective must be gauged against whether the measure ensures that the EU citizens do not participate as consumers in products derived from seals killed inhumanely regardless of the type of the hunt.

7.446. As noted above, virtually all of the seal products derived from the hunts in Canada and Norway are denied access to the EU market.⁷¹⁶ To the extent that seals derived from such hunts include seals killed inhumanely, the ban prevents the EU public from purchasing products derived from seals killed inhumanely in Canadian and Norwegian hunts.

7.447. As examined above, however, the animal welfare risks arising from seal hunts in general also exist in IC and MRM hunts, and thus seals may also be killed inhumanely in such hunts. Given that products are allowed on the market under the IC and MRM exceptions regardless of whether they derive from seals killed humanely, these exceptions, based on their design, are incapable of contributing to preventing consumers in the European Union from purchasing or being exposed to products that may have been made of seals killed inhumanely in IC or MRM hunts.⁷¹⁷ Moreover, we observe that the products allowed under the IC and MRM exceptions are not subject to any mechanism or labelling scheme through which consumers can be informed of the presence of seal products on the EU market in general and of whether a specific product contains seal. This suggests that EU consumers are exposed to, and may be purchasing, seal products without knowing that such products may be derived from seals killed inhumanely.

7.448. Overall, based on its design and expected operation, we find that the ban under the EU Seal Regime is capable of making a contribution to preventing the EU public from being exposed on the EU market to products that may have been derived from seals killed inhumanely in Canadian or Norwegian hunts. However, the IC and MRM exceptions under the Regime diminish the degree of the actual contribution made by the ban, as consumers are exposed to seal products that are allowed on the EU market under these IC and MRM exceptions, which may have been derived from seals killed inhumanely in IC or MRM hunts.

7.449. As regards the second aspect of the objective (i.e. the number of seals killed inhumanely), the European Union argues that the ban makes material, but partial, contribution to addressing the animal welfare aspect of the concerns by reducing global demand for seal products resulting from commercial hunts, with the consequence that less seals are killed in an inhumane way.

7.450. First, we look at the design, structure, and expected operation of the measure. A direct impact of the "ban" under the measure is a reduction of the demand for seal products within the European Union. COWI assessed that "the current legislation has been in the pipeline and has created uncertainty about the EU market. Hence, trade numbers are down substantially since 2007 and so is the market price of raw skin (less than half). ... Some European markets [for seal oil], Sweden, Denmark, Finland and Germany were emerging but halted in recent years due to the

⁷¹⁵ European Union's response to Panel question No. 9, para. 36.

⁷¹⁶ See section 7.3.2.2 above.

⁷¹⁷ We note that entities from Greenland and Sweden have been approved by the Commission to be recognized bodies under Article 6 of the Implementing Regulation that may issue attesting documents for seal products to be placed on the EU market pursuant to the IC and MRM exceptions. (See European Union's response to Panel question Nos. 156 and 158; Commission decisions of 18 December 2012recognising the Swedish County Administrative Boards, (Exhibit EU-159), and of 25 April 2013 recognizing the Greenland Department of Fisheries, Hunting and Agriculture (APNN), (Exhibit EU-149)).

Regulation."⁷¹⁸ The ban also appears to have had a negative influence on the seal products market in general. For example, according to Norway, "the mere expectation of the adoption of the EU Seal Regime hampered trade."71

7.451. We noted however in the context of our analysis of the measure's contribution to the first aspect of the objective that the IC and MRM exceptions have a certain negative impact on the degree of the actual contribution made by the ban. As noted earlier, seal products qualifying under the IC and MRM exceptions are allowed on the EU market regardless of the animal welfare outcomes in such hunts. To that extent, the exceptions would also reduce the degree of contribution made by the ban to reducing the overall demand for seal products within the European Union and consequently the number of seals that may be killed inhumanely in these hunts.

7.452. At the same time, there is evidence indicating that Inuit communities have been adversely impacted by the EU Seal Regime as a whole, in particular the ban under the Regime. There is also evidence showing that while Inuit and other indigenous communities could potentially qualify and export seal products under the IC exception, they have not been able to benefit from the exception. The interplay between the ban and the IC exception may thus limit, to a certain extent, any negative impact of the IC exception on the degree of the contribution of the ban to the objective.

7.453. Furthermore, as described in section 7.2.2 above, the EU Seal Regime allows certain commercial activities relating to the production of seal products derived from commercial hunts. Specifically, the European Union confirmed that the transit and the "inward processing" of seal products derived from commercial seal hunts can take place under the measure.⁷²⁰ Data provided by the parties confirm the continued trade of seal products after the EU Seal Regime entered into force. This has been explained by the parties as attributable to transhipment, resale, and/or processing activities and representing goods not released for free circulation in the EU customs territory.721

7.454. COWI further gives an indication of the economic significance of these activities, estimating that approximately 5 per cent of the global seal fur trade is actually consumed in the European Union, "while a much larger part is passing through the EU either in transit, through auction houses, or for tanning purposes".⁷²² A document produced by the European Parliamentary Assembly also states, "Europe is the main importer of raw products and *exporter of manufactured goods*".⁷²³ The European Union also clarified that it does not claim that allowing the transit and inward processing of seal products makes a positive contribution to the public morals objective pursued by the EU Seal Regime.⁷²⁴ In our view, these activities thus facilitate the processing of seals obtained in commercial hunts into final seal products, which in turn may be sold to other

⁷¹⁸ We recall that the first step in the legislative history of the current EU Seal Regime was the

EU Parliamentary Declaration on banning seal products in 2006. (See Parliament Declaration, (Exhibit JE-19)). See also COWI 2010 Report, Annex 5, pp. 16 and 19 (Traceability systems for trade in seal products -Briefing note for workshop participants, 20 October 2009). COWI had warned the European Union before the Commission tabled its original proposal that "[s]ince seal hunting mostly takes place outside the Community

territory, any restrictions to market access in the Community will have trade impacts". (COWI 2008 Report,

p. 102). ⁷¹⁹ Norway's first written submission, paras. 673-674 (referring to COWI 2010 Report, Annex 5, Briefing note of 2009).

⁷²⁰ The European Union defines inward processing as the processing under customs control of imported inputs into products intended for export. (European Union's response to Panel question No. 131).

⁷²¹ See parties' responses to Panel question No. 98. During 2011, the value of seal products entering the European Union under the inward processing regime amounted to €812,000, of which €713,000 originated in Canada and €99,000 originated in Norway. (European Union's response to Panel question No. 177). Furthermore, data provided by the European Union confirm that tanned or dressed seal skins have been exported from the European Union following the adoption of the EU Seal Regime (possibly as a result of inward processing), though we note that these activities have occurred at reduced levels from previous years. (See Eurostat, exports of seal products from the European Union (2001-2011), (Exhibit EU-87)).

⁷²² COWI 2010 Report, p. 42. See also Ibid. p. 37 (noting that seal hides and skins were "mainly imported to the EU for tanning or further processing" and then exported, such that "[o]nly limited quantities end up on the EU market"). ⁷²³ Council of Europe Recommendation, (Exhibit EU-117), para. 3 (emphasis added)

 $^{^{\}rm 724}$ European Union's response to Panel question No. 131.

markets in the world.⁷²⁵ Data from Canada's DFO, for example, show that a large value of seal products has been exported to markets outside the European Union⁷²⁶, primarily consisting of seal oils rather than skins.727

7.455. Therefore, while the measure prohibits certain seal products on the EU market based on their link to the potential incidence of inhumane killing of seals, the measure allows commercial activities within the European Union, which is directly connected to the processing of the same products. This incoherency in the measure, in our view, further reduces the contribution of the measure to the reduction of the global demand for seal products derived from inhumane killing. We also consider that this implicit exception under the measure exposes EU citizens to other types of commercial activities directly related to the production and supply of seal products that may have been derived from seals killed inhumanely.728 By allowing such activities, in our view, the EU Seal Regime undermines its intended objective of addressing the EU public concerns on seal welfare.

7.456. Turning to the actual operation of the measure, particularly in terms of trade statistics concerning seal products, we make a general observation that data provided by the parties are incomplete in terms of product types and import/export countries.⁷²⁹ As such, we are not in a position to draw any concrete conclusions based on the data before us. Nevertheless, the data show a general trend that seal product imports⁷³⁰ from the complainants into the EU market have decreased significantly over the last few years as illustrated by the table below.⁷³¹

 726 However, we note that these values do not exceed peak exports to the respective markets in previous years and further display sharp fluctuations between years. ⁷²⁷ DFO Commercial Seal Harvest Overview 2011, (Exhibit JE-27), Table 15.

⁷²⁸ Canada comments that allowing transit and inward processing, insofar as it exposes EU citizens to the presence of morally tainted products based on the EU's own logic, has a detrimental moral impact on the EU public. (Canada's comments on the European Union's response to Panel question No. 131).

The product classifications used by Norway to compile statistical trade data do not generally distinguish between products that may or may not contain seal. Accordingly, there are no available data on the major seal product exports from Norway. (See Norway's response to Panel question No. 96).

⁷³⁰ Seal skins have historically constituted the majority of seal products traded.

⁷³¹ See Canada's and Norway's responses to Panel question No. 99 (acknowledging the decline in trade with the European Union). See also Canada's response to Panel question No. 96.

⁷²⁵ The European Union points out that the exclusion of transit and inward processing from the ban benefits mainly the complainants' sealing industry and was indeed requested by that industry, with the support of the Canadian authorities. (European Union's response to Panel question No. 131).

There is some evidence indicating that seal products derived from seals hunted in Canada and Norway, including those produced through the inward processing allowed within the European Union, have been diverted to other markets than the European Union since the introduction of the EU Seal Regime. (See, e.g. DFO Commercial Seal Harvest Overview 2011, (Exhibit JE-27), Table 14 (showing destination of seal products outside the European Union); Eurostat, exports of seal products from the European Union (2001-2011), (Exhibit EU-87) (showing exports of seal products from the European Union to other destinations)).

 $^{^{729}}$ The Panel notes that discrete data are unavailable for many of the seal products listed in the European Commission's Technical Guidance Note (Exhibit JE-3), as such products are subsumed under subheadings of the Combined Nomenclature. (European Union's response to Panel question No. 97, para. 275; European Union's response to Panel question No. 166, para. 262). Specifically, data from the European Union do not include imports of raw seal skins (because they are mixed under a heading with other animal skins), which the European Union notes comprised the majority of Canada's exports to the European Union prior to the ban. (European Union's response to Panel question No. 97, para. 276). However, prior to 2007, there were tariff lines exclusively for raw seal skins and the European Union has provided data for these in Exhibit EU-143B (separate table provided in Annex). Conversely, data from Canada's DFO do not include exports of tanned skins because exportation of tanned and dressed seal skin products, whole seal skins or cuttings thereof, and seal skin apparel, are combined with such products derived from other animals. Thus, there are no discrete data for these seal products. (European Union's response to Panel question No. 98, para. 282 (citing Canada's first written submission, para. 78 and confirmed in Canada's response to Panel question No. 96, para. 353)).

1	33	-

EU imports of tanned/dressed seal skins from Canada and Norway⁷³²

Volume (# of units) and value ('000 euro)

	Can	ada	Norway			
2002	20,016	€689	23,753	€1,627		
2003	11,594	€455	10,996	€400		
2004	6,169	€347	8,156	€319		
2005	5,964	€396	9,046	€300		
2006	6,609	€415	3,226	€175		
2007	551	€44	5,437	€448		
2008	25,892	€464	2,811	€213		
2009	549	€48	3,225	€234		
2010	10	€1	81	€26		
2011	5	€1	36	€2		

7.457. In respect of the actual number of seals hunted in the main sealing countries, available data show some reduction in recent years although the data also indicate some fluctuations in the number of seals hunted in 2009 and 2010, for instance in Norway and Namibia.⁷³³

Number of seals hunted

	2004	2005	2006	2007	2008	2009	2010	2011	2012
Canada ⁷³⁴	365,971	329,829	354,400	224,745	217,850	76,668	69,101	40,393	-
Norway ⁷³⁵	14,800	18,000	17,000	14,000	1,263	8,035	4,652	10,334	5,593
Greenland ⁷³⁶	157,697	191,605	188,939	160,493	156,874	142,384	-	-	-
Namibia ⁷³⁷	59,500	65,000	83,047	34,728	47,603	41,145	67,799	-	-
Russia ⁷³⁸	-	22,474	7,107	5,476	-	_	-	-	-
EU ⁷³⁹	523	594	100	-	-	-	-	-	842

7.458. Further, the entry into force of the EU Seal Regime may not be the only factor explaining a reduction in the number of seals hunted; other factors may also have come into play, including weather conditions that may have contributed to reducing the duration of the hunting season and

⁷³² Data from Exhibits EU-88 and EU-143B for tanned/dressed fur skins, not assembled (Tariff line 43021949). Tables state "where available" for units (but not for data on value and tonnage), thus indicating possible uncertainty or incompleteness of these data.

⁷³³ Canada has also stated that 90,000 seals were harvested during the 2013 season. (Canada's comments on the European Union's response to Panel question No. 119, footnote 41).

 ⁷³⁴ DFO Commercial Seal Harvest Overview 2011, (Exhibit JE-27), Table 8 (data for harp seals).
 ⁷³⁵ EFSA Scientific Opinion; COWI 2008 Report; Statistics Norway, (Exhibit NOR-158); and Joint Norwegian/Russian Fisheries Commission, Report of the Working Group on Seals to the 42th Session -

Appendix 8, (Exhibit NOR-16) (data for harp and hooded seals, though from 2007 only a very small fraction of total seals caught were hooded seals, the large majority being harp seals).

⁷³⁶ Management and Utilization of Seals in Greenland, (Exhibit JE-26), Table 3 (data for harp, ringed, and hooded seals).

⁷³⁷ EFSA Scientific Opinion; COWI 2008 Report; and Statistics from Namibian Ministry of Fisheries and Marine Resources, (Exhibit NOR-159) (data for Cape fur seals). ⁷³⁸ EFSA Scientific Opinion and COWI 2008 Report (data for harp seals).

⁷³⁹ EFSA Scientific Opinion and COWI 2008 Report (data for grey seals).

the adoption of legislation on trade in seal products in other countries.⁷⁴⁰ Thus, based on the data before us, the extent of the connection between the ban aspect of the measure and the reduction in the number of seals killed is not clearly discernible.

7.459. In sum, based on its design, structure, and expected operation, we find that the ban under the measure makes a contribution to reducing the demand for seal products within the European Union and, to a certain extent, to reducing a global demand. Further, based on the data showing the application of the ban, we also observed a downward trend in seal products trade, which also suggests that the measure may have contributed to reducing the demand within the European Union. The degree of contribution made by the ban, however, appears to be diminished by the products imported under the IC and MRM exceptions. We also found that allowing commercial activities relating to the processing of the seal products that are otherwise prohibited under the ban further undermines the objective of the measure.

7.460. In conclusion, we find that the EU Seal Regime is capable of making and does make some contribution to its stated objective of addressing the public moral concerns. The Regime, through its prohibitive aspect, prevents to some extent EU citizens from being exposed to and participating as consumers in commercial activities related to the products derived from seals that may have been killed inhumanely. It also appears to be negatively affecting the demand for seal products within the European Union and globally. At the same time, we observe that the IC and MRM exceptions under the measure have the effect of diminishing the degree of the measure's actual contribution to both aspects of its objective. Further, we find that the capability of the EU Seal Regime to achieve its objective of addressing the EU public moral concerns is further negatively affected because other commercial activities such as the inward processing of seal products are allowed without regard to the welfare of seals from which the products are derived. We also note that the Travellers exception, albeit limited in its scope, does not contribute to achieving the objective of addressing the EU public moral concerns is further seals on the objective of addressing the EU public moral concerns is contribute to achieving the objective of addressing the EU public moral concerns is further negatively affected because other commercial activities such as the inward processing of seal products are allowed without regard to the welfare of seals from which the products are derived. We also note that the Travellers exception, albeit limited in its scope, does not contribute to achieving the objective of addressing the EU public moral concerns on seal welfare.

7.461. We continue our examination by assessing whether any less trade-restrictive alternative measures exist that can make an equivalent or greater contribution to the objective of the EU Seal Regime, taking into account the risks non-fulfilment of the objective would create.

7.3.3.3.3 Risks non-fulfilment of the objective would create

7.462. The European Union submits that the "risk of non-fulfilment of the objective of protecting public morals is that the EU public would experience the same moral feelings that prompted the adoption of the EU Seal Regime".⁷⁴¹ Specifically, the European Union frames the risks created by non-fulfilment of the objective in terms of the two separate ways that the EU Seal Regime is intended to address public moral concerns on seal welfare (i.e. moral concerns about the inhumane killing of seals as such and EU consumers' contribution to inhumane killing along with exposure to morally tainted products). Thus, non-fulfilment of the objective would create risks of poor seal welfare as well as the community's participation in morally offensive seal hunts and exposure to the by-products of such hunts.⁷⁴²

7.463. Canada submits that the risk is that seals would be killed in a way that causes avoidable pain and suffering.⁷⁴³ With regard to the alleged public moral concerns about the inhumane killing of seals, the risk is that EU citizens would be morally offended or upset about the inhumane killing of seals continuing and the products from those hunts being placed on the EU market. Norway argues that the prevention of certain "moral feelings" is "a remarkably undefined basis on which to base a measure that is both restrictive and discriminatory", and does not suggest that the measure is necessary to "protect" public morals themselves.⁷⁴⁴ Rather, the measure appears to be

⁷⁴⁰ There is some indication, however, that market demand and price levels can be a factor in the scale of participation in some hunts. (See, e.g. DFO Empirical Base for Canada's Seal and Seal Products Industry, (Exhibit CDA-17), p. 2 ("Participation [in the hunt] varies from year to year, and depends upon ice conditions, *price of pelts*, etc.") (emphasis added)) Canada has similarly argued that the market opportunities potentially afforded by the IC exception will give Greenlandic hunters "a strong incentive to increase the scale of their hunt in order to place more products on the EU market". (Canada's comments on the European Union's response to Panel question No. 119, para. 36).

⁷⁴¹ European Union's response to Panel question No. 44, para. 152.

 $^{^{\}rm 742}$ European Union's response to Panel question No. 44.

⁷⁴³ Canada's second written submission, para. 329.

⁷⁴⁴ Norway's second written submission, paras. 270-274.

necessary to protect EU consumers from certain negative "feelings" that they might not like when shopping.

7.464. The Appellate Body clarified that the "risks non-fulfilment would create" component of Article 2.2 requires an ascertainment of "the nature of the risks at issue and the gravity of the consequences that would arise from non-fulfilment of the legitimate objective".⁷⁴⁵

7.465. Given that the objective of the EU Seal Regime is to address the EU public moral concerns with respect to seal welfare per se as well as EU public's wish to not purchase or be exposed to the products derived from seals killed inhumanely, the risks of non-fulfillment of such objective can be linked to these two specific aspects. Failing to achieve these two aspects of the measure's objective will thus expose the EU public to their existing moral concerns on seal welfare and to the products derived from seals that may have been killed inhumanely.

7.466. As we examined in the previous section, however, we found that the degree of the EU Seal Regime's actual contribution to its identified objective is diminished by the scope of the exceptions under the measure and the allowance of certain commercial activities within the European Union. Accordingly, we do not consider that the level of protection actually achieved by the measure is as high as the European Union claims the measure initially aimed to achieve.⁷⁴⁶ We will bear this in mind in taking into account the risks non-fulfilment would create in our subsequent analysis of the reasonable availability of a less trade-restrictive alternative measure.

7.3.3.3.4 Whether a less trade-restrictive alternative measure is reasonably available, taking account of the risks non-fulfilment would create

7.467. The Appellate Body has interpreted the second sentence of Article 2.2 of the TBT Agreement to suggest that the existence of "an unnecessary obstacle to international trade" may be established through a comparative analysis with possible alternative measures as "a conceptual tool for the purpose of ascertaining whether a challenged measure is more trade restrictive than necessary".⁷⁴⁷ Following guidance by the Appellate Body on the framework of relevant considerations for this comparison⁷⁴⁸ we begin our analysis by identifying the alternative measure advanced by the complainants and its trade-restrictiveness. We next assess the degree of contribution by the alternative measure to the relevant objective. Finally, we examine whether the alternative measure is reasonably available taking account of the risks non-fulfilment would create.

7.3.3.3.4.1 Identification of an alternative measure(s)

7.468. Both Canada and Norway propose an alternative measure whereby market access for seal products would be conditioned on compliance with animal welfare standards combined with certification and labelling requirements.⁷⁴⁹ This proposed alternative consists of three related components: (1) the establishment of animal welfare requirements in the hunting of seals; (2) certification of conformity with the animal welfare requirements; and (3) labelling seal products to show the certified compliance with the animal welfare requirements.⁷⁵⁰

7.469. The European Union notes that the alternatives advanced by each complainant "appear to be the same" and therefore addresses them together.⁷⁵¹

⁷⁴⁵ Appellate Body Reports, US – Tuna II (Mexico), para. 321; and US – COOL, para. 377.

⁷⁴⁶ The European Union emphasizes a high level of protection it aims to achieve against the risk that seals will experience excessive suffering when they are killed. (European Union's first written submission, para. 39). The complainants however refer to the low level of contribution of the EU Seal Regime to its objective (i.e. it tolerates a high risk of non-fulfilment).

⁷⁴⁷ Appellate Body Report, US – Tuna II (Mexico), para. 320.

⁷⁴⁸ Appellate Body Report, US – Tuna II (Mexico), para. 322.

 ⁷⁴⁹ Canada's first written submission, para. 556; Norway's first written submission, para. 779.
 ⁷⁵⁰ Canada's first written submission, paras. 557-560; Norway's first written submission, para. 793. ⁷⁵¹ European Union's first written submission, para. 370. See also European Union's second written

submission, para. 301.

7.3.3.3.4.2 Trade-restrictiveness of the alternative measure

7.470. Canada describes this alternative as less trade restrictive because the current EU Seal Regime excludes all non-Inuit commercial seal products from the EU market, whereas the alternative regime would allow such non-Inuit commercial seal products provided they meet the animal welfare requirements.⁷⁵² Norway submits that the alternative is less trade restrictive because imports would be permitted provided they comply with animal welfare requirements.⁷⁵³

7.471. The European Union argues that seal products qualifying under the EU Seal Regime, namely those from IC hunts, might not have market access under a regime imposing animal welfare requirements.⁷⁵⁴

7.472. We found above that the EU Seal Regime limits trade in seal products, including those imported from the complainants, and thus is trade restrictive.⁷⁵⁵ The European Union also acknowledges that the EU Seal Regime was enacted precisely for its trade-restrictive effect.⁷⁵⁶ In comparison, the alternative measure could possibly permit seal products from the complainants that are prohibited under the EU Seal Regime.⁷⁵⁷ In view of the potentially large quantities of seal products derived from non-IC or MRM hunts, we consider that their potential allowance under the proposed alternative measure makes such proposed measure less trade restrictive.⁷⁵⁸

7.3.3.3.4.3 Degree of contribution of the alternative measure to the objective of the EU Seal Regime

Main arguments of the parties

Complainants

7.473. Both Canada and Norway contend that the EU Seal Regime fails to address animal welfare concerns in that it allows placement on the market of products without regard for the welfare of the seals from which the products are derived. By contrast, the proposed alternative measure would contain explicit animal welfare requirements upon which placement on the market is conditioned and, in combination with certification requirements, would directly contribute to fulfilment of the objective of animal welfare.⁷⁵⁹

7.474. Canada and Norway further argue that an alternative measure directly addressing animal welfare would thereby address any public moral concerns relating to animal welfare.⁷⁶⁰ The alternative measure would apply to all seal products consistently, prohibiting products from seals killed in an inhumane manner while granting access to products derived from seals harvested in a manner that respected the animal welfare criteria. This alternative would thus address animal

⁷⁵² Canada's first written submission, para. 638. See also Canada's second written submission, para. 326. Canada argues that given that such products comprise the "vast majority of seal products", their potential allowance under the proposed regime makes this a less trade-restrictive alternative measure.

⁷⁵³ See Norway's responses to Panel questions following the second substantive meeting, Annex I, first row.

⁷⁵⁴ European Union's comments on Canada's response to Panel question No. 115; European Union's opening statement at the second meeting of the Panel, para. 64.

⁷⁵⁵ The Appellate Body has recognized that an import ban is "by design as trade-restrictive as can be". (Appellate Body Report, *Brazil – Retreaded Tyres*, para. 150).

⁷⁵⁶ European Union's first written submission, paras. 36-37, 357, and 586.

⁷⁵⁷ See section 7.2.1 above.

⁷⁵⁸ We reach this conclusion notwithstanding variation in the precise amounts and proportions of humanely killed seals in commercial hunts. Nor is our conclusion altered by the possibility that some products from IC hunts qualifying under the EU Seal Regime might fail to satisfy animal welfare requirements, as market access would still potentially be open to products from humanely killed seals in IC hunts.

⁷⁵⁹ Canada's first written submission, paras. 564-569; second written submission, paras. 319-322, 338; Norway's first written submission, paras. 886-894; opening statement at the first meeting of the Panel, paras. 134 and 141.

⁷⁶⁰ Canada's second written submission, para. 318, 330 (referring to European Union's response to Panel question No. 9); opening statement at the second meeting of the Panel, para. 82; Norway's second written submission, para. 308 (referring to European Union's response to Panel question No. 44).

welfare and public moral concerns, as well as the alleged public moral concerns arising from consumers' participation in the economic activity that sustains the market.⁷⁶¹

7.475. The complainants further submit that the alternative measure would continue to allow market access to products from Inuit or indigenous hunts that were certified to meet animal welfare requirements.⁷⁶²

Respondent

7.476. The European Union submits that as it is not feasible to apply and enforce effectively and consistently a humane killing method⁷⁶³, the proposed alternative measure would allow products on the EU market that were obtained from seals killed in a manner that causes them excessive suffering.⁷⁶⁴ Allowing products from humanely killed seals would not meet the concerns that led to the adoption of the EU Seal Regime. Such concerns would persist because, "in order to kill the requisite number of certified seals in a humane way, it would be necessary to kill many other seals in an inhumane way", which in turn "could have the perverse effect of increasing the number of seals killed in an inhumane way".⁷⁶⁵

7.477. With respect to the application of animal welfare requirements to Inuit or other indigenous communities, the European Union argues that the nature of Inuit seal hunting poses particular challenges to applying and monitoring animal welfare.⁷⁶⁶

In this connection, Canada points out that there are animal welfare requirements applicable to Inuit engaged in sealing in Canada and specifically proposes that Greenlandic seal products could be segregated so that products satisfying animal welfare requirements could be commercially exported while other products (such as those derived from netting) could be diverted to local consumption. (See Canada's second written submission, para. 328; Canada's response to Panel question No. 114; Canada's response to Panel question No. 115, para. 72).

The complainants also submit that the proposed alternative measure would also fulfil what they claim to be other objectives of the EU Seal Regime. Specifically, Canada and Norway assert that the labelling requirement would meet the objective of preventing consumer confusion, whereas the EU Seal Regime allows unlabelled seal products on the market without any indication of their animal welfare compatibility. (Canada's first written submission, paras. 570-574; opening statement at the second meeting of the Panel, para. 84; Norway's first written submission, paras. 895-903; opening statement at the first meeting of the Panel, para. 135). Similarly, Canada and Norway contend that the objective of promoting consumer choice and personal use would be better served by the alternative measure as consumers would be given wider access to seal products on the EU market along with the relevant information as to animal welfare. (Canada's first written submission, paras. 575-577; opening statement at the second meeting of the Panel, para. 84; Norway's first written submission, paras. 906; second written submission, para. 276).

As to the sustainable management of marine resources, Canada and Norway claim that the EU Seal Regime contains requirements (e.g. exclusivity of purpose) that undermine this objective, whereas the alternative measure could encourage rather than create disincentives for the sustainable management of marine resources for large hunts conducted for multiple purposes. (Canada's first written submission, paras. 578-581; Norway's first written submission, paras. 912-917; opening statement at the first meeting of the Panel, para. 136).

⁷⁶³ See European Union's first written submission, Section 2.4.

⁷⁶⁴ European Union's first written submission, para. 375. See also European Union's response to Panel question No. 148 (arguing that "the contribution of any given certification system to the intended policy objective is, to a large extent, a function of the underlying substantive requirements and cannot be measured without taking into account the specific content of such requirements").

⁷⁶⁵ European Union's second written submission, para. 310.

⁷⁶⁶ The European Union further submits that Canada's suggestion of segregating products "would fail to make an equivalent contribution to the objective pursued by the IC exception" given the dynamics of the Greenlandic hunt and domestic market. (European Union's comments on Canada's response to Panel question No. 115. See also European Union's opening statement at the second meeting of the Panel, para. 64). The European Union also comments upon various areas of Canadian seal hunting regulations that exempt the Inuit

⁷⁶¹ Canada's second written submission, paras. 323-325; opening statement at the first meeting of the Panel, paras. 94-96; opening statement at the second meeting of the Panel, para. 83; Norway's second written submission, paras. 276, 305-312.

⁷⁶² Canada's first written submission, para. 563; second written submission, para. 328; opening statement at the second meeting of the Panel, para. 85; Norway's first written submission, para. 905; second written submission, para. 312. As to the capacity of Inuit or other indigenous communities to comply with such requirements, the complainants contend that there is no "inherent" reason why animal welfare requirements could not be applied to Inuit or other indigenous communities. (Canada's response to Panel question No. 115, para. 68; Norway's response to Panel question No. 114, para. 84).

- 138 -

Analysis by the Panel

7.478. As an initial matter, we recall that the degree of the actual contribution of the EU Seal Regime to the fulfilment of its objective is ascertained "from the design, structure, and operation of the [EU Seal Regime], as well as from evidence relating to its application".⁷⁶⁷ Based on such analysis, we found that the EU Seal Regime is capable of making and does actually make a contribution to the achievement of its stated objective of addressing the public moral concerns. The Regime, through its prohibitive aspect, prevents to a certain extent the EU public from being exposed to and participating as consumers in commercial activities related to the products derived from seals that may have been killed inhumanely. It also appears to have the effect of negatively impacting the global demand for seal products.

7.479. The question with respect to the alternative measure is whether it would make an equivalent or greater contribution to that actually achieved by the EU Seal Regime in the two respects described above.⁷⁶⁸

7.480. We first address the contribution of the alternative measure to preventing or reducing exposure of the EU public to products raising moral concerns. To the extent that the alternative measure could effectively distinguish and label products from humanely and inhumanely killed seals, the direct participation of EU consumers in the market for seal products would theoretically be confined to products that conform to the animal welfare concerns of the EU public. However, given the risks to animal welfare that we have found to exist in all seal hunts, the degree of contribution of the alternative measure would in part depend on the feasibility of meeting adequately defined animal welfare requirements.⁷⁶⁹ As acknowledged by the complainants⁷⁷⁰, and consistent with the findings of EFSA⁷⁷¹, an indefinite portion of hunted seals may experience pain, distress, or other forms of suffering. Thus, even if the alternative measure succeeded in limiting market access to exclusively those products derived from humanely killed seals, such products would originate in hunts that may have caused poor animal welfare outcomes for some other number of seals. Moreover, the capacity of an alternative measure to effectively (i.e. accurately) differentiate products from humanely and inhumanely killed seals will depend on the practical feasibility of the certification system proposed by the complainants as part of the alternative measure.

7.481. In sum, the degree of contribution achieved by the alternative measure to preventing participation of the EU public as consumers in the inhumane killing of seals depends on the reasonable availability of satisfying adequate animal welfare standards in seal hunts as well as the capability of accurately distinguishing the resulting products for placement on the EU market.

7.482. We next address the reduction of global demand for seal products and number of seals killed. We have found that the EU Seal Regime is capable of contributing to some extent to the decline of the market for seal products. This, in turn, contributes to reduced prices and consumer demand for some seal products (particularly seal skins) that has coincided with some reduction in the number of seals killed in major sealing countries. As mentioned in relation to trade-restrictiveness, the alternative measure would potentially afford market access to seal products from commercial hunts that are currently prohibited under the EU Seal Regime. Although the scale of market access that could be obtained under the alternative measure cannot be precisely determined in the abstract, the alternative measure would potentially reopen an outlet for the

in Canada from animal welfare hunting requirements. (See European Union's comments on Canada's response to Panel question No. 114).

⁷⁶⁷ See Appellate Body Reports, US – Tuna II (Mexico), para. 317; US – COOL, para. 373.

⁷⁶⁸ In this connection, we note the complainants' arguments that there is no requirement for the alternative measure to achieve the level of protection *selected* by the EU legislators if the measure at issue does not fully achieve the objective at that level. (Canada's opening statement at the second meeting of the Panel, para. 90; Norway's second written submission, paras. 274-275, 280-281; opening statement at the first meeting of the Panel, para. 132; response to Panel question No. 147).
⁷⁶⁹ See, e.g. Commission Proposal, Explanatory memorandum, (Exhibit JE-9), p. 12 ("Labelling alone of

⁷⁶⁹ See, e.g. Commission Proposal, Explanatory memorandum, (Exhibit JE-9), p. 12 ("Labelling alone of seal products is not an alternative to a ban on trade in those products as labelling would only be relevant to assuage the ethical animal welfare concerns of citizens and consumers as and when the killing and skinning methods in force in the sealing countries would accord with the criteria provided for in this Regulation.").

⁷⁷⁰ See Canada's and Norway's responses to Panel question No. 69.

⁷⁷¹ EFSA Scientific Opinion, p. 88 ("there is no perfect killing method that will work at all times, and under all circumstances").

marketing of goods derived from commercial seal hunts on the EU market. This may in turn contribute to sustaining or increasing the overall number of seals killed, which would have the consequence of subjecting a greater number of seals to the animal welfare risks incidental to seal hunting.⁷⁷²

7.483. At the same time, the alternative measure could potentially introduce an economic incentive for sealing countries to adopt and enforce the animal welfare standards established by the measure. Thus, although the alternative measure may generate economic incentives to subject a greater number of seals to the welfare risks of seal hunting, this may be counterbalanced in some measure by the encouragement of improved practices in the hunt. For example, the European Commission considered that its Proposal to allow trade in seal products based on compliance with animal welfare requirements would give "incentives to countries concerned to review and improve, where need be, their legislation and practice concerning the methods to be complied with when killing and skinning seals".⁷⁷³

7.484. On balance, we consider that the alternative measure may have the capacity of restoring the potential market in the European Union for seal products with the consequence of subjecting a greater number of seals to the risks of poor animal welfare. Although this in itself may be contrary to the European Union's stated objective of reducing global demand for seal products and consequently reducing the number of seals killed inhumanely, the imposition of animal welfare requirements may also promote humane killing practices in seal hunts that could reduce the number of inhumanely killed seals to some extent. The impacts of the alternative measure in this regard are however closely related to the type of animal welfare requirements to be imposed, the feasibility of enforcement of such requirements, and the attendant risks of inhumane killing in seal hunts.

7.485. In light of the inextricable link between the contribution of the alternative measure to the objective and the feasibility of its implementation, we next address the parties' contentions regarding the reasonable availability of the various components of the alternative measure.

7.3.3.3.4.4 Reasonable availability of the alternative measure

Main arguments of the parties

Complainants

7.486. Canada and Norway state that the alternative proposed measure is reasonably available to the European Union for the following four reasons. First, it is feasible to prescribe animal welfare criteria applicable to the hunting of seals based on existing scientific evidence that would ensure the minimization of suffering.⁷⁷⁴ Second, once established, animal welfare criteria can be effectively monitored and enforced in the context of seal hunting.⁷⁷⁵ Third, a system of certification

⁷⁷² We note that the European Parliament responded to the Commission Proposal by concluding that the risks to animal welfare posed by seal hunting were too high for humane killing requirements to be imposed. Parliament Report, (Exhibit JE-4), Justification for Amendment 28, p. 21:

Commercial seal hunts are inherently inhumane because humane killing methods cannot be effectively and consistently applied in the field environments in which they operate. Moreover, seal hunts occur in remote locations, and are conducted by thousands of individuals over large, inaccessible areas, making effective monitoring of seal hunting impossible. As such only a comprehensive ban without the derogation drafted by the Commission would meet citizens' demands to see an end to the trade in seal products.

⁷⁷³ Commission Proposal, Explanatory memorandum, (Exhibit JE-9), p. 5. See also Commission Impact Assessment, (Exhibit JE-16), p. 26 (noting that a prohibition subject to animal welfare exceptions "will hurt the economy where it is supposed to hurt – but at the same time benefit 'best practice' seal hunting and therefore provides an incentive to improve the welfare of hunted seal species".).

⁷⁷⁴ Canada's first written submission, paras. 582-623; Norway's first written submission,

paras. 802-832; second written submission, para. 287. See also Norway's response to Panel question No. 147. ⁷⁷⁵ Canada's first written submission, paras. 624-653; Norway's first written submission,

paras. 833-851; opening statement at the first meeting of the Panel, paras. 142-144, 147-199. See also Norway's response to Panel question No. 147.

of conformity with animal welfare requirements is feasible and reasonably available.⁷⁷⁶ Fourth, animal welfare labelling of seal products is reasonably available contrary to the preamble recital of the EU Seal Regime that a labelling scheme would not be cost effective.⁷⁷⁷

7.487. Further, both Canada and Norway reference the European Union's policy in related product areas as evidence of the feasibility of prescribing animal welfare requirements and monitoring killing for animal welfare compliance.⁷⁷⁸ With respect to the certification of animal welfare compliance, Canada and Norway specifically contend that certification would not need to be on a seal-by-seal basis to achieve a level of contribution to seal welfare that is equal to or greater than that of the EU Seal Regime. In this respect, the complainants suggest options drawing upon other certification schemes that include regional/geographic certification and hunter licensing.⁷⁷⁹

Respondent

7.488. The European Union responds that the complainants' proposed alternative is similar to the same measure which had been proposed by the European Commission during the legislative process.⁷⁸⁰ However, this measure was deliberately rejected by EU legislators because "although it could be possible, in theory, to prescribe a humane method for killing seals, in practice the unique conditions in which seal hunting takes place would render it impossible to apply and enforce such method in an effective and consistent manner."⁷⁸¹ In particular, the European Union disputes various distinct components of the proposed alternative.

7.489. First, as to the possibility of prescribing humane killing methods, the European Union argues that the animal welfare requirements in question must be capable of being "applied and enforced effectively and consistently, so as to achieve the level of protection selected by the EU legislators".⁷⁸² The European Union submits that while many veterinary experts agree that a humane killing method for seals could, in theory, be defined, there is disagreement as to the requirements of such a method as well as to what would constitute an acceptable level or means of effective practice in carrying out such requirements.⁷⁸³

7.490. Second, the European Union addresses what it considers to be "the crucial issue" of whether it is possible to effectively and consistently apply and enforce a humane killing method.⁷⁸⁴ On this point, the European Union emphasizes that a genuinely humane method cannot be applied on a consistent basis with adequate monitoring and enforcement as a result of inherent obstacles.⁷⁸⁵

7.491. Third, the European Union argues that its measures applied with regard to other animals are not indicative of available alternative measures for seals due to "the major differences between the situations concerned".⁷⁸⁶

⁷⁷⁶ Canada's first written submission, paras. 654-676; Norway's first written submission, paras. 852-868. See also Canada's and Norway's responses to Panel question Nos. 147; parties' responses to

Panel question No. 94; Norway's second written submission, paras. 302-304.

⁷⁷⁷ Canada's first written submission, paras. 677-685; Norway's first written submission, paras. 869-877.

⁷⁷⁸ Norway's first written submission, paras. 878-883. See also Norway's second written submission, paras. 278, 290-294; opening statement at the first meeting of the Panel, para. 145: Canada's first written submission, paras. 611-623, 642-653; Canada's second written submission, paras. 339-344.

⁷⁷⁹ See complainants' responses to Panel question No. 147; Canada's opening statement at the second meeting of the Panel, para. 89.

⁷⁸⁰ European Union's first written submission, paras. 372; second written submission, para. 304.

⁷⁸¹ European Union's first written submission, para. 373. See also European Union's second written submission, paras. 304-307 (explaining the concern of EU legislators with excessive *unavoidable* suffering in seal hunts as opposed to the Commission's proposal focusing on avoidable suffering).

⁷⁸² European Union's first written submission, para. 377.

⁷⁸³ European Union's first written submission, paras. 377-387; response to Panel question No. 63.

 ⁷⁸⁴ European Union's first written submission, para. 388. See also European Union's opening statement at the first meeting of the Panel, para. 11.
 ⁷⁸⁵ European Union's first written submission, paras. 389-403; second written submission,

 ⁷⁸⁵ European Union's first written submission, paras. 389-403; second written submission, paras. 312-313.
 ⁷⁸⁶ European Union's first written submission, para. 406. See also European Union's first written

⁷⁸⁶ European Union's first written submission, para. 406. See also European Union's first written submission, paras. 407-413; response to Panel question No. 64; second written submission, paras. 86-91.

- 141 -

7.492. Finally, the European Union disputes that the certification and labelling component of the alternative measure would be viable⁷⁸⁷ and submits that the availability of certification cannot be considered independently of the underlying welfare requirements.⁷⁸⁸ In particular, the European Union argues that certification would have to be made on a seal-by-seal basis in view of the impossibility of applying animal welfare requirements consistently in seal hunts⁷⁸⁹, and the various examples of certification and labelling systems cited by the complainants "lack pertinence" as "the animals concerned are different, the environment is different, the killing methods are different and, consequently, the risks to animal welfare are also very different".⁷⁹⁰

Analysis by the Panel

7.493. As the Appellate Body has stated, "[a]n alternative measure may be found not to be 'reasonably available' ... where it is merely theoretical in nature, for instance, where the responding Member is not capable of taking it, or where the measure imposes an undue burden on that Member, such as prohibitive costs or substantial technical difficulties."⁷⁹¹ In this dispute, the question of reasonable availability presents a key factual disagreement between the parties as to the feasibility of measures that can be taken under the conditions in which seal hunting occurs to fulfil the relevant policy objective. Specifically, we address the parties' contentions regarding: (1) the prescription of animal welfare criteria; (2) the application, monitoring, and enforcement of animal welfare criteria; and (3) certification and labelling of compliance with animal welfare criteria.

7.494. With respect to the prescription of animal welfare criteria, we note that there is some discrepancy between experts and other sources as to what would constitute adequate welfare standards and criteria. There is general recognition of the principle of minimizing animal pain and suffering prior to killing. However, this principle is understood and interpreted differently among expert conclusions, in particular regarding the specific elements of a three-step killing method as adapted to the conditions of seal hunts.⁷⁹² More significantly, the principle of *minimizing* animal pain and suffering gives rise to uncertainty regarding what should be considered an acceptable level of such suffering. Some sources have provided recommendations of humane killing to accommodate the practical demands of seal hunting⁷⁹³, thus tolerating risks to welfare that are rejected by others.⁷⁹⁴ One notable example in this regard pertains to delays in the killing process, and there has been explicit acknowledgement by some experts of subjectivity and divergence in what is to be considered an acceptable lapse of time between killing steps.⁷⁹⁵

⁷⁸⁹ European Union's response to Panel question No. 64; second written submission, para. 320; comments on complainants' responses to Panel question No. 147(a).

⁷⁸⁷ European Union's second written submission, paras. 309, 317-321.

⁷⁸⁸ European Union's comments on complainants' responses to Panel question No. 147.

⁷⁹⁰ European Union's second written submission, para. 319. See also European Union's comments on complainants' responses to Panel question No. 147(b).

⁷⁹¹ Appellate Body Report, *Brazil – Retreaded Tyres*, para. 156 (citing Appellate Body Report, *US – Gambling*, para. 308). ⁷⁹² See, above regarding, for example: disputes as to whether certain stunning techniques can be

⁷⁹² See, above regarding, for example: disputes as to whether certain stunning techniques can be regarded as achieving death on their own (stun/kill) and the implications of this for later steps; the dispute among experts as to the preferability for checking consciousness with blink tests, skull palpation, or second-stunning; and the suggestion by some that an adequate killing method would contain a fourth step of rechecking.

checking. ⁷⁹³ See, e.g. EFSA Scientific Opinion, p. 10 ("Terms of Reference as provided by the Commission"); IVWG Report (2005) (statement that recommendations should be made in accordance with what can be realistically achieved in the circumstances of the hunt); NAMMCO Report (2009), pp. 5-6.

⁷⁹⁴ See, e.g. Butterworth (2007) and Richardson (2007).

⁷⁹⁵ See Daoust (2002) and Daoust (2012). This is particularly reflected in the dispute between carrying out the steps "immediately" (e.g. Burdon (2001)) and "as soon as possible" (e.g. EFSA Scientific Opinion and IVWG Report (2005)).

The contentions of Canada and Norway with regard to certain conclusions of Daoust (2012) are illustrative of different levels of tolerance for the risks to animal welfare in characterizing the humaneness of the seal hunt. For instance, the study's most comprehensive observation of the shooting of 278 seals at the 2009 Front yielded the result that fourteen (5.0%) of the seals "were considered to have a poor welfare outcome; these animals were not killed immediately with the first shot and were not shot again before being retrieved, in at least 12 of these cases with a gaff from the vessel". (Daoust (2012), p. 450). Canada favourably cites this finding along with the authors' comment that "[t]his proportion of seals considered to have had a poor welfare outcome is comparable to, or lower than, that in other types of hunt." (Daoust (2012), p. 453; Canada's first written submission, para. 601). Conversely, the European Union highlights the particular

- 142 -

7.495. In our view, these differences in animal welfare requirements and standards stem in part from differing assessments and tolerance of the risks involved in the application and monitoring of killing methods. However, those risks would persist irrespective of the specific standard of animal welfare prescribed in market access requirements of the alternative measure. We therefore address the reasonable availability of the alternative measure in light of the risks incidental to seal hunting as identified above.⁷⁹⁶

7.496. Based on the differing views on what would constitute adequate welfare standards, and absent a clearly articulated standard from the complainants, the requirements under the alternative measure could possibly span a range of different levels of stringency or leniency. In the case of more stringent requirements for humane killing reflecting a high level of animal welfare⁷⁹⁷, such requirements may not be practical for hunters to consistently satisfy in light of the conditions in which seal hunting takes pace.⁷⁹⁸ Further, although adopting more stringent requirements for humane killing would in principle answer to animal welfare concerns, in order to genuinely assuage such concerns there would need to be a mechanism to verify that the requirements were actually satisfied for seals used to generate products. Assuming that more exacting welfare requirements were imposed that were capable of being verified in the course of a seal hunt, this would imply accurate differentiation between seals killed in accordance with the strict requirements and those falling short of the higher welfare standard. Assessed against the backdrop of the welfare risks of seal hunting, this could give rise to infliction of significant suffering in larger scale hunts in order to kill other seals in accordance with the higher standards of welfare.⁷⁹⁹ At the same time, to the extent that the animal welfare requirements and verification of humane killing would be made less stringent to accommodate practical challenges of seal hunting, such an alternative measure may thus directly compromise the welfare of seals. This in turn would diminish the degree of contribution to fulfilment of the objective of addressing public moral concerns.

7.497. Finally, as regards the certification and labelling of compliance with animal welfare criteria, we note that this component of the alternative measure necessarily corresponds to application of the animal welfare standards to be certified. We have found that the conditions and challenges of seal hunting pose the risk that some portion of hunted seals will experience poor animal welfare outcomes. Consequently, a certification system limiting market access to products from humanely killed seals would need to be capable of distinguishing between seals killed in accordance with the relevant standard of animal welfare, and those killed inhumanely. A certification system that did not make this distinction would undermine its own capability of assuring that animal welfare (and

welfare concern implicated by the documented delay in this study, and argues that favourable citation to these results indicates that the authors of Daoust (2012) and the Canadian government regard outcomes as humane that other authorities would consider inhumane. (European Union's first written submission, paras. 389-396; second written submission, paras. 50-53).

⁷⁹⁶ We note that the parties have developed their arguments as to humane killing in recognition of its importance to the question of the necessity of the EU Seal Regime and the availability of an alternative measure. (See European Union's first written submission, para. 388 and opening statement at the first meeting of the Panel, para. 11 (citing as "the crucial issue" whether it is possible to effectively and consistently apply and enforce a humane killing method); Canada's second written submission, para. 335 ("defence of the EU Seal Regime depends on acceptance of the notion that there are inherent obstacles to the humane killing of seals"); Norway's second written submission, para. 277 (considering that the European Union "concedes that 'it might be possible to design a genuinely humane method for killing seals'; however, it explains, it is *impossible to apply* such a standard consistently in 'commercial' seal hunts")).

⁷⁹⁷ More stringent requirements reflecting a higher standard of animal welfare could include, for example: that each seal have all steps of the killing process performed in immediate succession without any delay; that no seal be hooked or gaffed for retrieval without having first been checked for consciousness; and/or that each seal be irreversibly stunned in the first application of a stunning method.

⁷⁹⁸ The IVWG, for instance, which was specially convened to make recommendations for improving humane practice in the Canadian harp seal hunt, recognized that recommendations needed to be "realistic in the context of the hunt, so that sealers will accept and implement them". (IVWG Report (2005), p. 7). At the time of the most recent amendments to Canada's Marine Mammal Regulations, the director general for the DFO's resource management branch gave indications that more stringent requirements risked non-compliance by sealers to explain why only "very minimal changes" were being made to hunting regulations. (Transcript of statements to the radio station CBH-FM (28 December 2008), (Exhibit EU-105), p. 3 ("If you actually go out in any, you know, in any industry, and you make a large-scale change in regulations, the, there is probably a probability that a lot of the regs [*sic*] would not be abided by.")).

⁷⁹⁹ See European Union's second written submission, para. 310 (citing *amicus curiae* submission by Anima et al., (Exhibit EU-81)).

by extension public moral concerns) were being addressed.⁸⁰⁰ Conversely, an alternative scheme designed to certify that a given seal product was *in fact* derived from a humanely killed seal may impose large costs and/or logistical demands on those participating in the hunt and subsequent marketing of products.⁸⁰¹

7.498. In this regard, we also find it instructive to consider the Commission Proposal, which the parties have concurred is similar to the proposed alternative measure. The Commission Proposal would have required *inter alia* that "an appropriate scheme is in place whereby seal products ... are certified as coming from seals to which" animal welfare requirements have been effectively applied and enforced.⁸⁰² This scheme is framed so as to certify only that products "are obtained from seals killed and skinned *in a country where, or by persons to whom*," the stipulated animal welfare requirements would apply.⁸⁰³ The complainants have argued that a certification system need not be on a seal-by-seal basis, but rather could be achieved through country certification or hunter licensing.⁸⁰⁴ However, country or hunter-based schemes would potentially fail to convey accurate information in respect of the seal from which the product was derived⁸⁰⁵, thus diminishing the capability of the alternative measure of addressing EU public moral concerns on seal welfare.⁸⁰⁶ A

Moreover, even if the hunt itself has been accurately monitored for animal welfare, the maintenance of such accuracy would further require some form of traceability of the products to the market. COWI addressed the possibility of a "full chain of custody traceability system" as the most strict (and most effective) option but concluded that it could be costly and inefficient depending on the flexibility of the system. (See COWI 2010 Report, pp. 74-76 regarding Option 3). Although this evaluation was made in respect of the requirements of the Basic Regulation, we consider that it is indicative of the additional burdens of accurate transmission of information regarding regulatory compliance under the alternative measure. Indeed, Canada has also made indications that, after certifying and labelling conforming products from given hunts, there would be subsequent difficulties after the skins are sold and undergo secondary processing to preserve the identifying label. (See Canada's response to Panel question No. 85, para. 329 (regarding identification of seals harvested by Inuit hunters)).

⁸⁰² Commission Proposal, (Exhibit JE-9), Article 4(1)(c).

 803 Commission Proposal, (Exhibit JE-9), Article 4(1)(a). (emphasis added) The applicable animal welfare criteria are found in Annex II of the Proposal.

⁸⁰⁴ See Canada's and Norway's responses to Panel question No. 147. More specifically, it has been suggested that certification could take place at various levels, including the certification of countries, particular hunts, or vessels.

⁸⁰⁵ COWI advised that "identification requirements" comprise the first "key aspect" of being able to track conforming products to the market, and that different criteria (such as the IC and MRM requirements) would entail different "identification requirements" suited to the criteria in question. (See COWI 2010 Report, p. 80). Similarly, we consider that any *animal welfare* critera would demand "identification requirements" suited to the fact that the humaneness of killing will differ among individual seals within the same hunt (which is not the case for assessing compliance with IC or MRM requirements).

⁸⁰⁶ The complainants have pointed to various possible certification schemes as evidence for the feasibility of certification in their proposed alternative measure. However, these schemes are of limited assistance in determining the feasibility of a certification system for the alternative measure in that they: are country or hunter-based certification (e.g. the Agreement on International Humane Trapping Standards (Exhibit CDA-28), the EU's leghold trap certification scheme (Exhibits JE-8, NOR-80, NOR-105 and NOR-111) or Australia's kangaroo hunter licensing scheme); unrelated to animal welfare (e.g. the Friend of the Sea scheme for wild catch fisheries (Exhibits NOR-102, NOR-103 and NOR-104) and the Marine Stewardship Council fisheries scheme (Exhibits NOR-97, NOR-98, NOR-99, NOR-100 and NOR-101); or applied to situations and under circumstances that are significantly different from those of the seal hunt setting onto which they would have to be transposed (e.g. the EU Slaughter Regulation (Exhibit CDA-31) and the Agreement on the International Dolphin Conservation Programme (Exhibits NOR-95 and NOR-96) (concerning dolphins, which, unlike seals, are not the primary target of hunters pursued on a potentially large scale)).

Norway has argued that these schemes are intended to "*illustrate* the practical feasibility (i.e., reasonable availability) of implementing one or more of the *aspects* or *components* of Norway's proposed alternative measures. Needless to say, each scheme would have to be adapted to meet the needs of measure aimed at the welfare of seals." (Norway's response to Panel question No. 147, para. 245) (emphasis original) However, based on the fundamental differences in the requirements and/or subject matter of these schemes, it

⁸⁰⁰ We recall the arguments of Canada and Norway that the exceptions of the EU Seal Regime undermine the objective of animal welfare by allowing market access irrespective of the humaneness with which seals were killed. Inasmuch as inhumanely killed seals were not identified as such under the alternative measure, however, some products would similarly have market access despite being in conflict with the objective of protecting animal welfare.

⁸⁰¹ In this connection, we have also identified certain challenges to effective monitoring and enforcement of animal welfare standards in the context of seal hunting, which can occur over large territories with many participants. We consider that such difficulties of monitoring and enforcement compound the difficulties of aligning criteria for humane killing with the risks of seal hunting in a manner that does not unduly compromise animal welfare. (See paras. 7.219-7.221 above).

more rigorous certification scheme, on the other hand, could require exclusion of all seals killed in a way that did not meet the specified welfare requirements, which could lead to more overall hunting to obtain the desired amount of humanely killed seals.

7.499. Thus, certification schemes of greater specificity and rigor may be considered less reasonably available to the extent that they would require greater expenditure and practical challenges of implementation. At the same time, schemes that are not designed to account for the actual welfare outcomes of the seals from which products are derived may be considered comparatively more reasonably available.

7.3.3.3.4.5 Overall assessment of the reasonable availability of an alternative measure, taking account of risks of non-fulfilment would create

7.500. Both Canada and Norway assert the suitability of the alternative measure in light of the risks non-fulfilment of the European Union's objectives would create, primarily due to the alleged failure of the EU Seal Regime to fulfil its objectives and the lower risks and consequences that would arise under the alternative measure.⁸⁰⁷ Thus, in line with their arguments regarding the contribution to the measure's objective, Canada and Norway submit that the risks non-fulfilment of the objective of the measure would create are accepted under the EU Seal Regime (which allows products from inhumane hunts) and avoided by the alternative measure (which would only allow products meeting specified animal welfare criteria).808

7.501. The European Union submits that the "risk of non-fulfilment of the objective of protecting public morals is that the EU public would experience the same moral feelings that prompted the adoption of the EU Seal Regime".809

7.502. As described above, the alternative measure as proposed by the complainants appears to span a range of possible regimes of varying stringency and leniency with respect to animal welfare requirements and accuracy of certification. On the one hand, more stringent and accurate regimes would appear to pose precisely the "prohibitive costs or substantial technical difficulties" that can prevent an alternative measure from being considered to be reasonably available.⁸¹⁰ On the other hand, more lenient regimes would call into question the degree to which the alternative measure can contribute to the welfare of seals. Moreover, an alternative measure within this range may give rise to an increase in the number of seals hunted with the accompanying risks to seal welfare through restored market opportunities within the European Union. This may undermine the objective of the EU Seal Regime of reducing the overall number of seals killed inhumanely. We recall in this regard the Appellate Body's guidance that a responding Member cannot be reasonably expected to employ an alternative measure that involves a continuation of the very risk that the challenged measure seeks to halt.811

7.503. Further, the complainants' position rests on the premise that the alternative measure could calibrate the conditions of market access to the circumstances and risks existing in seal hunts. The complainants have not specified the substance of the exact regime (including the standard of animal welfare and method of certification) that would comprise their suggested alternative measure.⁸¹² Rather, they emphasize that the alternative measure focuses directly on animal

has not been clearly explained how such adaptation would take shape for seal hunts in a manner that would provide trustworthy assurance that animal welfare had been respected.

⁸⁰⁷ Canada's first written submission, paras. 691-695; Norway's first written submission,

paras. 909-911.

⁸⁰⁸ Canada's second written submission, paras. 329-333; Norway's second written submission, paras. 269-270. We note that Canada considers its view of the risks non-fulfilment would create to be similar to that of the European Union, with the difference that the risk is that seals would be killed in a way that causes avoidable (rather than excessive) pain and suffering. Canada's second written submission, para. 329.

⁸⁰⁹ European Union's response to Panel question No. 44, para. 152.

⁸¹⁰ Appellate Body Report, *Brazil – Retreaded Tyres*, para. 156, citing Appellate Body Report, US – Gambling, para. 308.

Appellate Body Report, EC – Asbestos, para. 174

⁸¹² See, e.g. Norway's second written submission, para. 283 ("It is not Norway's task to show precisely what standards should be adopted by the European Union in order to address animal welfare through a technical regulation conditioning access to the regulator's market."); Canada's comments on the European Union's response to Panel question No. 148, para. 111 (stating that the possible animal welfare requirements of its alternative measure "include the recommendations from EFSA, the recommendations of the IVWG, and standards applicable in hunts of other wild animals, such as deer and kangaroos").

- 145 -

welfare to allow products from commercial hunts that Canada and Norway contend achieve high levels of humane killing.⁸¹³ The complainants do not deny, however, and the evidence before us confirms, that inhumane killing and poor animal welfare outcomes do occur in seal hunts.⁸¹⁴ To that extent, the alternative measure would not be able to address the EU public's moral concerns with respect to their wish to not participate as consumers in products derived from seal hunts in general, and the reopening of the EU market could stimulate global demand so as to incentivize the killing of more seals. Although the contribution of the EU Seal Regime to the fulfilment of its objective is lowered by the implicit and explicit exceptions of the measure, the complainants have not clearly defined an alternative measure in respect of its separate components and their cumulative capability to address the moral concerns of the EU public.

7.504. In view of the evidence of the risks and challenges of seal hunting, and as a result of our assessment above of the proposed alternative with respect to its level of contribution to the relevant objective, we conclude that although the proposed alternative measure can be considered less restrictive of trade, the alternative measure is not reasonably available, taking account of the risks non-fulfilment would create.⁸¹⁵

7.3.3.3.5 Conclusion

7.505. In light of the above, we conclude that the EU Seal Regime is not more trade restrictive than necessary within the meaning of Article 2.2 of the TBT Agreement. The EU Seal Regime was found to be capable of making and does make contributions to the objective of addressing the EU public concerns on seal welfare. We further found that the alternative measure proposed by the complainants, which may be considered as less trade restrictive than the EU Seal Regime, is not reasonably available to the European Union, taking into account the risks that non-fulfilment of the European Union's objective would create.

7.3.4 Article 5

7.3.4.1 Conformity assessment procedure(s) (CAP): whether the EU Seal Regime is a CAP within the meaning of Annex 1

7.506. Annex 1.3 defines "conformity assessment procedures":

Any procedure used, directly or indirectly, to determine that relevant requirements in technical regulations or standards are fulfilled.

7.507. The explanatory note to the provision provides:

Conformity assessment procedures include, inter alia, procedures for sampling, testing, and inspection; evaluation, verification and assurance of conformity; registration, accreditation and approval as well as their combinations.

7.508. Canada argues that, given that the Basic Regulation constitutes a technical regulation, the process of evaluating whether seal products satisfy the conditions specified in the Regulation, particularly Articles 3, 5, and 6 of the Implementing Regulation, amounts to a conformity

⁸¹³ See Canada's and Norway's responses to Panel question No. 145.

⁸¹⁴ See EFSA Scientific Opinion, p. 3 ("[M]any seals can be, and are, killed rapidly and effectively without causing avoidable pain, distress, fear and other forms of suffering, using a variety of methods that aim to destroy sensory brain functions. However, there is strong evidence that, in practice, effective killing does not always occur ...").

⁸¹⁵ This conclusion is not changed by our consideration of measures applied to other product areas. We are mindful that examining enforcement measures applicable to the same behaviour relating to like products "may provide useful input in the course of determining whether an alternative measure which could 'reasonably be expected' to be utilized, is available or not". (Appellate Body Report, *Korea – Various Measures on Beef*, para. 170). However, we have described above the characteristics of seal hunting with respect to the physical environment, seal species, as well as the risks and challenges of seal hunting. We have noted that the evidence does not establish that effective stunning rates in seal hunts are comparable to those in commercial abattoirs, and in any case that the two situations differ significantly in areas of great relevance to the application of humane killing methods. Therefore, we do not consider that the situations to which other measures are applied are sufficiently similar to the circumstances of the seal hunt to assist in determining the availability of alternative measures.

assessment procedure.⁸¹⁶ Norway has not addressed the question of whether the EU Seal Regime is a conformity assessment procedure within the meaning of Annex 1.

7.509. The European Union argues that, because the EU Seal Regime is not a technical regulation within the meaning of Annex 1, the procedural provisions under the Implementing Regulation concerning the operation of the exceptions do not concern compliance with technical regulations and hence do not constitute "conformity assessment procedures" within the meaning of Annex 1.3.⁸¹⁷

7.510. The Panel found that the EU Seal Regime as a whole is a technical regulation laying down product characteristics. In addition, Articles 3, 5, and 6 of the Implementing Regulation establish the procedure for determining whether the specific requirements under the EU Seal Regime are fulfilled. Accordingly, we find that these provisions under the EU Seal Regime constitute a CAP within the meaning of the TBT Agreement.

7.3.4.2 Article 5.1.2: whether the CAP creates an unnecessary obstacle to international trade

7.511. Article 5.1.2 states:

Article 5

Procedures for Assessment of Conformity by Central Government Bodies

5.1 Members shall ensure that, in cases where a positive assurance of conformity with technical regulations or standards is required, their central government bodies apply the following provisions to products originating in the territories of other Members:

...

5.1.2 conformity assessment procedures are not prepared, adopted or applied with a view to or with the effect of creating unnecessary obstacles to international trade. This means, inter alia, that conformity assessment procedures shall not be more strict or be applied more strictly than is necessary to give the importing Member adequate confidence that products conform with the applicable technical regulations or standards, taking into account of the risks non-conformity would create.

7.512. The text and structure of Article 5.1.2 indicate that the provision consists of general obligations, set out in the first sentence, and an example of the general obligations, set out in the second sentence.⁸¹⁸

7.513. More specifically, the general obligations under the first sentence are not to prepare, adopt, or apply conformity assessment procedures with a view to or with the effect of creating unnecessary obstacles to international trade. The second sentence explains the meaning of the general obligations by prescribing a situation where a certain CAP may be found in violation of the obligation under the first sentence.⁸¹⁹ Therefore, a violation of the obligations set out in the first sentence could be established by demonstrating, for instance, that a given CAP has the effect of creating unnecessary obstacles to international trade or by showing a breach of the specific requirement in the second sentence.

⁸¹⁶ Canada's first written submission, paras. 705-708, and 716.

⁸¹⁷ European Union's first written submission, para. 419.

⁸¹⁸ In addition, the chapeau of Article 5.1 establishes the scope of the obligation as applying to

situations where "positive assurance of conformity with technical regulations or standards is required". In such cases, Members must ensure that their central government bodies apply the provisions of the sub-

paragraphs of Article 5.1. (See Canada's first written submission, para. 710; Norway's first written submission, para. 933; European Union's first written submission, para. 422).

⁸¹⁹ The term "inter alia" in the second sentence signifies that it is only one example of the requirements stemming from the general obligation set out in the first sentence.

7.514. Both Canada and Norway have developed arguments for their claim under Article 5.1.2 that may be analysed under both the first and second sentences of Article 5.1.2.⁸²⁰ We begin our examination of the complainants' claim under Article 5.1.2 with their contentions made based on the first sentence. We will then evaluate the complainants' arguments that have relevance under the second sentence of Article 5.1.2.

7.3.4.2.1 Whether the CAP creates an unnecessary obstacle to trade by failing to ensure the existence of a body to perform the CAP

7.3.4.2.1.1 Main arguments of the parties

Complainants

7.515. Canada submits that the failure by the European Union to ensure that a competent body exists to assess conformity with conditions that determine market access for qualifying seal products amounts to a violation of Article 5.1.2. Canada argues that in the absence of such a body, the CAP cannot function, thus preventing any trade in seal products satisfying the relevant conditions. Under the measure, third party entities are allowed to request that they be listed as "recognized bodies" that are authorized to verify compliance with the conditions, and to issue documents attesting to that compliance. Unless and until such applications are submitted and approved by the European Union, no attesting documents can be issued (i.e. no seal products can be imported into the European Union for the purpose of placing them on the market). Conditioning market access on the uncertain prospect that a third party entity will apply for, and be recognized by the European Union, as a body authorized to ascertain conformity, and issue a certificate to that effect, creates precisely the kind of uncertainty that the trade rules are meant to reduce.⁸²¹

7.516. Norway also argues that the Commission has prepared and adopted a CAP that lacks an essential element needed to enable trade to occur. Specifically, the Commission's failure to designate a body competent to assess and certify conformity results in an "institutional lacuna" in the CAP that leaves traders in conforming seal products reliant on a third party successfully seeking to become a recognized body. In this regard, Norway contends that a Member cannot make third parties responsible for the performance of its WTO obligations.⁸²² Consequently, the CAP is ineffective due to this institutional lacuna creating an effective ban on trade in conforming products. This ban is unnecessary because the Commission could have designated a "default" recognized body that would be competent, at all times, to assess and certify conformity.⁸²³

7.517. Further, Norway submits that an importing Member is responsible for ensuring that a body is available to assess conformity from the date of entry into force of a technical regulation. Thus, in designing and adopting a CAP, an importing Member is obliged to ensure that the system function from the date of its entry into force. If the European Union did not wish to establish a recognized body capable of functioning from the EU Seal Regime's entry into force, it should have given interested third parties an adequate opportunity to apply sufficiently far in advance so as to allow recognized body by that date, the European Union was obliged to designate a recognized body no later than the entry into force of the EU Seal Regime. Even if an importing Member were entitled, at the time of adopting a CAP, to await successful application from a third party to serve as a recognized body (*quod non*), such entitlement cannot endure indefinitely. In light of the enduring obligation under Article 5.1.2, if it becomes clear that third parties are unwilling or unable to serve as recognized bodies, an importing Member remains responsible for implementing a

⁸²⁰ In particular, the Panel considers that the complainants' arguments as to the failure to ensure the existence of a body to perform the CAP relate to obligations under the first sentence of 5.1.2. Additionally, Norway has developed specific arguments based on the elements of a necessity test under the second sentence of Article 5.1.2, and Canada has raised arguments relevant to less trade-restrictive alternative measures.

⁸²¹ Canada's first written submission, paras. 718-721.

⁸²² Norway refers to Appellate Body Report, *Thailand – Cigarettes (Philippines)*, para. 117; and Panel Report, *EC – Trademarks and Geographical Indications (Australia)*, paras. 7.309-7.310. See also Norway's second written submission, paras. 365-367 (citing the Appellate Body Reports in *Korea – Various Measures on Beef* and *US – Tuna II (Mexico)* to argue that an element of third-party choice does not relieve a Member from, or enable it to "contract out" of, its obligations under the covered agreements).

⁸²³ Norway's first written submission, paras. 944-951. See also Norway's second written submission, paras. 333-335.

- 148 -

conformity assessment system that functions to allow trade to occur in conforming products, for example by designating a recognized body.⁸²⁴

Respondent

7.518. The European Union considers that the complainants direct their claim under Article 5.1.2 against the fact that the Implementing Regulation establishes a third-party conformity assessment mechanism. As such, it submits that the Panel is called upon to determine whether Article 5.1.2 precludes the adoption of systems whereby the conformity assessment bodies need to be designated by the central government before issuing certificates of conformity and whereby they must continue to meet the designation conditions for as long as they issue such certificates. While the European Union acknowledges that the requirement to obtain a certificate under the EU Seal Regime, like any other regime requiring certification, constitutes an obstacle, this does not amount to the CAP having been prepared, adopted, or applied with a view to or with the effect of creating *unnecessary obstacles* to international trade.⁸²⁵ Further, the European Union argues that the CAP takes into account the particularities entailed by the certification of conformity with the IC and MRM exceptions. In the European Union's view, the TBT Agreement not only allows but encourages a number of features adopted in the Implementing Regulation.⁸²⁶

7.519. First, the European Union argues that the text of Article 5.1.2 does not impose an obligation to designate a public (central or local government) body in all cases where a positive assurance of conformity with technical regulations or standards is required.⁸²⁷ Second, the European Union contends that there is no basis in the text of Article 5.1.2 to argue that WTO Members should not allow government and non-governmental bodies from other WTO Members to apply to be recognized conformity assessment bodies.⁸²⁸

7.520. Finally, the European Union argues that there is no basis in the text of Article 5.1.2 to require a WTO Member to designate a "back-up" or "default" public (central or local government) body where it decides to put in place a system of designated (public and private) conformity assessment bodies. The European Union refers to the Explanatory note to point 3 of Annex 1 envisioning "registration, accreditation and approval as well as their combinations", concluding that the necessity of any system for accreditation/designation of certifying bodies must be assessed based on its own merits. This interpretation is supported by subsequent practice in the considerable diversity between the systems for accreditation/designation of conformity assessment bodies between WTO Members. While the European Union does not exclude the possibility that the designation of a public body may be a desirable approach in some cases, it calls on the Panel to reject a reading of the TBT Agreement whereby doing so would be a generalised obligation applicable to all conformity assessment procedures.⁸²⁹

⁸²⁴ Norway's second written submission, paras. 355-380. Specifically, Norway argues that "even if the European Union was not obliged to designate a recognized body when the conformity assessment procedures were adopted in August 2010 (*quod non*), the failings of the system have since become manifest, compelling the European Union to take action by designating a recognized body." (Ibid., para. 361. See also Norway's response to Panel question No. 87, para. 426).

⁸²⁵ Despite its concession that certification constitutes an obstacle on some level, the European Union argues that the specific requirement to be included on the list of recognized bodies under Article 6 of the Implementing Regulation does not pose an obstacle, but rather facilitates international trade by providing an accessible authoritative reference to all market operators both within and outside the territory of the European Union. The European Union also notes that Canada and Norway do not challenge the specific requirements that a recognized body must meet under Article 6 of the Implementing Regulation, and that the requirements for the issuance of attesting documents are necessary to give the European Union adequate confidence that imported seal products satisfy the relevant conditions. (European Union's first written submission, paras. 441-452).

⁸²⁶ European Union's first written submission, paras. 431-437.

⁸²⁷ The European Union refers to Article 8 of the TBT Agreement, which in its view constitutes relevant context for the interpretation of Article 5, to show that WTO Members may confer conformity assessment procedures to non-governmental (i.e. private) bodies.

procedures to non-governmental (i.e. private) bodies. ⁸²⁸ The European Union notes that Canada and Norway do not explicitly make such an argument, but submits that such a reading can be implied in their argument that a "default" body must exist at all times for as long as a CAP is in place. In this regard, the European Union cites Article 6 of the TBT Agreement concerning the recognition of conformity assessment in other Members by central government bodies.

⁸²⁹ European Union's first written submission, paras. 447-462.

- 149 -

7.3.4.2.1.2 Analysis by the Panel

7.521. To assess the complainants' claim under the first sentence of Article 5.1.2, we must examine the following points of contention between the parties: first, whether Article 5.1.2 permits a CAP that requires third-party accreditation and conformity assessment without creating or designating a default body independent of third-party approval; and, second, whether a CAP must be capable of allowing trade in conforming products to occur from the date of entry into force of a given measure.

7.522. Beginning with the first question, we first note that the text of Article 5.1.2 contains no precise indication of permitted and prohibited types of CAP. Thus, the text provides no direct prescription as to the permissibility of third-party accreditation, nor does it indicate whether such accreditation would require creation or designation of a default and/or back-up body.

7.523. Further, the context provided in other provisions of the TBT Agreement supports the view that there is some flexibility as to permissible CAP regimes, particularly with respect to the possibility of third-party accreditation. For example, the definition of a CAP in Annex 1 of the TBT Agreement encompasses, in addition to inspection and verification procedures, procedures for "registration, accreditation and approval as well as their combinations".⁸³⁰ We note that this explicit provision for accreditation does not contain any limitation as to the type of entity to be accredited. Moreover, the use of the term "inter alia" and the stipulation "as well as their combinations" suggest wide versatility in the types of regime that may be considered a CAP under the TBT Agreement.⁸³¹ We also note that Article 6 of the TBT Agreement provides for Members' *recognition* of conformity assessment from other Members "provided they are satisfied that those procedures offer an assurance of conformity ... equivalent to their own procedures".⁸³² To this end, it is explicitly contemplated that the system for recognizing conformity assessment from other Members may entail "limitation of the acceptance of conformity assessment results to those produced by designated bodies in the exporting Member".⁸³³

7.524. Therefore, based on our examination of the terms of Article 5.1.2, as well as its relevant context provided in other provisions of the TBT Agreement, we consider that Article 5.1.2 permits a system of third-party accreditation as part of a CAP. Accordingly, we do not consider that the third-party accreditation system under the EU Seal Regime (the CAP) violates Article 5.1.2. Nor do we find from the relevant text and context of Article 5.1.2 an obligation on the part of a responding Member to create or designate a default body pending accreditation or recognition of third-party entities to perform a CAP.

7.525. We next turn to the second question, namely whether, under the first sentence of Article 5.1.2, a CAP must be capable of allowing trade in conforming products to occur from the date of its entry into force. In other words, the question is whether failure to have in place a mechanism through which trade in regulated products can occur from the date of entry into force of a CAP results in a violation of the obligation under Article 5.1.2.

7.526. With respect to the circumstances of the measure at issue, the Implementing Regulation entered into force on 20 August 2010, which is the same day of the application of the Basic Regulation.⁸³⁴ This means that Article 3 of the Basic Regulation containing the IC and MRM exceptions also applied from 20 August 2010.⁸³⁵ The requirements of the CAP stipulated in the Implementing Regulation were published in the Official Journal of the European Union on 17 August 2010, three days prior to their entry into force along with Article 3 of the Basic Regulation. A system of third-party accreditation logically requires some time in processing the applications from their review and until ultimate approval. Given that the CAP requirements were published three days before their application with no other mechanism available, the earliest opportunity for

⁸³⁰ TBT Agreement, Annex 1.3, Explanatory Note.

⁸³¹ We note that the European Union has submitted evidence that other WTO Members, including Canada and Norway, have types of CAP in place with designation/accreditation of third-party certifying bodies. (See European Union's first written submission, para. 451; Exhibits EU-71, 72, 73). The complainants have also acknowledged that third-party conformity assessment schemes are "not uncommon". (See Canada's and Norway's responses to Panel question No. 87).

⁸³² Article 6.1 of the TBT Agreement.

⁸³³ Article 6.1.2 of the TBT Agreement.

⁸³⁴ See European Union's response to Panel question No. 149.

⁸³⁵ Basic Regulation, Article 8.

potential applicants to *initiate* this process would have been just shortly before the day of entry into force of the Implementing Regulation. Further, based on the numerous requirements in Article 6 of the Implementing Regulation, it would not have been reasonable to expect that the CAP could be completed prior to the Regime's entry into force.⁸³⁶

7.527. Consequently, as of the effective date of the EU Seal Regime, it was not possible for seal products to be examined or processed pursuant to the necessary CAP. Although third-party bodies could apply to become a recognized body under the EU Seal Regime upon its entry into force, the specific CAP established by the Implementing Regulation imposed the additional time necessary to examine and approve such a body according to specific criteria. Because of this period of review required for each application for inclusion in the list of recognized bodies, trade in qualifying seal products was practically not possible for some period of time following the entry into force of the EU Seal Regime.

7.528. The particular facts and circumstances described above therefore show that the measure in question was established such that the CAP was not capable of allowing trade in conforming products to occur on the date of its entry into force. In light of this, we conclude that the CAP had the effect of creating unnecessary obstacles to international trade inconsistently with the first sentence of Article 5.1.2.⁸³⁷

7.529. We next turn to whether the EU Seal Regime CAP creates an unnecessary obstacle to international trade because it is more strict or applied more strictly than necessary within the meaning of the second sentence.

7.3.4.2.2 Whether the CAP creates an unnecessary obstacle to trade because it is "more strict or ... applied more strictly than is necessary to give the importing Member adequate confidence" of conformity

7.3.4.2.2.1 Main arguments of the parties

Complainants

7.530. Norway submits that the parties do not disagree that the CAP set forth in the EU Seal Regime is, by definition, trade-restrictive. Moreover, given that applications to become a recognized body could not be made, much less approved, before entry into force of the EU Seal Regime, the CAP necessarily gave rise to a ban on trade in conforming seal products, which could not demonstrate compliance with the relevant requirements.⁸³⁸

7.531. Norway contends that the European Union's omission to establish a recognized body, which necessarily prevented lawful trade in conforming seal products, does not contribute to giving the European Union confidence that conforming seal products meet the relevant requirements. Rather than ensuring that a conformity assessment system operates effectively to give confidence to the

⁸³⁶ This finds additional support in the time taken to review and process applications actually received, as discussed in the subsequent section on Article 5.2.1.

⁸³⁷ We note in this regard the suggestion by Norway that the European Union could have provided for advance notice and/or opportunity for interested parties to apply for recognition. (See, e.g. Norway's response to Panel question No. 50, para. 264). Although the European Union has referenced consultations conducted during the legislative process, the evidence in this regard does not reveal any specific engagement as to conformity assessment. Further, we note that these consultations took place prior to the adoption of the Basic Regulation (and thus before the requirements of the CAP were established). (See Canada's and Norway's comments on the European Union's response to Panel question No. 149).

We are also mindful of the fact that certain entities have applied and been approved as recognized bodies by the Commission. This subsequent approval, however, does not alter the fact that the CAP under the EU Seal Regime was initially incapable of assessing conforming products, irrespective of third-party action or inaction. Given the circumstances of this dispute, we are not able to make specific findings as to measures that may have been available to the European Union to enable the assessment of conformity under the EU Seal Regime from its entry into force.

⁸³⁸ Norway's second written submission, paras. 340-345. See also Norway's first written submission, para. 949 (citing Panel Report, *Brazil – Retreaded Tyres*, para. 7.114) ("A ban on the importation of conforming seal products is, of course, the most trade-restrictive obstacle to trade in these products that can be envisaged"). We note that Norway equates "strictness" with "trade-restrictiveness". (See Norway's second written submission, para. 340 (focusing on "the strictness (or, put another way, trade-restrictiveness) of the measure or its application")).

importing Member, the European Union's omission necessarily renders the system ineffective. Thus, the procedures have been designed in a manner that deprives the European Union of any opportunity to verify that conforming seal products meet the relevant requirements. Further, traders seeking access to the EU market are deprived of any opportunity to demonstrate that their products meet the requirements.⁸³⁹

7.532. Norway proposes that the European Union could have adopted a less-trade restrictive alternative by designating a recognized body that would be competent, at all times (or at least in the absence of third-party recognized bodies), to assess and certify conformity. According to Norway, this body could have been designated at the level of the European Union, including the Commission itself, or the Commission could have established a series of regional bodies within the European Union. Such a system would ensure that the CAP would always function to enable traders to secure approval for conforming seal products, irrespective of third party action, and would fully achieve the European Union's objective of giving itself confidence that imported seal products meet the relevant requirements.⁸⁴⁰

7.533. Canada states that evaluation of the CAP would include consideration of reasonably available, less trade-restrictive alternative measures, "such as supplier declaration of conformity, rather than a third party conformity assessment (3PCA) procedure".⁸⁴¹ Canada further asserts that where product safety is not the central concern of a technical regulation, schemes based on supplier declaration of conformity are more common than third-party conformity assessment.⁸⁴²

Respondent

7.534. The European Union argues that the very requirement to obtain a certificate constitutes an obstacle. In the case of the EU Seal Regime CAP, however, the degree of trade-restrictiveness is not more than necessary for the relevant purpose.⁸⁴³

7.535. The European Union argues that the CAP serves the purpose of providing adequate assurance that the only seal products placed on the market are those that comply with the exceptions established under the EU Seal Regime. In particular, the requirements of the Implementing Regulation ensure that recognized bodies are both impartial and capable of verifying and attesting that the requirements to benefit from an exception have been fulfilled.⁸⁴⁴

7.536. Regarding the alternative measures proposed by the complainants, the European Union argues that Canada neither presents any concrete alternative nor shows how such an alternative would be equally effective and less trade restrictive than the mechanism put in place by the Implementing Regulation.⁸⁴⁵ Specifically, Canada has failed to establish that supplier declaration of conformity would give the European Union, "as the importing Member, adequate confidence that products conform with the applicable regulation" taking into account the risks of non-conformity.⁸⁴⁶

7.537. The European Union argues that Norway failed to demonstrate why the proposed alternative would be equally effective in determining the product's conformity with the regulation concerned, and less trade restrictive than the CAP at issue.⁸⁴⁷ The European Union contends that in a context, like the one at issue, where certification can entail inspections of compliance with requirements (such as those of the IC and MRM exceptions) at the place of origin of the product, the designation of a default public authority in the European Union could have a greater trade distortive effect than the CAP under the Implementing Regulation. In particular, the issuance of certificates of compliance by a single central government authority would probably entail a less efficient and costlier certification mechanism for operators.⁸⁴⁸ In this light, what Norway describes

⁸⁴⁷ European Union's first written submission, para. 439.

⁸⁴⁸ The European Union notes in this respect that pursuant to Article 5.2.5 of the TBT Agreement,

conformity assessment authorities are entitled to charge for "communication, transportation and other costs

⁸³⁹ Norway's second written submission, paras. 346-348.

⁸⁴⁰ Norway's second written submission, paras. 349-354.

⁸⁴¹ Canada's response to Panel question No. 49, para. 192.

⁸⁴² Canada's response to Panel question No. 87, para. 333.

⁸⁴³ European Union's first written submission, paras. 431-434, 445.

⁸⁴⁴ European Union's first written submission, paras. 432, 441-446.

⁸⁴⁵ European Union's first written submission, para. 438.

⁸⁴⁶ European Union's second written submission, para. 323.

as an "institutional lacuna" is rather a mechanism to ensure a level playing field and avoid giving an inherent systemic advantage to trade in seals products that would originate in the European Union or its immediate proximity.⁸⁴⁹

7.538. Lastly, the European Union comments upon the complainants' acknowledgement that no Canadian or Norwegian entities have applied to become recognized bodies, and interprets the reasons given to be grounded in the lack of desire of the potential beneficiaries to make use of the system rather than in alleged deficiencies in the set-up of the system itself. On this basis, the European Union contends that the existence of a default recognized body would not have altered the considerations motivating the decision of Canadian and Norwegian entities not to submit a request.850

7.3.4.2.2.2 Analysis by the Panel

7.539. Given the similarities in its text and structure to the second sentence of Article 2.2 of the TBT Agreement, the Panel considers, and the parties do not dispute, that the requirement under the second sentence of Article 5.1.2 calls for a relational analysis similar to that applied in Article 2.2, namely a weighing and balancing of a measure's trade-restrictiveness, degree of its contribution to an objective, and possible less trade-restrictive alternative measures. In the context of a claim under Article 5.1.2, however, the analysis relates to the fulfilment of only one objective: giving positive assurance that the relevant requirements of the technical regulation are fulfilled.

7.540. With respect to trade-restrictiveness, it is undisputed that the CAP necessarily has some restrictive effect to the extent that it imposes additional conditions in order for the trade in seal products to be permitted. In this case, the obligation to obtain attesting documentation from a body that has applied and been approved for recognition comprises an obstacle for those wishing to place seal products on the EU market pursuant to the exceptions of the EU Seal Regime. However, the question of whether the CAP amounts to an *unnecessary* obstacle to international trade depends on the other factors of the analysis to be weighed and balanced.

7.541. Turning to the contribution of the CAP to the assurance of conformity, we recall that the categories of seal products allowed on the EU market are addressed under Article 3 of the Basic Regulation and Articles 3, 4, and 5 of the Implementing Regulation. For seal products potentially qualifying under these exceptions, we initially note that the CAP covers the assessment of products from activities that are conducted in locations outside and remote from the European Union.

7.542. Any entity seeking approval to be a recognized body under the EU Seal Regime must demonstrate that it meets the requirements set out in Article 6 of the Implementing Regulation. Given the inherent nature of third-party accreditation, we consider that the degree of contribution to assurance of conformity is to be judged with regard for the *capability* and *credibility* of an authorized body in providing positive assurance of conformity with the substantive requirements of the EU Seal Regime.⁸⁵¹ In this vein, the CAP expressly addresses the relevant capabilities of an applicant body by requiring "the capacity to ascertain that the requirements of" the IC or MRM exceptions are met; "the ability to monitor compliance with the requirements" of the IC and MRM exceptions; and "the capacity to issue and manage attesting documents ... as well as process and archive records".⁸⁵² Further, other provisions answer to the credibility of an applicant entity by requiring that the entity be able to avoid conflicts of interest in addition to being subject to an independent third-party audit.853

arising from differences between the location of facilities of the applicant and the conformity assessment body". ⁸⁴⁹ European Union's first written submission, paras. 463-468.

⁸⁵⁰ European Union's second written submission, paras. 330-332.

⁸⁵¹ See European Union's first written submission, para. 442 (explaining that the purpose of the application and criteria for recognition "is to ensure that the entity is and remains: capable of verifying and attesting that the requirements to benefit from an exception have been fulfilled; and impartial"). (emphasis

original) ⁸⁵² Implementing Regulation, Article 6(1)(b), (c) and (e).

⁸⁵³ Implementing Regulation, Article 6(1)(d) and (g).

7.543. In this connection, we observe that the primary function of recognized bodies under the CAP pertains to inspection and certification of conformity with IC and MRM requirements.⁸⁵⁴ In view of the particular function of recognized bodies, we consider that the EU Seal Regime CAP contributes to the assurance of conformity with the relevant requirements of the EU Seal Regime through its provision for the capacity and impartiality of applicant entities.

7.544. We next consider the reasonable availability of less-trade restrictive alternative measures. The comparison of a possible alternative CAP under Article 5.1.2 should examine whether the alternative CAP is less trade restrictive than the CAP in question and would provide an equivalent assurance of conformity. We recall in this regard that the complainants bear the burden of identifying a possible alternative measure.⁸⁵⁵ Canada proposes a supplier declaration of conformity, and Norway proposes the designation of a recognized body in the absence of third-party recognized bodies.

7.545. With respect to Canada's suggested alternative of supplier declaration of conformity, such an alternative would appear to be less restrictive in that it would dispense with suppliers' dependence on third-party accreditation to assess conformity. However, Canada has not provided specific arguments as to how such an alternative would provide an equivalent assurance of conformity as the current CAP. Moreover, Canada has not indicated whether such suppliers would be subject to some form of approval based on capacity/credibility, or instead would have automatic and undifferentiated eligibility to make conformity declarations. Therefore, we do not consider that Canada has made a *prima facie* case establishing that an alternative of supplier declaration of conformity would be less trade restrictive and make an equivalent contribution to assuring conformity.

7.546. We observe that Norway's suggested alternative does not fundamentally differ from the EU Seal Regime CAP, but rather would supplement the existing CAP with a designated entity to enable trade in conforming seal products. In this regard, Norway specifies that the designated body in the alternative CAP would exist within the European Union. As mentioned, the function of a recognized body under the EU Seal Regime CAP pertains to the verification and inspection of seal products for compliance with IC and MRM requirements. Further, as pointed out by the European Union, certification of conformity in the present context may require verification and inspection at the place of origin of the product in order to obtain positive assurance that all requirements are met. An entity based in the European Union would therefore be required to manage the added difficulties of assessing the conformity of products that are potentially derived from hunts occurring at considerable distances outside the European Union. In our view, this may have implications for the level of contribution to fulfilment of the relevant objective by Norway's alternative CAP, and we have not been provided any evidence or arguments as to how such an entity would make an equivalent contribution to confidence of conformity. Furthermore, the difference of location between applicants and the conformity assessment body may result in the imposition of additional costs and burdens in order to verify compliance.⁸⁵⁶

7.547. In conclusion, we do not consider that Canada and Norway have established that an alternative CAP would make a contribution to confidence of conformity at the same level as the current CAP. We therefore reject the complainants' claim that the EU Seal Regime CAP is more strict or applied more strictly than is necessary to give adequate confidence of conformity with the applicable technical regulations within the meaning of the second sentence of Article 5.1.2.

⁸⁵⁴ This role is distinct from the more limited responsibilities of "competent authorities" under Article 9 of the Implementing Regulation, which provides for narrowly defined functions for verification, control, and repository. The European Union also notes that competent authorities would be precluded from serving as recognized bodies due to the Implementing Regulation's prohibitions on conflicts of interest. (European Union's response to Panel question No. 86, paras. 243-247).
⁸⁵⁵ See, e.g. Appellate Body Reports, US – Tuna II (Mexico), para. 323; US – Gambling, para. 309;

⁸⁵⁵ See, e.g. Appellate Body Reports, *US* – *Tuna II (Mexico)*, para. 323; *US* – *Gambling*, para. 309; *China* – *Publications and Audiovisual Products*, para. 319.

⁸⁵⁶ In this regard, we also note that Article 5.2.5 of the TBT Agreement addresses "fees imposed for assessing the conformity of products originating in the territories of other Members" and provides that such fees may be imposed "taking into account communication, transportation and other costs arising from differences between location of facilities of the applicant and the conformity assessment body". (See also European Union's first written submission, para. 467).

7.3.4.3 Article 5.2.1: whether the EU Seal Regime is "undertaken and completed as expeditiously as possible"

7.3.4.3.1 Main argument of the parties

7.3.4.3.1.1 Complainants

7.548. The complainants contend that the obligation in Article 5.2.1 is similar to that found in Annex C(1)(a) of the SPS Agreement which requires control, inspection and approval procedures to be "undertaken and completed without undue delay". The complainants accordingly cite to the interpretation of the panel in *EC – Approval and Marketing of Biotech Products* in the context of the SPS Agreement to further develop the meaning of the provisions in Article 5.2.1. In particular, the complainants note that the SPS Agreement was found to allow Members "the time that is reasonably needed to determine with adequate confidence whether their relevant SPS requirements are fulfilled".⁸⁵⁷ Notwithstanding this allowance, the requirement to carry out such actions "without undue delay" means that "approval procedures be undertaken and completed with no unjustifiable loss of time".⁸⁵⁸ They also describe the ordinary meaning of the word "expeditiously" to refer to the performance of an action as quickly as possible without compromising the effectiveness of the action.⁸⁵⁹

7.549. Against the interpretive benchmark of unjustifiable delay or loss of time, Canada and Norway emphasize that Article 5.2.1 requires that conformity assessment procedures be undertaken *and* completed "as expeditiously as possible".⁸⁶⁰ The complainants contend that the EU Seal Regime violates this obligation by failing to create or designate a body capable of conducting the conformity assessment, instead leaving it to other entities to seek authorization for the performance of this task. In the complainants' view, this results in the effective impossibility of determining conformity with the requirements set out in the EU Seal Regime.⁸⁶¹

7.550. Canada additionally asserts that, even if the "failure to create a designated body itself does not *per se* give rise to a violation of Article 5.2.1", the sheer lapse of time during which no bodies have been recognized as competent to issue attesting documents "amounts to a failure to ensure that the conformity assessment procedure established by the Implementing Regulation can be undertaken and completed as expeditiously as possible".⁸⁶²

7.551. Highlighting the lapse of more than two years for the approval of a Greenlandic applicant, Norway argues that Article 5.2.1 suggests that a violation of this provision is established only if the more rapid conduct of a CAP is "possible". Norway reiterates its view that it would be "possible" for the European Union to conduct its procedures more rapidly than under the current Regime by designating a body that could act in a timely fashion, without making its procedures depend on the desire of a third party entity to seek, and secure, approval as a recognized body.⁸⁶³

⁸⁵⁷ Canada's first written submission, para. 725 (citing Panel Report, *EC – Approval and Marketing of Biotech Products*, para. 7.1498); Norway's first written submission, para. 955 (citing a similar statement made in Panel Report, *EC – Approval and Marketing of Biotech Products*, para. 7.1499).

⁸⁵⁸ Canada's first written submission, para. 725 (citing Panel Report, *EC – Approval and Marketing of Biotech Products*, para. 7.1495). See also Norway's first written submission, para. 954.

⁸⁵⁹ Canada's first written submission, para. 724 ("a requirement to undertake and complete something 'as expeditiously as possible' means that the action must be performed as quickly as possible without compromising the effectiveness of the action, having regard to its purpose and the surrounding circumstances"); Norway's first written submission, para. 954 ("The ordinary meaning of the term 'expeditiously' refers to action taken as speedily as possible, without compromising the quality or effectiveness of the action at issue".).

⁸⁶⁰ Canada's first written submission, para. 727; Norway's first written submission, para. 957; Norway's second written submission, para. 385.

⁸⁶¹ Canada's first written submission, paras. 728-731 (asserting that "it is legally and practically impossible for seal products to be certified as conforming with the conditions, even if, as a matter of fact, they do"); Norway's first written submission, paras. 958-961 (describing the practical consequences of the "institutional lacuna" left by the EU Seal Regime conformity assessment procedures).

⁸⁶² Canada's first written submission, para. 732.

⁸⁶³ Norway's second written submission, paras. 387-393.

- 155 -

7.3.4.3.1.2 Respondent

7.552. The European Union agrees that the jurisprudence on Annex C(1)(a) of the SPS Agreement is relevant in interpreting the obligation under Article 5.2.1.⁸⁶⁴ Specifically, the European Union refers to the panel's findings in *EC – Approval and Marketing of Biotech Products* that "the phrase 'undertake and complete' covers all stages of approval procedures and should be taken as meaning that, *once an application has been received*, approval procedures must be started and then carried out from beginning to end".⁸⁶⁵ The European Union concurs with the complainants in citing to the requirement that there be "no unjustifiable loss of time"⁸⁶⁶, but additionally cites to the finding that "*delays attributable to action, or inaction, of an applicant must not be held against a Member*" in a panel's determination of "undue delay".⁸⁶⁷

7.553. On these foundations, the European Union addresses the complainants' arguments as consisting of claims against the Implementing Regulation *as such* and *as applied.*⁸⁶⁸ Regarding the claim against the Implementing Regulation as such, that European Union argues that the TBT Agreement does not oblige Members to create "default" conformity assessment bodies. Moreover, the phrase "undertake and complete" as interpreted by the panel in *EC – Approval and Marketing of Biotech Products* establishes that obligations are triggered only once an application for conformity assessment has been received. According to the European Union, despite providing for broad potential eligibility under the Implementing Regulation, it has received "only twelve applications to be added on the list of recognized bodies". The European Union contends it has discharged the duties of good faith under Article 5.2.1 and the low interest of other public authorities and private entities is not attributable to the European Union.⁸⁶⁹

7.554. As to the separate claim by Canada against the Implementing Regulation as applied, the European Union first argues that such a claim is outside the Panel's terms of reference based on the text of Canada's panel request.⁸⁷⁰ Alternatively, the European Union argues that any delays in processing applications to date are not imputable to the European Union as they are due to the deficiency of the application referenced by Canada (i.e. from Greenland).⁸⁷¹

7.3.4.3.2 Analysis by the Panel

7.555. Article 5.2.1 provides:

5.2 When implementing the provisions of paragraph 1, Members shall ensure that:

5.2.1 conformity assessment procedures are undertaken and completed *as expeditiously as possible* and in a no less favourable order for products originating in the territories of other Members than for like domestic products; (emphasis added)

7.556. The chapeau of Article 5.2 directly references Article 5.1 and clarifies the relationship between the obligations in the sub-paragraphs of the two provisions. Specifically, Article 5.2 provides that "[w]hen *implementing* the provisions of" Article 5.1, Members must adhere to the specific obligations laid out in the sub-paragraphs of Article 5.2 with respect to the implementation of the CAP.⁸⁷² Among the detailed rules contained in Article 5.2, the complainants have raised a challenge as to whether the EU Seal Regime CAP is "undertaken and completed as expeditiously as possible" within the meaning of the first clause of Article 5.2.1.

⁸⁶⁴ European Union's first written submission, para. 475.

 ⁸⁶⁵ European Union's first written submission, para. 476 (citing Panel Report, *EC – Approval and Marketing of Biotech Products*, para. 7.1494) (emphasis added by the European Union)
 ⁸⁶⁶ European Union's's first written submission, para. 477 (citing Panel Report, *EC – Approval and*

 ⁸⁶⁶ European Union's's first written submission, para. 477 (citing Panel Report, *EC – Approval and Marketing of Biotech Products*, para. 7.1495).
 ⁸⁶⁷ European Union's's first written submission, para. 478 (citing Panel Report, *EC – Approval and*

⁸⁶⁷ European Union's's first written submission, para. 478 (citing Panel Report, *EC – Approval and Marketing of Biotech Products*, para. 7.1497) (emphasis added by the European Union)

⁸⁶⁸ European Union's's first written submission, paras. 471-472.

⁸⁶⁹ European Union's's first written submission, paras. 479-484; second written submission,

paras. 330-332.

⁸⁷⁰ European Union's's first written submission, para. 485.

⁸⁷¹ European Union's first written submission, paras. 486-487.

⁸⁷² See also Canada's response to Panel question No. 50; Norway's response to Panel question No. 50; second written submission, paras. 381-382.

7.557. To assess the complainants' claim under Article 5.2.1, we must therefore examine the meaning of the phrase "undertaken and completed as expeditiously as possible".

7.558. First, we address the question concerning the precise point in time when the obligation to "undertake and complete" a CAP is triggered during the implementation process of a CAP. The parties dispute whether this obligation predates the receipt of an application for recognition under the CAP.

7.559. In our view, the chapeau of Article 5.2 dictates that the detailed obligations of the subparagraphs are confined to the *implementation* of the more general obligations under Article 5.1. While Article 5.1.2 covers the entire process in which a CAP is "prepared, adopted or applied", Article 5.2.1 applies only to the implementation stage of the process. This means that the obligations of Article 5.2 are not coterminous with those of Article 5.1, but limited to the application of a CAP.

7.560. We further note that the SPS Agreement contains a similar obligation as that contained in Article 5.2.1 of the TBT Agreement; Annex C(1)(a) of the SPS Agreement provides:

Members shall ensure, with respect to any procedure to check and ensure the fulfilment of sanitary or phytosanitary measures, that ... such procedures are undertaken and completed *without undue delay*. (emphasis added)

7.561. Given the similarity in the text, we agree with the parties that there are certain parallels in the terms and scope of Article 5.2.1 of the TBT Agreement and Annex C(1)(a) of the SPS Agreement. Both provisions pertain to procedures adopted to ensure fulfilment of specific requirements contained in a measure falling under the TBT Agreement or the SPS Agreement, respectively.⁸⁷³

7.562. Regarding the phrase "undertaken and completed" in particular, the panel in *EC – Approval* and *Marketing of Biotech Products* concluded:

[t]he verb 'undertake' makes clear that Members are required *to begin, or start, approval procedures after receiving an application for approval.*⁸⁷⁴ ... Thus, in our view, the phrase 'undertake and complete' covers all stages of approval procedures and should be taken as meaning that, *once an application has been received*, approval procedures must be started and then carried out from beginning to end.⁸⁷⁵ (emphasis added)

7.563. Based upon the relevant text and context of Article 5.2.1, and consistent with the interpretive guidance of the same phrase in the SPS Agreement, we consider that "undertaken and completed" in Article 5.2.1 applies to the implementation of a CAP from the moment when an application for recognition has been received and through the completion of the process. In the present dispute, the application in question is for inclusion on the list of recognized bodies under Article 6 of the Implementing Regulation. Our understanding of the temporal scope of the obligation in Article 5.2.1 is unaltered by the fact that the application is for accreditation to perform conformity assessment, rather than a direct application for certification of conformity to an existing authorized entity.⁸⁷⁶

⁸⁷³ There is also overlap in the indicative terms provided in the explanatory notes for "conformity assessment procedures" under the TBT Agreement and "control, inspection and approval procedures" under the SPS Agreement. In particular, these terms and their explanatory notes coincide with respect to "sampling", "testing", and "inspection", and the inclusion of "*inter alia*" to indicate the non-exhaustive nature of the list.
⁸⁷⁴ The dictionary meanings of the verb "undertake" include "[t]ake on (an obligation, duty, task, etc.);

⁸⁷⁴ The dictionary meanings of the verb "undertake" include "[t]ake on (an obligation, duty, task, etc.); commit oneself to perform; begin (an undertaking, enterprise, etc.)". The *New Shorter Oxford English Dictionary*, L. Brown (ed.) (Clarendon Press, 1993), Vol. II, p. 3476. The French and Spanish versions of Annex C(1)(a), first clause, also support this reading. The French version uses the verb "engager", the Spanish version the verb "iniciar". (footnote original)

⁸⁷⁵ Panel Report, EC – Approval and Marketing of Biotech Products, para. 7.1494 (emphasis added)

 $^{^{876}}$ See, e.g. Japan's third-party submission, para. 58 (distinguishing the present dispute from *EC* – *Approval and Marketing of Biotech Products* where "the issue did not relate to the designation of the certifying body but rather to a *de facto* moratorium on approvals").

7.564. Further, we observe that the adverb "expeditiously" indicates that the obligation relates to the speed and/or timing of the performance of a CAP.⁸⁷⁷ At the same time, the term "expeditiously" is qualified by the phrase "as possible". We take this qualification to be based on the fundamental purpose of any CAP to secure "a positive assurance of conformity with technical regulations"⁸⁷⁸, and recognition that doing so may necessarily entail some time to determine that relevant requirements are fulfilled.

7.565. In this connection, we also take note of the interpretation by the panel in *EC – Approval and Marketing of Biotech Products* of the phrase "without undue delay" to mean that approval procedures were required to be undertaken and completed "with no unjustifiable loss of time".⁸⁷⁹ The panel similarly accounted for the function of approval procedures to check and ensure fulfilment of SPS requirements, and reasoned on this basis that "Members applying such procedures must in principle be allowed to take the time that is reasonably needed to determine with adequate confidence whether their relevant SPS requirements are fulfilled".⁸⁸⁰

7.566. We agree with the approach of the panel in EC - Approval and Marketing of Biotech Products. While the duty of expeditious conformity assessment prescribed in Article 5.2.1 must be carried out so as not to create an unnecessary obstacle to trade, such duty of the regulating Members must be balanced against the regulating Members' need and practical ability to make an adequate conformity assessment.⁸⁸¹ Therefore, in our view, Article 5.2.1 permits the time that is reasonably required to assess conformity with technical requirements.

7.567. Turning to the question of whether the CAP in question has been undertaken "as expeditiously as possible" in the particular circumstances of this dispute, the parties' arguments mainly concern the issue of the justifiability and attribution of delay caused under the CAP.

7.568. First, the complainants criticize the design of the CAP as creating an absence, attributable to the European Union, of any body competent to "undertake and complete" the required CAP. In other words, we understand the complainants' arguments to be based on the same grounds as those raised in the context of their claim under Article 5.1.2, namely the absence of a default body (i.e. an "institutional lacuna"). We recall our finding above on this question that Article 5.1.2 does not impose an obligation to create or designate a default body pending accreditation or recognition of third-party entities to perform a CAP. In light of this finding, we are not persuaded of the complainants' contention that the absence of a default body in the CAP leads to a violation of Article 5.2.1.⁸⁸²

7.569. Next, regarding the arguments based on the actual time taken for the European Union to process conformity assessment applications, the European Union attributes delay in the application process to the complainants themselves for failing to initiate the procedures that would prompt the European Union's obligation to undertake and complete the CAP as expeditiously as possible. The European Union also contends that delays in the processing of requests from Sweden and Greenland cannot be considered attributable to the European Union.⁸⁸³ We must thus assess the evidence relating to the application of the CAP with respect to those applications that were submitted to the Commission. On the basis of our assessment, we evaluate whether the CAP, as applied in those cases, was not undertaken and completed as expeditiously as possible in compliance with Article 5.2.1.

⁸⁸⁰ Panel Report, *EC* – *Approval and Marketing of Biotech Products*, para. 7.1498. The panel further stated: "Put another way, we view Annex C(1)(a), first clause, essentially as a good faith obligation requiring Members to proceed with their approval procedures as promptly as possible, taking account of the need to check and ensure the fulfilment of their relevant SPS requirements."

⁸⁸³ See, e.g. European Union's second written submission, para. 327.

⁸⁷⁷ Shorter Oxford English Dictionary, 6th edn, A. Stevenson (ed.) (Oxford University Press, 2007), Vol. 1, p. 898 ("1. Speedily performed or given; conducive to speed performance").

⁸⁷⁸ Article 5.1 of the TBT Agreement.

⁸⁷⁹ Panel Report, EC – Approval and Marketing of Biotech Products, para. 7.1495.

⁸⁸¹ See Annex 1.3 of the TBT Agreement (definition of "conformity assessment procedures").

⁸⁸² We note that that there has not been any contention that the European Union has prevented the submission or receipt of applications in its administration of the CAP, but rather that there were inadequate incentives to apply because of the substantive requirements of the EU Seal Regime. (See Canada's and Norway's responses to Panel question Nos. 84 and 85). Because no applications for recognition from Canadian or Norwegian entities have ever been submitted, we do not consider the obligations of Article 5.2.1 to have been breached by the European Union in respect of the non-approval of such entities where it was not possible to "undertake" the CAP in the first place.

- 158 -

7.570. Regarding the application from Greenland, on 21 February 2011, the Greenlandic government notified that its Department for Fisheries, Hunting, and Agriculture (APNN) would serve as certifying authority under Article 6 of Implementing Regulation.⁸⁸⁴ On 7 July 2011, the Commission sent a formal deficiency letter stating that, although there was no evidence that the entity was ineligible, there was insufficient documentary evidence to demonstrate fulfilment of all of the requirements of Article 6 of Implementing Regulation.⁸⁸⁵ The Greenlandic authorities provided an initial reply letter on 5 January 2012 with supporting documentation⁸⁸⁶ and subsequently provided supplementary information on 1 November 2012⁸⁸⁷ and 29 January 2013.⁸⁸⁸ The Commission issued its decision recognising the APNN of Greenland on 25 April 2013.889

7.571. With respect to the applications received from Swedish entities under the CAP, the first applications were received on 20 January 2011 from 11 county administrative boards designated by the Swedish government.⁸⁹⁰ A formal deficiency letter was sent on 7 July 2011 (the same date as the letter to Greenland) similarly stating that there was no evidence that the applicants were ineligible and detailing documentary insufficiencies of the application.⁸⁹¹ The Swedish authorities replied on 6 October 2011 with an item-by-item response to the Commission's deficiency letter⁸⁹², and the Commission issued its decision recognising the Swedish bodies on 18 December 2012.⁸⁹³

7.572. It is thus clear that there were some delays of varying length in the exchanges between the Commission and the respective applicant bodies in Sweden and Greenland.

7.573. In the case of Greenland, we observe the multiple exchanges of information with Greenlandic authorities in the course of the recognition process. The formal deficiency letters from the Commission, sent several months after the applications, itemize the deficiencies of each application with specific reference to the enumerated requirements of Article 6 of the Implementing Regulation. The letters also contain references to information provided in the original applications and request elaboration and additional documentation on certain points with indications of the type of evidence that would be suitable, such as legal references or factual explanations.

7.574. The first response from Greenland to the deficiency letter was submitted approximately six months afterwards and acknowledges "the tardy reply, which is the result of internal discussion and the number and scope of the questions". This response goes on to provide extensive explanation of export authorizations as well as the licensing and monitoring scheme of the commercial seal hunt.

7.575. Subsequent communications elaborate upon relevant laws and monitoring functions, while also indicating that there may have been meetings and exchanges outside the exchange of written correspondence and documentation.⁸⁹⁴ Given the multiple submissions of documentation made to the Commission as well as indications of other forms of engagement, the evidence regarding the Greenlandic application does not enable us to make a precise assessment of the cause or attribution of the overall length of the proceedings. Nor have the complainants provided a sufficient basis for us to conclude that the procedures under the CAP with respect to the specific applications from Greenland were not undertaken and completed as expeditiously as possible within the meaning of Article 5.2.1

⁸⁸⁴ Greenland's request pursuant to Article 6(2) of the Implementing Regulation, (Exhibit EU-148).

⁸⁸⁵ Deficiency letter to Greenland of 7 July 2011, (Exhibit EU-150).

⁸⁸⁶ Greenland's response of 5 January 2012, (Exhibit EU-151) and Supporting documentation to the submission of 5 January 2012, (Exhibit EU-152).

 $^{^{887}}$ Supplementary document received from Greenland on 1 November 2012, (Exhibit EU-153).

⁸⁸⁸ Submission by Greenland of 29 January 2013, (Exhibit EU-154) and annexes, (Exhibit EU-155). ⁸⁸⁹ Commission decision of 25 April 2013 recognising the Greenland Department of Fisheries, Hunting

and Agriculture (APNN), (Exhibit EU-149). Sweden's request pursuant to Article 6(2) of the Implementing Regulation, (Exhibit EU-156).

⁸⁹¹ Deficiency letter to Sweden of 7 July 2011, (Exhibit EU-157) (also indicating the decision to treat the applications of the 11 county administrative boards jointly).

⁸⁹² Sweden's response of 6 October 2011, (Exhibit EU-158).

⁸⁹³ Commission decision of 18 December 2012 recognising the Swedish County Administrative Boards, (Exhibit EU-159). ⁸⁹⁴ See, e.g. Submission by Greenland of 29 January 2013, (Exhibit EU-154) (referring to a meeting

with a Commission official in Brussels in December 2012).

- 159 -

7.576. With respect to the Swedish application, the Commission sent a formal deficiency letter to the applicants about 6 months from receipt of the applications. The nature and content of the deficiency letter appears to be similar to that sent to the applicants from Greenland. The applicants then sent a reply to this deficiency letter roughly three months later containing an item-by-item response with extensive reference to Swedish legislation and an explanation of the function of county administrative boards. As was the case with respect to the applications from Greenland, we are not presented with any specific explanation regarding the time taken from receiving the applications to responding to the applicants with a deficiency letter. Nor do we find any additional evidence of activity regarding the Swedish application prior to recognition over one year later.⁸⁹⁵

7.577. As a general observation based on the factual circumstances described above, the CAP conducted with respect to both the Greenlandic and the Swedish applications took some time until their completion. Particularly, the circumstances surrounding the Swedish applications reveal a delay greater than one year from receipt of the applicants' responses to the Commission's deficiency letter until the final approval by the Commission. The amount of time taken in this specific instance, without sufficient justification, would not therefore seem "expeditious" within the meaning of Article 5.2.1.

7.578. In addressing the length of time taken to approve a recognized body under the CAP, the complainants have referred simply to the lapse of time from the effective date of the EU Seal Regime until the approval of the applications from Greenland and Sweden. They have not otherwise explained how specifically the CAP was not as expeditious as possible in undertaking and completing the applications concerned in this dispute.

7.579. We first recall our consideration above that the obligations of Article 5.2.1 apply upon *receipt* of an application. Accordingly, we do not consider the effective date of the EU Seal Regime, as suggested by the complainants, to be the correct benchmark against which the time taken for the undertaking and completion of the CAP is to be judged. Further, in our view, a violation of Article 5.2.1 must be examined in light of the specific circumstances relating to a given CAP. This would entail an evaluation of not only the entire period of time taken for each procedural step (e.g. correspondences between an applicant and the Commission) during the course of the undertaking and completion of a CAP. This would allow us to objectively assess whether the time taken for the conformity assessment of a given application was "as expeditious as possible". As noted above, in the present dispute, however, the complainants have not provided any specific argument as to how the CAP was *not* conducted in the concerned instances as expeditiously as possible within the meaning of Article 5.2.1.⁸⁹⁶

7.580. Therefore, in spite of our concern expressed above regarding the time taken with respect to the Swedish applications, we have not been provided a sufficient basis to conclude that the CAP was not undertaken and completed as expeditiously as possible within the meaning of Article 5.2.1 of the TBT Agreement.

7.4 Non-discrimination claims under the GATT 1994

7.4.1 Relationship between the GATT 1994 and the TBT Agreement

7.581. The complainants in this dispute presented claims under both the GATT 1994 and the TBT Agreement, including claims concerning the non-discrimination obligations. As we already addressed Canada's non-discrimination claim under Article 2.1 of the TBT Agreement, we find it useful to review the relationship between, and the legal standards under, the GATT 1994 and the

⁸⁹⁵ Although the European Union has asserted that it "explained why the processing of the request made by entities from Sweden took as long as it took", the reference to its first written submission allegedly doing so merely recounts what transpired in October 2011 and then in December 2012, without accounting for the intervening period of time. (European Union's second written submission, para. 327).

⁸⁹⁶ We observe that the panel in *EC* – *Approval and Marketing of Biotech Products* considered that, although "a Member is not legally responsible for delays which are not attributable to it", it would be sufficient to establish that the general *de facto* moratorium on approvals "caused undue delay in at least one instance". (Panel Report, *EC* – *Approval and Marketing of Biotech Products*, paras. 7.1497 and 7.1504). However, in contrast to the present dispute, the parties in that case had developed extensive arguments concerning specific actions that could have been taken to prevent the delays at issue and the justifiability of the reasons for which such actions were not taken.

TBT Agreement before we begin our examination of the parties' non-discrimination claims under the GATT 1994.

7.582. In the recent trilogy of disputes involving the claims under the TBT Agreement and the GATT 1994, the Appellate Body provided guidance on the relationship between the obligations under these two Agreements.⁸⁹⁷ Based on the text of the second recital of the preamble of the TBT Agreement, the Appellate Body observed, "the TBT Agreement expands on the pre-existing GATT disciplines and emphasizes that the two Agreements should be interpreted in a coherent and consistent manner."898

7.583. More specifically, the Appellate Body stated that the balance set out in the preamble of the TBT Agreement between, on the one hand, the desire to avoid creating unnecessary obstacles to international trade under the fifth recital and, on the other hand, the recognition of Members' right to regulate under the sixth recital, is not, in principle, different from the balance set out in the GATT 1994, where obligations such as national treatment in Article III are qualified by the general exceptions provision of Article XX.899

7.584. In the context of addressing the national treatment obligation under Article 2.1 of the TBT Agreement , the Appellate Body made the following observations: first, the similar formulation of the provisions under Article 2.1 of the TBT Agreement and Article III:4 of the GATT 1994⁹⁰⁰; second, the overlap in their scope of application in respect of technical regulations⁹⁰¹; and third, the absence of a general exceptions clause in the TBT Agreement that resembles a general exceptions clause in Article XX contained in the GATT 1994.⁹⁰² Based on these observations, the Appellate Body considered that Article III:4 of the GATT 1994 provides relevant context for the interpretation of the national treatment obligation of Article 2.1 of the TBT Agreement. However, noting the scope of the TBT Agreement as an agreement governing "technical regulations"903; the sixth recital of the preamble of the TBT Agreement⁹⁰⁴; and the object and purpose of the TBT Agreement to strike a balance between the objective of trade liberalization and Members' right to regulate, the Appellate Body concluded that Article 2.1 should not be interpreted as prohibiting any detrimental impact on competitive opportunities for imports in cases where such detrimental impact on imports stems exclusively from a legitimate regulatory distinction.

7.585. Therefore, the Appellate Body clarified the legal standards for the non-discrimination provisions under the GATT 1994 (Articles I:1 and III:4) and the TBT Agreement (Article 2.1): under the GATT 1994, the "treatment no less favourable" standard prohibits WTO Members from modifying the conditions of competition in the marketplace to the detriment of the group of imported products vis-à-vis the group of domestic like products, whereas under the TBT Agreement , the "treatment no less favourable" standard does not prohibit detrimental impact on imports that stems exclusively from a legitimate regulatory distinction rather than reflecting discrimination against the group of imported products.⁹⁰⁵ The additional element (i.e. legitimate regulatory distinction) that the Appellate Body considered as necessary to complete an analysis under Article 2.1 of the TBT Agreement reflects the Appellate Body's earlier observation regarding the absence in the TBT Agreement of a general exceptions clause equivalent to Article XX in the GATT 1994.

905 Appellate Body Report, US - Clove Cigarettes, paras. 180-182, 215; US - Tuna II (Mexico), para. 215.

⁸⁹⁷ See Appellate Body Reports, US – Clove Cigarettes, US – Tuna II (Mexico), and US – COOL.

⁸⁹⁸ Appellate Body Report, US – Clove Cigarettes, paras. 90-91. The Appellate Body further read the sixth recital of the preamble of the TBT Agreement as counterbalancing the trade-liberalization objective expressed in the fifth recital. Specifically it found that the sixth recital "recognizes" Members' right to regulate versus the "desire" to avoid creating unnecessary obstacles to international trade, expressed in the fifth recital. Thus, according to the Appellate Body, the sixth recital suggests that Members have a right to use technical regulations in pursuit of their legitimate objectives, provided that they do so in an even-handed manner and in a manner that is otherwise in accordance with the provisions of the TBT Agreement. (Appellate Body Report, US - Clove Cigarettes, para. 95).

⁸⁹⁹ Appellate Body Report, US – Clove Cigarettes, para. 96.

⁹⁰⁰ Appellate Body Report, US – Clove Cigarettes, para. 99.

⁹⁰¹ Appellate Body Report, US - Clove Cigarettes, para. 100. ⁹⁰² Appellate Body Report, US - Clove Cigarettes, para. 100.

 $^{^{903}}$ Appellate Body Report, US – Clove Cigarettes, para. 169. The Appellate Body also found support for this interpretation in the obligations under Article 2.2 as well as the sixth recital of the preamble of the TBT Agreement (paras. 170-173).

⁹⁰⁴ Appellate Body Report, US – Clove Cigarettes, para. 173.

- 161 -

7.586. Therefore, we do not consider that the legal standard with respect to the nondiscrimination obligation under Article 2.1 of the TBT Agreement "equally applies" to claims under Articles I:1 and III:4 of the GATT 1994 as argued by the European Union.⁹⁰⁶ As noted by the Appellate Body, under the GATT 1994, the objective of trade liberalization, including Members' obligation to respect non-discrimination obligations as embodied in Articles I:1 and III:4, are balanced against Members' right to regulate under the separate general exceptions clause of Article XX.

7.587. Bearing the above in mind, we begin our analysis of whether certain aspects of the EU Seal Regime modify the conditions of competition for Canadian and Norwegian imports of seals products on the EU market *vis-à-vis* Greenlandic and EU domestic seal products within the meaning of Articles I:1 and III:4 respectively. If we find in the affirmative, we will then examine whether the European Union has demonstrated why such aspects of the EU Seal Regime are nevertheless justified under Article XX.

7.4.2 Article I:1

7.4.2.1 Main arguments of the parties

7.588. As Canada's arguments under Article I:1 of the GATT 1994 largely resemble those under Article 2.1 of the TBT Agreement, we summarize in this section Norway's arguments only. To recall, Norway did not present a claim under Article 2.1 of the TBT Agreement. We will however cross-reference Canada's arguments to the extent relevant and appropriate in the context of our analysis.

7.589. Like Canada, Norway argues that the EU Seal Regime violates Article I.1 of the GATT 1994 because it grants a market access advantage to certain seal products from Greenland without extending such advantage "immediately and unconditionally" to "like" seal products from Canada and Norway.⁹⁰⁷

7.590. Norway further argues that the conditions of the IC exception discriminate on grounds of origin by establishing explicit links between importation and the territory of production.⁹⁰⁸ In respect of both its express wording and the necessary implications of the terms used, the EU Seal Regime restricts market access advantages to a "limited" and "closed" group of countries under the IC exception.⁹⁰⁹ Norway argues that the IC exception benefits predominantly one single country out of the list of countries identified, namely Greenland.⁹¹⁰ Further, the measure is expected to operate, in practice, in a manner that confers little or no benefit on seal products

⁹⁰⁶ European Union's first written submission, para. 528. See, for instance, Canada's opening statement at the first meeting of the Panel, paras. 51-53, 55; Norway's opening statement at the first meeting of the Panel, paras. 23-25.

See, for instance, *US – Tuna II (Mexico)*, para. 405 where the Appellate Body notes that the scope and content of Article 2.1 of the TBT Agreement and Articles I:1 and III:4 of the GATT 1994 are not the same.

⁹⁰⁷ Canada's opening statement at the second meeting with the Panel, para. 31; Norway's first written submission, para. 286. In making its argument, Norway distinguishes between "finished" and "intermediate" products. (See Norway's first written submission, paras. 369-371 and 372-375, respectively).

⁹⁰⁸ Norway's first written submission, para. 286. It should be noted that Norway is not making a claim of *de jure* discrimination *per se*. However, Norway notes in its submission that "[t]he facts set forth ... in support of Norway's claim that the European Union Seal Regime *de facto* discriminates contrary to Article I:1 of the GATT 1994 would also support a finding by the Panel that the European Union Seal Regime is indeed *de jure* inconsistent with Article I:1 of the GATT 1994, since it necessarily limits the extension of a relevant "advantage" to a defined and closed group of countries." (Norway's first written submission, para. 377, footnote 595).

⁹⁰⁹ Norway notes that the Basic Seal Regulation expressly names certain Members or territories of Members as qualifying under this aspect of the IC exception. In particular, Article 2(4) of the Basic Regulation lists the following six Inuit Communities: Inupiat, Yupik (Alaska), Inuit, Inuvialuit (Canada), Kalaallit (Greenland) and Yupik (Russia). As regards the definition of "indigenous communities" in Article 2(1) of the Implementing Regulation, Norway argues that it also establishes a "closed group" because an indigenous community must have inhabited the territory of the Member in question "at the time of the conquest or colonisation or the establishment of present State boundaries". (Norway's first written submission, paras. 378-380).

⁹¹⁰ Norway's first written submission, para. 389.

- 162 -

originating in Norway.⁹¹¹ Norway further contends that by conditioning market access on the existence of a tradition of producing certain goods in the country; of belonging to a certain people that has long resided in the country; or on factors such as the partial use of the product in the country of production, the European Union conditions market access on the "situation or conduct" of the exporting countries.⁹¹² This reflects a failure to extend the advantage of market access "unconditionally" to like products originating in all WTO Members, as required by Article I:1 of the GATT 1994.⁹¹³

7.591. The European Union does not contest that the EU Seal Regime, through the IC exception, confers an "advantage" in the sense of Article I:1 of the GATT 1994 to the extent that it permits the placing on the EU market of seal products that would otherwise be excluded through the general prohibition.⁹¹⁴ In response to Norway's argument, the EU asserts that the conditions under the IC exception are origin-neutral as they refer to the type and purpose of the hunt, rather than to a defined origin.⁹¹⁵ The reference to country names where Inuit and indigenous communities currently live does not imply that the IC exception applies only to a "limited" or "closed" list of countries.⁹¹⁶ In the EU's view, the fact that hunts conducted by Inuit and other indigenous communities in countries such as Canada and Norway represent a lower percentage than in other countries, such as Greenland, cannot be found to be discriminatory *per se.*⁹¹⁷ The European Union also disagrees with Norway's contention that the advantage under Article I:1 may not be granted subject to conditions relating to the situation or conduct of other countries; if the conditions to obtain a certain advantage are drafted in an origin-neutral manner, the requirement of such conditions would not be discriminatory.⁹¹⁸

7.4.2.2 Analysis by the Panel

7.592. Article I:1 of the GATT 1994 provides that:

With respect to customs duties and charges of any kind imposed on or in connection with importation or exportation or imposed on the international transfer of payments for imports or exports, and with respect to the method of levying such duties and charges, and with respect to all rules and formalities in connection with importation and exportation, and with respect to all matters referred to in paragraphs 2 and 4 of Article III, any advantage, favour, privilege or immunity granted by any [Member] to any product originating in or destined for any other country shall be accorded immediately and unconditionally to the like product originating in or destined for the territories of all other [Members].

7.593. Based on this provision, three elements must be satisfied in order to demonstrate an inconsistency with Article I:1 of the GATT 1994: (i) there must be an "advantage, favour, privilege or immunity" of the type covered by Article I:1; (ii) the advantage is not granted "immediately and unconditionally"; (iii) to like products originating in or destined for all other WTO Members.⁹¹⁹

7.594. In the context of Canada's claims under Article 2.1 of the TBT Agreement, the Panel addressed the question of whether Canadian seal products are "like" seal products of other origin (Greenland). We found in section 7.3.2.1 above that the seal products belonging to these different groups are like products, irrespective of whether they conform or not to the requirements under the EU Seal Regime. We recall that the parties do not dispute that conforming and non-conforming seal products are like.

⁹¹¹ It is not clear from Norway's submission whether they are referring to their seal products derived from Inuit or from commercial hunts.

⁹¹² Norway's first written submission, para. 388 (referring to *Canada – Autos*, para. 10.23).

⁹¹³ Norway's first written submission, para. 388.

⁹¹⁴ European Union's first written submission, para. 542.

⁹¹⁵ According to the European Union, the terms "Inuit" and "indigenous communities" defined in the Basic and Implementing Regulations, respectively, are not indicative of a particular origin. In fact, these communities are widely spread around the world. (See European Union's first written submission, paras. 280-283 and 547).

⁹¹⁶ European Union's first written submission, para. 557.

⁹¹⁷ European Union's first written submission, para. 289.

⁹¹⁸ European Union's first written submission, para. 538.

⁹¹⁹ Panel Reports, Indonesia – Autos, para. 14.138; and EU – Footwear, para. 7.99.

7.595. The term "advantage" in Article I:1 is broad and applies to all matters referred to in paragraphs 2 and 4 of Article III of the GATT 1994. Article III:4 applies in turn to all "laws, regulations and requirements affecting the internal sale, offering for sale, purchase, transportation, distribution or use [of a product]." The EU Seal Regime is undoubtedly a "law or regulation" within the meaning of Article III:4 of the GATT 1994 and therefore, the measure also falls within the scope of Article I:1 of the GATT 1994.

7.596. Furthermore, the advantage granted by the EU Seal Regime is in the form of market access; it is granted to seal products that meet the conditions under the IC exception. The EU Seal Regime affects the placing on the market of seal products and therefore the "internal sale", "offering for sale", "distribution" and "purchase" of seal products.

7.597. With respect to the third element of Article I:1, the MFN obligation contained therein requires that once seal products from Greenland are granted the advantage of access to the European Union market, such advantage be extended "immediately and unconditionally" to Canadian and Norwegian seal products that are found to be "like".⁹²⁰ As explained above, the evidence suggests that this has not been the case because the vast majority of seal products from Canada and Norway do not meet the IC requirements for placing on the market under the EU Seal Regime.⁹²¹ In contrast, virtually all of Greenlandic seal products are likely to qualify under the IC exception for placing on the market.⁹²² Thus, in terms of its design, structure, and expected operation, the EU Seal Regime detrimentally affects the conditions of competition on the market of Canadian and Norwegian origin as compared to seal products of Greenlandic origin.

7.598. Bearing in mind this finding, we address Norway's argument that the IC requirements give rise to origin-based discrimination. Norway argues that in order to gualify under the IC exception, seal products must originate in one of a limited number of countries inhabited by an indigenous community that meets the specific terms of the conditions.⁹²³ According to Norway, the origin of seal products that are likely to qualify under the IC exception can be determined by necessary implication.⁹²⁴ First, based on Norway's arguments in this regard, it is not clear whether Norway is presenting a *de jure* claim with respect to the IC exception in this dispute. Although Norway refers to the term "de jure" in a footnote in the context of its arguments relating to Article I:1, it has not fully developed a *de jure* claim, as a separate and additional claim from its *de* facto claim with respect to the IC exception. Furthermore, to the extent we already found a de facto violation with respect to the IC exception above, we do not find it necessary to make a finding on a *de jure* claim.

7.599. We note that several countries, including non-EU member States, have Inuit or indigenous communities living in their territory.⁹²⁵ For instance, Canada's Inuit and Inuvialuit populations are specifically mentioned in the Basic Regulation⁹²⁶; there is also evidence that Norway's Sami

⁹²⁰ For the purpose of this analysis, the groups of products compared are all Norwegian seal products and all Greenlandic seal products (i.e. with reference to Table 1 above, cells B+G are compared to cells D+I).

 $^{^{921}}$ The evidence before us shows that roughly 5% of Canada's seal harvest and less than 4.5% of Norway's seal harvest are likely to comply with the requirements of the IC exception. (See Canada's first written submission, para. 286 and footnote 391; Norway's first written submission, para. 391, Table 1 citing COWI 2010 Report, (Exhibit JE-21), pp. 27, 30-31; Nunavut Report (2012), (Exhibit JE-30), p. 1; and Norwegian Ministry of Fisheries and Coastal Affairs, Facts about Fisheries and Aquaculture 2010 (Exhibit NOR-63), p. 21).

⁹²² It is estimated that 100% of Greenland's seal products would qualify under the IC exception. (See Norway's first written submission, paras. 391-394, citing COWI 2010 Report, (Exhibit JE-21), section 3.1, pp. 28-30 and Management and Utilization of Seals in Greenland (Exhibit JE-26), p. 13; and European Union's response to Panel question No. 156 (where the European Union notes that entities from Greenland have been authorized as "designated bodies" that could deliver attesting documents for the purpose of placing Greenlandic seal products on the EU market)).

⁹²³ We note that Norway submits this argument in the context of its *de facto* discrimination claims while noting that the same arguments would support a claim of de jure discrimination to the extent that the EU Seal Regime limits the extension of a relevant "advantage" to a defined and closed group of countries. (See Norway's first written submission, para. 377, footnote 595).

⁹²⁴ Norway's opening statement at the first meeting of the Panel, paras. 43-44. As noted, it is unclear to the Panel whether Norway is arguing that the EU Seal Regime is de jure discriminatory against Norwegian imports of seal products. (Norway's first written submission, footnote 595; opening statement at the first meeting of the Panel, para. 43). ⁹²⁵ European Union's first written submission, paras. 283-284.

⁹²⁶ The definition of "Inuit" in Article 2(4) contains a reference to these two populations.

- 164 -

population would qualify as an "indigenous community" within the meaning of the definition in Article 2(1) of the Implementing Regulation. The COWI 2010 Report identifies a number of countries whose Inuit or indigenous populations are likely to meet the requirements of the IC exception.⁹²⁷ The IC requirements permit the placing on the market based on conditions relating to the characteristics of seal hunts as opposed to a "closed list" of countries. We also note that some of the communities whose seal products may qualify under the exception are not concentrated in one single country.⁹²⁸ In light of these considerations, we do not believe that the EU Seal Regime gives rise to discrimination based on origin *per se*.

7.600. Based on our findings in paragraph section 7.3.2.2.2 above, we consider that the measure at issue does not "immediately and unconditionally" extend the same market access advantage on the EU market to the complainants' imports as they do to seal products originating from Greenland and thus is inconsistent with the obligation under Article I:1 of the GATT 1994.

7.4.3 Article III:4

7.4.3.1 Main arguments of the parties

7.601. As in the case of Canada's claim under Article 1:1, Canada's arguments under Article III:4 largely resemble those under Article 2.1 of the TBT Agreement. Thus, we summarize below Norway's arguments under Article III:4 of the GATT 1994 by cross-referencing as appropriate to Canada's arguments under Article 2.1 of the TBT Agreement.

7.602. Like Canada, Norway argues that the EU Seal Regime violates Article III:4 of the GATT 1994 because through the requirements of the MRM exception, the Regime accords to imported seal products from Norway (and Canada) a treatment that is less favourable to that accorded to like domestic seal products.⁹²⁹ Specifically, Norway argues that by introducing the "non-systematic", "non-profit", and "sole purpose" requirements, the European Union has tailored the MRM exception to the realities of the seal hunt in the European Union.⁹³⁰ In contrast, Norway's seal products are ineligible under the MRM exception despite the fact that the hunts are conducted in accordance with sustainability principles.⁹³¹ Unlike the seal hunts conducted in the European Union, the hunt in Norway is not merely incidental to other fishing activities; as such, it cannot be carried out on condition that no profit is derived from the hunt.⁹³² In addition, the treatment of subsidies in the EU definition of "non-profit" serves to further exclude Norwegian products from the EU market while allowing seal products from the European Union.⁹³³

⁹³² Norway's first written submission, paras. 444-445.

⁹²⁷ COWI 2010 Report, p. 30.

⁹²⁸ The evidence before the Panel suggests that Inuit and indigenous populations may occupy the territories of adjacent countries. (See, e.g. European Union's first written submission, paras. 283-284). The European Union gives the example of Sami communities who live and hunt in both Norway and Sweden. (See COWI 2010 Report, p. 33).

⁹²⁹ Norway's first written submission, para. 424.

⁹³⁰ Norway's first written submission, para. 430 (citing Comments from EU Member States on Proposal for a Regulation of the European Parliament and of the Council concerning trade in seal products, (Exhibit JE-10), pp. 16 and 18; and European Parliament Debates, (Exhibit JE-12), p. 72). Norway notes that according to scientific literature, the problem posed by seals to the fishing activities of Finland, Latvia, Lithuania, and Sweden relates essentially to seals' attacks on fishing gear. (Norway's first written submission, para. 431 (referring to Finnish Game and Fisheries Research Institute, *Symposium on Biology and Management of Seals in the Baltic Area held in Helsinki* (2005), (Exhibit NOR-64)).

⁹³¹ Norway's first written submission, paras. 434-441. Norway does not contend that the first requirement under the MRM exception (i.e. that seal products must derive from hunts conducted under a natural resources management plan which uses scientific population models of marine resources and applies the ecosystem-based approach, and that the seal catch must not exceed the total allowable catch quota established under the plan) is discriminatory. Norway further notes that it expects the products of its seal hunt to meet this requirement under the MRM exception because the hunt in Norway is conducted "based on ecosystem management principles". (Norway's second written submission, paras. 81-82 (citing COWI 2010 Report, Annex 5, p. 13)).

⁹³³ Norway notes that "[i]n order to allow the long-term viability of the seal hunt and maintain the professional capabilities necessary to carry out the hunt, Norway does provide a subsidy in relation to the hunt." Norway adds that "[t]he purpose of the subsidy is to ensure that the recommended TAC quotas are taken." (Norway's first written submission, paras. 448-449 and footnote 708).

7.603. The European Union argues that the MRM exception does not modify the conditions of competition to the detriment of Norway's (and Canada's) seal products.⁹³⁴ The EU Seal Regime equally affects seal products resulting from hunts for commercial purposes (non-conforming) and seal products resulting from hunts for marine resource management purposes (conforming).935 The European Union further argues that the MRM exception is based on considerations that are completely unrelated to the domestic origin of seal products.936

7.4.3.2 Analysis by the Panel

7.604. Article III:4 of the GATT 1994 provides that:

The products of the territory of any Member imported into the territory of any other Member shall be accorded treatment no less favourable than that accorded to like products of national origin in respect of laws, regulations and requirements affecting their internal sale, offering for sale, purchase, transportation, distribution or use.

7.605. There are three elements that must be examined to assess a measure's consistency with Article III:4 of the GATT 1994: (i) whether the measure is a law, regulation or requirement affecting the internal sale, offering for sale, purchase or use of goods; (ii) whether the products at issue are like; and (iii) whether imported products are accorded less favourable treatment than that accorded to like domestic products.⁹³⁷

7.606. With respect to the first element of Article III:4, we note that the EU Seal Regime is undoubtedly a "law" or "regulation" affecting the internal sale, offering for sale, purchase, distribution and use of seal products within the meaning of Article III:4.

7.607. As regards the second element of Article III:4, we found in the context of Canada's claims under Article 2.1 of the TBT Agreement that seal products are "like" irrespective of whether they conform or not to the requirements under the EU Seal Regime. We also recall that the parties do not dispute that conforming and non-conforming seal products are like.

7.608. With respect to the third element under Article III:4, the national treatment obligation contained therein requires that imported products from Canada and Norway receive a treatment no less favourable than that accorded to domestic seal products.⁹³⁸ Based on the evidence before the Panel, it appears that the vast majority of seal products from Canada and Norway are excluded from the EU market by the terms of the MRM exception.⁹³⁹ In contrast, evidence shows that virtually all domestic seal products are likely to qualify for placing on the market.⁹⁴⁰

7.609. In light of the above considerations, we conclude that the measure at issue grants Canadian and Norwegian seal products a treatment less favourable than that accorded to EU seal products within the meaning of Article III:4 of the GATT 1994.

7.4.4 The European Union's justification of the EU Seal Regime under Article XX

7.4.4.1 Introduction

7.610. Article XX of the GATT 1994 provides:

⁹³⁴ European Union's first written submission, paras. 500-525; second written submission, paras. 235-263 and 369-372.

⁹³⁵ European Union's second written submission, para. 370.

⁹³⁶ European Union's first written submission, para. 323; second written submission, paras. 207, 247 and 370. ⁹³⁷ Appellate Body Report, *Korea – Various Measures on Beef*, para. 133.

⁹³⁸ For the purpose of this analysis, the groups of products compared are all Norwegian seal products and all EU seal products (i.e. with reference to Table 1 above, cells B+G are compared to cells A+F).

³⁹ We note, for instance, that although Norway's hunts take place under a natural resources management plan which uses scientific population models of marine resources and applies the ecosystembased approach, Norway's seal products would not meet the "non-profit", "non-systematic" and "sole purpose" requirements under the MRM exception (See COWI 2010 Report, Annex 4, p. 3 as discussed in Norway's first written submission, paras. 263-265). Regarding Canada's arguments that its seal products would not be eligible under the MRM exception, we refer to our analysis in section 7.3.2.2 above.

⁹⁴⁰ See, for instance, COWI 2008 Report, p. 117.

Subject to the requirement that such measures are not applied in a manner which would constitute a means of arbitrary or unjustifiable discrimination between countries where the same conditions prevail, or a disguised restriction on international trade, nothing in this Agreement shall be construed to prevent the adoption or enforcement by any contracting party of measures:

- (a) necessary to protect public morals;
- (b) necessary to protect human, animal or plant life or health ...

7.611. As noted above, under the GATT 1994, the objective of trade liberalization through principles such as non-discrimination is balanced against Members' right to regulate to pursue the policy objectives listed in Article XX.

7.612. We found that the IC exception and the MRM exception under the EU Seal Regime are inconsistent with the obligations under Articles I:1 and III:4 by modifying the conditions of competition for Canadian and Norwegian seal products vis-à-vis like products of Greenlandic and EU domestic origins on the EU market. The European Union alleges that these exceptions are based on "regulatory differences"⁹⁴¹ that are "necessary" to achieve the objectives invoked by the European Union under Article XX(a) and XX(b). Accordingly, we must examine whether the European Union has established its case under Article XX by demonstrating two elements: first, the measure falls within the scope of, and meets the requirements under paragraph (a) and/or paragraph (b) of Article XX; and, second, the measure satisfies the requirements of the chapeau of Article XX.942

7.613. In this connection, we recall that "the GATT 1994 and the TBT Agreement should be interpreted in a coherent and consistent manner".⁹⁴³ Under the TBT Agreement, we found that the EU Seal Regime as a technical regulation, contributes, to a certain extent, to the objective of addressing the public moral concerns on seal welfare within the meaning of Article 2.2.944 We further found under Article 2.1 of the TBT Agreement that the IC and MRM exceptions under the Regime caused detrimental impacts on Canadian and Norwegian seal products, which was found not to stem exclusively from legitimate regulatory distinctions.⁹⁴⁵ All of these considerations under the TBT Agreement are therefore relevant to our examination of the European Union's claim under Article XX of the GATT 1994. We are however mindful that the obligations under these two Agreements are not the same.946

7.614. Having the above in mind, we start by examining the specific aspects of the measure that must be considered in our analysis of Article XX in this dispute.

7.4.4.2 Aspects of the EU Seal Regime to be considered

7.4.4.2.1 Main arguments of the parties

7.615. According to the European Union, the analysis should focus on whether the "regulatory differences" upon which the finding of less favourable treatment is based are "necessary" in order to achieve the objectives at the level of protection chosen by the responding Member. In this case, the alleged less favourable treatment "results from the interplay between the General Ban and the IC and MRM exceptions".⁹⁴⁷ Thus, the Panel should examine whether the "regulatory differences"

⁹⁴¹ We note that the European Union uses the term "regulatory *differences*" rather than "regulatory distinctions" in the context of Article XX of the GATT 1994.

⁹⁴² Appellate Body Report, US – Shrimp, para. 119.
⁹⁴³ Appellate Body Report, US – Clove Cigarettes, para. 91.

⁹⁴⁴ See section 7.3.3.3.2 above.

⁹⁴⁵ See section 7.3.2 above.

⁹⁴⁶ For instance, Article XX of the GATT 1994 is a general exceptions provision under which a measure found inconsistent with one or more provisions of the GATT 1994 can be justified. Accordingly, the burden of proving a case under Article XX of the GATT 1994 rests with the respondent. As observed by the Appellate Body, the TBT Agreement does not contain a general exceptions provision. In light of the similarities in the text between the GATT 1994 and the TBT Agreement, and given the overlap in their coverage, the Appellate Body addressed the absence of a general exceptions provision in the TBT Agreement by clarifying the specific requirements to be established under Article 2.1 of the TBT Agreement. See section 7.4.1 above.

⁹⁴⁷ European Union's second written submission, para. 382.

between the General Ban and the two exceptions are "necessary" in order to achieve the objectives invoked by the European Union.⁹⁴⁸ Specifically, the European Union submits that the following must be shown: (a) the treatment provided by the General Ban is "necessary" to achieve the objective at the selected level of protection; and (b) it is not "necessary", in order to achieve the objective at the same level of protection, to extend the same treatment provided under the General Ban to seal products falling under the MRM exception or the IC exception.⁹⁴⁹

7.616. Canada similarly submits that the focus of the analysis under Article XX(a) and (b) justifying an inconsistency with provisions under the GATT 1994 should be on whether the discriminatory treatment based on the regulatory differences is necessary.⁹⁵⁰ To Canada, the relevant regulatory differences are found in the expected operation of the three categories under which seal products will qualify for market access. Thus, the conditions that restrict market access for seal products under the IC, MRM, and Travellers categories must be necessary in order for the EU Seal Regime to be found provisionally justified under Article XX. Specifically, the European Union is required to show how the discriminatory treatment of Canadian seal products makes a material contribution to the achievement of its objectives.⁹⁵¹

7.617. Norway contends that "what the EU must justify under Articles XX(a) and XX(b) is the *discrimination on grounds of origin* that violates Article I:1 and III:4".⁹⁵² Norway considers that the European Union only defends the "General Ban" aspect of the measure, whereas "it is the restrictive conditions of the IC and [MRM] requirements that favour certain origins".⁹⁵³

7.4.4.2.2 Analysis by the Panel

7.618. In examining Members' right to regulate under Article XX of the GATT 1994, a question arises as to what aspects of a measure must be analysed under the legal framework of Article XX.⁹⁵⁴ Based on their submissions, the parties in this dispute appear to be in agreement that it is the aspect of a measure infringing the GATT 1994 that must be *justified* under Article XX. However, this does not mean that a panel can consider only that aspect of the measure to determine the measure's justifiability under Article XX.

7.619. In our view, the question concerning the specific aspects of a measure that must be considered for a claim under Article XX should be dealt with in light of the specific circumstances of a given dispute.⁹⁵⁵ Such circumstances include the nature and characteristics of a measure at

⁹⁵² Norway's response to Panel question No. 43, para. 224 (relying on Appellate Body Report, *Thailand – Cigarettes (Philippines)*, para. 177) (emphasis original)

⁹⁵⁵ The Appellate Body in US – Gasoline states:

The relationship between the affirmative commitments set out in, e.g. Articles I, III and XI, and the policies and interests embodied in the "General Exceptions" listed in Article XX, can be given meaning within the framework of the *General Agreement* and its objective and purpose by a treaty interpreter only *on a case-by-case basis, by careful scrutiny of the factual and legal context in a given dispute,* without disregarding the

⁹⁴⁸ European Union's second written submission, para. 383.

⁹⁴⁹ European Union's response to Panel question Nos. 43 and 139.

⁹⁵⁰ Canada's response to Panel question No. 43, para. 167.

⁹⁵¹ Canada's response to Panel question No. 43

⁹⁵³ Norway's response to Panel question No. 43, para. 225.

⁹⁵⁴ In *US* – *Gasoline*, the Appellate Body touched upon the meaning and scope of the term "measures" as referenced in the chapeau and paragraph (g) of Article XX. (See Appellate Body Report, *US* – *Gasoline*, pp. 12-13). While not providing a concrete view on this question, the Appellate Body observed that "the Panel [in that dispute] was following the practice of earlier panels in applying Article XX to provisions found to be inconsistent with Article III:4: the 'measures' to be analyzed under Article XX are the same provisions infringing Article III:4. These earlier panels had not interpreted "measures" more broadly under Article XX to include provisions not themselves found inconsistent with Article III:4."

See also Appellate Body Report in *Thailand – Cigarettes (Philippines)*, citing GATT Panel Report, *US – Section 337 Tariff Act*. The Appellate Body stated that when Article XX(d) is invoked to justify an inconsistency with Article III:4, "what must be shown to be 'necessary' is the treatment giving rise to the finding of less favourable treatment." (Appellate Body Report, *Thailand – Cigarettes (Philippines)*, paras. 177 and 179). The Appellate Body further stated, "when less favourable treatment is found based on differences in the regulation of imports and of like domestic products, the analysis of an Article XX(d) defence should focus on whether those regulatory differences are 'necessary' to secure compliance with 'laws or regulations' that are not GATT-inconsistent." The Appellate Body observed that "in putting forth its defence, Thailand sought to justify administrative requirements relating to VAT liability generally, rather than to justify the *differential treatment* afforded to imported versus domestic cigarettes under its measure".

issue, the manner in which the complainants present their claims with respect to the measure, and the relationship between the GATT 1994 and other WTO covered agreements that were also examined with respect to the measure concerned.

7.620. Regarding the nature and characteristics of the measure at issue in the present dispute, the two components (i.e. a ban and exceptions) comprising the EU Seal Regime are closely connected to each other. As discussed in section 7.3.1 above, we considered that the IC and MRM exceptions cannot operate in isolation without the ban. For example, similar factual circumstances were also present in US - Gasoline. In that dispute, the Appellate Body examined how the discriminatory aspect of the measure (baseline establishment rules for refiners, blenders, and importers of gasoline) was to be analysed under Article XX(g) with respect to other parts of the measure. The Appellate Body stated:

The baseline establishment rules, taken as a whole ..., need to be related to the "nondegradation" requirements set out elsewhere in the Gasoline Rule. Those provisions can scarcely be understood if scrutinized strictly by themselves, totally divorced from other sections of the Gasoline Rule which certainly constitute part of the context of these provisions. ... Without baselines of some kind, such scrutiny would not be possible and the Gasoline Rule's objective of stabilizing and preventing further deterioration of the level of air pollution prevailing in 1990, would be substantially frustrated. The relationship between the baseline establishment rules and the "nondegradation" requirements of the Gasoline Rule is not negated by the inconsistency, found by the Panel, of the baseline establishment rules with the terms of Article III:4. We consider that, given that substantial relationship, the baseline establishment rules cannot be regarded as merely incidentally or inadvertently aimed at the conservation of clean air in the United States for the purposes of Article XX(g). 956

7.621. Although the analysis above was conducted in the context of paragraph (g) rather than (a) or (b) of Article XX, we do not see any reason why it would not be relevant to our examination of the EU Seal Regime under paragraphs (a) and (b) of Article XX. Specifically, we find guidance in the manner in which the Appellate Body weighed and considered the interrelationship of different components of the measure for the purpose of a sub-paragraph of Article XX.

7.622. Like the measure in question in US – Gasoline, the IC and MRM exceptions have a "substantial" relationship with the ban given that the exceptions cannot exist without the ban and that the exceptions inform the scope of the ban. Accordingly, the IC and MRM exceptions can "scarcely be understood if scrutinized strictly by themselves, totally divorced from" the ban of the EU Seal Regime.⁹⁵⁷ We also consider that this relationship between the exceptions and the ban "is not negated by the inconsistency" of the exceptions aspect of the EU Seal Regime with the terms of Articles I:1 and III:4.958

7.623. Furthermore, as discussed in section 7.3.1 above, the EU Seal Regime in its entirety, comprising both the ban and the exceptions, was found to constitute a technical regulation within the meaning of Annex 1.1 of the TBT Agreement. A legitimate policy objective pursued by the EU Seal Regime as a technical regulation was determined to be "addressing the EU public moral concerns on seal welfare". Having found as such, the EU Seal Regime as a whole was examined for the necessity of its trade-restrictiveness in fulfilling the identified policy objective under Article 2.2, whereas the justification of the detrimental impacts caused by the IC and MRM exceptions (regulatory distinctions) on the imported products was examined under Article 2.1.

7.624. Under these circumstances, we consider that although it is the aspects of the EU Seal Regime infringing the GATT 1994 (i.e. the IC and MRM exceptions) that must be justified under Article XX, our *analysis* under paragraphs (a) and (b) should first focus on the EU Seal Regime as a whole; it is the EU Seal Regime as a whole that pursues the European Union's identified objective, rather than the exceptions on their own independently from the ban. We must then examine whether the EU Seal Regime, in particular the distinctions drawn in the form of the IC and

words actually used by the WTO Members themselves to express their intent and purpose. (Appellate Body Report, *US* – *Gasoline*, p. 17) (emphasis added) ⁹⁵⁶ Appellate Body Report, *US* – *Gasoline*, p. 19.

⁹⁵⁷ Appellate Body Report, US – Gasoline, p. 17.

⁹⁵⁸ See Appellate Body Report, US – Gasoline, p. 19.

MRM exceptions, are applied in a manner consistent with the requirements under the chapeau of Article XX. This approach is justified in our view by the fact that an exception to a general rule, by definition, would hardly be considered as "necessary", when considered on its own, to achieve a policy objective of the general rule.⁹⁵⁹

7.4.4.3 Paragraph (a) of Article XX

7.4.4.3.1 Main arguments of the parties

7.625. In support of its claim that the EU Seal Regime is "necessary to protect public morals" within the meaning of Article XX(a)⁹⁶⁰, the European Union largely refers back to its arguments on Article 2.2 of the TBT Agreement.⁹⁶¹ The European Union emphasizes that moral concern with regard to the protection of animals is regarded as a value of high importance in the European Union, which is now expressly enshrined in its constitutional treaties.⁹⁶² Regarding the measure's restrictive effect on trade, the European Union concedes that the EU Seal Regime aims at being very trade-restrictive, consistently with the desired high level of fulfilment of the policy objective.⁹⁶³

7.626. The European Union argues that the EU Seal Regime makes a substantial contribution to its policy objective. Further, none of the alternative measures as suggested by the complainants in the context of Article 2.2 is apt to make an equivalent contribution to the policy objective sought by the EU Seal Regime.⁹⁶⁴

7.627. Canada argues that the EU Seal Regime does not fall within the scope of the protection of public morals. Article XX(a) of the GATT 1994 requires a moral norm that is a standard of conduct applied generally throughout the community or society and broadly accepted within the community. Canada contends that the European Union has not set out a clearly discernible and unambiguous rule of moral conduct, particularly with respect to the claimed distinction between "commercial" and "non-commercial" hunts. The public "concerns" cited by the European Union do not rise to the level of public morals, and the idea that the EU Seal Regime addressed public moral concerns rests on a false premise that the commercial seal hunt is inherently inhumane.⁹⁶⁵

7.628. Canada further submits that, even if the EU Seal Regime falls within the scope of Article XX(a), it is not necessary to protect public morals. The *protection* of public morals requires the prevention of some type of harm to a public moral within the territory of the Member whose measure is at issue. Although this is in principle a highly important interest or value, Canada questions the seriousness of the harm that might be expected to arise to public morals in the absence of the EU Seal Regime. In addition, the claim that the measure makes a material contribution to the EU public's moral concerns is dependent on the artificial distinction between commercial and non-commercial seal hunts, and between Inuit and non-Inuit hunts. Finally, the EU Seal Regime imposes the most trade-restrictive type of measure and a less trade-restrictive

⁹⁶² European Union's first written submission, para. 585. Moreover, the measure at issue was adopted in response to longstanding public demands and with the support of the vast majority of the members of both the European Parliament and the EU Council. The measure is also supported by a very large majority of the European population.

⁹⁵⁹ See footnote 454 above.

⁹⁶⁰ European Union's first written submission, paras. 578-590. See also European Union's response to Panel question No. 43 (noting that what must be found to be necessary are the "regulatory differences" in the EU Seal Regime, namely that it is not necessary to extend the same treatment provided under the General Ban to seal products falling under the MRM or IC exceptions).

⁹⁶¹ See European Union's response to Panel question No. 139 (explaining that it cross-references its arguments under Article 2.2 of the TBT Agreement to show a substantial contribution to the policy objective and that alternative measures do not make an equivalent contribution; specifically that "the arguments put forward under Article 2.2 of the TBT Agreement are equally relevant and valid under Article XX(a) of the GATT"); European Union's comments on the complainants responses to Panel question No. 139, para. 93 ("The reasons why it is not "necessary", in order to achieve the objectives mentioned in Article XX(a) or XX(b), to extend the General Ban to the marketing of seal products falling within the IC exception and the MRM exception are essentially the same as the reasons that render the regulatory distinction between those categories "legitimate" for the purposes of both Articles I:1 and III:4 of the GATT and Article 2.1 of the TBT Agreement.").

⁹⁶³ European Union's first written submission, para. 586.

⁹⁶⁴ European Union's first written submission, paras. 587-589.

⁹⁶⁵ Canada's second written submission, paras. 130-169.

alternative of animal welfare criteria with a certification and compulsory labelling scheme is reasonably available.⁹⁶⁶

7.629. Norway argues that the European Union has failed to meet its burden to show that the violation of the GATT 1994 resulting from the IC and MRM exceptions is necessary to protect public morals. With respect to the arguments of the European Union under Article 2.2 of the TBT Agreement, Norway submits that the legal standards and the allocation of the burden of proof are not the same under Article XX of the GATT 1994 and Article 2.2 of the TBT Agreement.⁹⁶⁷

7.4.4.3.2 Analysis by the Panel

7.630. The necessity of a measure within the meaning of Article XX(a) is determined through "a process of weighing and balancing" of "all the relevant factors, particularly the extent of the contribution to the achievement of a measure's objective and its trade restrictiveness, in the light of the importance of the interests or values at stake".⁹⁶⁸ The more vital or important the values or interests furthered by a measure are, the easier it will be to accept that measure as necessary.⁹⁶⁹ According to the Appellate Body, "if this analysis yields a preliminary conclusion that the measure is necessary, this result must be confirmed by comparing the measure with its possible alternatives, which may be less trade restrictive while providing an equivalent contribution to the achievement of the objective pursued".⁹⁷⁰

7.631. With this guidance in mind, we begin our analysis by considering whether the European Union's policy objective pursued through the EU Seal Regime falls within the range of policies designed to protect public morals as prescribed in Article XX(a). We recall our conclusion in section 7.3.3.1 that the European Union seeks to address the public moral concerns on seal welfare through the EU Seal Regime. We reached this conclusion by confirming, based on the evidence before us, the existence of the EU public concerns on seal welfare in general and by finding that such concerns are of a moral nature within the European Union. In this connection, we followed the guidance provided by previous panels concerning the scope of the notion "public morals" as stipulated in Article XX(a) of the GATT 1994 and Article XIV(a) of the GATS. In light of these considerations, we find that the policy objective pursued by the European Union ("addressing the EU public moral concerns on seal welfare") falls within the scope of Article XX(a) ("to protect public morals").

7.632. The European Union submits that the "moral concern with regard to the protection of animals" is regarded as a value of high importance in the European Union.⁹⁷¹ We consider, and the parties do not dispute, that the protection of such public moral concerns is indeed an important value or interest.⁹⁷²

7.633. Next, in assessing a measure's contribution to the objective pursued under Article XX of the GATT 1994, the Appellate Body stated that such a contribution exists when there is a genuine relationship of ends and means between the objective pursued and the measure at issue.

⁹⁶⁶ Canada's second written submission, para. 170-203. See Canada's second written submission, para. 196 ("Although this alternative measure was asserted in Canada's claim under Article 2.2 of the TBT Agreement, the reasons why the alternative is a less trade-restrictive alternative under that provision apply *mutatis mutandis* to the less trade-restrictive alternative element of the the necessity test under Article XX(a)."). See also Canada's response to Panel question No. 43.

 ⁹⁶⁷ Norway's second written submission, paras. 126-134. See also Norway's response to Panel question Nos. 17, 18, 43, and 139.
 ⁹⁶⁸ Appellate Body Reports, *Brazil – Retreaded Tyres*, para. 156; *Korea – Various Measures on Beef*,

⁹⁶⁸ Appellate Body Reports, *Brazil – Retreaded Tyres*, para. 156; *Korea – Various Measures on Beef*, para. 164; *EC – Asbestos*, para. 172, *US – Gambling*, para. 306, *Dominican Republic – Import and Sale of Cigarettes*, para. 70.

⁹⁶⁹ Appellate Body Report, *Korea – Various Measures on Beef*, para. 162.

⁹⁷⁰ Appellate Body Report, *Brazil – Retreaded Tyres*, para. 156.

⁹⁷¹ This part of the Article XX analysis may be comparable to the risks of non-fulfilment under Article 2.2 of the TBT Agreement.

⁹⁷² The European Union also refers to the panel statement in *China – Audiovisual Products* that the protection of public morals "ranks among the most important values or interests pursued by members as a matter of public policy". (Panel report, *China – Audiovisual Products*, para. 7.817).

The panel elaborated that "we do not consider it simply accident that the exception relating to 'public morals' is the first exception identified in the ten subparagraphs of Article XX. We therefore concur that the protection of public morals is a highly important value or interest." (Panel Report, *China –Audiovisual Products*, para. 7.817).

Regarding the methodology used to make an assessment of the measure's contribution, the Appellate Body clarified that it ultimately depends on the nature, quantity, and quality of evidence existing at the time the analysis is made.⁹⁷³ Further, the analysis of the contribution can be done either in quantitative or qualitative terms.⁹⁷⁴

7.634. Given the close relationship between the GATT 1994 and the TBT Agreement and the need to interpret relevant provisions under both Agreements in a consistent and harmonious manner, we consider that an analysis of a measure's contribution to an objective under Article 2.2 of the TBT Agreement is also relevant to such analysis under Article XX of the GATT 1994. The Appellate Body in US - Tuna II (*Mexico*) recalled that in assessing the necessity of a measure under Article 2.2, a panel must assess the contribution to the legitimate objective *actually achieved* by the measure at issue as in other situations, such as for instance when determining the contribution of a measure to the achievement of a particular objective in the context of Article XX.⁹⁷⁵ Accordingly, we will refer back to our relevant analysis under Article 2.2 of the TBT Agreement to the extent necessary for the analysis of the measure's contribution under Article XX(a) of the GATT 1994.

7.635. Before beginning our analysis, we address one additional element in relation to the measure's contribution in the context of Article XX of the GATT 1994: trade-restrictiveness of a measure. The Appellate Body stated in the context of Article XX(b) that when a measure produces restrictive effects on international trade as severe as those resulting from an import ban, it would be difficult for a panel to find that measure necessary unless it is satisfied that the measure is apt to make a *material* contribution to the achievement of its objective.⁹⁷⁶ Thus, the trade-restrictiveness of a measure is closely linked to the extent of the measure's contribution to the objective that must be proved in assessing the overall necessity of a measure in the context Article XX of the GATT 1994.⁹⁷⁷

7.636. In light of the particular circumstances of the present dispute, and given the guidance by the Appellate Body, we consider that, for a preliminary finding that the measure as a whole is "necessary" within the meaning of paragraph (a), the contribution made by the "ban" to the identified objective must be shown to be at least material given the extent of its trade-restrictiveness. As discussed under Article 2.2 of the TBT Agreement, the actual degree of the contribution by the ban must also be assessed by taking account of the impact of the explicit and implicit exceptions on the operation of the ban. This will allow us to objectively evaluate the necessity of the measure against any reasonably available GATT-consistent or less trade-restrictive alternative measures that make an equivalent or greater contribution to the objective than the measure in question.

7.637. With respect to the ban aspect of the EU Seal Regime, we recall our earlier finding that the ban does contribute to the European Union's objective by reducing, to a certain extent, the global demand for seal products and by helping the EU public avoid being exposed to seal products on the EU market that may have been derived from seals killed inhumanely.⁹⁷⁸ To the extent that such seal products are prohibited from the EU market⁹⁷⁹, we find that the ban makes a material contribution to the objective of the measure.

⁹⁷³ Appellate Body Report, *Brazil – Retreaded Tyres*, paras. 145-157.

⁹⁷⁴ According to the Appellate Body, the same applies to the analysis of the existence of a risk (in the case of Article XX(b)) that the measure aims to prevent. (Appellate Body Report, *Brazil – Retreaded Tyres*, para. 146).

para. 146). ⁹⁷⁵ Appellate Body Report, *US – Tuna II (Mexico)*, para. 317. (emphasis added) Accordingly, we followed the analytical approach taken for the assessment of a measure's contribution under Article XX in previous disputes for the purpose of our analysis of the measure's contribution under Article 2.2 of the TBT Agreement. ⁹⁷⁶ Appellate Body Beart, Brazel, Detraceded Types areas 150, 151

⁹⁷⁶ Appellate Body Report, *Brazil – Retreaded Tyres*, paras. 150-151.

⁹⁷⁷ By contrast, under Article 2.2 of the TBT Agreement, the Appellate Body did not consider that a certain threshold degree of contribution exists. For example, in US - Tuna II (*Mexico*) and US - COOL, the necessity test by the Appellate Body in the context of Article 2.2 entailed an assessment of the "*degree* of contribution" by a measure without necessarily determining whether the actual degree of contribution by the measure reaches a certain minimum threshold such as a *material* or *significant* contribution.

⁹⁷⁸ See section 7.3.3.3.2 above.

⁹⁷⁹ We recall that the EU market constituted an important market for seal products from Canada and Norway prior to the introduction of the EU Seal Regime. (See, for example, Canada's response to Panel question No. 96). For Canada, Norway and the European Union were the two main markets for raw seal skins,

- 172 -

7.638. However, as discussed in section 7.3.3.3.2 in detail, the degree of the contribution made by the ban is diminished by both the explicit and implicit exceptions under the measure. Specifically, with respect to the IC and MRM exceptions, we considered that the exceptions, combined with the absence of any mechanism under the measure to inform consumers of the presence of seal products on the EU market, reduced the effectiveness of the ban under the measure by allowing seal products access to the EU market. Further, the implicit exceptions provided under the measure through certain commercial activities such as transit and inward processing for export were also found to undermine the measure's fulfilment of the objective. Overall, with respect to the EU Seal Regime as a whole, however, we found that it contributed to a certain extent to its objective of addressing the EU public moral concerns on seal welfare.

7.639. Based on the assessment above, the EU Seal Regime can be provisionally deemed "necessary" within the meaning of Article XX(a) of the GATT 1994, unless it is demonstrated that the European Union could have adopted a GATT-consistent or less trade-restrictive measure as an alternative to the EU Seal Regime. For the reasons discussed in detail in section 7.3.3.3.4 above, however, we concluded that the alternative measure proposed by the complainants was not reasonably available to the European Union given inter alia the animal welfare risks and challenges found to exist in seal hunting in general. Therefore, we consider that the EU Seal Regime is provisionally deemed necessary within the meaning of Article XX(a) of the GATT 1994.

7.4.4.4 Paragraph (b) of Article XX

7.640. We have taken note of the European Union's argument that the EU Seal Regime, to the extent that it addresses the moral concerns of the EU population by reducing the number of seals that are killed, also falls within the scope of Article XX(b) because it contributes to protecting the health of seals.⁹⁸⁰ The European Union never submitted in this dispute that the protection of seal welfare as such was the objective of the EU Seal Regime. Based on the examination of the measure at issue as well as other available evidence before us, we determined that the objective of the EU Seal Regime was to address the EU public moral concerns on seal welfare.⁹⁸¹ We further found this objective to fall within the scope of the policy objective governed by Article XX(a). Under these circumstances, and given the limited extent of the European Union's arguments under Article XX(b), we consider that the European Union has failed to establish a prima facie case for its claim under Article XX(b).

7.4.4.5 Chapeau of Article XX

7.4.4.5.1 Main arguments of the parties

7.641. The European Union argues that the EU Seal Regime is not applied in a manner that constitutes "arbitrary or unjustifiable discrimination between countries where the same conditions prevail" because it applies indistinctly irrespective of the country of origin of the products.⁹⁸²

7.642. Canada notes that the focus under the chapeau is on the cause or rationale for the discrimination in the light of the policy objective, and contends that the regulatory distinctions introduced by the IC and MRM categories do not bear any rational relationship to the two policy objectives advanced by the European Union. According to Canada, the rigidity with which the EU Seal Regime is applied and the disregard for differing regulatory conditions in sealing countries demonstrates that the EU Seal Regime, as applied, constitutes arbitrary and unjustifiable discrimination between countries where the same conditions prevail.⁹⁸³

7.643. Norway submits that under the chapeau of Article XX a panel may consider both the substantive content and the application of a measure. Although the chapeau of Article XX refers to the application of a measure, this does not mean that substantive content (design and structure) is irrelevant. In Norway's view, the manner of application of a measure will be heavily influenced,

exports. ⁹⁸⁰ European Union's first written submission, para. 591; Canada's second written submission, para. 204; Norway's second written submission, paras. 126-134.

fats and oils, etc. during the period of 2000-2010 with the European Union receiving 50% of all Canadian

⁹⁸¹ See section 7.3.3.1 above.

⁹⁸² European Union's first written submission, para. 590.

⁹⁸³ Canada's second written submission, paras. 218-226.

- 173 -

possibly decisively so, by the substantive content of the measure. Thus, while the legal standards of discrimination under the provisions are not the same, in this case the same elements considered under Articles I and III also lead to the measure failing to meet the requirements of the chapeau of Article XX. Norway also cites arbitrary and unjustifiable discrimination between countries where the same animal welfare and resource management conditions prevail, and the European Union's failure to engage in international negotiations (referencing US - Shrimp).⁹⁸⁴

7.4.4.5.2 Analysis by the Panel

7.644. Having found that the EU Seal Regime as a whole is "necessary" within the meaning of Article XX(a), we examine its consistency with the requirements under the chapeau of the provision. The chapeau of Article XX provides:

Subject to the requirement that such measures are not applied in a manner which would constitute a means of arbitrary or unjustifiable discrimination between countries where the same conditions prevail, or a disguised restriction on international trade, nothing in this Agreement shall be construed to prevent the adoption or enforcement by any contracting party of measures ...

7.645. The focus of the chapeau, according to the Appellate Body, is on the application of a measure already found inconsistent with an obligation of the GATT 1994 but falling within one of the paragraphs of Article XX.⁹⁸⁵ Specifically, the existence of one of the three types of situations regarding the application of measures might lead to an inconsistency with the chapeau of Article XX: (a) arbitrary discrimination between countries where the same conditions prevail; (b) unjustifiable discrimination between countries where the same conditions prevail; or (c) a disguised restriction on international trade.⁹⁸⁶ We observe that in previous disputes, the first two situations (i.e. arbitrary or unjustifiable discrimination) were often addressed together.⁹⁸⁷ We are also mindful of the Appellate Body's guidance that "the fundamental theme - when interpreting the chapeau - is to be found in the purpose and object of avoiding abuse or illegitimate use of the exceptions to substantive rules available in Article XX".988

7.646. Based on this guidance, the focus of our analysis under the chapeau with respect to the EU Seal Regime is whether the EU Seal Regime, in particular the distinctions drawn in the form of the IC and MRM exceptions, are applied in a manner which would constitute a means of arbitrary or unjustifiable discrimination.

⁹⁸⁴ Norway's response to Panel guestion No. 146.

⁹⁸⁵ Appellate Body Report, Brazil - Retreaded Tyres, para. 215. Further, the Appellate Body emphasizes the principle of good faith embodied in the chapeau: "[t]he chapeau serves to ensure that Members' rights to avail themselves of exceptions are exercised in good faith to protect interests considered legitimate under Article XX, not as a means to circumvent one Member's obligations towards other WTO Members." (Appellate Body Report, Brazil – Retreaded Tyres, paras. 215 and 224).

The Appellate Body further states in this regard:

[[]t]he function of the chapeau is the prevention of abuse of the exceptions specified in the paragraphs of Article XX. ... "[t]he chapeau of Article XX is, in fact, but one expression of the principle of good faith." ... "[o]ne application of this general principle ... prohibits the abusive exercise of a state's rights and enjoins that whenever the assertion of a right 'impinges on the field covered by [a] treaty obligation, it must be exercised bona fide, that is to say, reasonably'. Accordingly, the task of interpreting and applying the chapeau is "the delicate one of locating and marking out a line of equilibrium between the right of a Member to invoke an exception under Article XX and the rights of the other Members under varying substantive provisions (e.g. Article XI) of the GATT 1994 ..." The location of this line of equilibrium may move "as the kind and the shape of the measures at stake vary and as the facts making up specific cases differ". (Appellate Body Report, Brazil - Retreaded Tyres, para. 224 (citing Appellate Body Report, US -*Shrimp*)) (footnotes omitted)

⁹⁸⁶ Appellate Body Report, US – Shrimp, para. 184.

⁹⁸⁷ See, for instance, Appellate Body Reports, US – Gasoline, US – Shrimp (Article 21.5 – Malaysia), US - Gambling, and Brazil - Retreaded Tyres; and Panel Reports, US - Gambling, EC - Tariff Preferences, EC -Asbestos, and Brazil – Retreaded Tyres. 988 Appellate Body Report, US – Gasoline, p. 25 (emphasis added); Panel Report, Brazil – Retreaded

Tyres, para. 7.221.

7.647. First, with respect to the existence of "discrimination" within the meaning of the chapeau, we find it useful to recall the panel's finding in *Brazil – Retreaded Tyres*:

[t]he initial violation identified in relation to this measure is a prohibition or restriction on importation within the meaning of Article XI. This type of measure (an import ban in this instance), does not necessarily *ipso facto* result in discrimination, as an inconsistency with Articles I or III would. Thus, any discrimination alleged to exist in the application of the measure would arise, in this case, in addition to the restriction that is inherently present in the measure by its very nature.⁹⁸⁹

7.648. As in that dispute, the measure in question here also consists of a ban, albeit formulated in positive forms, and exceptions. We note however that, unlike the measure in *Brazil – Retreaded Tyres*, the IC and MRM exceptions are embedded in the measure itself rather than arising from the subsequent application of the measure. In our view, this is merely a formal difference that does not change the substantive character of a measure consisting of a ban with discriminatory exceptions.⁹⁹⁰ Therefore, under the circumstances of this dispute, we consider that discrimination alleged to exist in the application of the EU Seal Regime within the meaning of the chapeau results from the discriminatory impact found in the IC and MRM exceptions under Articles I:1 and III:4.⁹⁹¹

7.649. Accordingly, we proceed to examine whether such discrimination is arbitrary or unjustifiable within the meaning of the chapeau of Article XX and arises between countries where the same conditions prevail. An assessment of this question leads us to recall our analysis of the legitimacy of the regulatory distinctions drawn in the IC and MRM exceptions under Article 2.1 of the TBT Agreement. As noted in section 7.4.1, the Appellate Body clarified the legal standard under Article 2.1 of the TBT Agreement (i.e. "legitimate regulatory distinction") based on *inter alia* its observation regarding the relationship between the GATT 1994 and the TBT Agreement and the absence in the TBT Agreement of a general exceptions clause equivalent to Article XX in the GATT 1994.⁹⁹² This means, in our view, that our analyses under Article 2.1 of the TBT Agreement regarding the legitimacy of the IC and MRM exceptions are relevant and applicable to the assessment of the exceptions for its consistency with the requirements under the chapeau.

7.650. More specifically, under Article 2.1 of the TBT Agreement, we examined, first, whether the regulatory distinctions drawn between commercial and IC/MRM hunts were rationally connected to the objective of the EU Seal Regime as a whole or otherwise based on justifiable grounds, and, second, whether such distinctions, as reflected in the IC and MRM exceptions under measure, were designed and applied in an even-handed manner among the potential beneficiaries.⁹⁹³ For the IC exception, given the recognized benefits to Inuit, we found the distinction between commercial and IC hunts (and hence the products regulated based on that distinction) to be justifiable despite the lack of a rational connection to the measure's objective. For the same reasons discussed under the TBT Agreement, however, we find that due to the lack of even-handedness in the design and application of the IC exception, the IC exception does not meet the requirements under the chapeau of Article XX. Regarding the MRM exception, we found that the distinction between commercial and MRM hunts is neither rationally connected to the objective nor based on any justifiable grounds. Further, consistent with our view under the TBT Agreement, the MRM exception is not designed and applied in an even-handed manner and hence is inconsistent with the requirements of the chapeau of Article XX.

⁹⁸⁹ Panel Report, *Brazil – Retreaded Tyres*, para. 7.229. We note that in *Brazil – Retreaded Tyres* the complainant raised a claim against the ban under Article XI of the GATT 1994.

⁹⁹⁰ This was also taken into account in our consideration of the EU Seal Regime's qualification as a technical regulation. (See, above section 7.3.1).

⁹⁹¹ We are mindful that "the 'nature and quality' of the discrimination referenced in the chapeau of Article XX is different from the discrimination in the treatment of products that might already have been found to be inconsistent with one of the substantive obligations of the GATT 1994". Panel Report, *Brazil – Retreaded Tyres*, para. 7.229 (referring to Appellate Body Report, *US – Gasoline*, p. 21) However, we consider that the circumstances in the present dispute support our approach here. Specifically, such circumstances include the following: the character of the measure having exceptions to a ban that result in discrimination; the manner in which the complainants framed their contentions by focusing on the exceptions aspect of the measure; and the need to read relevant provisions of the TBT Agreement (e.g. preamble and Articles 2.1 and 2.2) and of the GATT 1994 in a consistent and harmonious manner.

⁹⁹² See section 7.3.2.3 above.

 $^{^{993}}$ See sections 7.3.2.3.3 and 7.3.2.3.4 above.

- 175 -

7.651. Therefore, we conclude that the European Union has failed to establish that the discriminatory impact found in the IC and MRM exceptions under the EU Seal Regime is justified under Article XX(a) of the GATT 1994.

7.5 Claims under Article XI:1 of the GATT 1994 and Article 4.2 of the Agreement on Agriculture

7.5.1 Article XI:1 of the GATT 1994

7.5.1.1 Summary of the parties' arguments

7.652. In their respective first written submissions, the complainants argue that the "EU Seal Regime" is inconsistent with the European Union's obligations under Article XI:1 of the GATT 1994.⁹⁹⁴ Specifically, Canada argues that the EU Seal Regime is a border measure that operates as a quantitative restriction for foreign seal products, and as an import prohibition on Canadian seal products derived from commercial hunts.⁹⁹⁵ Norway also asserts that the EU Seal Regime effectively operates as a border measure that is inconsistent with Article XI:1 to the extent that importation of seal products is permitted only if the products conform to one of the three exceptions imposed under the Seal Regime.⁹⁹⁶

7.653. In the subsequent stages of the proceedings, however, the complainants argue that each of the IC, MRM, and Travellers exceptions under the Regime, rather than the EU Seal Regime as a whole, violates Article XI:1 of the GATT 1994 because they all impose quantitative restrictions on the importation of Canadian and Norwegian seal products.⁹⁹⁷ Seal products can only be placed on the EU market if they satisfy the conditions of one of the exceptions; as such, each of the exceptions has a "limiting effect" on the importation of seal products, which amounts to a restriction on imports within the meaning of Article XI:1.⁹⁹⁸

7.654. More specifically, according to the complainants, the IC and Travellers exceptions do not apply to domestic seal products as the conditions only apply to seal products at the point of import.⁹⁹⁹ The nature and quantity of the seal products imported under the Travellers exception cannot be such as to indicate that the products are being imported for commercial reasons¹⁰⁰⁰, and the conditions of the IC exception effectively apply only to imported seal products.¹⁰⁰¹

⁹⁹⁴ In its panel request, Canada states, "each of the measures referred to above [the Regulations] is inconsistent with ... Article XI:1 of the GATT 1994 because the measures result in a prohibition or restriction on the importation of seal products from Canada into the European Union." (Canada's request for the establishment of a panel, p. 2). We also observe that Canada argued that "the measure" operates *de facto* as a border measure imposing a discriminatory restriction on the importation of seal products, in violation of Articles I:1 and XI:1 of the GATT 1994, rather than as an internal measure. (Canada's response to Panel question No. 1).

Norway states in its panel request, "by restricting the importation of seal products, the EU Seal Regime appears to violate Article XI:1 of the GATT 1994 and Article 4.2 of the Agreement on Agriculture." For its claims under Articles I:1 and III:4, by contrast, Norway refers to "the general prohibition and the exceptions set out [in the EU Seal Regime]" in its panel request. (Norway's request for the establishment of a panel, p. 2).

²⁹⁵ Canada's first written submission, para. 281.

⁹⁹⁶ Norway's first written submission, para. 457.

⁹⁹⁷ Canada's second written submission, para. 121; Norway's second written submission, para. 94; Parties' responses to Panel question No. 3. The complainants submit that different aspects of a measure may affect the competitive opportunities of imported products "in different ways"; it is therefore appropriate to examine these aspects under different provisions. The IC, MRM, and Travellers exceptions under the EU Seal Regime may thus be assessed on their own for the purpose of determining the specific GATT article under which they should be examined.

⁹⁹⁸ Canada's second written submission, paras. 124-129; Norway's second written submission, paras. 114-125.

⁹⁹⁹ Canada's second written submission, paras.124-126; Norway's second written submission, paras. 102, 105-106; response to Panel question No. 3, paras. 16-17. Seal products qualifying under the Travellers exception are acquired outside of the European Union and can only be "imported" for personal use of travellers or their families. (Canada's second written submission, para. 124; Norway's second written submission, para. 115 (referring to Basic Regulation, Article 3(2)(a))).

¹⁰⁰⁰ Canada's second written submission, para. 124; Norway's second written submission, para. 116.

¹⁰⁰¹ Canada and Norway argue that in any event, the IC category violates Article I:1 of the GATT 1994. The complainants note in respect of the IC exception that their claims under Article I:1 are not alternative to

7.655. Further, the complainants indicate that their claims under Article III:4 and Article XI:1 of the GATT 1994 with respect to the MRM exception are alternative claims. The complainants assert that the application of Article III:4 or XI:1 of the GATT 1994 depends on a factual determination as to whether the restrictive conditions of the MRM exception apply effectively to the placing on the market of seal products of EU origin.¹⁰⁰²

7.656. The European Union argues that the measure does not fall within the scope of Article XI:1; the EU Seal Regime is not a border measure but an internal measure that applies indistinctly to foreign and domestic products.¹⁰⁰³ The Regime imposes a "General Ban" on seal products, and the IC and MRM exceptions apply to both imported and "like" domestic seal products. With respect to the Travellers exception, which applies only to imports, it provides "more favourable treatment" by way of derogation from the "General Ban".¹⁰⁰⁴ The exceptions under the EU Seal Regime are not trade restrictive and thus cannot amount to import restrictions under Article XI:1 of the GATT 1994.¹⁰⁰⁵ The European Union notes that "the fact that the EU Seal Regime is enforced at the border is merely for administrative convenience in order to ensure effective enforcement."¹⁰⁰⁶

7.5.1.2 Analysis by the Panel

7.657. Article XI:1 of the GATT 1994 requires that WTO Members not institute or maintain any "prohibitions or restrictions other than duties, taxes, or other charges, whether made effective through quotas, import or export licences or other measures ... on the importation of any product of the territory of any other [Member]."¹⁰⁰⁷

7.658. The complainants contend that each of the IC, MRM, and Travellers exceptions, considered individually, imposes quantitative limitations on imported seal products in violation of Article XI:1 of the GATT 1994.¹⁰⁰⁸ Therefore, we do not understand the complainants to be claiming that the EU Seal Regime as a whole has a restrictive impact on imported seal products inconsistently with Article XI:1 of the GATT 1994; rather, they contend that each exception results in a limiting effect on imports and thus violates Article XI:1.

- ¹⁰⁰⁴ European Union's second written submission, para. 378.
- ¹⁰⁰⁵ European Union's second written submission, para. 379.
- ¹⁰⁰⁶ European Union's second written submission, para. 376.

¹⁰⁰⁷ The guidance provided in previous disputes regarding its scope suggests that Article XI:1 does not apply to internal regulations affecting imported products that also apply to the like domestic products; instead, according to the *Ad* Note to Article III of the GATT 1994, these are dealt with under Article III.

The *Ad* Note to Article III of the GATT 1994 reads as follows:

Any internal tax or other internal charge, or any law, regulation or requirement of the kind referred to in paragraph 1 which applies to an imported product and to the like domestic product and is collected or enforced in the case of the imported product at the time or point of importation, is nevertheless to be regarded as an internal tax or other internal charge, or a law, regulation or requirement of the kind referred to in paragraph 1, and is accordingly subject to the provisions of Article III.

See, for instance, Panel Report, *India – Autos*, para. 7.220 (referring to GATT Panel Report, *Canada – FIRA*, para. 5.14) ("... the General Agreement distinguishes between measures affecting the 'importation' of products, which are regulated in Article XI:1, and those affecting 'imported products', which are dealt with in Article III. If Article XI:1 were interpreted broadly to cover also internal requirements, Article III would be partly superfluous.").

¹⁰⁰⁸ With respect to the MRM exception, the complainants explain that their claim under Article XI:1 is alternative to their claim under Article III:4 in light of the *Ad* Note to Article III, to the extent that the MRM exception can be considered as applying to domestic products. At the same time, in the context of Article XI:1, the complainants argue that the conditions under the MRM exception do not apply *de facto* to the European Union's domestic seal products.

their claims under Article XI:1. (Canada's second written submission, para. 126; Norway's response to Panel question No. 3).

¹⁰⁰² See the complainants' responses to Panel question No. 3; Canada's second written submission, paras. 127-129; Norway's second written submission, paras. 94-95.

In the complainants' view, the MRM exception operates "in effect" as a border measure. The design, structure and expected operation of the MRM exception are such that none of the restrictive conditions under that exception "applies to ... the like domestic product" in the sense of the *Ad* Note to Article III; therefore, the "real impact" of the MRM requirements can only be felt by imported products.

¹⁰⁰³ European Union's first written submission, paras. 488-499; second written submission, para. 374.

- 177 -

7.659. In our understanding, the complainants' claims under Article XI:1 of the GATT 1994 are linked to their position, as discussed in section 7.2.2 above, that the EU Seal Regime consists of three sets of requirements (IC, MRM, and Travellers), rather than a ban with exceptions. For the complainants, each of these three sets of requirements can independently violate different provisions of the GATT 1994 such as Articles I:1, III:4, and XI:1, as they "affect the competitive opportunities for imported products in different ways".¹⁰⁰⁹ For example, the allegedly discriminatory impact of the IC and MRM exceptions was presented in the complainants' claims under Articles I:1 and III:4 of the GATT 1994, respectively.

7.660. We recall our finding above regarding the characterization of the EU Seal Regime as a measure containing both prohibitive and permissive aspects, namely a ban and exceptions.¹⁰¹⁰ In reaching this conclusion, we found that the prohibitive aspect of the measure, namely the ban on the placing on the market and importation of seal products, was implied in the terms and expected operation of the provisions under the Basic Regulation. Specifically, we observed that the importation and placing on the market of seal products are allowed under the measure only through the three exceptions.¹⁰¹¹ Consequently, the measure works effectively as a ban on seal products, including imports, in any other situation.

7.661. Hence, it is the implicit prohibitive aspect of the measure, the scope of which is informed by the exceptions, that restricts imported seal products rather than the exceptions considered individually.¹⁰¹² In the factual circumstances of this dispute, a restriction on imported products is imposed in the form of an implicit ban under the measure rather than through the individual exceptions as claimed by the complainants.¹⁰¹³ In other words, it is the EU Seal Regime as a whole, providing for specific exceptions to a ban, that results in a restrictive impact on the importation of products from certain sources.

7.662. In this dispute, the complainants focused on the discriminatory aspects of the Regime, particularly with respect to the IC and MRM exceptions of the Regime, rather than the measure in its entirety as an import "prohibition or restriction" within the meaning of Article XI:1 of the GATT 1994.¹⁰¹⁴ The complainants consider that each individual exception is an independent source

See Panel Report, India - Autos, para. 7.223. The panel in India - Autos noted that it saw "merit in the proposition that there may be circumstances in which specific measures may have a range of effects. In appropriate circumstances they may have an impact both in relation to the conditions of importation of a product [Article XI:1] and in respect of the competitive conditions of imported products on the internal market within the meaning of Article III:4." (Panel Report, India - Autos, paras. 7.223 and 7.296). Accordingly, added the panel, any analysis of the applicability of either Article III:4 or XI:1 should thus be based on the principles within Article 3.2 of the DSU. (Panel Report, India - Autos, para. 7.224)

The panel in India - Autos exercised judicial economy for the United States' claim under Article III:4 with respect to the measure that was already found inconsistent with Article XI:1; the panel did not consider it necessary to separately consider the United States' general claim that the measure was also inconsistent with Article III:4. While the panel thus refrained from further consideration of the broader application of Article III:4 to the same features of the measure dealt with in the Article XI analysis, it nevertheless found it appropriate to make a separate ruling on one distinct element of the (same) measure, which was alleged by the complainants to constitute a violation of Article III:4 but was not considered in the panel's Article XI analysis.

¹⁰¹³ See section 7.2.2 above.

¹⁰¹⁴ As noted in above, in its first submission, both Canada and Norway argued that the EU Seal Regime as a whole is a quantitative restriction on importation for purposes of Article XI:1. However, in the subsequent phases of the proceedings, Canada and Norway focus on the restrictive effect of the requirements under the IC, MRM, and Travellers exceptions respectively.

We note that in EC - Asbestos, where the measure at issue also consisted of a ban and exceptions to the ban, Canada argued that Article XI:1 applies to the ban on imports (one component of the EC measure), whereas the European Union considered that as the import ban is merely the logical corollary of the general prohibition on the use of asbestos and asbestos-containing products, Article III:4 must be assessed in the light of Note Ad Article III. Canada put forward several propositions in sequential order with respect to its claims: (a) first, the EC measure comes in part under Article III:4 and in part under Article XI:1; (b) if the panel were to reject that proposition, then the whole of the EC measure at issue should fall under Article XI:1; or (c) if the panel were also to reject that approach, the whole of the EC measure should then fall under Article III:4.

¹⁰⁰⁹ See, e.g. Canada's second written submission, para. 121; Norway's second written submission, para. 106 ¹⁰¹⁰ See section 7.2.2 above.

¹⁰¹¹ We also observed above that certain commercial activities such as transit and inward processing of seal products that are otherwise prohibited under the measure are implicitly allowed under the Regime. (See para. 7.53 above).

¹⁰¹² This is not to say that different aspects of a measure cannot be examined under different provisions of the WTO Agreement, as we clarified in para. 7.27 above.

- 178 -

of restrictiveness for imports. For the reasons explained above, however, we disagree with the complainants. With respect to the Travellers exception, the complainants did not present any other specific claim than Article XI:1 of the GATT 1994. However, we do not consider that the Travellers exception, considered on its own, imposes an import restriction within the meaning of Article XI:1 of the GATT 1994. As a derogation from the implicit ban, the Travellers exception allows travellers to bring into the European Union seal products that are otherwise prohibited under the measure.¹⁰¹⁵

7.663. Based on our considerations above, we are not persuaded by the complainants' argument that each of the IC, MRM, and Travellers exceptions individually imposes an import restriction in violation of Article XI:1 of the GATT 1994. Thus, the Panel rejects the complainants' claims under Article XI:1 of the GATT 1994 with respect to all three exceptions under the EU Seal Regime.

7.5.2 Article 4:2 of the Agreement on Agriculture (Norway)

7.664. Norway argues that the "EU Seal Regime" violates Article 4.2 of the Agreement on Agriculture.¹⁰¹⁶ Norway asserts that if the "EU Seal Regime" is found to violate Article XI:1 of the GATT 1994, then it would also violate Article 4.2 of the Agreement on Agriculture, because a quantitative restriction on importation for purposes of Article XI:1 would also constitute a "quantitative import restriction" on agricultural products prohibited under Article 4.2 of the Agreement on Agriculture.¹⁰¹⁷ The European Union requests the Panel to reject Norway's claim because the EU Seal Regime is not a border measure subject to Article XI:1 of the GATT 1994, but an internal regulatory measure that applies to both domestic and imported seal products.¹⁰¹⁸

7.665. In light of Norway's reliance on Article XI:1 of the GATT 1994 for its claim under Article 4.2 of the Agreement of Agriculture, and given our finding above in the context of Article XI:1 of the GATT 1994, the Panel rejects Norway's claim under Article 4.2 of the Agreement on Agriculture.

7.6 Non-violation claim under Article XXIII:1(b) of the GATT 1994

7.666. Having addressed the complainants' claims of inconsistency with the TBT Agreement and the GATT 1994, we now turn to the complainants' non-violation claims under Article XXIII:1(b) of the GATT 1994.¹⁰¹⁹

The panel in EC - Asbestos commented that it was difficult to tell, based on Canada's arguments, whether Canada was claiming a cumulative application of Articles III:4 and XI:1 to the part of the measure banning imports. It then further stated that "[i]f Canada does in fact make such a claim ..., we do not consider that this forms part of the terms of reference given to the Panel by the DSB and, even if that were the case, Canada's arguments do not make a prima facie case in the sense given to this concept by the Appellate Body." The panel consequently did not consider it necessary to examine this point any further. (Panel Report, EC - Asbestos, paras. 8.83-8.100).

¹⁰¹⁵ Further, see footnote 71 above for the definition of the term "import" under the EU Seal Regime. Given the scope and use of the term "import" in the measure, combined with the nature of the Travellers requirements, the scope of seal products allowed to "enter into the customs territory of the Community" under the Travellers exception seems to have a fundamentally different character (i.e. being "exclusively ... for ... *personal use*") than seal products governed by the ban and the IC and MRM exceptions of the EU Seal Regime.

¹⁰¹⁶ Norway argues that for the same reasons that these requirements constitute restrictions prohibited under Article XI:1, they also constitute a "quantitative import restriction" on agricultural products that is prohibited under Article 4.2 of the Agreement on Agriculture. (Norway's second written submission, para. 118). Norway considers that the Agreement on Agriculture is applicable to seal products restricted by the EU Seal Regime because all the products covered by the EU Seal Regime are listed in Annex 1 of the Agreement on Agriculture as falling within the scope of the Agreement. These products are listed in the European Commission's Technical Guidance Note issued pursuant to Article 3(3) of the Basic Regulation (Exhibit JE-3).

¹⁰¹⁷ Norway refers to the finding of the panel in *Korea – Various Measures on Beef* that "... the general prohibition against import restrictions contained in Article XI and its *Ad* Note find a more specific application in Article 4.2 of the Agreement on Agriculture together with its footnote with regard to agricultural products." (Norway's first written submission, paras. 465-466 (citing Panel Report, *Korea – Various Measures on Beef*, footnote 400)).

¹⁰¹⁸ European Union's first written submission, paras. 623 and 625.

¹⁰¹⁹ We note that this follows the sequence in which the complainants have presented their claims, and is in keeping with the general priority to be accorded to addressing violation as opposed to non-violation claims. (See Panel Reports, *US* – *COOL*, para. 7.888; *Japan* – *Film*, para. 10.26 ("traditionally in cases involving both violation and non-violation claims, panels first address claims of inconsistency with a covered

7.6.1 Main arguments of the parties

7.667. As referenced by all three parties¹⁰²⁰, the panel in Japan – Film laid out the three required elements of a claim under Article XXIII:1(b) that are applicable in the current dispute:

The text of Article XXIII:1(b) establishes three elements that a complaining part must demonstrate in order to make out a cognizable claim under Article XXIII:1(b): (1) application of a measure by a WTO Member; (2) a benefit accruing under the relevant agreement; and (3) nullification or impairment of the benefit as a result of the application of the measure.¹⁰²¹

7.668. With regards to the first element, both Canada and Norway argue, and the European Union does not dispute, that the EU Seal Regime is a measure within the meaning of Article XXIII:1(b) that has been applied by the European Union.¹⁰²²

7.669. Regarding the second element, both Canada and Norway rely on the conception of "benefit" under Article XXIII:1(b) as "that of legitimate expectations of improved market-access opportunities arising out of relevant tariff concessions" ¹⁰²³ Additionally, both complainants recognize in their arguments that "in order for expectations of a benefit to be legitimate, the challenged measures must not have been reasonably anticipated at the time the tariff concession was negotiated."1024

7.670. As to the relevant tariff concessions, Canada and Norway identify concessions granted by the European Union for the seal products listed in Exhibit JE-42.¹⁰²⁵ Regarding the legitimacy of their expectations of improved market-access opportunities, both Canada and Norway invoke the rebuttable presumption articulated by the panel in Japan – Film that measures introduced after the conclusion of tariff negotiations are *not* reasonably anticipated by a complainant.¹⁰²⁶ In the case at hand, the complainants indicate that the relevant tariff concessions were granted at the close of the Tokyo and Uruguay negotiating rounds, both of which were concluded years before the adoption of the Basic Regulation in 2009.¹⁰²⁷ Consequently, the complainants state that the burden is on the European Union to show that Canada and Norway should have anticipated the adoption of measures similar to the EU Seal Regime.¹⁰²⁸

7.671. The European Union counters that "recourse to GATT Article XXIII:1(b) should be treated as 'particularly exceptional' in relation to measures justified by Article XX(b)".¹⁰²⁹ The European Union generally relies on the conclusions of the panel in EC - Asbestos, which found that "situations that fall under Article XX justify a stricter burden of proof being applied..., particularly with regard to the existence of legitimate expectations".¹⁰³⁰ As a consequence, the panel declined

¹⁰²² Canada's first written submission, paras. 738-739; Norway's first written submission, paras. 1007-1008.

¹⁰²³ Panel Report, Japan – Film, para. 10.61. See Canada's first written submission, para. 741; Norway's first written submission, para. 980.

¹⁰²⁴ Panel Report, *Japan – Film*, para. 10.76. See Canada's first written submission, para. 741; Norway's first written submission, para. 985.

¹⁰²⁵ Canada's first written submission, para. 743; Norway's first written submission, para. 1009.

¹⁰²⁶ Panel Report, Japan – Film, para. 10.79. See Canada's first written submission, para. 741; Norway's first written submission, para. 990.

In this connection, the complainants dispute the position taken by the panel in EC - Asbestos that there is no such rebuttable presumption for measures justified under Article XX of the GATT 1994, arguing inter alia that this is contrary to relevant treaty text and past jurisprudence. (See Canada's and Norway's responses to Panel question No. 51). ¹⁰²⁷ Canada's first written submission, para. 743; Norway's first written submission, para. 1011.

¹⁰²⁸ Canada's first written submission, para. 744; Canada's second written submission, para. 352;

Norway's first written submission, para. 1012; Norway's second written submission, paras. 406-409. ¹⁰²⁹ European Union's first written submission, para. 597, citing Panel Report, *EC – Asbestos*,

para. 8.272. ¹⁰³⁰ Panel Report, *EC – Asbestos*, para. 8.282; European Union's first written submission, para. 601. The panel went on to state that "Members have recognized a priori the possibility that the benefits they derive from

agreement pursuant to Article XXIII:1(a), before moving on to consider claims of non-violation nullification or impairment under Article XXIII:1(b)").

¹⁰²⁰ See Canada's first written submission, para. 736; Norway's first written submission, paras. 975; European Union's first written submission, para. 592.

¹⁰²¹ Panel Report, *Japan – Film*, para. 10.41.

to apply the rebuttable presumption of Japan - Film as it did "not consider such a presumption to be consistent with the standard of proof ... applicable ... in the case of an allegation of a nonviolation nullification concerning measures falling under Article XX of the GATT 1994".¹⁰³¹

7.672. Under this burden of proof, and absent the rebuttable presumption described in Japan -Film, the European Union contends that the complainants could have reasonably anticipated the measure due to long-standing public concern about seal hunts and corresponding legislation, both internationally and within Canada and Norway specifically.¹⁰³²

7.673. Canada submits that the history of anti-sealing activities cited by the European Union to show concerns about the Canadian hunt fails to explain the nature of those concerns and to acknowledge that they were addressed by the Canadian government.¹⁰³³ Norway argues that the circumstances surrounding the adoption of the EU Seal Regime¹⁰³⁴ as well as its "design, structure, and operation"1035 demonstrate that such measures could not have been reasonably anticipated, and distinguishes various past measures relating to seals from the EU Seal Regime.¹⁰³⁶

7.674. Finally, under the third element, Canada and Norway adhere to the explanation of this element by the panel in Japan - Film that "it must be demonstrated that the competitive position of the imported products subject to and benefitting from a relevant market access (tariff) concession is being *upset by* ('nullified or impaired ... as the result of') the application of a measure not reasonably anticipated".¹⁰³⁷ The complainants likewise follow the interpretation that this element concerns a causal connection between the measure and the nullification or impairment which must be more than *de minimis*.¹⁰³⁸

7.675. Norway cites the statement by the panel in EC - Asbestos that "[b]y its very nature, an import ban constitutes a denial of any opportunity for competition"¹⁰³⁹, and both complainants argue that the EU Seal Regime indeed results in the denial of competitive opportunity for their products in the European Union market.¹⁰⁴⁰ Canada and Norway allege that the discriminatory nature and impacts of the EU Seal Regime are further evidence of the contribution to nullification or impairment of benefits¹⁰⁴¹, with Canada adding that European Union legislators specifically intended to target imports of Canadian seal products.¹⁰⁴²

7.676. In response, the European Union reiterates its position that "the EU Seal Regime does not discriminate, either de iure or de facto, between domestic and imported like products" and

certain concessions may eventually be nullified or impaired at some future time for reasons recognized as being of overriding importance. [...] Moreover, the nature and importance of certain measures falling under Article XX can also justify their being taken at any time, which militates in favour of a stricter treatment of actions brought against them on the basis of Article XXIII:1(b)". Panel Report, EC – Asbestos, para. 8.281; European Union's first written submission, para. 602.

¹⁰³¹ Panel Report, *EC – Asbestos*, para. 8.291; European Union's first written submission, para. 604.

¹⁰³² European Union's first written submission, paras. 608-618.

¹⁰³³ Canada's second written submission, para. 353-357. Canada refers to the focus of past anti-sealing activity on seal pups that are no longer hunted in Canada, as well as regulatory improvements that Canada has

made over the years. ¹⁰³⁴ Norway's first written submission, paras. 1015-1024. In particular, Norway discusses the contemporary recognition of the novelty of the matter and substantial amendment over the course of the legislative process

¹⁰³⁵ Norway's first written submission, paras. 1025-1028. Norway specifically argues that it could not have anticipated a measure having contradictory results in the pursuit of animal welfare (i.e. allowing products from inhumanely killed seals while banning products from humanely killed seals).

¹⁰³⁶ See Norway's second written submission, paras. 410-431.

¹⁰³⁷ See Panel Report, Japan – Film, para. 10.82 (emphasis in original); Canada's first written submission, para. 745; Canada's second written submission, para. 346; Norway's first written submission, para. 1002.

¹⁰³⁸ See Panel Report, Japan – Film, para. 10.84; Canada's first written submission, paras. 745, 747; Canada's second written submission, para. 346; Norway's first written submission, para. 1000.

¹⁰³⁹ Norway's first written submission, para. 1001, citing Panel Report, *EC – Asbestos*, para. 8.289. See also Norway's second written submission, para. 434 (clarifying that the effect of the EU Seal Regime is to ban

imports of seal products from Norway, even though other products are permitted). ¹⁰⁴⁰ Canada's first written submission, para. 747; Canada's second written submission, para. 347;

Norway's first written submission, para. 1034. ¹⁰⁴¹ Canada's first written submission, paras. 748-749; Canada's second written submission, para. 347; Norway's first written submission, para. 1035; Norway's second written submission, paras. 435-437. 1042 Canada's first written submission, para. 750.

therefore the complainants have not shown that the EU Seal Regime upsets the competitive relationship between them. $^{\rm 1043}$

7.6.2 Analysis by the Panel

7.677. The Appellate Body has endorsed the rationale for non-violation complaints identified in GATT precedent that "[t]he idea underlying [the provisions of Article XXIII:1(b)] is that the improved competitive opportunities that can legitimately be expected from a tariff concession can be frustrated not only by measures proscribed by the General Agreement but also by measures consistent with that Agreement."¹⁰⁴⁴ At the same time, however, the Appellate Body has confirmed that "the remedy in Article XXIII:1(b) 'should be approached with caution and should remain an exceptional remedy".¹⁰⁴⁵

7.678. In this connection, we also note that the panel in Japan – Film underscored that under Article XXIII:1(b) "each case should be examined on its own merits" involving an assessment of the particular circumstances of each dispute.¹⁰⁴⁶

7.679. We recall that we have found that the IC and MRM exceptions violate Articles I:1 and III:4 of the GATT 1994, respectively, and are not justified under Article XX(a) of the GATT 1994. Additionally, both exceptions were also found to be in violation Article 2.1 of the TBT Agreement.¹⁰⁴⁷ In light of these findings, we must consider whether it is necessary for us to additionally address the complainants' non-violation claim under Article XXIII:1(b) of the GATT 1994.

7.680. In previous disputes where a similar non-violation claim was addressed, a finding of violation of the provisions of the GATT 1994 was considered to render an examination of non-violation nullification or impairment of benefits unnecessary, for example "if compliance by the [respondent] with the finding on Article III:4 would necessarily remove the basis of the ... claim of nullification or impairment".¹⁰⁴⁸ Similarly, the relevant question in the present dispute is whether compliance by the European Union with the above-mentioned findings of violation would remove the basis of the complainants' non-violation claims.

7.681. We observe certain parallels between the elements of the legal tests under Articles I:1 and III:4, and Article XXIII:1(b) of the GATT 1994. The panel in *Japan – Film*, after reviewing WTO/GATT case law under Articles I and III of the GATT 1994 concerning *de facto* discrimination, considered that "the reasoning contained therein appears to be equally applicable in addressing the question of *de facto* discrimination with respect to claims of non-violation nullification or impairment".¹⁰⁴⁹ In identifying this similarity, the panel stated that non-violation claims relate not to "equality of competitive conditions" but to the upsetting of "relative conditions of competition" created by tariff concessions.¹⁰⁵⁰ The panel further elaborated this statement as follows:

[I]t could be argued that the standard we enunciated and applied under Article XXIII:1(b) – that of 'upsetting the competitive relationship' – may be different from the standard of 'upsetting effective equality of competitive opportunities' applicable to Article III:4. However, we do not see any significant distinction between the two standards apart from the fact that this Article III:4 standard calls for no less favourable treatment for imported products in general, whereas the Article XXIII:1(b) standard calls for a comparison of the competitive relationship between foreign and

¹⁰⁴³ European Union's first written submission, paras. 606-607.

 ¹⁰⁴⁴ Appellate Body Report, *EC – Asbestos*, para. 185 (citing GATT Panel Report, *EEC – Oilseeds*, para. 144). See also Panel Report, *Japan – Film*, para. 10.35.
 ¹⁰⁴⁵ Appellate Body Report, *EC – Asbestos*, para. 186 (citing Panel Report, *Japan – Film*, para. 10.37).

¹⁰⁴⁵ Appellate Body Report, *EC – Asbestos*, para. 186 (citing Panel Report, *Japan – Film*, para. 10.37). The Appellate Body cited with approval the explanation given by the panel in *Japan – Film* that "Members negotiate the rules that they agree to follow and only exceptionally would expect to be challenged for actions not in contravention of those rules". Panel Report, *Japan – Film*, para. 10.36.

¹⁰⁴⁶ Panel Report, Japan – Film, para. 10.37. See also Panel Report, US – COOL, para. 7.902.

¹⁰⁴⁷ This finding is confined to the dispute brought by Canada only.

¹⁰⁴⁸ Panel Report, US – COOL, para. 7.904 (citing GATT Panel Report, ECC – Oilseeds I, para. 142).

¹⁰⁴⁹ Panel Report, *Japan – Film*, para. 10.86.

¹⁰⁵⁰ Panel Report, Japan – Film, para. 10.86.

domestic products at two specific points in time, i.e. when the concession was granted and currently.¹⁰⁵¹

7.682. We note that the parties have each cross-referenced their own arguments regarding the discriminatory nature of the EU Seal Regime under Articles I:1 and III:4 of the GATT 1994 to support their respective positions as to the nullification or impairment of benefits under Article XXIII:1(b) of the GATT 1994.¹⁰⁵² Moreover, on the specific question of the upsetting of "relative conditions of competition", both complainants premise their claim of nullification or impairment on the fact that domestic and other foreign products may continue to have access to the EU market under the IC and MRM exceptions.¹⁰⁵³ In our view, the "relative conditions of competition" that the complainants claim are upset by the IC and MRM exceptions are precisely those that have been addressed in our findings of violations under Articles I:1 and III:4 of the GATT 1994 and Article 2.1 of the TBT Agreement.

7.683. Therefore, compliance by the European Union with our findings of violations under Articles I:1 and III:4 of the GATT 1994 and Article 2.1 of the TBT Agreement would remove the basis of the complainants' non-violation claims of nullification or impairment. Accordingly, we refrain from examining the complainants' non-violation claims under Article XXIII:1(b) of the GATT 1994.

¹⁰⁵¹ Panel Report, Japan – Film, para. 10.380.

¹⁰⁵² See Canada's first written submission, para. 748; European Union's first written submission, para. 607; Norway's first written submission, para. 1035. ¹⁰⁵³ See Canada's second written submission, para. 347; Norway's second written submission,

paras, 435-436.

We note that Norway has also raised the competitive position of Norwegian seal products vis-à-vis "nonseal products that compete with seal products". Norway's second written submission, para. 435. Although Norway asserts that "[s]eal products are in competitive relationships with other products" (para. 437), Norway has not developed arguments or put forward evidence as to the nature and extent of such competitive relationship. In the absence of such evidence and argument, we are unable to determine the "relative conditions of competition" between such products and, consequently, whether such conditions have been upset as a result of the EU Seal Regime.

8 CONCLUSIONS AND RECOMMENDATIONS

8.1 Complaint by Canada (DS400): Conclusions and recommendations

8.1. Canada has made claims with regard to certain aspects of the EU Seal Regime under Articles 2.1, 2.2, 5.1.2, and 5.2.1 of the TBT Agreement and Articles I:1, III:4, XI:1, and XXIII:1(b) of the GATT 1994.

- 8.2. With respect to Canada's claims under the TBT Agreement, we conclude that:
 - a. the EU Seal Regime is a "technical regulation" within the meaning of Annex 1.1 to the TBT Agreement;
 - b. the IC exception and the MRM exception under the EU Seal Regime are inconsistent with Article 2.1 because the detrimental impact caused by these exceptions does not stem exclusively from legitimate regulatory distinctions and consequently the exceptions accord imported seal products treatment less favourable than that accorded to like domestic and other foreign seal products;
 - c. the EU Seal Regime is not inconsistent with Article 2.2 because it fulfils the objective of addressing the EU public moral concerns on seal welfare to a certain extent, and no alternative measure has been demonstrated to make an equivalent or greater contribution to the fulfilment of the objective as the EU Seal Regime;
 - d. the European Union has acted inconsistently with its obligations under Article 5.1.2 because the conformity assessment procedures under the EU Seal Regime were incapable of enabling trade in qualifying products to take place as from the date of entry into force of the EU Seal Regime; and
 - e. Canada has not demonstrated that the European Union acted inconsistently with its obligations under Article 5.2.1.
- 8.3. With respect to Canada's claims under the GATT 1994, we conclude that:
 - a. the IC exception under the EU Seal Regime is inconsistent with Article I:1 because an advantage granted by the European Union to seal products originating in Greenland is not accorded immediately and unconditionally to the like products originating in Canada;
 - b. the MRM exception under the EU Seal Regime is inconsistent with Article III:4 because it accords imported seal products treatment less favourable than that accorded to like domestic seal products;
 - c. each of the IC, MRM, and Travellers exceptions is not inconsistent with Article XI:1;
 - d. the IC exception and the MRM exception under the EU Seal Regime are not justified under Article XX(a) because they fail to meet the requirements under the chapeau; and
 - e. the IC exception and the MRM exception under the EU Seal Regime are not justified under Article XX(b) because the European Union has failed to make a *prima facie* case for its claim.

8.4. Finally, in light of the above findings of violation, we have refrained from examining Canada's non-violation claim under Article XXIII:1(b) of the GATT 1994.

8.5. Under Article 3.8 of the DSU, in cases where there is an infringement of the obligations assumed under a covered agreement, the action is considered *prima facie* to constitute a case of nullification or impairment of benefits under that agreement. Accordingly, we conclude that to the extent that the European Union has acted inconsistently with Articles 2.1 and 5.1.2 of the TBT Agreement and Articles I:1 and III:4 of the GATT 1994, it has nullified or impaired benefits accruing to Canada under these agreements.

8.6. Pursuant to Article 19.1 of the DSU, we recommend that the Dispute Settlement Body request the European Union to bring the inconsistent measures into conformity with its obligations under the TBT Agreement and the GATT 1994.

8.1 Complaint by Norway (DS401): Conclusions and recommendations

8.1. Norway has made claims with regard to certain aspects of the EU Seal Regime under Articles 2.2, 5.1.2, and 5.2.1 of the TBT Agreement; Articles I:1, III:4, XI:1, and XXIII:1(b) of the GATT 1994; and Article 4.2 of the Agreement on Agriculture.

- 8.2. With respect to Norway's claims under the TBT Agreement, we conclude that:
 - a. the EU Seal Regime is a "technical regulation" within the meaning of Annex 1.1 to the TBT Agreement;
 - b. the EU Seal Regime is not inconsistent with Article 2.2 because it fulfils the objective of addressing the EU public moral concerns on seal welfare to a certain extent, and no alternative measure has been demonstrated to make an equivalent or greater contribution to the fulfilment of the objective as the EU Seal Regime;
 - c. European Union has acted inconsistently with its obligations under Article 5.1.2 because the conformity assessment procedures under the EU Seal Regime were incapable of enabling trade in qualifying products to take place as from the date of entry into force of the EU Seal Regime; and
 - d. Norway has not demonstrated that the European Union acted inconsistently with its obligations under Article 5.2.1.
- 8.3. With respect to Norway's claims under the GATT 1994, we conclude that:
 - a. the IC exception under the EU Seal Regime is inconsistent with Article I:1 because an advantage granted by the European Union to seal products originating in Greenland is not accorded immediately and unconditionally to the like products originating in Norway;
 - b. the MRM exception under the EU Seal Regime is inconsistent with Article III:4 because it accords imported seal products treatment less favourable than that accorded to like domestic seal products;
 - c. each of the IC, MRM, and Travellers exceptions is not inconsistent with Article XI:1;
 - d. the IC exception and the MRM exception under the EU Seal Regime are not justified under Article XX(a) because they fail to meet the requirements under the chapeau; and
 - e. the IC exception and the MRM exception under the EU Seal Regime are not justified under Article XX(b) because the European Union has failed to make a *prima facie* case for its claim.

8.4. Given our finding on Article XI:1 of the GATT 1994 above, we reject Norway's claim under Article 4.2 of the Agreement on Agriculture that the EU Seal Regime constitutes a quantitative import restriction on agricultural products.

8.5. Finally, in light of the above findings of violation, we have refrained from examining Norway's non-violation claim under Article XXIII:1(b) of the GATT 1994.

8.6. Under Article 3.8 of the DSU, in cases where there is an infringement of the obligations assumed under a covered agreement, the action is considered *prima facie* to constitute a case of nullification or impairment of benefits under that agreement. Accordingly, we conclude that to the extent that the European Union has acted inconsistently with Article 5.1.2 of the TBT Agreement and Articles I:1 and III:4 of the GATT 1994, it has nullified or impaired benefits accruing to Norway under these agreements.

8.7. Pursuant to Article 19.1 of the DSU, we recommend that the Dispute Settlement Body request the European Union to bring the inconsistent measures into conformity with its obligations under the TBT Agreement and the GATT 1994.